## TOWN OF BROADWAY, VIRGINIA LAND DEVELOPMENT REGULATIONS

### TABLE OF CURRENT ARTICLES

Revision date appears in the lower right hand corner of each article.

Title		Revision Date
Ι	AUTHORITY AND ENACTMENT	December 6, 2011
II	PURPOSES OF THE REGULATIONS	December 6, 2011
III	DEFINITIONS OF TERMS USED IN THIS ORDINANCE	May 2, 2023
IV	ESTABLISHMENT OF DISTRICTS	December 6, 2011
V	APPLICATION OF ZONING REGULATIONS	December 6, 2011
VI	USES IN DISTRICTS	October 3, 2023
VII	USE REGULATIONS	October 3, 2023
VIII	ADMINISTRATION	February 7, 2023
IX	SUBDIVISION REGULATIONS	July 7, 2015

## ARTICLE 1 AUTHORITY AND ENACTMENT

#### 101.00 AUTHORITY TO ESTABLISH ZONING

Whereas, by act of the General Assembly of Virginia as recorded in Title 15.2, Chapter 22, Article 7, Section 15.2-2280 *et seq.*, Code of Virginia, 1950, as amended, the governing body of any county or municipality may, by ordinance, classify the territory under its jurisdiction into districts of such number, shape, and size as it may deem best suited to carry out the purpose of zoning, and in each district it may regulate, restrict, prohibit, and determine the following: (a) the use of land, buildings, structures, and other premises for agricultural, business, industrial, residential, floodplain and other specific uses; (b) the size, height, area, bulk, location, erection, construction, reconstruction, alteration, repair, maintenance, razing, or removal of structures; (c) the areas and dimensions of land, water, and air space to be occupied by buildings, structures, and uses and of courts, yards, and other open spaces to be left unoccupied by uses and structures, including variations in the sizes of lots based on whether a public or community water supply or sewer system is available and in use; and (d) the excavation or mining of soil or other natural resources.

#### 102.00 AUTHORITY TO ESTABLISH SUBDIVISION REGULATIONS

Whereas, by act of the General Assembly of Virginia as recorded in the Code of Virginia, 1950, as amended, as Article 6, Section 15.2-2240 et seq. requires the Town Council of the Town of Broadway, Virginia adopt regulations to assure the orderly subdivision of land and its development, to provide for the harmonious and economic development of the Town, for the coordination of streets within subdivisions with other existing or planned streets, for adequate open spaces for traffic, recreation, light and air, and for the distribution of population and traffic which will tend to create conditions favorable to health, safety, convenience, and prosperity including reasonable regulations and provisions that apply to or provide (a) for size, scale, and other plat details; (b) for the coordination of streets within and contiguous to the subdivision with other existing or planned streets within the general areas as to location, widths, grades, and drainage; (c) for adequate provisions for drainage and flood control and other public purposes, and for light and air; (d) for the extent to which and the manner in which streets shall be graded, graveled, or otherwise improved and water and storm and sanitary sewer and other public utilities or other community facilities are to be installed; (e) for the acceptance of dedication for public use of any right-of-way located within any subdivision which has been constructed or proposed to be constructed within the subdivision, any street, curb, gutter, sidewalk, bicycle trail, drainage or sewerage system, water line as a part of a public system, or other improvement, financed or to be financed in whole or part by private funds in accordance with certain conditions prescribed herein; (f) for monuments of specific types to be installed establishing street and property lines; (g) for provisions regarding the time period within which plats approved in accordance with the provisions herein must be recorded; (h) for the administration and enforcement of such ordinance, not inconsistent with provisions contained in this chapter, and specifically for the imposition of reasonable fees and charges for the review of plats and plans, and for the inspection of facilities required by any such ordinance to be installed; such fees and charges shall in no instance exceed an amount commensurate with the services rendered taking into consideration the time, skill and administrator's expense involved; (i) for payment by a subdivider or developer of land of his pro rata share of the cost of providing reasonable and necessary sewerage and drainage facilities, located outside the property limits of the land owned or controlled by him but necessitated or required, at least in part, by the construction, or improvement of his subdivision or development in accordance with certain provisions herein; (j) for reasonable provisions permitting a single division of a lot or parcel for the purpose of sale or gift to a member of the immediate family of the property owner, subject only to any express requirement contained in the Code of Virginia; (k) for the partial or complete release of any bond, escrow, letter of credit, or other performance guarantee required by the Town under the provisions herein; and (l) for the exercise of other subdivision powers prescribed by general law.

#### 103.00 ENACTMENT

Therefore, be it ordained by the Town Council of the Town of Broadway, Virginia, for the purpose of promoting the health, safety, or general welfare of the public and of further accomplishing the objectives of Title 15.2, Chapter 22, Articles 6 and 7 of the Code of Virginia, 1950, as amended that the following Ordinance be adopted, constituting the Town of Broadway Zoning Ordinance, as amended and Subdivision Ordinance - Town of Broadway, Virginia, as amended. This Ordinance shall be known and may be cited as the "Land Development Regulations of the Town of Broadway, Virginia (2009)". The Subdivision Ordinance shall consist of Article 9, and the Zoning Ordinance shall consist of all other articles of the Land Development Regulations. A reference herein to "this Ordinance" shall refer to the Land Development Regulations as a whole, including both the Zoning Ordinance and the Subdivision Ordinance.

## ARTICLE 2 PURPOSES OF THE REGULATIONS

#### **201.00 PURPOSES**

The Planning Commission of the Town of Broadway, Virginia and the Town Council of the Town of Broadway, Virginia have undertaken to achieve the delicate balance between the individual property rights of its citizens and the health, safety, and general welfare of the public and accomplish the objectives of Section 15.2-2200 by reasonable restrictions on those property rights. The purposes of these regulations are: (a) to provide for adequate light, air, convenience of access, and safety from fire, flood, and other dangers; (b) to reduce or prevent congestion in the public streets; (c) to facilitate the creation of a convenient, attractive, and harmonious community; (d) to expedite the provision of adequate police and fire protection, disaster evacuation, civil defense, transportation, water, sewerage, flood protection, schools, parks, forests, playgrounds, recreational facilities, airports, and other public requirements; (e) to protect against destruction of, or encroachment upon, historic areas; (f) to protect against one or more of the following: overcrowding of land, undue density of population in relation to the community facilities existing or available, obstruction of light and air, danger and congestion in travel and transportation, or loss of life, health, or property from fire, flood, panic, or other dangers; (g) to encourage economic development activities that provide desirable employment and enlarge the tax base; and (h) to establish certain subdivision standards and procedures to assure the orderly subdivision of the land and its development for the Town of Broadway, Virginia, and to implement the Town's Comprehensive Plan.

**Ordinance of December 6, 2011** added the phrase "and to implement the Town's Comprehensive Plan" to the end of this section.

#### 202.00 INTERPRETATION OF ORDINANCE

The Town Council finds and declares that the standards and procedures of this Ordinance are part of a long-range plan to guide and facilitate the orderly and beneficial growth of the Town, and to promote the public health, safety, convenience, comfort, prosperity, and general welfare. More specifically, the purposes of these standards and procedures are to provide a guide for the change that occurs where land and acreage become urban in character as a result of development for residential, business, or industrial purposes, to provide assurance that the purchasers of lots are buying a commodity that is suitable for development and use; and to make possible the provision of public services in a safe, adequate, and efficient manner. This Ordinance shall be liberally construed to effectuate the purposes expressed herein.

#### 203.00 NON-EXCLUSIONARY INTENT

It is not the intent of these regulations to exclude any economic, racial, religious, or ethnic group from enjoyment of residence, land ownership, or tenancy within the Town of Broadway. Further, it is not the intent of this ordinance to use public powers in any way to promote the separation of economic, racial, religious, or ethnic groups within the Town of Broadway, except as may be an incidental result of meeting the other purposes outlined herein.

# ARTICLE 3 DEFINITIONS OF TERMS USED IN THIS ORDINANCE

#### **301.00 GENERAL**

Except as otherwise provided herein, all words shall have the customary dictionary meaning. The present tense includes the future tense. The singular number includes the plural and the plural includes the singular. The masculine gender includes the feminine and neuter genders. The word "person" includes a firm, corporation, association, organization, trust, or partnership. The word "lot" includes "plot" or "parcel." The word "building" includes "structure." The word "shall" is always mandatory; the term "may" is discretionary. The word "used" or "occupied" as applied to any land or building shall be construed to include the words "intended, arranged, or designed to be used or occupied."

#### 302.00 SPECIFIC DEFINITIONS

When used in this Ordinance the following words and phrases shall have the meaning given in this Section:

- 302.01 **Abandoned Vehicles.** Any vehicle which is deemed inoperable by the fact that any of the following requirements are not met: valid license, inspection sticker, town tag, or cannot operate under its own power.
- 302.02 *Accessory Building*. A building that is intended to be used or is used for an accessory use. An accessory building is also an accessory structure.
- 302.03 Accessory Structure. A structure that is intended to be used or is used for an accessory use. An accessory building is an accessory structure, but not all accessory structures are accessory buildings.
- 302.04 *Agriculture*. The tilling of the soil, the raising of crops, horticulture, and forestry, including the keeping of animals and fowl, and including any agricultural industry or business, such as fruit packing plants, dairies or similar use, not including commercial slaughterhouses.
- 302.05 *Alley*. A platted service way providing a secondary means of access to abutting properties.
- 302.06 *Alteration*. Any change in the total floor area, use, adaptability, or external appearance of an existing structure.

- 302.07 *Animal or Poultry Husbandry*. Any keeping, boarding, breeding, or raising of any number of horses, goats, sheep, poultry, or other customary farm animals for any purpose; or of more than five (5) dogs and seven (7) cats or other customary pet animals for noncommercial purposes.
- 302.08 *Animal Hospital or Clinic*. Establishment where treatment is received and no activity is conducted outside the main building. Kennels are not included.
- 302.09 **Apartment**. A unit in a multi-family dwelling providing living quarters for a single family, in which separate access to the outside is usually not provided, and in which the major orientation of the unit is horizontal rather than vertical; or any condominium unit of similar physical character, appearance, and structure.
- 302.10 *Auction House*. A building or place where a public sale is held which property or goods, excluding livestock, are sold to the highest bidder.
- 302.11 **Automobile Graveyard.** Any lot or place which is exposed to the weather upon which more than three (3) motor vehicles of any kind not displaying current Commonwealth of Virginia inspection certification are placed, located, or found.
- 302.12 Bed and Breakfast or Short Term Rental. A Single Family Dwelling, a unit within a Two-Family Dwelling or a Multi-Family Dwelling, or an Accessory Building to a Dwelling (a) in which guests are provided with an abiding place for a fee on a more or less transient and temporary basis, (b) in which the average length of stay is one week or less in duration; and (c) in a structure and on a lot in which no more than four (4) guest rooms under common ownership are available for guests, with no more than eight (8) guests are at one (1) time, except for minors in the company of adult guests. To be a Bed and Breakfast or Short Term Rental, it is not necessary that the facility provide breakfast or any other meals. A structure used as a Bed and Breakfast or Short Term Rental does not thereby cease to be a Dwelling, and the requirements of the Land Development Regulations applicable to such Dwellings continue to apply to the Bed and Breakfast or Short Term Rental. The particular type of Dwelling constituting the Bed and Breakfast or Short Term Rental must be a structure that is permitted (either by right, or through a duly granted special use permit) in the underlying zoning district.

**Ordinance of May 2, 2023** amended this section to add the reference to "Short Term Rentals" and adjusted other aspects of the definition. The section previously read as follows: "Bed and Breakfast. A bed and breakfast facility is a structure in which guests are provided with

sleeping quarters for a fee, but the proprietor also resides in the facility, and in which no more than four (4) guest rooms are available for guests, and no more than eight (8) guests are at one (1) time, except for minors in the company of adult guests."

302.13 **Boarding House (Rooming House).** A building or part thereof where lodging is provided for compensation to unrelated persons on a more or less transient or

temporary basis and in which the average length of stay exceeds one (1) week in duration

**Ordinance of May 2, 2023** adjusted several aspects of this definition. The section previously read as follows: "*Boarding House (Rooming House)*. A building or part thereof, other than a hotel, motel, or restaurant, where meals and/or lodging are provided for compensation for

three (3) to ten (10) unrelated persons where no cooking or dining facilities are provided in individual rooms and in which the length of stay usually exceeds one (1) week in duration. A lodging house is also included in this definition."

- 302.14 *Building*. Any structure designed or intended for support, enclosure, shelter, or protection of persons, animals, or property.
- 302.15 *Building Code*. The Virginia Uniform Statewide Building Code, as adopted by the Town of Broadway and as amended.
- 302.16 **Building Inspector**. The building official appointed by the Rockingham County to administer and enforce the provisions of the Building Code, or his designated representative or agent.
- 302.17 *Building, Main*. A building in which is conducted the main or principal use of the lot on which said building is situated.
- 302.18 *Child Care Center*. Any facility other than a Family Day Care Home, providing care, protection, and guidance to a group of children during only part of the day.
- 302.19 Repealed by Ordinance of December 6, 2011.

Former Section 302.19 read: "Church or House of Worship. A building where persons regularly assemble for religious worship, and which is maintained and

controlled by a religious body organized to conduct public worship."

- 302.20 *Community Center*. Community entertainment, recreation, or meeting place operated by a non-profit organization.
- 302.21 *Communications Equipment*. Any tower, dish or other equipment used to send or receive electronic transmissions for public or private use.
- 302.22 *Condominium*. A unit in a building or development which is individually owned, but in which the common areas are owned, controlled, and maintained through an organization consisting of all individual owners.
- 302.23 *Cottage Industry*. A small, non-polluting business or industry employing fewer than five (5) workers.
- 302.24 *Cul-de-Sac*. A circular turning area at the end of a dead end street.

302.241 *Density, gross*: The number of dwelling units per gross acre of land on the site or lot.

Ordiannce of December 6, 2011 added this section.

302.242 **Density, net**: The number of dwelling units per net acre, which is the gross site or lot area less floodplain, steep slopes in excess of 25%, public right-of-way, and any nonresidential use areas in residential or mixed-use developments.

Ordiannce of December 6, 2011 added this section.

- 302.25 **Developer**. An owner of property being subdivided, whether or not represented by an agent.
- 302.26 **Development**. A tract of land developed or to be developed as a unit under single ownership or unified control which is to contain three or more residential dwelling units. The term "development" shall not be construed to include any property which will be principally devoted to agricultural production.
- 302.27 **Driveway**. Any private way provided for the principal purpose of providing vehicular access to an off-street parking area or service in the case of drive-in type uses.
- 302.28 **Dwelling**. Any building or portion thereof which is designed for or used for residential purposes, except hotels, boarding houses, manufactured homes, and automobile trailers.

**Ordinance of May 2, 2023** deleted the reference to "lodging houses" and "tourist cabins".

302.29 *Dwelling, Multi-Family*. A building designed for, or occupied exclusively by, three (3) or more families living independently of each other; the term includes condominiums of similar physical appearance, character, and structure. The term also includes Townhouses and Apartments.

**Ordinance of July 7, 2015** deleted the phrase "or more" after the word "three (3)".

**Ordinance of May 2, 2023** restored the language that the Ordinance of July 7, 2015 deleted, and added the final sentence.

- 302.30 *Dwelling, Single-Family*. A building designed for, or occupied exclusively by, one (1) family.
- 302.31 *Dwelling, Two-Family (Duplex)*. A building designed for, or occupied exclusively by, two (2) families living independently of each other.
- 302.32 **Dwelling Unit**. One or more rooms in a dwelling designed for living or sleeping purposes, and having at least one (1) kitchen.

- 302.33 *Easement*. A grant by a property owner of the use of land for a specific purpose or purposes by the general public, a corporation, or a certain person or persons.
- 302.34 *Engineer, Civil.* An engineer currently licensed by the Commonwealth of Virginia.
- 302.35 *Family*. One or more persons occupying a dwelling and living as a single housekeeping unit, as distinguished from persons occupying a boarding house, lodging house, or hotel, as herein defined.
- 302.36 *Family Day Care Home*. Any private family home providing care, protection, and guidance to not more than ten (10) children during only part of the day. Children related by blood or marriage to the person who maintains the home shall not be counted.
- 302.37 *Flood Proofing*. A combination of structural provisions, changes, or adjustments to properties and structures subject to flooding required for new construction in the floodplain by the Virginia Uniform Statewide Building Code, Section 135.6.
- 302.38 *Floor Area*. The sum of the gross horizontal areas of the total number of floors of a building measured from the exterior faces of the exterior walls or from the center line of walls separating two buildings, but not including any attic space providing headroom of less than seven (7) feet, unusable basement or cellar space not used for retailing, uncovered steps or fire escapes, open porches, accessory water or cooling towers, accessory off-street parking spaces, and accessory off-street loading berths.
- Frontage. The minimum width of a lot measured from one side lot line to the other, along a straight line on which no point shall be farther away from the street upon which the lot fronts, or from the front edge of the lot, than the building setback line as defined and required herein.
- 302.40 *Garage, Private*. Accessory building designed or used for the storage vehicles owned and used by the occupants of the building to which it is accessory.
- 302.41 *Garage, Public*. A building or portion thereof, other than a private garage, designed or used for servicing, repairing, painting, equipping, renting, selling, or storing motor-driven vehicles.
- Gardening. Any use of land unenclosed except for fencing for the raising of grass, flowers, vegetables, crops, trees, or other botanical objects of natural growth, generally for the use and/or consumption by the occupants of the premises, but not including accessory structures used for the same purpose.

- 302.43 *Golf Course*. Any golf course, publicly or privately owned, on which the game of golf is played, including accessory uses and buildings customary thereto, but excluding golf driving ranges as defined herein.
- 302.44 *Golf Driving Range*. A limited area on which golf players do not walk, but onto which they drive golf balls from a central driving tee.
- 302.441 *Group Home.* A residential facility in which no more than eight mentally ill, mentally retarded or developmentally disabled persons reside with one or more resident counselors or other staff persons. A "residential facility" means any group home or other residential facility for which the Department of Mental Health, Mental Retardation and Substance Abuse Services is the licensing authority pursuant to the Virginia Code. For the purposes of these regulations a group home shall be treated as residential occupancy by a single family per Virginia Code § 15.2-2291. Mental illness and developmental disability shall not include current illegal use of or addiction to a controlled substance as defined in Virginia Code § 54.1-3401, as amended.

**Ordinance of December 6, 2011** added this section. Pursuant to authority granted in Section 805.05, and by letter dated January 3, 2012, to the Town Clerk, the Town Attorney corrected a typographical error in this section by

capitalizing the word "Home" and first letter of the first sentence, and changing the colon after the title of the section to a period.

- 302.45 *Group Housing Project*. A Group Housing Project shall consist of two (2) or more buildings located on a site where the building arrangement is such that the property cannot be subdivided into conventional streets and lots that meet the requirements of the Town of Broadway Land Development Regulations.
- 302.46 *Group Housing Project for Older Persons*. A Group Housing Project provided under any state or federal program specifically designed and operated to assist elderly persons (as defined in the state or federal program) or intended for, and solely occupied by, persons sixty-two (62) years of age or older.
- 302.47 *Guest Room*. A room which is intended, arranged or designed to be occupied, or which is occupied, by one or more guests paying direct or indirect compensation therefore, but in which no provision is made for cooking. Dormitories are excluded.
- 302.48 *Health Department*. The Rockingham County Health Department or its designated agent or representative.
- 302.49 *Height*. The vertical distance measured from the level of the edge of the pavement opposite the middle of the front of the structure to the highest point of the roof if a flat roof; to the deck line of a mansard roof; or to the mean height level between the eaves and the ridge of a gable, hip, or gambrel roof, or to the tallest point if no roof. For

structures set back from the road line, the height shall be measured from the average elevation of the ground surface along the front of the structure.

- 302.50 *Highway Engineer*. The official designated by the Virginia Department of Highways and Transportation to inspect subdivision streets and alleys, and other public ways.
- 302.51 *Home Occupation*. An occupation, trade, or business conducted as an accessory use within a dwelling or within an accessory building located on the same lot as the dwelling unit, in a residential zoning district, and conducted by the resident thereof.
- 302.52 *Hog Pen*. An enclosure for concentrated confinement or housing of swine.
- 302.53 *Hospital*. An institution rendering medical, surgical, obstetrical, or convalescent care, including any institution licensed as a hospital by the State Hospital Board.
- 302.54 *Hotel*. A building or group of buildings that is (a) not a Bed and Breakfast/Short Term Rental; (b) in which lodging, or board and lodging, are provided and offered to the public for compensation on a more or less transient or temporary basis; and (c) in which the average length of stay is one week or less in duration. The term "hotel" includes the term "motel".

Ordinance of May 2, 2023 adjusted several aspects of the definition. The section previously read as follows: *Hotel*. A building that is not an inn or bed and breakfast and in which lodging, or board and lodging, are provided and offered to the public for compensation and in which

cooking facilities may be provided or in which lodging facilities are provided primarily for travelers and in which the length of stay is primarily less than one week in duration. The term "hotel" includes the term "motel".

#### 302.55 Repealed by Ordinance of May 2, 2023

**Former Section 302.55 read** as follows; "*Inn*. A building or buildings designed and occupied as the more or less temporary abiding place for individuals who are, for compensation, lodged with or without meals, and in which provision is not generally made for cooking in

individual rooms or suites and in which no more than four (4) guest rooms are available for guests, and no more than eight (8) guests are at one (1) time, except for minors in the company of adult guests."

- 302.56 **Job Trailer**. A vehicular portable unit built on a chassis, utilized as a temporary building on construction sites, which is either designed or converted for uses incidental to construction work.
- Junk Yard (Automobile Wrecking Yard). A lot, land, or structures or part thereof, used primarily for the collecting, storage, and sale of waste paper, rags, scrap metal, or discarded material; or for the collecting, dismantling, storage, and salvaging of machinery or vehicles not in running conditions, or for the sale of parts thereof.
- 302.58 *Kennel*. Any location where breeding, raising, grooming, caring for or boarding of dogs, cats, or other similar small animals for commercial purposes is carried on.

- 302.59 *Light Industry*. Includes warehousing and light manufacturing uses which produce some noise, traffic congestion or danger, but which are of such limited scale or character that they present no serious hazard to neighboring properties from fire, smoke, noise, or odors.
- 302.60 *Livestock*. Animals kept or raised for sale, use, or pleasure.
- 302.61 *Livestock Market*. A commercial establishment wherein livestock is collected for sale, sold, or auctioned off.
- 302.62 *Living Unit.* Another term for "dwelling unit".
- 302.63 **Loading Space**. A space within the main building or on the same lot, providing for the standing, loading, or unloading of trucks and other carriers.
- 302.64 **Lot.** A numbered and measured portion or parcel of land separated from other portions or parcels by description in a site plan or a recorded plat, or by metes and bounds, intended to be a unit for the purpose, whether immediate or future, or transfer of ownership, or of development or separate use.
- 302.65 *Lot Area*. The total horizontal area within the lot lines of a lot. No alley, public way, public land, or area proposed for future street purposes is included within the net area of the lot.
- 302.66 *Lot, Corner*. A lot abutting upon two (2) or more streets at their intersection. Of the two sides of a corner lot the front shall be deemed to be the shorter of the two sides fronting on streets.
- 302.67 *Lot Coverage*. The ratio of the horizontally projected area of the main and accessory buildings on a lot to the total area of the lot, except where otherwise defined herein.
- 302.68 Lot, Depth of. The average horizontal distance between the front and rear lot lines.
- 302.69 *Lot of Record*. A lot or parcel of land whose existence, location, and dimensions have been recorded in the Office of the Clerk of the Circuit Court of Rockingham County at the time of the adoption of this Ordinance.
- 302.70 *Lot*, *Width*. The average horizontal distance between side lot lines.
- 302.71 *Main Use*. The primary purpose for which land or a building is used.

- 302.72 *Manufacture and/or Manufacturing*. The processing and/or converting of raw, unfinished materials, or products or either of them into articles of substances of different character or for use for a different purpose.
- Manufactured Home. A manufactured dwelling unit designated for long-term occupancy and constructed originally with wheels for movement (whether or not such wheels have later been removed) and which has plumbing and electrical connections provided for attachment to outside systems. A camping vehicle or travel trailer shall not be considered a manufactured home.
- 302.74 *Manufactured Home Park*. Any development in which space is leased providing for three (3) or more manufactured homes intended for residential use for a period of time longer than thirty (30) days.
- 302.75 *Manufactured Home Stand*. A plot of ground within a manufactured home park designed to accommodate one manufactured home.
- 302.76 *Manufactured Home Subdivision*. Any area designated to accommodate three (3) or more manufactured homes intended for residential use on lots owned by the manufactured home owner.
- 302.761 *Mobile Food Service Establishment*. Any mobile vehicle located on any lot from which, for compensation, food or beverages are dispensed to persons for consumption, whether or not on the premises. If a vehicle is immobilized on a lot such that it becomes a fixture, then such vehicle is not a Mobile Food Service Establishment, but is instead a Restaurant. A vehicle operating primarily in transit on a public street (such as an ice cream truck) is not a Mobile Food Service Establishment.

Ordinance of December 6, 2011 added this section.

- 302.77 *Motor Home or Camper*. A unit or subunit which is or becomes self-propelled and is designed for human habitation on a short-term basis.
- Motor Vehicle. Any vehicle which is self-propelled or designed for self-propulsion. Any device designed, used, or maintained primarily to be loaded on or affixed to a motor vehicle to provide a mobile dwelling, a sleeping place, storage, or office or commercial space shall be considered a part of a motor vehicle.
- Non-Conforming Lot. An otherwise legally platted lot that does not conform to the minimum area or width requirements of this Ordinance for the District in which it is located either at the effective date of this Ordinance or as a result of subsequent amendments to the Ordinance.

- 302.80 **Non-Conforming Use of Structures**. The otherwise legal use of a building or structure that does not conform to the use of regulations of this ordinance for the district in which it is located, either at the effective date of this ordinance or as a result of subsequent amendments to the ordinance.
- Non-Conforming Structure. A structure existing at the time of enactment or amendment of this Ordinance which does not conform to the requirements of this Ordinance by reason of height or condition, or by reason of its impingement upon required yard areas.
- 302.82 *Non-Conforming Use of Land*. A use of land existing at the time of the enactment of this Ordinance, or at the time of an amendment thereto, which does not conform with the regulations of the use district in which it is located.
- Nursing Home. Any facility or any identifiable component of any facility in which the primary function is the provision, on a continuing basis, of nursing services and health-related services for the treatment and inpatient care of two or more nonrelated individuals, including facilities known by varying nomenclature or designation such as convalescent homes, skilled care facilities, intermediate care facilities, extended care facilities, and infirmaries.
- 302.84 *Off-Street Parking Area*. Space provided for vehicular parking outside the dedicated street right-of-way.
- 302.85 *One Hundred Year Flood*. A flood that, on the average, is likely to occur once every 100 years.
- 302.86 *Overhang*. Any projection, either roof, bay, window, or similar cantilevered construction, which extends beyond the foundation of a structure.
- 302.87 *Parks, Playgrounds, and Outdoor Recreation Areas*. Land publicly or privately owned devoted to recreational pursuits, usually an open area reserved for outdoor activities such as play, hiking, exercise, or competitive sport not requiring structures for habitation.
- 302.88 **Pen.** A small enclosure used for the concentrated confinement and housing of animals or poultry; a place for feeding and fattening animals; a coop. Enclosed pasture or range with an area in excess of one hundred (100) square feet for each small animal or two hundred (200) square feet for each larger animal shall not be regarded as a pen.
- 302.89 **Personal Service Home Occupation**. A home occupation that generally provides onpremises goods or services to individual clients. Personal service home occupations include (but are not limited to) occupations related to the care of a person, a person's

- apparel, or the training and development of a person, such as beauty or barber shops, nail salons, dressmakers or tailors, photography or portrait painting.
- 302.90 *Plat*. Includes the terms: map, plan, plot, replat, or replot; a map or plan of a tract or parcel of land which is to be, or which has been subdivided. When used as a verb "plat" is synonymous with "subdivide".
- 302.91 *Professional*. When used in connection with "use" and "occupancy", a use or occupancy by persons generally engaged in rendering personal, executive, sales, or administrative services or activities, including accountants, architects, professional engineers and land surveyors, doctors, lawyers, insurance offices, real estate offices, religious organizations, stockbrokers, and administrative agencies considered professional in character. The term does not include repair or sale of tangible personal property stored or located within the structure nor any use which would create any loud noises or noxious odors.
- 302.92 *Property*. Any tract, lot, parcel, or several of the same collected together for the purpose of subdividing.
- 302.93 **Property Owners Association**. A corporation or other legal entity or a non-profit organization which has as its purpose maintenance of streets and/or other common areas.
- 302.931 *Public Assembly Hall*. A meeting halls for any church or other religious institution, club, lodge or similar use in which people gather to meet for public or private functions, and including attendant educational or recreational facilities.

Ordinance of December 6, 2011 added this section.

- 302.94 *Public Service or Storage Buildings*. Governmental facilities necessary for public health, safety, and welfare.
- 302.95 *Public Utilities*. Public service structures such as power plants or substations; treatment plants, sewage treatment plants; or such similar operations publicly or privately owned furnishing electricity, gas, rail transport, communications, or related services to the general public. The term Utilities does not include Telecommunications Antennae or Telecommunications Towers.
- 302.96 *Public Utility Transmission Systems*. To include water, sewer, electricity, gas lines, or related transmission facilities for public use, telephones, etc. The term does not include Telecommunications Antennae or Telecommunications Towers.
- 302.97 *Ramada*. A structure erected over a manufactured home for the purpose of providing shade or shelter.

- 302.98 **Required Open Space**. Any space required in any front, side, or rear yard.
- 302.99 **Residential Use**. Any place, building, or establishment used in whole or in part as a dwelling or manufactured home.
- 302.100 **Restaurant**. Any building in which, for compensation, food or beverages are dispensed to persons not residing on the premises for consumption on the premises, including, among other establishments, cafes, delicatessens, or refreshment stands.
- 302.101 **Restaurant, Drive-In.** An eating and/or drinking establishment which caters to motor-driven vehicle business where the person being served may consume his food and/or drink while sitting in a motor-driven vehicle, as opposed to a restaurant serving exclusively inside or adjacent to the main building.
- 302.102 **Retail Stores and Shops**. Buildings for display and sale of merchandise at retail for for the rendering or personal services (but specifically exclusive of coal, wood, and lumber yards), such as the following, which will serve as illustrations: drug store, newsstand, food store, candy shop, milk dispensary, dry-goods and notions store, antique store and gift shop, hardware store, household appliance store, furniture store, florist, optician, music and radio store, tailor shop, and beauty and barber shop.
- 302.103 *Right-of-Way*. Access over or across particularly described property for a specific purpose or purposes.
- 302.104 *Right-of-Way Line*. The dividing line between a lot, tract, or parcel of land and a contiguous street, railroad, or public utility right-of-way.
- 302.105 **School, Business or Commercial**. Privately owned and operated educational institution or educational organization, no matter how titled, maintained or conducting classes for the purpose of offering instruction, for a consideration, profit or tuition, to prepare individuals to pursue any occupation for profit in business administration, bookkeeping, accounting, data processing, stenography, clerical, secretarial, receptionist, or other office occupations.
- 302.106 *School, Private*. Privately owned and operated educational institution or educational organization, maintained or conducting classes for the purpose of offering instruction of students from kindergarten to twelfth grade level.
- 302.107 *School, Public*. Publicly owned and operated educational institution or educational organization regulated by the Commonwealth of Virginia and maintained or conducting classes for the purpose of offering instruction of students from kindergarten to twelfth grade level.

- 302.108 **School, Vocational.** Privately or publicly owned and operated educational institution or educational organization maintained or conducting classes for the purpose of offering instruction to pursue any occupation for profit in any skilled trade, electronics, data processing or industry, or to give occupational training, or to give training in public or other service occupations, or to give vocational training designed to prepare an individual for, or to upgrade an individual in, technical occupations and technical phases of other occupations.
- 302.109 **Screening**. A barrier to vision or noise consisting of trees, bushes, shrubbery, or fences.
- 302.110 **Setback**. The minimum distance by which any building structure must be separated from the front lot line.
- 302.111 **Setback Line**. A line generally parallel with and measured from the lot line, defining the limits of a yard in which no building or structure may be located above ground.
- 302.1111 *Short Term Rental.* See definition of Bed and Breakfast or Short Term Rental in Section 302.12.

Ordinance of May 2, 2023 added this section.

- 302.112 **Sign**. Any words, lettering, parts of letters, figures, numerals, phrases, sentences, emblems, devices, designs, trade names or marks, or combinations thereof, by which anything is made known, such as the designation of an individual, a firm, an association, a profession, a business, a commodity, or product, which are visible from any public way and used as an outdoor display.
- 302.113 **Sign Area**. The smallest square, rectangle, triangle, circle, or combination thereof encompassing the entire advertising area, excluding architectural trim and structural supports.
- 302.114 *Sign, Business*. A sign painted, electrical, or otherwise, erected for the purpose of conveying information, knowledge, or ideas to the public about a subject related to the premises upon which said sign is located.
- 302.115 **Sign, Directional**. A directional sign is one (one end of which may be pointed or on which an arrow may be painted) indicating the direction to which attention is called giving only the name of the firm or business responsible for the erection of same and distance.
- 302.116 *Sign, Locational*. A sign which directs attention to the approximate location of an establishment from which an advertised product or service may be obtained.

- 302.117 **Sign, Outdoor Advertising.** A structural poster panel or painted sign, either free standing or attached to a building, for the purpose of conveying information, knowledge, or ideas to the public about a subject unrelated to the premises upon which it is located. Also known as a "billboard".
- 302.118 *Sign Structure*. A structure, including the supports, uprights, bracing and framework be it single-faced, double-faced, V-type, or otherwise, which is located on the ground or on top of another structure and which supports no more than two (2) signs.
- 302.119 *Sign Structure Facing*. The surface of the sign upon, against, or through which the message of the sign is exhibited, not including architectural trim and structural supports.
- 302.120 *Site Plan*. The proposal for a development or a subdivision including all covenants, grants or easements and other conditions relating to use, location and bulk of buildings, density of development, common open space, public facilities, and such other information as is required in applicable sections of this Ordinance.
- 302.121 **Storage Trailer**. A vehicular portable unit built on a chassis, utilized as a temporary storage unit, which is designed or converted for uses incidental to the principal permitted use on the property.
- 302.122 *Storey*. That portion of a building, other than the basement, included between the surface of the floor next above it, if there be no floor above it, the space between the floor and the ceiling next above it.
- 302.123 *Street*. The principal means of access to abutting properties.
- 302.124 *Street Center Line*. A line generally parallel to the right-of-way lines that equally divide the street right-of-way.
- 302.125 *Street, Half.* A street that does not meet the minimum right-of-way width requirements set forth or referenced in this Ordinance.
- 302.126 *Street, Internal*. A private street providing access to lots within a development, but not including driveways.
- 302.127 *Street Line*. The dividing line between a street or road right-of-way and the contiguous property.
- 302.128 *Street, Major*. A heavily traveled thoroughfare or highway that carries a large volume of through traffic, or anticipated traffic exceeding five hundred (500) vehicles per day.

- 302.129 **Street, Other.** A street that is used primarily as a means of public access to the abutting properties with anticipated traffic of less than five hundred (500) vehicles per day.
- 302.130 *Street Width*. The total width of the strip of land dedicated or reserved for public travel, including roadway, curbs, gutters, sidewalks, planting strips, and bikeways.
- 302.131 *Structure*. Anything constructed or erected, the use of which requires a location on the ground, or attached to something having a location on the ground. Does not include streets, driveways, sidewalks, curbs, or street level guttering, or objects entirely underground.
- 302.132 **Subdivider**. Any individual, corporation or registered partnership owning any tract, lot, or parcel of land to be subdivided, or a group of two or more persons owning any tract, lot, or parcel of land to be subdivided who have given their power of attorney to one of their groups or another individual to act on their behalf in planning, negotiating for, in representing, or executing the legal requirements of the subdivision.
- 302.133 *Surveyor*. A land surveyor currently certified by the Commonwealth of Virginia.
- 302.134 *Television and/or Radio Stations*. A broadcasting facility licensed in the public interest, convenience, and necessity by the Federal Communications Commission, which includes transmitting and receiving equipment, studios, offices, utility buildings, and other necessary accessories required to operate a station.

#### 302.135 Repealed by Ordinance of May 2, 2023

Former Section 302.135 read as follows; "Tourist Court, Auto Court, Motel, Hotel, Cabin, or Motor Lodge. Building or buildings containing individual sleeping rooms, designed for, or used temporarily by,

automobile tourists or transients, with garage or parking space conveniently located to each unit. Cooking facilities may be provided for each unit.

#### 302.136 Repealed by Ordinance of May 2, 2023

Former Section 302.136 read as follows; "Tourist Home. A dwelling where only lodging is provided for

compensation for up to fourteen (14) persons (in contrast to hotels and boarding houses) and open to transients."

302.137 **Townhouse**. A unit separated from adjacent units by a vertical wall with no openings, providing a dwelling for a single family, in which separate access to the outside is provided, and in which the major orientation of the unit is vertical rather than horizontal, and which is part of a structure containing at least four (4) such units.

- 302.138 *Townhouse Development*. One or more single-family dwellings consisting of townhouses, with accessory parking, open space, and recreational and management facilities.
- 302.1381 **Townhouse group**. A cluster or grouping of townhouse units containing not less than four nor more than six individual townhouse dwelling units contiguous to one another.

Ordinance of July 7, 2015 added this section

302.1382 *Townhouse project site*. The entire parcel or parcels of land for which individual townhouse units are proposed prior to the creation of any townhouse lots.

Ordinance of July 7, 2015 added this section

- 302.139 *Travel Trailer Park or Travel Trailer Camp*. Premises where travel trailers are parked temporarily in conjunction with travel, recreation, or vacation.
- 302.140 *Tree*. A woody perennial plant having a single main stem.
- 302.141 *Warehouse*. A structure for storing goods, wares or merchandise.
- 302.142 **Wholesale Sales**. An operation which sells chiefly to retailers, other merchants, or industrial, institutional and commercial uses for resale or business use.
- 302.143 **Yard**. A space on the same lot with a main building, such space being open, unoccupied, and unobstructed by structures from ground to sky except where encroachments and accessory structures are expressly permitted.
- 302.144 **Yard, Front**. An open, unoccupied space, of a minimum depth equal to the setback, on the same lot with the main building, extending the full width of the lot and situated between the right-of-way line and the front line of the building projected to the side lines of the lot. On corner lots, the depth of the front yard shall be considered as parallel to the street upon which the lot has its least dimension.
- 302.145 **Yard, Rear**. An open space, excluding steps, on the same lot with the main building, such space being unoccupied except possibly by an accessory structure and extending the full width of the lot and situated between the rear line of the lot and the rear line of the main building projected to the side lines of the lot. On all corner lots the rear yard shall be the opposite end of the lot from the front yard.
- 302.146 *Yard, Side*. An open, unoccupied space, excluding steps, on the same lot with a main building, situated between the side line of the building and the adjacent side line of the lot extending from the rear line of the front yard to the front line of the rear yard.

If no front yard is required, the front boundary of the side yard shall be the front line of the lot and if no rear yard is required, the rear boundary of the side yard shall be the rear line of the lot. On corner lots, the side yard shall be considered as parallel to the street upon which the lot has its greatest dimension.

#### 303.00 CONTIGUOUSNESS.

Whenever in these regulations, the substantive rights or obligations of a party depend on two or more parcels or uses being contiguous, such contiguousness shall be determined while ignoring any streets or rights of way that intervene between two property lines.

Ordinance of December 6, 2011 added this section.

## ARTICLE 4 ESTABLISHMENT OF DISTRICTS

#### 401.00 DIVISION OF THE TOWN OF BROADWAY INTO DISTRICTS

For the purposes of this Ordinance, the Town of Broadway is divided into zoning districts named and described in the following sections. The boundaries of said zoning districts are hereby established and shown on the Official Zoning Map maintained in the office of the Zoning Administrator and shown as originally adopted within Article 12, Zoning Map et seq.

#### 402.00 INCORPORATION OF THE ZONING MAP

The zoning map entitled the "Official Zoning Districts Map for the Town of Broadway, Virginia", dated July 5, 1988, as amended from time to time since that time, hereinafter referred to as the Official Zoning Map, with all notations, references, amendments, and dates thereof, and other information shown thereon, shall constitute a part of this Ordinance. Said map shall be made a public record and shall be kept permanently in the office of the Zoning Administrator, where it shall be accessible to the general public.

#### 403.00 MAP AMENDMENT

If in accordance with the provisions of Article 8, herein, changes are made in the district boundaries or other information portrayed in the Official Zoning Map, such changes shall be entered on the Official Zoning Map within ten (10) days after the amendment has been approved, together with a numerical entry referring to the application for the amendment, submitted in accordance with Article 8, herein, which shall be kept as a public record by the Zoning Administrator. Said numerical entry shall state the reference number of the application in the records of the Zoning Administrator and the date of the approval of the amendment by the Governing Body. Amendments to this Ordinance, which involve matter portrayed on the Official Zoning Map, shall become effective immediately upon being entered onto the Official Zoning Map. The Town of Broadway Official Zoning Map, which shall be located in the office of the Zoning Administrator, shall be the final authority in determining the current zoning status of land and water areas, buildings, and other structures in the Town. No changes of any nature shall be made in the Official Zoning Map except in accordance with law.

#### 404.00 REPLACEMENT OF THE OFFICIAL ZONING MAP

In the event that the Official Zoning Map becomes damaged, destroyed, lost or difficult to interpret because of the nature or number of changes and additions, the Governing Body may, by resolution, adopt a new Official Zoning Map which shall supercede the prior Official Zoning

Map. The new Official Zoning Map may correct drafting or other errors or omissions in the prior Official Zoning Map, but no such corrections shall have the effect of amending the original Official Zoning Map or any subsequent amendment thereof. Unless the prior Official Zoning Map has been lost or has been totally destroyed, the prior map, or any significant parts thereof remaining shall be preserved, together with all available records pertaining to its adoption or amendment.

#### 405.00 RULES FOR DETERMINING BOUNDARIES

Unless district boundary lines are fixed by dimensions, and where uncertainty exists with respect to the boundaries of any of the aforesaid districts as shown on the Official Zoning Map, the following shall apply:

- 405.01 Unless otherwise indicated, district boundaries indicated as approximately following property lines, landlines, centerlines of streams, roads, highways, alleys, the shorelines of reservoirs, or other bodies of water or civil boundaries, shall be construed to follow such lines.
- District boundaries indicated as approximately parallel to the centerline of streams, roads, highways, or rights-of-way of the same, or the shorelines of reservoirs, or other bodies of water, or said lines extended, shall be construed as being parallel thereto and at such distance therefrom as indicated on the Official Zoning Map. If no distance is given, such dimensions shall be determined by the use of the scale shown on the Official Zoning Map.
- Where a public road, street, or alley is officially vacated or abandoned, the regulations applicable to the property to which it is reverted shall apply to such vacated or abandoned road, street, or alley.
- Where a district boundary is indicated to follow a river, creek, branch, or other body of water, said boundary shall be construed to follow the centerline at low water or at the limit of the jurisdiction, and in the event of change in the shoreline, such boundary shall be construed as moving with the actual shoreline with its reestablished center or channel.
- 405.05 If no distance, curvature description, or other means is given to determine a boundary line accurately and the foregoing provisions do not apply, the same shall be determined by the use of the scale shown on the Official Zoning Map. In case of subsequent dispute the matter shall be referred to the Board of Zoning Appeals, which shall determine the boundary in accordance with the provisions of this Ordinance.

## ARTICLE 9 SUBDIVISION REGULATIONS

#### 901.00 SUBDIVISION REQUIREMENTS

Under the authority to establish subdivision regulations under Virginia law, the regulations established herein constitute minimum requirements which shall apply to all subdivisions, except as hereinafter provided. This Subdivision Ordinance is part of a long range plan to guide and facilitate the orderly, beneficial growth of the community and to promote the public health, safety, convenience, comfort, and general welfare, and to implement the Town's Comprehensive Plan. More specifically, the purposes of the standards and procedures in this chapter are to provide a guide for the change that occurs when land becomes urban in character as a result of development for residential, business, or industrial purposes, to provide assurance that the purchasers of lots are buying a commodity that is suitable for development and use, and to make possible the provision of public services in a safe, adequate, and efficient manner.

**Ordinance of December 6, 2011** added the phrase "and to implement the Town's Comprehensive Plan" to the end of the second sentence of this section.

#### 902.00 GENERAL PROVISIONS

- Ordinance Mandatory. No person shall subdivide land, relocate or adjust the boundary line between any parcels of land, or vacate any boundary line, without making a plat of such subdivision, relocation, adjustment, or vacation, and recording it in the office of the Circuit Court of Rockingham County and without fully complying with the provisions of this Subdivision Ordinance.
- Approval Required. No such plat of land located in the Town shall be recorded unless and until it shall have been submitted to and approved by the appropriate municipal authorities of the Town, and, if the land in question is not situated entirely within the Town, by the appropriate municipal authorities of each county or municipality having a subdivision ordinance, in which any part of the land lies.
- 902.03 <u>Sales Prohibited Without Compliance</u>. No person shall sell or transfer any such land before such plat has been duly recorded as provided herein, unless such subdivision was lawfully created prior to the adoption of a subdivision ordinance applicable thereto.
- 902.04 <u>Penalty for Illegal Conveyance</u>. Any person violating the foregoing provisions of this section shall be guilty of a misdemeanor and upon due conviction thereof shall be subject to a fine of not more than five hundred dollars (\$500) for each lot or parcel of land so subdivided or transferred or sold; and the description of such

lot or parcel by metes and bounds in the instrument of transfer or other document used in the process of selling or transferring shall not exempt the transaction from such penalties or from the remedies herein provided. This section shall not be in lieu of any civil action to compel compliance with any provision of this Subdivision Ordinance. No building or Zoning permit shall issue in connection with any land divided or conveyed in violation of this Subdivision Ordinance.

- 902.05 <u>Recordation</u>. No clerk of any court shall file or record a plat of a subdivision required by this article to be recorded until such plat has been approved as required herein; and the penalties provided by Virginia law shall apply to any failure to comply with the provisions of this subsection.
- 902.06 <u>Improvements</u>. All required improvements shall be installed by the subdivider at his cost. Specifications and requirements set forth in this regulation shall be followed. The subdivider's performance bond shall not be released until construction has been inspected and approved by the appropriate official. Nothing herein shall be construed as creating an obligation upon the Town to pay for grading or paving, or for sidewalks, sewers, water systems, curb and gutter improvements, or other construction.
- Necessary Changes. No change, erasure or revision shall be made on any final plat, nor on accompanying data sheets after approval thereof has been endorsed in writing on the plat or sheets, unless authorization for such changes has been granted in writing by the party granting such approval.
- 902.08 Private Contracts. This Subdivision Ordinance bears no relation to any private easement, covenant agreement, or restriction, nor is the responsibility of enforcing such private easement, covenant, agreement or restriction implied herein to any public official. When this Subdivision Ordinance calls for more restrictive standards than are required by private contract, the provisions of this Subdivision Ordinance shall control.
- 902.09 Payment of Pro Rata Share of Certain Facilities. The subdivider or developer shall pay a pro rata share of the cost of providing reasonable and necessary water, sewerage and drainage facilities, located outside the property limits of the land owned or controlled by him but necessitated or required, at least in part, by the construction or improvement of his subdivision or development; provided, however, that no such payment shall be required until such time as the Town Council or a designated department or agency thereof shall have established a general water, sewer and drainage improvement program, in compliance with Virginia law, for an area having related and common water, sewer and drainage conditions and within which the land owned or controlled by the subdivider or developer is located. Each such payment received shall be expended only for the construction of those facilities for which the payment was required, and until so

expended shall be held in an interest-bearing account for the benefit of the subdivider or developer; in lieu of such payment other methods of performance guarantee satisfactory to the Governing Body shall be posted conditioned on payment at the commencement of such construction.

#### 903.00 LOTS TO WHICH ORDINANCE APPLICABLE; APPROVAL PROCESS

903.01 <u>General Applicability</u>. The term "subdivision" shall include any of the following:

- The division of any single parcel of land into not more than three (3) lots including the remnant, all of which front on an existing street, which do not involve any new road, the extension or installation of public facilities, or the creation of any public improvements, which do not adversely affect the remainder of the parcel or adjoining property (a "Minor Subdivision"). The term "Minor Subdivision" shall not include a subdivision if the Subdivision Administrator determines that a subdivider is using repeated or successive minor subdivisions in an attempt to evade the provisions of this Subdivision Ordinance regarding Major Subdivisions.
- The relocation or adjustment of any boundary line between lots, that does not result in the creation of a larger number of lots than existed before such relation or adjustment (a "Boundary Line Adjustment"); and
- 903.01-3 The division of any parcel or parcels of land into two (2) or more lots, when such division does not constitute a Minor Subdivision (a "Major Subdivision").
- Administrative Approval. An applicant may obtain administrative approval of certain plats. The Subdivision Administrator may approve such plats detailing the foregoing if he finds that such plats comply with all requirements of this Subdivision Ordinance, without approval from the Planning Commission or Town Council. If the Subdivision Administrator disapproves such plat, he shall transmit to the applicant a statement of the reasons for such disapproval, either by indications on the plat or separately. From the decision of the Subdivision Administrator approving or disapproving such plats, an appeal shall lie to the Circuit Court of Rockingham County in accordance with Virginia law. If he so desires, the Subdivision Administrator may seek advice from the Planning Commission on any proposed plat, and, in addition, may refer such plat to the Town Council for action instead of taking action himself, in which event all

With revisions through July 7, 2015 Land Development Regulations – IX-3 procedures shall apply as with a plat that does not qualify for administrative approval. Administrative approval is permitted for:

- 903.02-1 A Minor Subdivision, provided that no other application for a minor subdivision of the same parcel or parcels has been considered within five (5) years from the date of approval;
- A Boundary Line Adjustment that does not involve the relocation or alteration of streets, alleys, easements for public passage, or other public areas.
- 903.03 <u>Planning Commission and Council Approval</u>. Any plat not eligible for administrative approval shall be submitted for approval to the Planning Commission and Town Council in accordance with the provisions of this Subdivision Ordinance.
- Partition Suits. This Subdivision Ordinance shall not apply to the final decree of a court of competent equity jurisdiction that partitions land among the parties of a bona fide chancery suit for the partition of land within the Town, and such decree may be recorded without the approval of the Subdivision Administrator or Town Council. Notwithstanding the foregoing, this Subdivision Ordinance shall apply if the Subdivision Administrator finds that such decree was procured by parties acting collusively in order to evade the requirements of this Subdivision Ordinance. Furthermore, all requirements of the Zoning Ordinance shall apply to any lot created by a decree exempt from the application of this Subdivision Ordinance under this paragraph.

#### 904.00 PROVISIONS REGARDING LOTS

- 904.01 Lot Size and Width. No lot shall be created (through subdivision, relocation or adjustment of a boundary line, or vacation of a boundary line) that does not meet all the requirements of the Zoning Ordinance for the zoning district in which the lot is situate, including provisions therein regarding the minimum lot size and lot width requirements.
- 904.02 <u>Lot Shape</u>. Lots shall not contain peculiarly shaped elongations solely to provide necessary square footage by adding area which would be unusable for normal purposes.
- 904.03 Remnants. Land subject to flooding and all remnants of lots below minimum size left over after subdividing a tract must be added to adjacent lots, or may become the property of a homeowners association, but may not in any event be allowed to remain as unusable parcels.

### 905.00 PROVISIONS REGARDING STREETS AND RELATED PUBLIC IMPROVEMENTS

905.01 <u>Street Orientation</u>. The following regulations shall apply:

Dot Shall Abut on a Street. Each lot (except a lot on which a townhouse will be constructed) shall abut on a developed public street dedicated by the subdivision plat, or on a street which has become public by right of use. If the existing streets are not fifty (50) feet in width, the subdivider shall make provisions in the deeds to the lots for all buildings to be so constructed as to permit the widening by dedication of said roads or streets to a width of fifty (50) feet. No lot shall embrace any portion of a street or alley.

Blocks. Except in Minor Subdivisions, streets shall be designed in such a way as to allow blocks with a maximum length of six hundred (600) feet, a maximum total perimeter length of sixteen hundred (1,600) feet, and the minimum length of blocks upon which lots have frontage of three hundred (300) feet. Such blocks shall be wide enough to allow two tiers of lots of minimum depth, except where fronting on major streets.

**Ordinance of December 6, 2011** amended this section to change the phrase "twelve hundred (1200)" to "six hundred (600), to add the phrase ", a maximum total perimeter

length of sixteen hundred (1,600) feet, and" immediately after that phrase, and to change "five hundred (500)" to "three hundred (300)".

905.01-3 Street Alignment and Layout. The arrangement of streets shall make provision for the continuation of existing streets in adjoining areas. The street arrangement must be such as to cause no unnecessary hardship to owners of adjoining property when they plat their own land and seek to provide for convenient access to it. Where it is called for in the Comprehensive Plan, or would further the orderly development of the Town for adjacent or contiguous subdivisions, whether existing or contemplated, to provide for street access to adjoining property, proposed streets shall be extended by dedication to the boundary line of such property. Half streets along the boundary of land proposed for subdivision are not permitted. Wherever possible, streets should intersect at right angles.

Ordinance of December 6, 2011 added the phrase "is called for in the Comprehensive

Plan, or" after the phrase "where it" in the third sentence of this section.

905.01-4 <u>Alleys</u>. The creation of new alleys is permitted.

**Ordinance of December 6, 2011** deleted the word "not" before the word "permitted" in this section.

- 905.01-5 <u>Reserve Strips</u>. There shall be no reserve strips controlling access to public streets.
- Cul-De-Sacs. Dead-end streets are prohibited unless designed as 905.01-6 cul-de-sacs, in accord with VDOT standards. The number or percentage of dead end streets in any subdivision shall not exceed the number that will produce a connectivity measurement meeting the standards set in VDOT's Secondary Street Acceptance Standards (SSARs)... Streets designed to have one end permanently closed must be terminated by a turn-around of not less than one hundred (100) feet in diameter. The street shall not exceed 500 feet in length and shall not furnish access to more than 20 dwelling units. Drainage of cul-de-sac streets shall preferably be towards the open end unless such requirement is not practical. If drainage is toward the closed end, adequate provisions shall be made for runoff to be carried away. Drainage easements shall be required where necessary.

Ordinance of December 6, 2011 added the phrase ", in accord with VDOT standards. The number or percentage of dead end streets in any subdivision shall not exceed the number that will produce a connectivity measurement meeting the standards set in

VDOT's Secondary Street Acceptance Standards (SSARs)." to the end of the first sentence, and changed the phrase "one hundred (100)" to "eighty (80)" in the second sentence of this section.

- 905.01-7 <u>Intersections</u>. In all hillside areas, streets running with contours shall be required to intersect at angles of not less than sixty (60) degrees.
- 905.01-8 <u>Curve Radii</u>. The minimum curve radius at intersections of local streets shall be ten (10) feet, and twenty-five (25) feet for major streets.

Ordinance of December 6, 2011 added this section.

- 905.02 Street Design. The following regulations shall apply:
  - 905.02-1 <u>Street Width</u>. For all subdivisions, all streets shall have, as a minimum, a right-of-way width of fifty (50) feet and a pavement width of thirty (30) feet except that pavement width may be a 26 feet for streets that will carry less than 200 vehicles per day.

**Ordinance of December 6, 2011** added the phrase "except that pavement width may be a 26 feet for streets that will carry less than 200 vehicles per day." to the end of the first sentence of this section.

**Ordinance of July 7, 2015** repealed the final sentence of this section as originally enacted. The former language read: "The

foregoing sentence shall not apply to a private access easement for townhouses, but in lieu of the foregoing sentence, a proposed subdivision for townhouses shall comply with all the requirements of the Zoning Ordinance regarding private access easements for townhouses."

- Onstruction and Surfacing. All streets and alleys, subgrade and base material, shall be developed according to standards in the Virginia Department of Transportation Highway's Street Standards. Streets shall be surfaced in accordance with State standards. The subdivider shall construct and surface all platted streets to the width and in accordance with the requirements of this Subdivision Ordinance.
- Ourbs and Gutters. Concrete curbs and gutters shall be installed on both sides of all streets shown on the subdivision plat in accordance with State standards. In the case of newly platted streets in a subdivision in which each lot contains one acre or more, curb and gutter is not required, but the development plan must nevertheless provide satisfactory storm drainage provisions. Curbs, gutters, and any required sidewalks shall be constructed such that there will be a driveway entrance for each lot not being used for a townhouse.
- 905.02-4 Street Names. Proposed streets which are obviously in alignment with other already existing and named streets shall bear the names of the existing streets. The name of a proposed street shall not duplicate or confuse an existing street in the town or in the postal district. The use of the following suffixes shall constitute a duplication if used after a name which is repeated before each suffix: street, road, avenue, boulevard, drive, way, place, court, lane, etc. Street names shall be indicated on the preliminary and final plats, and shall be approved by the Town. Names of existing streets shall not be changed except by specific approval.
- Monuments. Upon completion of subdivision streets, sewers and other improvements, the subdivider shall make certain that all monuments required by this regulation are clearly visible for inspection and use. Such monuments shall be inspected and approved by the administrator before any improvements are accepted. All street corners, all points where the street line intersects the exterior boundaries of the subdivision, at right angle points, and points of curve in each street, and all other lot corners shall be marked with iron pipe not less than

three-fourths (3/4) inch in diameter and twenty-four (24) inches long and driven so as to be flush with the finished grade. When rock is encountered, a hole shall be drilled four (4) inches deep in the rock into which shall be cemented a steel rod one-half (1/2) inch in diameter, the top of which shall be flush with the finished grade line.

#### 905.04 <u>Related Required Improvements</u>. The following improvements shall be required:

- 905.04-1 <u>Sidewalks</u>. Concrete sidewalks shall be provided on one (1) or both sides of all new streets. Furthermore, when a new lot is created that fronts on an existing street, and adjacent property on either side has an existing sidewalk, the applicant shall construct a sidewalk that connects to the existing sidewalk.
- 905.04-2 <u>Street Identification Signs</u>. Street identification signs of an approved design shall be installed at all intersections.
- 905.04-3 <u>Lighting</u>. In Major Subdivisions, streetlights of adequate type and intensity shall be required to promote public health and safety in any subdivision. Streetlights shall be provided at all intersections. In order to ensure compliance with the foregoing, the design proposal for such streetlights shall be submitted with the preliminary or final plat of the proposed subdivision.
- 905.04-4 <u>Railroad Crossing</u>. In Major Subdivisions, the applicant shall be responsible for providing flashing lights and short-arm gates for any new or improved road within a subdivision which may cross a railroad track.

#### 906.00 UTILITY AND STORMWATER REQUIREMENTS

- Plans and Specifications for Utility Fixtures and Systems to be Submitted for Approval. If the applicant for any subdivision desire to construct in, on, or under any streets or alleys located in such subdivision any gas, water, sewer, or electric light or power works, pipes, wires, fixtures, or systems, then with the submission of their plat they shall present plans or specifications therefore, which shall be considered as part of the consideration of such plat.
- 906.02 <u>Public Water and/or Sewer Required</u>. Public water and sewer service shall be extended to all lots within a subdivision by the developer. Such systems shall be designed to connect with such systems already in existence in accordance with state department of health standards. Such systems shall furnish an adequate supply of water to each lot, with adequate main sizes and fire hydrants located to

meet the specifications of the insurance services office. The distance between sewer manholes shall not exceed three hundred (300) feet. A copy of the approval of such system by the appropriate public agency or utility company shall be submitted with the final plat. Any such property with existing septic tanks must connect to the public systems already in existence in accordance with state department of health standards, immediately after the failure of the individual septic tank.

- 906.03 <u>Fire Protection</u>. The installation of adequate fire hydrants in a subdivision at approved locations is required, provided that the necessary water supply is available.
- 906.04 Flood Control. If any portion of the proposed subdivision is determined by the agent to be in the one hundred (100) year floodplain, the subdivider shall provide the necessary information to demonstrate that the presence of the one hundred (100) year floodplain was considered in the layout of the subdivision. The subdivider shall also provide the plans for meeting the statewide stormwater management criteria, and, if necessary, and provisions of the Zoning Ordinance regarding flood control. The Flood Control information shall include a properly certified engineer's statement that such improvements, when properly installed will be adequate to meet the criteria as applied to the proposed development.
- 906.05 <u>Storm Drainage</u>. In order to provide adequate provisions for drainage control, the following rules shall apply:
  - 906.05-1 <u>Existing Storm Sewers</u>. Where existing storm sewers are reasonably accessible, the proposed subdivision shall connect with such storm sewers.
  - 906.05-2 <u>New Storm Sewers</u>. New storm sewers and related installations shall be required only when the runoff of stormwater cannot be satisfactorily handled within the street pavement.
  - Design of Storm Drainage Facilities. Storm drainage facilities shall be designed to avoid the concentration of stormwater runoff onto adjacent developed or undeveloped properties or the collection of water at low points in the subdivision and along streets. Water shall be drained away from onsite sewage disposal facilities. Storm drainage facilities should be designed not only to handle the anticipated peak discharge from the property being subdivided, but also the anticipated increase in runoff that will occur when all the property at a higher elevation in the watershed is fully developed.

- Watercourses. Where a subdivision is traversed by a watercourse, drainage way, channel, or stream, there shall be provided a drainage easement conforming substantially with the line of such watercourse, drainage way, channel, or stream and of such width as will be adequate to preserve the unimpeded flow of natural drainage, or for the purpose of widening, deepening, relocating, improving, or protecting such drainage facilities. Any changes in the existing drainage way shall be subject to the approval of the state water control board.
- 906.05-5 <u>Street Design</u>. All streets shall be so designed as to provide for the discharge of surface water from their rights-of-way.
- 906.05-6 <u>Certification of Required Improvements</u>. The applicant shall provide for all such drainage improvements together with a certified engineer's or surveyor's statement that the contemplated improvements (or lack thereof) will be adequate for proper development. The highway resident engineer shall then approve or disapprove the plans.
- 906.05-7 <u>Maintenance of Systems, etc.</u> All stormwater detention, retention, and conveyance systems and facilities must be conveyed to, and maintained by a homeowners association or other entity approved by the agent approving the plat.
- 906.05-8 <u>Stormwater and Sewer Systems</u>. Sanitary sewers shall not be used to carry stormwater.
- 906.06 <u>Easements</u>. The Town may require that easements for drainage through adjoining property be provided by the subdivider. When necessary to provide access, easements of not less than ten (10) feet in width shall be provided for water, sewer, power lines, and other utilities to serve the subdivision. Easements may also be required in, along or adjacent to natural water courses as drains for sanitary sewers and water diversion purposes.

#### 907.00 RELATION TO EROSION AND SEDIMENT CONTROL LAWS

907.01 <u>Incorporation of Rockingham County Ordinance</u>. The Town of Broadway adopts the Erosion and Sediment Control Ordinance of Rockingham County for all subdivisions except Minor Subdivisions that disturb less than 10,000 square feet of land.

907.02 <u>Plan.</u> In any instance in which the Rockingham County Erosion and Sediment Control Ordinance is applicable, an applicant shall submit with his or her final

Control Ordinance is applicable, an applicant shall submit with his or her final plat a valid and approved Rockingham County Erosion and Sediment Control Plan.

#### 908.00 PRELIMINARY SKETCHES AND PLATS

908.01 Preliminary Sketch. The subdivider may, if he or she so chooses, submit to the

Subdivision Administrator a preliminary sketch of the proposed subdivision prior to his preparing detailed preliminary and final plats. The purpose of such preliminary sketch is to permit the Administrator to advise the subdivider whether his plans, in general, are in accordance with the requirements of this Ordinance. Upon submission of any such preliminary sketch, it shall be studied and the subdivider advised where it appears that changes would be necessary. The agent may mark the preliminary sketch indicating necessary changes and any such marked sketch shall be submitted with the preliminary plat. The preliminary sketch shall be as follows: It shall be drawn on white paper, or on a print of a topographic map of the property. It shall be drawn to an appropriate scale i.e., one hundred (100) feet to the inch. It shall show the same location and dimensions of all streets entering the property, adjacent to the property, or terminating at the boundary of the property to be subdivided. It shall show the location of all proposed streets, lots, parks, playgrounds, and other proposed uses

of the land to be subdivided and shall include the approximate dimensions.

Preliminary Plat. In the case of every subdivision that is not a Minor Subdivision or a Boundary Line Adjustment, the subdivider shall present to the Subdivision Administrator four (4) copies of a preliminary layout at an appropriate scale of at least one hundred (100) feet to the inch. In the case of a Minor Subdivision or a Boundary Line Adjustment, the subdivider may (but shall not be required to) present a preliminary plat to the Subdivision Administrator, who may consider it and approve or disapprove it as provided herein without the involvement of the Planning Commission. All preliminary plats shall include the following

information:

908.02

908.02-1 Name of subdivision, owner, subdivider, surveyor, or engineer,

date of drawing, number of sheets, north point, and scale;

Docation of proposed subdivision by an inset map at a scale of not less than one inch equal to two thousand (2,000) feet showing adjoining roads, their names and number, towns, subdivisions, and

other landmarks;

- 908.02-3 The boundary survey or existing survey of record provided such survey shows a closure with an accuracy of not less than one in twenty-five hundred; total acreage, acreage of subdivided area, number and approximate area and frontage of all building sites, existing buildings within the boundaries of the tract, names of owners and their property lines within the boundaries of the tract, and adjoining such boundaries;
- All existing, platted, and proposed streets, their names, numbers, and widths; existing utility or other easements, public areas, and parking spaces; culverts, drains, and water courses, their names and other pertinent data;
- 908.02-5 All parcels of land to be dedicated for public use and the conditions of such dedication;
- 908.02-6 Topography at an appropriate interval; and for this purpose, the Subdivision Administrator may require the subdivider to furnish topographic maps, floodplain profile elevation information or other relevant information;
- 908.02-7 Elevations of existing and proposed ground surface at all street intersections and at points of major grade change along the center line of streets together with proposed grade lines connecting therewith;
- 908.02-8 Proposed connections with existing sanitary sewers and existing water supply. The location, type, profile, percentage of slope, pipe size, and location of manholes for all sewers shall be shown. The location, type and sizes of all waterlines shall be shown as well as the location of necessary control valves and fire hydrants;
- Provisions for collecting and discharging surface drainage and preliminary designs of any structure that may be required.
- Procedure for Consideration. The Planning Commission and agent shall discuss the preliminary plat with the subdivider in order to determine whether or not his preliminary plat generally conforms to the requirements of the Land Development Code. The subdivider shall then be advised in writing, which may be by formal letter or by legible markings on his copy of the preliminary plat, concerning the approval or disapproval of the preliminary plat, any additional data that may be required, the character and extent of public improvements that will have to be made, and the amount of the performance bond which will be required as a prerequisite to approval of the final subdivision plat. In the case of a disapproval,

the written advice shall describe specifically the reasons for disapproval and shall state what corrections or modifications would render the plat acceptable.

- 908.04 <u>Time Limits for Consideration</u>. The Planning Commission or Subdivision Administrator shall act on all preliminary plats (including any required or convenient submission of plats to other agencies) within the maximum time limits prescribed in Virginia law, taking into account any tolling periods provided in Virginia law and the time periods during which other agencies may comments on such plats.
- Amount of Bond. In determining the cost of required improvements and the amount of the performance bond, the Subdivision Administrator shall require an amount equal to the estimate of the cost of improvements to be furnished by the subdivider.
- 908.06 No Guarantee. Approval of the preliminary plat does not constitute a guarantee of approval of the final plat.
- 908.07 Six Months Limit. The subdivider shall have not more than six (6) months after receiving official notification concerning the preliminary plat to submit to the Subdivision Administrator a final subdivision plat in accordance with this Subdivision Ordinance. Failure to do so shall make preliminary approval null and void. The Subdivision Administrator may, on written request by the subdivider, for good cause shown, grant an extension of this time limit.
- 908.08 Review by Engineer. The Subdivision Administrator may, at his option, submit a preliminary plat to a professional engineer of his choice, retained by the Town, to obtain review of the plat by such engineer and such engineer's advice thereon. The proponent of the subdivision shall bear all fees associated with such engineer's review, regardless whether the subdivision is ultimately approved.

## 909.00 REQUIREMENTS FOR FINAL PLATS

- Open 209.01 Characteristics of Final Plats. Four (4) copies of the final plat shall be submitted to the Subdivision Administrator. The subdivision plats submitted for final approval and subsequent recording shall be clearly and legibly drawn in ink upon tracing cloth at an appropriate scale, i.e., one hundred (100) feet to the inch on sheets not exceeding seventeen (17) inches by twenty-two (22) inches in size. When a subdivision cannot be platted on sheets of this size, it is suggested that it be platted in sections, numbering the sections numerically, as Section 1, 2, etc., of \_\_\_\_\_\_ subdivision.
- 909.02 <u>Requirements</u>. A final plat should contain at least the following information:

- Name of subdivision, ward, town, state, owner, north point, scale of drawing, and number of sheets. If shown on more than one (1) sheet, match lines shall clearly indicate where the several sheets join. A space containing the Certificate of Approval shall be provided for the use of the approving authority, containing spaces for the endorsement of the Subdivision Administrator and any other official whose approval is necessary, as well as a Certification by a surveyor or professional engineer, or both, that all requirements of this Subdivision Ordinance have been met to the best of his knowledge and belief;
- Docation of proposed subdivision by an insert map, at a scale of not less than one (1) inch equals two thousand (2,000) feet, indicating adjoining roads, their names and numbers, towns, subdivisions, and other landmarks;
- A boundary survey with an error of closure within the limits of one (1) in five thousand (5,000) related to the true meridian and showing the location of all monuments and their type of material. The survey may be related to the State Plan Coordinate grid, if the Coordinates of two (2) adjacent corners of the subdivision are shown;
- A statement to the effect that the subdivision as it appears on this plat is with the free consent and in accordance with the desires of the owners, proprietors, and trustees, if any, which shall be signed by the owners, proprietors, and trustees, if any, and shall be duly acknowledged before some officer authorized to take acknowledgements of deeds;
- When the subdivision consists of land acquired from more than one source of title, the outlines of the various tracts shall be indicated by dashed lines, and identification of the respective tracts shall be placed on the plat;
- When any existing boundary line is being vacated, the location of such boundary line shall be indicated by a dashed line and a notation that such line is hereby vacated;
- 909.02-7 The accurate location and dimensions by bearings and distances with all curve data on all lots and street lines and center lines of streets; boundaries of all proposed or existing easements, parks, school sites, or other public areas; the number and area of all

building sites; all existing public and private streets, their names, numbers, and widths; existing utilities, and those to be provided such as sanitary sewers, storm drains, water mains, manholes, and underground conduits including their size and type; water courses and their names, names of owners and their property lines, both within the boundary of the subdivision and adjoining said boundaries;

All dimensions shown shall meet the standards published by the State Board of Licensing;

909.02-9 The data of all curves along the street frontages shall be shown in detail at the curve or in a curve data table containing the following: Delta, radius, arc, tangent, chord, and chord bearings.

## 910.00 CONSIDERATION OF FINAL PLATS

Action on Plat. The Town Council (or, for plat on which administrative approval is permitted under this Subdivision Ordinance, the Subdivision Administrator) shall act on proposed final plats within the maximum period permitted under state law after it has been officially submitted for approval (taking into account any tolling periods or extensions permitted under state law) by either approving or disapproving such plat in writing, and giving with the latter specific reasons therefore. The specific reasons for disapproval may be contained in a separate document or may be written on the plat itself, and shall relate in general terms such modifications or corrections as will permit approval of the plat.

Time Limit. The subdivider shall have not more than six (6) months after receiving final approval to file the subdivision plat for recordation. If a plat is not filed for recordation within the time limit such approval shall be withdrawn and the plat marked void and returned. Notwithstanding the foregoing, in any case where construction of facilities to be dedicated for public use has commenced pursuant to an approved plan or permit with surety approved under the terms herein, or where the developer has furnished surety to the Town by certified check, cash escrow, bond, or letter of credit in the amount required herein, the time for plat recordation shall be extended to one year after final approval or to the time limited specified in the surety agreement, whichever is greater.

910.03 Required Surety. Before the acceptance of dedication for public use of any right-of-way located within any subdivision which has been constructed or proposed to be constructed within the subdivision, any street, curb, gutter, sidewalk, bicycle trail, drainage or sewerage system, waterline as part of a public system, or other improvement dedicated for public use or for the provision of any

other site-related improvement required for vehicle ingress and egress, including traffic signalization and control, for public access streets, for structures necessary to ensure stability of critical slopes, and for storm water management facilities, in any case financed or to be financed in whole or in part by private funds, the owner or developer must: (1) certify to the Town that the construction costs have been paid to the person constructing such facilities; or (2) furnish to the Subdivision Administrator a certified check, letter of credit, cash escrow, or contract for construction in the amount of the estimated costs of construction or a personal, corporate or property bond, with surety satisfactory to the Subdivision Administrator, in an amount described in § 907.05, and conditioned upon the construction of such facilities.

910.04

Additional Bond for Highway. Should the Town have accepted the dedication of a road for public use and such road is not acceptable into the State Highway System due to factors other than its quality of construction, the Town may require the subdivider or developer to furnish a maintenance and indemnifying bond or letter of credit with surety satisfactory to the Subdivision Administrator, in an amount sufficient for, and conditioned upon, the maintenance of such road until such time as it is accepted into the State Highway System.

910.05

<u>Recordation</u>. Without further action by the Town, the recordation of such plat shall operate to transfer, in fee simple, to the Town of Broadway such portion of the premises platted as is on such plat set apart for streets, alleys, or other public use and to transfer to the Town any easement indicated on such plat to create public right of passage over the same or to provide for the conveyance of stormwater, domestic water, and sewage, including the installation and maintenance of any facilities utilized for such purposes.

910.06

<u>Endorsement</u>. The plat shall not be approved until the subdivider has complied with the general requirements and minimum standards of design in accordance with this Ordinance, and has made arrangements for performance bonds satisfactory to the Subdivision Administrator. Approval of the final plat shall be written by the agent on the face thereof.

## 911.00 VACATION OF PLAT OR BOUNDARY LINE

911.01

Administrative Approval. An applicant may obtain administrative approval from the Subdivision Administrator of a plat the sole purpose of which is to vacate a boundary line between one or more lots under common ownership, when such vacation does not involve the relocation or alteration of streets, alleys, easements for public passage, or other public areas. To obtain such approval, the applicant shall submit a plat that complies with the provisions of §§ 908.01 and 908.02. The Subdivision Administrator shall examine such plat to determine its

compliance with the foregoing sections and shall approve the plat if he finds that the vacation of such boundary lines does not worsen any nonconformity with the provisions of this Subdivision Ordinance.

911.02 <u>Vacation of Plat</u>. Any vacation of boundary lines or of a plat that does not qualify for administrative approval under § 911.01 may be vacated by any method permissible under Virginia law, provided, however, that no such vacation shall be recorded without the consent of the Town Council of the Town and such other parties as Virginia law requires, and without the due holding of any public hearings that Virginia law requires.

## 912.00 ADMINISTRATION

- 912.01 <u>Subdivision Administrator</u>. The Zoning Administrator shall serve as Subdivision Administrator. The Subdivision Administrator shall have all authority permitted by Virginia law to enforce the provisions of this Subdivision Ordinance, including, to the full extent permissible under Virginia law, the powers that the Zoning Administrator has to enforce the Zoning Ordinance.
- 912.02 <u>Variances to Subdivision Ordinance</u>. Upon application of a party proposing a subdivision, the Town Council may authorize a variance the provisions of this Subdivision Ordinance if it finds that strict adherence to the general regulations would result in substantial injustice or hardship. The Subdivision Administrator shall maintain a record of all variances granted to this Subdivision Ordinance, noting for each variance the reasoning that justified the departure from normal subdivision regulations. In granting variances, the Town Council may require such conditions as will in its judgment, secure substantially the objectives of the standards or requirements so varied or modified.
- Delinquent Taxes. The Subdivision Administrator shall not act on any application for a subdivision if Town property taxes on the property that is the subject of the application have not been paid.

# ARTICLE 5 APPLICATION OF ZONING REGULATIONS

The regulations established herein within each district shall be minimum regulations and shall be uniformly applied to each class of structure or land, except as hereinafter provided:

#### 501.00 USES

No building or land shall hereafter be used or occupied and no building or part thereof shall be erected, constructed, moved, or structurally altered except in conformity with the regulations herein specified for the district in which it is or is to be located.

- Permitted Uses. A permitted use is one which is allowed by right in the district in which the land is situated. Where the proposed use is permitted and is in accordance with other regulations herein, the Zoning Administrator shall issue a Zoning Permit without a public hearing and in accordance with the procedures of this Ordinance.
- 501.02 <u>Special Uses</u>. Where uses are permitted by special use permit, such uses shall require a Special Use Permit in lieu of a Zoning Permit. A Certificate of Occupancy is still required. Procedures for applying for a Special Use Permit shall be as described in Article 8.
- Accessory Use. An accessory use is a subordinate use, customarily incidental to and located upon the same lot occupied by a permitted main use. Unless an accessory use is specifically limited to certain zoning districts, the listing of an accessory use in this ordinance shall not be construed to limit other accessory uses not mentioned, so long as they are clearly accessory to the principal permitted use of land and do not create a threat to public health, safety, or the welfare of the community. In the case of any dispute over whether a use is a permitted accessory use, the burden of proof is on the landowner to show that these criteria are met. No accessory building shall be constructed upon a lot until the construction of the main building has actually commenced, and no accessory building shall be used (other than for temporary storage of construction materials) unless the main building on a lot is completed and used.
- 501.04 Ordinance Exclusive. No use shall be permitted in the Town unless it is an enumerated permitted use, a special use for which a valid Special Use Permit has issued, or an accessory use.

## 502.00 BUILDINGS

No building shall hereafter be erected, constructed, or altered so as to violate this Zoning Ordinance.

## 503.00 LOTS AND YARDS

No new lot or yard shall hereafter be created, nor shall any lot or yard existing at the time of enactment of this Ordinance be altered, nor shall any building or structure, whether new or existing be moved, so that lot width, depth, or area requirements; front, side, or rear yard requirements; or inner or outer court requirements; or other requirements of this Ordinance are not maintained, except when a portion of a lot is acquired for public use. No part of a yard or other open space required for any building for the purpose of complying with the provisions of this Ordinance shall be included as part of a yard or other open space similarly required for another building. Every part of a required yard or court shall be open from its lowest point to the sky unobstructed, except as otherwise permitted herein.

## 504.00 GARDENING

Gardening shall be exempt from zoning permit requirements in any district allowing residential uses provided that such gardening shall not be objectionable by reason of odor, dust, noise, pollution, soil erosion, sedimentation, or drainage.

## 505.00 PERMITS ISSUED PRIOR TO ADOPTION OF ORDINANCE

Nothing contained herein shall require any change in the plans or construction of any building or structure for which a permit was granted prior to the effective date of this Ordinance. However, if such construction does not commence within thirty (30) days after this Ordinance becomes effective, or if construction is discontinued for a period of six (6) months or more, further construction shall be in conformity with the provisions of this Ordinance for the district in which the operation is located.

## 506.00 EXISTING NON-CONFORMING LOTS, BUILDINGS, AND USES

506.01

It is the intent of this Ordinance to recognize that the elimination of existing lots, buildings, and structures or uses that are not in conformity with the provisions of this Ordinance is as much a subject of health, safety, and general welfare as is the prevention of the establishment of new uses that would violate the provisions of this Ordinance. It is, therefore, the intent of this Ordinance to permit these non-conformities to continue, but not to encourage their survival, permit their

expansion, or permit their use as grounds for adding other structures or uses prohibited elsewhere in the same district. Therefore, any structure or use of land existing at the time of the enactment of this Ordinance, and amendments thereto, but not in conformity with its regulations and provisions, may only be continued subject to the provisions herein.

506.02 <u>Lots of Record</u>. Where a lot of record at the time of enactment of this Ordinance does not contain land of sufficient area or width to permit conformity with the dimensional requirements of this Ordinance, the following provisions shall apply:

- Two or More Lots Not Zoned R-1. When two (2) or more adjoining and vacant lots with continuous frontage, excluding those zoned R-1, are in single ownership at the time of enactment of this Ordinance or amendments thereto, and each of such lots have a width or lot area less than is required by the district in which they are located, but taken together such lot would conform to such standards, such lots shall be platted and resubdivided so as to create one (1) or more lots which conform to the minimum lot width and area requirements of the district prior to use or transfer of ownership of such parcels.
- Two or More Lots Zoned R-1. When two (2) or more adjoining and vacant lots with continuous frontage on land zoned R-1 are in single ownership at the time of enactment of this Ordinance or amendments thereto, such lots shall be platted and reparcelled so as to create one (1) or more lots which conform to a minimum frontage at setback line of seventy-five (75) feet and a minimum area requirement of nine thousand seven hundred fifty (9,750) square feet prior to use or transfer of ownership of such parcels.
- Single Nonconforming Lots. Where a single non-conforming lot of record at the time of enactment or amendment of this Ordinance is not of continuous frontage with other lots in the same ownership, such lot may be used as a building site, provided notwithstanding the failure of the lots to meet frontage or lot size requirements, nevertheless yard dimensions, and requirements other than those applying to area or width of the lot shall conform to the regulation for the district in which such lot is located.
- Nonconforming Structures. Where a lawful structure exists at the time of enactment or amendment of this Ordinance that could not be built in the district in which it is located by reason of restrictions on lot coverage, height, yard dimensions, or other requirements, such structure may be continued so long as it remains otherwise lawful, subject to the provisions of this article.

- Nonconforming Unsafe Structures. Any structure or portion thereof declared unsafe by the Building Inspector may be restored to a safe condition, provided that the requirements of this section are met, and that the cost of restoration of the structure to a safe condition shall not exceed fifty (50) percent of its replacement cost at the time of the Building Inspector declaration.
- 506.03-2 <u>Enlargement of Nonconforming Structures</u>. No nonconforming structure may be enlarged or altered in any way which increases its non-conformity. Any structure or portion thereof may be altered to decrease its non-conformity.
- Installation of Plumbing in Nonconforming Structure.

  Notwithstanding the provisions of Section 506.04-2 above, whenever repairs on or installation of plumbing fixtures in residential structures is required by law or administrative action of the Health Official or the Building Inspector, such alterations shall be permitted, provided that where such alterations require an addition to the structure, such addition shall be no nearer the lot line than permitted by the requirements of this Ordinance. If such addition cannot be made in accordance with the proceeding sentence, then the said addition shall extend no nearer the lot line than the existing building line.
- Movement of Nonconforming Structures. If a nonconforming structure is moved, it shall be moved only to a location where it conforms to the yard dimension requirements of the district in which it is located after it is moved.
- 506.03-5 Damage or Destruction of Nonconforming Structure by Natural Disaster or Act of God. If a nonconforming structure or nonconforming portion of a structure is damaged or destroyed by natural disaster or other act of God, it shall be repaired, rebuilt, or replaced to eliminate or reduce the nonconforming features to the full extent possible. If such building cannot be repaired, rebuilt or replaced except to restore it to its original nonconforming condition, the owner shall have the right to do so. The owner shall apply for a building permit and any work done to repair, rebuild or replace such building shall be in compliance with the provisions of the Uniform Statewide Building Code and any work done to repair, rebuild or replace such building shall be in compliance with the provisions of Section 641, which was adopted as a condition of participation in the National Flood Insurance Program. Unless

such building is repaired or rebuilt within two years of the date of the natural disaster or replaced within two years of the date of the natural disaster or other act of God, such building shall only be repaired, rebuilt or replaced in accordance with the provisions of this Ordinance. However, if the nonconforming structure is in an area under a federal disaster declaration and the building has been damaged or destroyed as a direct result of conditions that gave rise to the declaration, then this paragraph shall be read by substituting the term "four years" for the term "two years" in every place it occurs in this paragraph.

- Damage or Destruction of Nonconforming Structure other than by Natural Disaster or Act of God. If a nonconforming structure is damaged or destroyed other than by a natural disaster or by Act of God, it shall be repaired, rebuilt, or replaced only in compliance with the provisions of this Ordinance.
- Nonconforming Uses of Land. Where a lawful use of land (that is, a use not carried on inside a structure) exists at the time of enactment or amendment of this Ordinance that would not be permitted by the regulations imposed herein and where such is either (1) an accessory use involving the use of no separate accessory structure or (2) a principal use involving no individual structure, then such use may be continued as long as it remains otherwise lawful, subject to the provisions of this article.
  - 506.04-1 <u>Extension of Nonconforming Use</u>. No such nonconforming use shall be enlarged or increased nor extended to occupy a greater area of land than was occupied at the time of enactment or amendment of this Ordinance.
  - Return to Nonconforming Use Prohibited. Such nonconforming use, if changed to a conforming use, shall not thereafter to changed back to any nonconforming use.
  - Relocation of Nonconforming Use on Same Lot. No such nonconforming use shall be moved in whole or in part to any portion of the lot or parcel other than that occupied by such use at the time of enactment or amendment of this Ordinance.
  - Cessation of Nonconforming Use. In the event that such use, other than agriculture, ceases for a period of more than two years (including cessations by reason of a natural disaster or Act of God), any subsequent use shall conform to all requirements of this Ordinance for the district in which the land is located.

Additional Structures in Support of Nonconforming Use. No additional structure not conforming to the requirements of this Ordinance shall be constructed in connection with such nonconforming use.

Nonconforming Annexed Agricultural Land. To the extent that the Town annexes any land under its agreement with Rockingham County dated January 1, 1991, and such land is being used for agricultural purposes at the time of such annexation ("Nonconforming Annexed Agricultural Land"), such agricultural purposes are an existing nonconforming use of land pursuant to this section 506, and shall be subject to the provisions herein. Notwithstanding the foregoing, such Nonconforming Annexed Agricultural Land shall be subject to the following provisions:

- (a) Nonconforming Annexed Agricultural Land shall be eligible for land use taxation.
- (b) Crop fields and grazing areas on Nonconforming Annexed Agricultural Land shall be exempt from general ordinances of the Town regarding mowing.
- (c) Owners of agricultural operations on Nonconforming Annexed Agricultural Land shall be permitted to erect, rebuilt, repair, and replace farm outbuildings so long as the use does not increase any nuisance to neighboring permitted uses.

Nonconforming Uses of Structure. Where a lawful use involving an individual structure or structures in combination, exists at the time of enactment or amendment of this Ordinance, that would not be permitted in the district in which it is located under the requirements of this Ordinance, such use may be continued as long as it remains otherwise lawful, subject to the provisions of this article.

Expansion of Structure. No structure existing at the time of enactment or amendment of this Ordinance devoted to a nonconforming use shall be enlarged, extended, moved, or structurally altered, except repairs on or installation of plumbing fixtures required by law or administrative action of the Health Official or the Building Official, or the changing of interior partitions or interior remodeling; or in changing the use of the structure to a conforming use.

- 506.05-2 <u>Expansion Within Structure</u>. A nonconforming use of a structure may be extended to include use of the entire structure, but shall not be extended to include either additional structures or land outside the structure.
- Discontinuance of Use. When a nonconforming use of a structure or structures and premises in combination is discontinued or abandoned (including by reason of the destruction of such structure by natural disaster or Act of God) for two years the structure and premises shall not thereafter be used except in conformity with the regulations of the district in which it is located.
- Return to Nonconforming Use Prohibited. When a nonconforming use of a structure is changed to a conforming use, such use shall not thereafter to changed back to any nonconforming use.
- Change from Nonconforming Use of Structure to Another Nonconforming Use. Upon the issuance of a Special Use Permit, any nonconforming use of a structure or structures and premises in combination may be changed to another nonconforming use if the Town Council finds that the proposed use is equally appropriate or more appropriate to the district than the existing use. The Town Council may impose such conditions relating to the proposed use as it may deem necessary in the public interest and may require a cash bond or equivalent, a surety bond of a surety company, or a certified check payable to the Treasurer of the Town of Broadway, be supplied in an amount equal to the estimated cost of complying with the conditions imposed to insure that the conditions imposed are being and will continue to be met.
- Nonconforming Manufactured Homes. Notwithstanding any other provision herein, the land owner or home owner of any manufactured home that is an existing nonconforming structure, or is an existing nonconforming use in the zoning district in which it is located, may remove a valid nonconforming manufactured home from a mobile or manufactured home park and may replace that home with another comparable manufactured home that meets the current HUD manufactured housing code. In such mobile or manufactured home park, a single-section home may replace a single-section home and a multi-section home may replace a multi-section home. The owner of a valid nonconforming mobile or manufactured home not located in a mobile or manufactured home park may replace that home with a newer manufactured home, either single- or multi-section, that meets the current HUD manufactured housing code. Any such replacement home shall retain the valid existing nonconforming status of the prior home.

506.07

Condominium Conversions. When a landowner proposes to convert an existing nonconforming structure, or an existing structure with a nonconforming use, to condominium status, the landowner shall obtain a Special Use Permit before such conversion. Notwithstanding any other provision herein, the Town Council shall grant the Special Use Permit if the applicant can demonstrate to the reasonable satisfaction of the local authority that the nonconformities are not likely to be adversely affected by the proposed conversion. No action on any such request shall be unreasonably delayed. In the event of an approved conversion to condominium ownership, the Town will impose such charges and fees as it lawfully imposes as a result of construction of new structures to the extent that such charges and fees, or portions of such charges and fees, imposed upon property subject to such conversions may be reasonably related to greater or additional services provided by Town as a result of the conversion.

## 507.00 PRIVATE ACCESS EASEMENTS

Each lot in use shall abut on a developed public street, except lots used as a townhouse, which may employ a private access easement in accordance with Section 708.03. If a lot presently in use does not abut on a developed public street, any use of such lot is an existing nonconforming use of the land and is subject to the provisions of Section 506.04. Provided, however, that notwithstanding the foregoing sentence, when the use of a lot is an existing nonconforming use of the land because of this Section 507 and the landowner proposes to convert the lot to a different use for which the lack of frontage on a developed public street will cause no greater disruption to the orderly development of the Town than the previous use, such landowner may apply for a Special Use Permit to allow the new use. The Town Council shall adjudicate such request in accordance with the other provisions of this Ordinance regarding Special Use Permits.

## ARTICLE 6 USES IN DISTRICTS

<04.00	CONCEDED MANAGED FOR CO	
601.00	CONSERVATION DISTRICT C-1	l

601.02-11

- Intent of the Conservation District C-1. This district covers portions of the Town which are occupied by various open uses such as agriculture, National Forests and Parks, state owned forest and park lands, and local government owned lands. This district is established for the specific purpose of providing recreation and open space uses, conservation of water and other natural resources, reducing soil erosion, protecting watersheds, and reducing hazards from flood and fire.
- 601.02 <u>Permitted Uses.</u> Within the Conservation District C-1, the following uses are permitted as of right:

601.02-1	Wildlife areas or game refuges;	
601.02-2	Flood control and watershed structures;	
601.02-3	Parks, playgrounds, and outdoor recreation areas;	
601.02-4	Public utilities;	
601.02-5	Public service or storage buildings;	
601.02-6	Nursery or tree farms;	
601.02-7	Fish hatcheries;	
601.02-8	Cemeteries;	
601.02-9	Timber production, forests;	
601.02-10	Agriculture uses such as general farming, pasture, grazing, outdoor plant nurseries, horticulture, truck farming, forestry, sod farming, and wild crop harvesting. General farming includes the raising of livestock and field crops. However, poultry houses and hog pens are prohibited.	

Public water and sewage systems.

- 601.03 <u>Permitted Accessory Uses in C-1 District.</u> The following accessory uses are permitted in the C-1 District:
  - Off street parking for uses permitted in this district, subject to the regulations of Article 7.

#### 611.00 LOW DENSITY RESIDENTIAL DISTRICT R-1

- Intent of the Low Density Residential District R-1. This district is composed of certain quiet, low-density residential areas plus certain open areas where similar residential development appears likely to occur. The regulations for this district are designed to stabilize and protect the essential characteristics of the district, to promote and encourage a suitable environment for family life. To these ends, development is limited to relatively low concentration and permitted uses are limited basically to single unit dwellings providing homes for the residents plus certain additional uses such as schools, parks, churches, and certain public facilities that serve the residents of the district.
- 611.02 <u>Permitted Uses.</u> Within Low Density Residential District R-1 the following uses are permitted as of right:
  - 611.02-1 Single-family dwellings;
  - Public parks, playgrounds, recreational buildings and grounds, tennis courts, swimming pools and outdoor recreational activities, all of a noncommercial nature;
  - 611.02-3 Public utility transmission systems;
  - Raising and keeping of livestock, including private stables, provided that the following conditions are met:
    - (a) Such use shall take place only on a lot that is not less than two (2) acres in area;
    - (b) Such use shall involve the keeping of not more than one animal per acre on the premises;
    - (c) In any such use no pen or structure for the keeping or livestock (other than grazing areas) and no area for the storage of manure or other odor or dust causing materials, shall be located within 100 feet of any side or rear lot line.
  - 611.02-5 Group Home, as defined herein

#### 611.02-6 Bed and Breakfasts or Short Term Rentals

Ordinance of December 6, 2011 added Section 611.02-5.

Ordinance of May 2, 2023 added Section 611.02-6.

- 611.03 <u>Special Uses</u>. The following uses are permitted in an R-1 zoning district after the issuance of a Special Use Permit in accordance with § 501.02:
  - Child care centers and family day care homes, provided that for any such use the main structure shall not be located closer than fifty (50) feet from any residential lot;
  - 611.03-2 Public utilities (other than transmission systems);
  - Public elementary, middle, and high schools and private schools having substantially the same curriculum as such public schools;
  - 611.03-4 Public service or storage buildings;
  - Public Assembly Hall, provided, however, that no such facilities shall be located closer than one hundred (100) feet from any other residential lot;
  - 611.03-6 Repealed by Ordinance of May 2, 2023
  - Agricultural and forestry uses and stands for seasonal retail sale of agricultural products produced on the property;
  - 611.03-8 Cemeteries:
  - 611.03-9 Fire stations and rescue squad facilities
  - 611.03-10 Repealed by Ordinance of May 2, 2023

Ordinance of December 6, 2011 amended Section 611.03-5 to replace "Churches and other places of worship with attendant educational and recreational facilities" with "Public Assembly Hall", and deleted the phrase "church or attendant" before the word "facilities", and added the word "such" before the word "facilities"

**Ordinance of March 4, 2014** added Section 611.03-10 to permit Inns as a special use in the R-1 zoning district.

**Ordinance of May 2, 2023** repeated 611.03-6 and 611.03-10, which had provided that Bed and Breakfasts and Inns were used permitted by Special Use in the R-1 zoning district.

- 611.04 <u>Accessory Uses</u>. The following accessory uses are permitted in the R-1 zoning district:
  - Home occupations in accordance with the provisions of § 705;

	611.04-2	Living quarters in main building for persons employed on the premises;	
	611.04-3	Travel trailers in accordance with the provisions of § 707;	
	611.04-4	Temporary buildings for uses incidental to construction work; provided that such buildings shall be removed upon completion or abandonment of the construction work;	
	611.04-5	Parking in accordance with § 703;	
	611.04-6 Private parking garage;		
	611.04-7	Shelter for house pets, but not exceeding two (2) shelters, to house not more than one (1) adult per shelter plus dependent animals of up to six (6) months of age;	
	611.04-8	Private swimming pool;	
	611.04-9	Fences in accordance with § 709;	
	611.04-10	Satellite receivers or dishes.	
611.05	Lot Area Req	uirements. Lots in the R-1 district shall not be less than 15,000 area.	
611.06	<u>Lot Width Requirements</u> . The minimum lot width at the setback line in an R-1 district shall be 100 feet. If the lot is located in a cul-de-sac, than the minimum lot width at the setback line in an R-1 district shall be 75 feet.		
611.07	<u>Setback Requirements</u> . Primary structures in an R-1 district shall be located 35 feet or more from any street right-of-way.		
611.08	<u>Yard Requirements</u> . The following yard requirements shall be applicable in the R-1 District:		
	611.08-1	Rear Yard. Each rear yard shall be a minimum of 30 feet.	
	611.08-2	Side Yard. Together the side yards shall be a minimum of 25 feet. Each side yard shall not be less than 10 feet.	
611.09	Height Requirements grade.	rements. Buildings may be erected up to 35 feet in height from	

Accessory Structure Requirements. No accessory structure shall be nearer than 5 feet to any lot line. If an accessory structure is over one story in height, such accessory structure shall not be nearer than 10 feet to any lot line. No accessory building shall be nearer a principal building than 10 feet. No accessory structure shall exceed 15 feet above grade height. No accessory building shall be larger than two hundred fifty (250) square feet in area without a special use permit; and such an accessory building shall be a permissible special use in the R-1 zoning district.

#### 612.00 RESIDENTIAL DISTRICT R-2

- Intent of Residential District R-2. The intent of the R-2 Residential District is to encourage residential neighborhoods and to stabilize and protect the essential character of such neighborhoods. The regulations for this district tend to protect against encroachment of commercial or industrial uses and other uses likely to generate noise, crowds, concentrations of traffic, light, dust, odors, smoke, or other obnoxious influences.
- 612.02 <u>Permitted Uses.</u> Within Residential District R-2 the following uses are permitted as of right:
  - 612.02-1 Single-family dwellings;
  - Two-family dwellings where each living unit contains a minimum of nine hundred (900) square feet;
  - Public parks, playgrounds, recreational buildings, and grounds, tennis courts, swimming pools, and outdoor recreational activities, all of a noncommercial nature. No swimming pool or structure shall be located closer than one hundred (100) feet from any residential lot:
  - 612.02-4 Public utility transmission systems;
  - Buildings originally constructed as bona fide single family dwellings that have been converted into two family dwellings.
  - 612.02-6 Group Home, as defined herein
  - 612.02-7 Bed and Breakfasts or Short Term Rentals

Ordinance of December 6, 2011 added Section Ordinance of May 2, 2023 added Section 612.02-7. 612.02-6.

612.03	Special Uses. The following uses are permitted in an R-2 zoning district as issuance of a Special Use Permit in accordance with § 501.02:		=
	612.03-1	Public utilities;	
	612.03-2		ly day care homes; provided that the ocated closer than fifty (50) feet from
	612.03-3	Homes for adults;	
	612.03-4	Schools;	
	612.03-5	Public service or storage buil	dings;
	612.03-6	•	ovided, however, that no attendant located closer than one hundred (100) al lot.
	612.03-7	Repealed by Ordinance of M	May 2, 2023
	612.03-8	Cemeteries;	
		Dwellings constituting a Ze provisions of § 714.	ro Lot Line Development under the
	612.03-10	Repealed by Ordinance of M	May 2, 2023
	612.03-6 to replace worship with atten facilities" with "Pub	ember 6, 2011 amended Section e "Churches and other places of dant educational and recreational slic Assembly Hall" and to add the before the words "recreational	Ordinance of March 4, 2014 added Section 612.03-10 to permit Inns as a special use in the R-2 zoning district. Ordinance of May 2, 2023 repeated 612.03-7 and 612.03-10, which had provided that Bed and Breakfasts and Inns were used permitted by Special Use in the R-2 zoning district
612.04 <u>Accessory Uses</u> . The following accessory uses are permitted in district:		uses are permitted in the R-2 zoning	
	612.04-1	Home occupations in accorda	ance with the provisions of § 705;
	612.04-2	Living quarters in main bu premises;	ilding for persons employed on the
	612.04-3	Travel trailers in accordance	with the provisions of § 707;

612.04-4	Temporary buildings for uses incidental to construction work; provided that such buildings shall be removed upon completion or abandonment of the construction work;	
612.04-5	Parking in accordance with § 703;	
612.04-6	Private parking garage;	
612.04-7	Shelter for house pets, but not exceeding two (2) shelters, to house not more than one (1) adult per shelter plus dependent animals of up to six (6) months of age;	
612.04-8	Private swimming pool;	
612.04-9	Fences in accordance with § 709;	
612.04-10	Satellite receivers or dishes.	
Lot Area Requirements. The following lot area requirements shall be applicable in the R-2 District: All such lots shall be at least 12,500 square feet in area. Provided, however, that a lot upon which is sited one-half of a duplex may be a minimum of 6,250 square feet, provided that the other half of such dwelling is also on a lot with a minimum size of 6,250 square feet, such that the structure sits on lots that total a minimum of 12,500 square feet in area. Notwithstanding the foregoing, in any lot that is part of a Zero Lot Line Development under the provisions of Section 714, the minimum lot size shall be 6,250 square feet.  Ordinance of April 6, 2010, added the final sentence		
of this section.	to, 2010, added the final sentence	

612.06 <u>Lot Width Requirements</u>. In the R-2 district, the minimum lot width at the setback line shall be 75 feet. Notwithstanding the foregoing, in any lot that is part of a Zero Lot Line Development under the provisions of Section 714, the minimum lot width at the setback line shall be 60 feet.

Ordinance of April 6, 2010, added the final sentence of this section.

612.05

- 612.07 <u>Setback Requirements</u>. In the R-2 district primary structures shall be located 30 feet or more from any street right-of-way.
- 612.08 <u>Yard Requirements</u>. The following yard requirements shall be applicable in the R-2 District:
  - 612.08-1 Rear Yard. Each rear yard shall be a minimum of 25 feet.

- 612.08-2 <u>Side Yard</u>. Each side yard be a minimum of 10 feet.
- 612.09 <u>Height Requirements</u>. Primary buildings may be erected up to 35 feet in height from grade.
- Accessory Structure Requirements. No accessory structure shall be nearer than 5 feet to any lot line. If an accessory structure is over one story in height, such accessory structure shall not be nearer than 10 feet to any lot line. No accessory building shall be nearer a principal building than 10 feet. No accessory structure shall exceed 15 feet above grade height. No accessory building shall be larger than two hundred fifty (250) square feet in area without a special use permit; and such an accessory building shall be a permissible special use in the R-2 zoning district.

## 613.00 MEDIUM DENSITY RESIDENTIAL DISTRICT R-3

Intent of Medium Density Residential District R-3. The intent of the Medium Density Residential District is to provide for a range of development densities in accordance with the Broadway Comprehensive Plan. The regulations for this district provide for development which is not completely residential because it includes public and semipublic, institutional and other related uses. However, it is basically residential in character, and as such, is protected against encroachment of heavy commercial, industrial, and other uses likely to generate noise, crowds, large concentrations of traffic, light, dust, odors, smoke, and other obnoxious influences. This District is also intended to help to implement the Urban Development Areas as shown in the Town Comprehensive Plan, to provide compact, interconnected, pedestrian-friendly areas.

Ordinance of December 6, 2011 added the final sentence to this section.

- 613.02 <u>Permitted Uses.</u> Within Medium Density Residential District R-3, the following uses are permitted as of right:
  - 613.02-1 Single-family dwellings;
  - 613.02-2 Two-family dwellings;
  - [Repealed by Ordinance of July 7, 2015]
  - 613.02-4 Public utility transmission systems;

613.02-5	Public parks, playgrounds, recreational buildings and grounds, tennis courts, swimming pools and outdoor recreational activities, all of a noncommercial nature; provided, however, that no swimming pool or structure shall be located closer than one hundred (100) feet to any other residential lot;			
613.02-6	Professional offices in structures similar in character with surrounding neighborhoods;			
613.02-7	Bed and Breakfasts or Short Term Rentals;			
613.02-8	Libraries.	Libraries.		
613.02-9	Group Home, as defined her	Group Home, as defined herein.		
Ordinance of December 6, 2011, added Section 613.02-9 but originally numbered it as Section 613.02-8. The Town Attorney corrected this typographical error by letter to the Town Clerk dated January 3, 2012, under the authority provided in Section 810.05.		Former Section 613.02-3 read as follows: "Multi family dwellings, apartments, townhouses, (as regulated in Section 708 of this Ordinance) and condominiums;" Ordinance of May 2, 2023 removed references to Tourist Homes and Inns in 613.02-7, and added the reference to Short Term Rentals.		
<u>Special Uses</u> . The following uses are permitted in an R-3 zoning district after the issuance of a Special Use Permit in accordance with § 501.02:				
613.03-1	General hospitals;			
613.03-2	Public utilities;			
613.03-3	Commercial operations that: (1) will not adversely affect the health or safety of persons residing in the neighborhood of the proposed use; (2) will not be detrimental to the public welfare or injurious to property or improvements; (3) will not be in conflict with the intent of this district; and (4) will comply with all other provisions regulating such uses;			
613.03-4	Boarding houses;			
613.03-5	Family day care homes, foster homes or group homes serving the mentally retarded, developmentally disabled or others, rest homes, homes for adults, or nursing homes;			
613.03-6	Townhouses (as regulated in Section 708 of this Ordinance)			
613.03-7	Schools;			

613.03

613.03-8 Public service or storage buildings; 613.03-9 Public Assembly Hall, provided, however, that no attendant recreational facility shall be located closer than one hundred (100) feet from any other residential lot. Former Section 613.06-6 read as follows: "Clubs, word "attendant" before the words "recreational fraternities, lodges and meeting places of other facility". The Ordinance originally referred to that organizations, provided that the buildings in which such section as "Section 613.03-8", but the Town Attorney corrected this typographical error by letter to the Town meetings are housed shall be located at least fifty (50) feet from any other lot;" This section was repealed by Clerk dated January 3, 2012, under the authority provided in Section 810.05. Ordinance of December 6, 2011. Ordinance of December 6, 2011 amended Section Ordinance of July 7, 2015 added current § 613.03-6. 613.03-9 to replace "Churches and other places of worship with attendant educational and recreational facilities" with "Public Assembly Hall" and to add the Accessory Uses. The following accessory uses are permitted in the R-3 zoning district: 613.04-1 Home occupations in accordance with the provisions of § 705; 613.04-2 Living quarters in main building for persons employed on the premises; 613.04-3 Travel trailers in accordance with the provisions of § 707; 613.04-4 Temporary buildings for uses incidental to construction work; provided that such buildings shall be removed upon completion or abandonment of the construction work; 613.04-5 Parking in accordance with § 703; 613.04-6 Fences in accordance with § 709; Satellite dishes: 613.04-7 613.04-8 Garages; 613.04-9 Pools. <u>Lot Area Requirements</u>. The following lot area requirements shall be applicable

Area. Lots shall not be less than 10,000 square feet in area, plus an

additional 2,000 square feet per unit.

613.04

613.05

in the R-3 District:

613.05-1

- 613.05-2 <u>Maximum Density</u>. A maximum density of (10) ten units per every (1) one acre shall be permitted.
- 613.06 <u>Lot Width Requirements</u>. The minimum lot width at the setback line in the R-3 district shall be 50 feet for single family dwellings and two family dwellings and 75 feet for three family (or greater than three family) dwellings.
- 613.07 <u>Setback Requirements</u>. In the R-3 District structures shall be located 20 feet or more from any street right-of-way however, covered porches may encroach into the front yard to within 10 feet of the front lot line.

**Ordinance of December 6, 2011** amended this section by changing "30" to "20" and by added the phrase

"however, covered porches may encroach into the front yard to within 10 feet of the front lot line".

- 613.08 <u>Yard Requirements</u>. The following yard requirements shall be applicable in the R-3 District.
  - 613.08-1 Rear Yard. Each rear yard shall be a minimum of 25 feet.
  - 613.08-2 <u>Side Yard</u>. Each side yard shall not be less than 10 feet.
- 613.09 <u>Height Requirements</u>. Buildings may be erected up to 35 feet in height from grade.
- Accessory Structure Requirements. No accessory structure shall be nearer than 5 feet to any lot line. If an accessory structure is over one story in height, such accessory structure shall not be nearer than 10 feet to any lot line. No accessory building shall be nearer a principal building than 20 feet. No accessory structure shall exceed 15 feet above grade height. No accessory building shall be larger than two hundred fifty (250) square feet in area without a special use permit; and such an accessory building shall be a permissible special use in the R-3 zoning district.
- According to functional street purposes and projected traffic flow; (b) To distribute traffic; (c) To assure safe and convenient sight distances; (d) To complement the natural topography; (e) In coordination with existing and planned streets; (f) To accommodate and promote pedestrian convenience and safety through appropriate sidewalks, buffers, curve radii, and pavement widths, and (g) In accordance with State standards.

Ordinance of December 6, 2011 added this section.

613.12 <u>Street Lighting</u>. The developer shall provide street lighting on all streets in the development and shall be built to the current requirements of the utility company.

Ordinance of December 6, 2011 added this section.

Pedestrian Circulation. The developer shall provide for sidewalks and pedestrian walkways that will enable residents, visitors and patrons to walk safely and conveniently between the various functional areas of the project and adjacent circulation systems. Sidewalks contiguous and parallel to streets shall be the primary pedestrian circulation mode; trails and walkways not associated with streets may supplement the sidewalk system.

Ordinance of December 6, 2011 added this section.

Parking. In addition to the regulations of § 703 for providing off-street parking in adequate amounts and in convenient locations, parking areas shall be located to the rear of principal structures, designed to preserve natural amenities and provide a safe pedestrian movement in and around the site, and should avoid excessive concentrations of pavement by scattered landscaping and tree planting. Alleys providing rear access to lots are permitted and encouraged. On-street parking contiguous to the lot may be used to meet the total parking requirements for the site.

Ordinance of December 6, 2011 added this section.

#### 614.00 RESIDENTIAL DISTRICT R-4

- 614.01 <u>Intent of the Residential District R-4</u>. The intent of the Residential District R-4 is to allow low cost residential development in the form of manufactured home parks and manufactured home subdivisions.
- 614.02 <u>Permitted Uses.</u> Within Residential District R-4 the following uses are permitted as of right:
  - Manufactured home parks in accordance with § 706;
  - Manufactured home subdivisions in accordance with § 706 and meeting the requirements of Article 9;
  - Permanent buildings housing management offices, child care centers, laundry facilities, or indoor recreational facilities or other service facilities, provided that: (a) parking requirements for such facilities are met; and (b) such uses are subordinate to the residential use and character of the park.

612.02-4 Group Home, as defined herein

Ordinance of December 6, 2011 added Section 612.02-4.

- 614.03 <u>Special Uses</u>. The following uses are permitted in an R-4 zoning district after the issuance of a Special Use Permit in accordance with § 501.02:
  - Neighborhood Commercial uses that are designed and intended to meet the service needs of persons residing in the park or subdivision and its immediate neighborhood.
- 614.04 <u>Accessory Uses</u>. The following accessory uses are permitted in the R-4 zoning district:
  - Home occupations in accordance with the provisions of § 705;
  - 614.04-2 Living quarters in main building for persons employed on the premises;
  - Travel trailers in accordance with the provisions of § 707;
  - Temporary buildings for uses incidental to construction work; provided that such buildings shall be removed upon completion or abandonment of the construction work;
  - 614.04-5 Parking in accordance with § 703;
  - 614.04-6 Fences in accordance with § 709;
  - 614.04-7 Satellite dishes;
  - 614.04-8 Garages;
  - 614.04-9 Pools.
- Lot Area Requirements. In any manufactured home park in the R-4 zoning district, the total area of the park shall be at least 3 acres, and the minimum area of any individual stand within the park shall be 4,000 square feet. In any manufactured home subdivision in the R-4 zoning district, the aggregate area of the lots in the subdivision shall be at least 4 acres, and the minimum area of any individual lot within the park shall be 7,000 square feet.

- 614.06 <u>Lot Width Requirements</u>. In any manufactured home park in the R-4 zoning district, minimum lot width at the setback line shall be 150 feet, and the minimum width of each individual stand within the park shall be 50 feet. In any manufactured home subdivision in the R-4 zoning district, the minimum aggregate width of the lots within the subdivision shall be at least 150 feet at the setback line, and the minimum lot width of each lot within the subdivision shall be 50 feet.
- Setback Requirements. In any manufactured home park in the R-4 zoning district, structures shall not be located within 25 feet of any street right of way, and each individual stand shall be set back at least ten (10) feet from the stand line. In any manufactured home subdivision in the R-4 zoning district, no structure shall be located within 20 feet of the location of any street right of way, and, further, no structure shall be located within 25 feet of the location any street right of way computed without regard to the internal lot lines of the subdivision.
- 614.08 <u>Yard Requirements</u>. The following yard requirements shall be applicable to in the R-4 District.
  - Rear Yard. In a manufactured home park, the rear yard of the park shall be a minimum of 25 feet and the rear yard of each stand within the park shall be a minimum of 10 feet. In a manufactured home subdivision, the rear yard of the subdivision (computed without regard to internal lot lines) shall be 25 feet, and the rear yard of each lot within the subdivision shall be 25 feet.
  - Side Yard. In a manufactured home park, the side yard of the park shall be a minimum of 25 feet, the two side yards of each stand within the park shall be a minimum of 25 feet combined and a minimum of 10 feet on each side. In a manufactured home subdivision, the side yard of the subdivision (computed without regard to internal lot lines) shall be 25 feet, and the side yard of each lot within the subdivision shall be 11 feet.
- 614.09 <u>Height Requirements</u>. Buildings may be erected up to 35 feet in height from grade.
- Accessory Structure Requirements. No accessory structure shall be nearer than 5 feet to any lot line. If an accessory structure is over one story in height, such accessory structure shall not be nearer than 10 feet to any lot line. No accessory building shall be nearer a principal building than 10 feet. No accessory structure shall exceed 15 feet above grade height. No accessory building shall be larger than two hundred fifty (250) square feet in area without a special use permit, and

such an accessory building shall be a permissible special use in the R-4 zoning district.

614.11 <u>Density Requirement</u>. In any manufactured home park in the R-4 district, the overall density shall not exceed seven (7) units per gross acre. The density of any particular acre within such park shall not exceed eight (8) units per gross acre.

## 615.00 PLANNED UNIT DEVELOPMENT DISTRICT R-5

- Intent of Planned Development District R-5. The intent of the Planned Development District R-5 is to provide for larger scaled development and clustering of single-family residential dwelling units through design innovation to provide for a neighborhood with a variety of housing types and densities, neighborhood shopping facilities, schools, parks, playgrounds, off-street parking and, where necessary, land reserved to provide local employment opportunities.
- 615.02 <u>Permitted Uses.</u> Within Planned Development District R-5 the following uses are permitted as of right:
  - 615.02-1 Single-family dwellings;
  - 615.02-2 Two-family dwellings;
  - 615.02-3 Multi-family dwellings, apartments, townhouses;
  - 615.02-4 Neighborhood shopping facilities.
  - 615.02-5 Group Home, as defined herein

Ordinance of December 6, 2011 added Section 615.02-5.

- Other Uses. In addition to the principal uses, other commercial or non-commercial service uses may be permitted provided: (a) that such uses are intended primarily to serve the needs of the project area residents; (b) that such uses are designed and located for the convenience of project area residents and to protect the character of the district; (c) that all subsequent changes in use shall be approved by the Zoning Administrator; (d) that construction of commercial facilities shall not begin until twenty-five (25) percent of the residential units or two hundred fifty (250) dwelling units, whichever is less, of the total planned development has been completed.
- 615.04 <u>Uses Permitted by Approval</u>. The development authorized within this district is regulated by a comprehensive development and management plan proposed by the developer and where the Town declines to impose certain general zoning

restrictions applicable in other districts because of its ability to examine and approve a detailed site plan and dedication of common open space. No use shall be permitted except uses in an approved development plan, and no chance of use shall occur except in accordance with the procedures described herein.

- 615.05 <u>Qualifying Requirements.</u> No land shall be zoned R-5 unless it meets the following requirements:
  - Ownership Requirements. The project area must be owned by one person or the application must be filed jointly by the owners of all land and within the project area. The holder of a written option to purchase land shall, for the purposes of such application, be deemed to be an owner of such land; however, each and every project area in a Planned Residential District must be owned by one person or by individuals holding a single common law joint tenancy before the final development plan is approved.
  - Availability of Public Utilities. The project area must be located where public water and sewer systems are available or where a community water and sewer system can be developed as part of the project.
  - 615.05-3 <u>Land Suitability</u>. The land must not be inappropriate for such development because of inadequate road access, inadequate community facilities, excessive distance to employment area, non-conformity to city development plans, or other public health, welfare or safety objectives. The Zoning Administrator may obtain opinions from any other government agency concerning the suitability of proposed development.
- 615.06 <u>Site Design Requirements</u>. The following are the site design requirements for the R-5 Residential District:
  - 615.06-1 <u>Maximum Density</u>. The gross residential density shall not exceed ten (10) dwelling units per acre.
  - Common Open Space. Minimum open space shall be not less than thirty (30) percent of the total area exclusive of buildings, streets, alleys, roads, parking areas, walks, patios, and other similar improvements but inclusive of swimming pools and other active and passive recreational areas. At least 15% of the site shall be usable open space, including squares, greens and plazas. Areas used for low-impact stormwater management facilities can be credited toward meeting the minimum 30% total open space.

**Ordinance of December 6, 2011** added the final sentence of Section 615.06-2.

- 615.06-3 <u>Functional Relationships</u>. The site development plan shall be designed for convenient relationships between the various functional areas of the project such as residential, recreational, shopping, etc.
- 615.06-4 <u>Lot Design</u>. The lot design, arrangement, and shape shall be such that lots will provide satisfactory and desirable sites for buildings, and be properly related to topography, and provide convenient and safe access.
- 615.06-5 <u>Street Design</u>. The street system within the project area shall be designed:
  - (a) According to functional street purposes and projected traffic flow;
  - (b) To distribute traffic;
  - (c) To assure safe and convenient sight distances;
  - (d) To complement the natural topography;
  - (e) In coordination with existing and planned streets; and
  - (f) To accommodate and promote pedestrian convenience and safety through appropriate sidewalks, buffers, curve radii, and pavement widths, and
  - (g) In accordance with State standards.

**Ordinance of December 6, 2011** replaced the phrase "discourage through" with the word "distribute" in Section 615.06-5(b),

added present Section 615.06-5(f), and renumbered former Section 615.06-5(f) as new Section 615.06-5(g).

Street Names and Signs. The name of proposed streets shall not duplicate existing street names irrespective of the use of the suffix street, avenue, boulevard, driveway, place, lane or court. Proposed streets that are obviously in alignment with other already existing and named streets shall bear the names of the existing streets. Street names shall be indicated on the preliminary plan and final subdivision plat. The developer shall provide street signs at all intersections.

615.06-7 <u>Street Lighting</u>. The developer shall provide street lighting on all streets in the development and shall be built to the current requirements of the utility company.

Pedestrian Circulation. The developer shall provide for sidewalks and pedestrian walkways that will enable residents, visitors and/or patrons to walk safely and conveniently between the various functional areas of the project and adjacent circulation systems. Sidewalks contiguous and parallel to streets shall be the primary pedestrian circulation mode; trails and walkways not associated with streets may supplement the sidewalk system.

Ordinance of December 6, 2011 added the final sentence of this section.

Parking. The regulations of § 703 shall not apply in an R-5 zoning district. The developer shall nevertheless provide off-street parking in adequate amounts and in convenient locations. Wherever feasible, parking areas should be located to the rear of principal structures, designed to preserve natural amenities and provide a safe pedestrian movement in and around the site, and should avoid excessive concentrations of pavement by scattered landscaping and tree planting. Generally, two parking spaces should be provided for each dwelling unit. Alleys providing rear access to lots are permitted and encouraged. On-street parking contiguous to the lot may be used to meet the total parking requirements for the site.

**Ordinance of December 6, 2011** added the phrase "located to the rear of principal structures," after the word "be" in the second sentence, added the phrase "and

provide a safe pedestrian movement in and around the site," after the word "amenities" in the second sentence, and added the final two sentences of the section.

- 615.06-10 <u>Water and Sewer</u>. All R-5 zoning districts shall be served by collective water and sewer systems as follows:
  - (a) Wherever feasible the project area water and sewer lines shall be connected to existing public systems.
  - (b) Where connection to existing public water or sewer systems are not feasible, the developer shall provide community water or sewer systems.
- 615.06-11 <u>Community Facilities</u>. The developer shall reserve or dedicate land for community facilities if the need is created by the project area development or if proposed on the Town development plan.

- 615.06-12 <u>Fire Hydrants</u>. The developer shall provide fire hydrants throughout the project area in such locations to provide adequate fire protection.
- Drainage. The site development plan shall include a plan for adequate drainage. The street and lot plan shall be designed to avoid drainage problems. Where storm drains or drainage ditches are required, or where an existing waterway or drainage way traverses the project area, an easement or right-of-way shall be provided with adequate improvements to contain the drainage flows from the tributary area upstream of the watershed.
- Floodways. Land subject to flooding and land deemed to be topographically unsuitable shall not be platted for residential occupancy, nor for such other uses as may increase danger of health, life, or property, or aggravate erosion or flood hazard. Such land within the project area shall be used as common open space or other uses which would not be endangered by periodic or occasional inundation or shall not produce conditions contrary to public welfare.
- 615.06-15 <u>Easements</u>. The developer shall provide easements through the project area for water, sewer, gas, telephone, power and other utilities as required by the respective utility departments, agencies or companies.
- 615.06-16 <u>Grading</u>. The site development plan shall be designed to minimize the amount of grading required for development. To the extent feasible, the natural lay of the land shall be maintained except where grading is required for public health or safety.
- 615.06-17 <u>Natural Amenities</u>. The developer shall make every reasonable effort to protect and preserve the natural amenities of the site such as tree cover, waterways, scenic overlooks, etc. The site development plan shall be designed to maximize the use and enjoyment of natural amenities by project residents.
- 615.06-18 <u>Landscaping and Screening</u>. The developer shall provide landscaping and screening to improve the project appearance or to provide a buffer between potentially conflicting uses.

- 615.07 <u>Data to Accompany Application</u>. In an application to zone or rezone land to an R-5 zoning district, the applicant shall submit a tentative, overall development plan which shall include:
  - Scale accurate proposed development plan mapping of the project to include:
    - (a) Proposed land uses including residential types, commercial types, recreation and any other proposed use;
    - (b) Proposed street system including public and private right-of-way;
    - (c) Proposed parking areas and parking space delineations;
    - (d) Proposed plat showing subdivision lot lines;
    - (e) Proposed utility rights-of-way or easements including water, sewer, gas, power, and telephone;
    - (f) Proposed drainage plan;
    - (g) Proposed location of buildings, structures, and improvements;
    - (h) Property lines of proposed common property;
    - (i) Proposed pedestrian circulation system;
    - (j) Proposed landscaping plan;
    - (k) Proposed treatment of the project perimeter such as screening or landscaping;
    - (l) Relationships and tie-ins to adjacent property.
  - Supporting documentation including the following minimum data:
    - (a) A legal description of the project boundaries;
    - (b) A statement of existing and proposed property owners;
    - (c) Names and addresses of all adjacent property owners;
    - (d) A statement of project development objective and character to be achieved;

- (e) An approximate development schedule including dates of proposed construction beginning and completion and staging plan, if appropriate;
- (f) A statement of intent regarding future selling or leasing of land areas, dwelling units, commercial area, etc.;
- (g) Quantitative data including the number and type of dwelling units; parcel sizes, gross and net residential densities, total amount and percentage of open space, residential, commercial, and other land use types;
- (h) Proposed building types including architectural style, height, and floor area;
- (i) Approvals from the Virginia Department of Highways and the Health Official (if appropriate);
- (j) Proposed agreements, provisions, or covenants which govern the use, maintenance, and continued protection of property to be held in common ownership;
- (k) A statement of proposed temporary and permanent erosion and sedimentation control measures to be taken.
- Application. An applicant for R-5 zoning shall file ten (10) copies of its preliminary plan application with the Zoning Administrator. The Zoning Administrator shall forward the application and data to the Planning Commission for its review and recommendation. The Planning Commission shall consider the general plan for the community, the location, arrangement and size of lots, parks, school sites and other reservations of open space; the location, width and grade of streets; the location and arrangement of parking spaces; the location, arrangement and height of buildings; the location, arrangement and design of neighborhood business areas and accessory parking spaces; the gross densities proposed for the area; and such other features as will contribute to the orderly and harmonious development of the area, with due regard to the type and the character of adjoining neighborhoods and the peculiar suitability of the proposed uses.
- 615.09 Processing Fee. At the time of filing the preliminary development plan application, the applicant shall deposit with the Zoning Administrator a check payable to the Treasurer in the amount of the application fee prescribed in Article 10.

## 615.10 Preliminary Plan Approval.

- 615.10-1 By Planning Commission. Within sixty (60) days after the filing of the preliminary development plan, the Planning Commission shall report to Town Council giving its recommendation to approve the plan as presented, to approve the plan as revised by the concurrence of the developer, or to disapprove the plan. The Planning Commission may not make its formal, final recommendation to the Town Council before holding any public hearing that Virginia law requires.
- By Town Council. The Town Council shall consider the application after holding any public hearings that Virginia law requires. If the preliminary development plan is approved, or approved with modifications by the Town Council, the Zoning Map shall be amended to show the R-5 Planned Unit Development. If the preliminary development plan is approved with modifications, the Town Council shall not amend the Zoning Map until the applicant has filed with the Zoning Administrator written consent to the plan as modified.
- Status of Approval. No building permits shall be issued within the project area until the Town has approved the final development plan under the procedures in the following sections and until the required bond is posted to the Zoning Administrator's satisfaction.
- Final Plan Application. Within six (6) months following the approval of the preliminary development plan, the applicant shall file with the Zoning Administrator ten (10) copies of a final development plan containing in final form, the information required in the preliminary plan, including but not limited to, final maps and documents specified above. In its discretion and for good cause, the Zoning Administrator may, upon receipt of a written application, extend for six (6) months the period for filing of the final development plan.
- Phasing Plan. If the project area is to be developed in stages, a phasing plan shall be submitted with the final development plan. The phasing plan shall delineate the areas to be developed in each phase and the approximate development schedule of each phase.
- 615.14 <u>Compliance with Preliminary Plan</u>. The final development plan shall be in substantial compliance with the preliminary development plan. The final development plan shall not be deemed in substantial compliance if modification does involve any of the following or if such modification exceeds the limitations of this District's regulations:

- (a) Variation of the proposed residential density or intensity of use by more than ten (10) percent;
- (b) Reduction of more than ten (10) percent of the area reserved for common open space;
- (c) Increase of the floor area proposed for nonresidential use by more than ten (10) percent; and
- (d) Increase of the total ground area covered by buildings by more than five (5) percent.
- 615.15 <u>Final Plan Approval</u>. The Town Council shall review the final development plan and shall approve the final development plan if it is in substantial compliance with the preliminary development plan. The Clerk of the Circuit Court of Rockingham County will record the final development plan in the manner provided for recording plats or subdivisions.
- 615.16 <u>Subdivision Plat Requirements</u>. Final subdivision plats shall be submitted and recorded before the granting of building permits or before the sale of any lots. Subdivision plats may be submitted for portions of the project area in accordance with the phasing plan. Subdivision plats shall be drawn according to the specifications contained in Article 9.
- Required Improvements. All improvements shown on the final development plan shall be installed by the developer at his cost. In cases where specifications have been established by state departments or local ordinances, such specifications shall be followed. The developer's performance bond shall not be released until construction has been inspected and approved by the appropriate official. The bonding requirements of Article 9 shall apply to development within an R-5 district.
- Monuments. Monuments shall be provided to identify lot and right-of-way lines permanently.
- Plans and Specifications of Physical Improvements. Two (2) blue or black line prints of the plans and specifications of all required physical improvements to be installed shall be prepared by a licensed engineer as certified by the Commonwealth of Virginia and shall be submitted to the Zoning Administrator for review. The Zoning Administrator shall approve or disapprove of the construction plans within forty-five (45) days of submission. If approved, one (1) copy bearing certification of such approval shall be returned to the developer. If disapproved, all papers shall be returned to the developer with the reason for disapproval stated in writing.

- Maintenance of Common Property. The developer shall create a property owners association to be responsible for maintaining all common property. The cost of maintaining common property shall be paid by property owner assessments and such assessments shall constitute a lien upon the individual properties in accordance with restrictive covenants that the developer prepares and causes to be recorded with the subdivision plat.
- 615.21 <u>Advertising and Sale</u>. The developer shall not advertise for sale or sell any tract or lot within the project area until any required surety bond has been posted with the Town and an approved plat has been properly recorded.
- 615.22 <u>Changes in Final Development</u>. No changes may be made in the approved final plan during the construction of the Planned Development except upon application under the procedures provided below:
  - Minor changes in the location, setting, and character of buildings and structures may be authorized by the Zoning Administrator if required by engineering or other circumstances not foreseen at the time the final plan was approved. No changes authorized by this section may increase the cubic volume of any building or structure by more than ten (10) percent.
  - All other changes in use, and rearrangement of lots, blocks, and buildings tracts, any changes in the provisions of common open spaces, and all other changes in the approved final plan by the developer or any succeeding owner, or agent must be made by application to the Town Council. In considering such amendment, the Town Council may refer such application to the Planning Commission for a recommendation and may consider whether such changes are required by changes in conditions that have occurred since the final plan was approved or by changes in the development policy of the Town. The Town Council may call for an additional public hearing if Virginia law requires it or if the Town Council deems it appropriate.
- Development Schedule and Review. The construction and provision of all facilities and improvements on common property, which are shown on the final development plan, must proceed at the same rate as the construction of dwelling units. At least once every six months following the approval of the final development plan the Zoning Administrator shall review all of the building permits issued for the Planned Development and examine the construction which has taken place on the site. If he shall find that the rate at which facilities and improvements on common property have been constructed and provided, is not in

accordance with the original development schedule, he may withhold additional building permits until the scheduled facilities and improvements on common property have been provided.

# Repealed by Ordinance of December 6, 2011

Former Section 615.24 read: "Failure to Begin Development. If no construction has begun or no use established in the Planned Development within one year from the approval of the final development plan, the final development plan shall lapse and be of no further effect. For good cause, the Town Council may, upon receipt of written application, extend for one additional year the period for the beginning of

construction or the establishment of a use. If a final development plan lapses under the provisions of this section, the Zoning Administrator shall record a notice of revocation with the recorded subdivision plat, the rezoning shall be null and void, and the zoning regulations applicable before the final development plan was approved shall then be in effect."

#### 621.00 GENERAL BUSINESS DISTRICT B-1

Intent of the General Business District B-1. Generally, this district covers that portion of the Town intended for the conduct of general business to which the public requires direct and frequent access, but which is not characterized either by constant heavy trucking other than stocking and delivery of light retail goods, or by any nuisance factors other than occasioned by incidental light and noise of congregation of people and passenger vehicles. This includes such uses as retail stores, banks, theaters, business offices, newspaper offices, printing presses, restaurants and taverns, and garages and service stations.

It is the further intent of the B-1 district to foster a streetscape that is oriented to pedestrian use of the sidewalks for shopping and interactions within the community. It shall be a continuous driving experience for residents and tourists, so as to make travel into and through the Town a safe and scenic experience. The streetscape shall maintain and enhance the current historical character of the corridor, so as to increase the town's pride and its attractiveness as a primary tourist destination in Rockingham County. Buildings within this district are encouraged to be multiuse. Uses, such as commercial on the first floor and residential or offices on the second floor, will help to provide activity in the downtown area throughout the day. This District is also intended to help to implement the Urban Development Areas as shown in the Town Comprehensive Plan, to provide compact, interconnected, pedestrian-friendly areas.

**Ordinance of December 6, 2011** capitalized the word "Town" in the second sentence of the second paragraph and added the final sentence of the second paragraph.

621.02 <u>Permitted Uses.</u> Within the General Business District B-1 the following uses are permitted as of right:

621.02-1 Animal hospitals or clinics;

621.02-2	Apartments in existing structures;
621.02-3	Auto accessory stores (but not servicing);
621.02-4	Automobile sales; provided that vehicles stored for sale must be stored on a surface with an asphaltic, concrete, or other properly bound pavement so as to provide a durable and dustless surface;
621.02-5	Automobile service stations and public garages (with major repair under cover);
621.02-6	Bakeries with total building square footage devoted to such use not to exceed five thousand (5,000) square feet;
621.02-7	Banks and loan and finance offices, including drive in types;
621.02-8	Barber and beauty shops (not including tattoo parlors);
621.02-9	Boardinghouses, rooming houses, nursing homes;
621.02-10	Bookstores;
621.02-11	Bus stations and taxi stands;
621.02-12	Business and professional offices;
621.02-13	Business and vocational schools;
621.02-14	Carpenter, electrical, plumbing, heating, appliance, bicycle, watch and shoe repair, residential painting, publishing, lithographing, upholstering, gunsmith or similar shops provided that any use shall be conducted within a completely enclosed building and provided that no part of a building for such use shall have any opening other than stationary windows or required fire exits within one hundred (100) feet of any Residential District;
621.02-15	Cemeteries;
621.02-16	Child care centers;
621.02-17	Churches and other places of worship, and church school buildings;

621.02-18	Clubs and lodges;
621.02-19	Department stores, variety stores, specialty shops, discount pawn shops, and appliance stores;
621.02-20	Drugstores;
621.02-21	Fences as provided in § 709;
621.02-22	Florist shops;
621.02-23	Fraternal and auxiliary organizations;
621.02-24	Funeral home and/or mortuary;
621.02-25	Greenhouses;
621.02-26	Hospitals;
621.02-27	Hotels, and Bed and Breakfasts or Short Term Rentals, provided that any Bed and Breakfast or Short Term Rental that is constructed as a one or two family Dwelling must receive a Special Use Permit under Section 621.03-13, or be part of a Dwelling that is an existing nonconforming use;
621.02-28	Household electrical appliance repair and services;
621.02-29	Job trailer;
621.02-30	Laundries, dry cleaning shops, and clothes dyeing establishments;
621.02-31	Libraries;
621.02-32	Massage Establishments;
621.02-33	Mini storage facilities;
621.02-34	Museums;
621.02-35	Newspaper offices and printing shops;
621.02-36	Off street parking;

621.02-37	Office buildings;
621.02-38	Parking garages and parking lots;
621.02-39	Personal and professional services;
621.02-40	Pet shops (excluding boarding kennels);
621.02-41	Police, fire, and rescue squad stations;
621.02-42	Post offices;
621.02-43	Power equipment, power lawn and garden equipment repair (under cover);
621.02-44	Printing shops, sign shops;
621.02-45	Public buildings and properties of a cultural, administrative, or service type;
621.02-46	Public service and storage buildings;
621.02-47	Public utilities;
621.02-48	Public utility transmission systems;
621.02-49	Radio and television broadcasting studios;
621.02-50	Recording studios, dance studios, and other music related instructional facilities;
621.02-51	Restaurants including dairy product stores and soda fountains, and drive in restaurants;
621.02-52	Retail stores and shops;
621.02-53	Theaters, assembly halls, playhouses and dinner theaters.

**Ordinance of May 2, 2023** amended Section 621.02-27 to remove the reference to Inns, to add the reference to Short Term Rentals, and to add the text that appears after "Short Term Rentals".

**Ordinance of October 3, 2023** added to Section 621.02-4 the requirement that vehicles held for sale be stored on a paved surface.

621.03		. The following uses are permitted in a B-1 zoning district after the Special Use Permit in accordance with § 501.02:
	621.03-1	Wholesale and processing not objectionable because of dust, noise, or odors;
	621.03-2	Public billiard parlors and pool rooms, bowling alleys, dance halls, and similar forms of public amusement;
	621.03-3	Athletic fields, stadiums, and arenas;
	621.03-4	Beverage manufacturing, bottling or distribution stations and food processing, packaging, or distribution stations;
	621.03-5	Circuses, carnivals, fairs, and sideshows;
	621.03-6	Drive in theaters provided all parts of such drive in shall be distant at least two hundred (200) feet from any Residential District and provided that the screen shall be located as not to be visible from adjacent streets or highways and it shall be set back not less than two hundred (200) feet from the established right of way of said street or highway;
	621.03-7	Livestock market and sales pavilions;
	621.03-8	Overnight recreational vehicle park;
	621.03-9	Shooting range or gallery;
	621.03-10	Wholesale business, storage or warehouse provided that any such use shall be distant at least fifty (50) feet from any Residential District;
	621.03-11	Kennels and animal hospitals provided that any structure or premise used for such purposes shall be located at least two hundred (200) feet from any residential lot;
	621.03-12	Swimming pools, skating rinks, golf driving ranges, miniature golf courses, or similar recreational use or facility if located at least two hundred (200) feet from any residential lot;

	621.03-13	One or two family dwellings on a lot that meets the same requirements that would be necessary for such use in the R-2 zoning district;
	621.03-14	Lumber and building supply (with storage under cover or concealed from public view);
	621.03-15	Plumbing and electrical supply (with storage under cover or concealed from public view);
	621.03-16	Automobile painting, upholstering, rebuilding, reconditioning, and body and fender works;
	621.03-17	Wholesale businesses, and storage warehouses;
	621.03-18	Electronic assembly and repair;
	621.03-19	Auction house;
	621.03-20	Tattoo parlor.
	621.03-21	Mobile Food Service Establishment
	Ordinance of De 621.03-21.	cember 6, 2011 added Section
621.04	Accessory Use district:	ses. The following accessory uses are permitted in the B-1 zoning
	621.04-1	Living quarters in the main building of persons employed on the premises;
	621.04-2	Private parking garage;
	621.04-3	Temporary buildings for uses incidental to construction work, such buildings shall be removed upon completion or abandonment of the construction work;
	621.04-4	Parking under the regulations of § 703.
621.05	Lot Area Req	uirements. No lot area requirements apply to lots in the B-1 district.
621.06		nd Setback Requirements. In the absence of a sidewalk, buildings ed at least (5) five feet from the edge of any street right-of-way and

at least 30 feet from the center line of any street right-of-way less than 50 feet in width. There shall be no setback for buildings in this district adjoining a sidewalk. In addition, any business adjoining a residential district must have a setback equal to half of the residential requirement for the residential district that it adjoins. No minimum lot width is required.

- 621.07 <u>Yard Requirements</u>. The following yard requirements shall be applicable in the B-1 district:
  - 621.07-1 Rear yard. All lots located in the B-1 district must have a rear yard of 20 feet.
  - 621.07-2 <u>Side yard</u>. Property located in the B-1 district that adjoins any residential district, or is separated from any residential district only by a public street or way, shall have a 10-foot side yard on the side or sides adjoining or adjacent to the residential district. Otherwise, no side yard is required.
- 621.08 <u>Height Requirements</u>. Buildings may be erected up to 35 feet in height from grade.
- Accessory Structure Requirements. No accessory structure shall be nearer than 10 feet to any lot line. No accessory building shall be nearer than 20 feet to any principal building. No accessory building shall be larger than two hundred fifty (250) square feet in area without a special use permit; and such an accessory building shall be a permissible special use in the B-1 zoning district.
- Additional Parking Requirements. Wherever feasible, parking areas should be located to the rear of principal structures, designed to preserve natural amenities and a safe pedestrian movement in and around the site, and should avoid excessive concentrations of pavement by scattered landscaping and tree planting.

Ordinance of December 6, 2011 added this section.

## 622.00 PLANNED BUSINESS DISTRICT B-2

Intent of the Planned Business District B-2. The B-2 Planned Business District is intended to permit the development of neighborhood business areas under one ownership or control in those areas of the Town where there are areas of sufficient size in heavily populated sections and where sanitary sewers, street access, and public water supply are adequately provided. Within this district the location of buildings, design of buildings, parking areas, and other open spaces shall be controlled in such a manner that it will not be a detriment to adjoining residential property or to the neighborhood in general. This District is also intended to help to

implement the Urban Development Areas as shown in the Town Comprehensive Plan, to provide compact, interconnected, pedestrian-friendly areas.

Ordinance of December 6, 2011 added the final sentence of this section.

- Permitted Uses. Within the B-2 district, no building, structure, or premises shall be used and no building or structure shall be erected or altered until and unless the Planning Commission has considered the same and the Town Council has approved the same in accordance with the conditions herein. The following rules apply to such consideration:
  - Uses permitted in the B-2 district as of right will be the same as those permitted as of right in the B-1 Zone, except that residential uses are not permitted in the B-2 district.
  - No building permit shall issue in the B-2 district, no construction shall commence, and no zoning permit shall issue for a new use that did not previously involve a building permit, unless the applicant thereof shall file with the Zoning Administrator a written application for approval of a contemplated development plan within said district, which application shall be accompanied with the following information:
    - (a) A plot plan indicating the location of present and proposed buildings, driveways, parking lots, landscaping, screening, and other necessary uses;
    - (b) Preliminary architectural plans for the proposed building or building;
    - (c) A description of the business operations proposed in sufficient detail to indicate the effects of those operations in producing excessive auto or traffic congestion or problems of noise, glare, odor, fire, or safety hazards, or other factors detrimental to the health, safety, and welfare of the area;
    - (d) Engineering or architectural plans for the handling of any of the problems of the type outlined in the paragraph above, including the handling of storm water and sewers and necessary plans for the controlling of smoke or other nuisances such as those enumerated under Item (c) above;
    - (e) Any other information the Planning Commission or Town Council may need to adequately consider the effect that the

proposed uses may have upon the area, and/or the cost of providing municipal services to the area.

- Special Uses. Any use permitted by Special Use Permit in the B-1 district shall also be permitted in the B-2 district if the Town Council concludes that such use is appropriate after applying the factors normally considered in the consideration of Special Use Permits.
- Procedures for Consideration. The Planning Commission and Town Council shall consider the application under the same procedures that apply to the consideration of Special Use Permits. In reviewing and acting upon the application, the Planning Commission and Town Council shall consider the factors enumerated in §§ 611.01 and 611.02.
- Various Regulations. In the B-2 district the area regulations, yard requirements, accessory building requirements, and height regulations shall generally comply with the requirements of the B-1 District unless the Town Council expressly approves the variation thereof in its approval of the site plan.
- 622.06 Setback Requirements. There shall be a twenty five (25) foot setback from all adjoining residential property and from streets that front residential property contiguous to the site. This setback shall act as a buffer between the business and residential uses. It shall be fully landscaped and maintained with grass and with trees or shrubbery of sufficient height and density to serve as a screen between the business zone and the residential zone. The buffer zone shall not constitute a site-distance obstruction at street intersections. The buffer zone shall be considered as part of the lot area but shall not be used for any business purpose such as buildings, parking lots, signs, or any accessory use. The buffer zone upon completion of development of the project shall be at or near the same grade or plane which existed prior to the development of the planned business district property, unless the Town Council expressly approves the variation thereof in its approval of the site plan. These restrictions shall not apply to that portion of the lot fronting on a major business thoroughfare or thoroughfares. Entrances to the property will be allowed from such arterial streets only.

**Ordinance of December 6, 2011** deleted the phrase "all streets and" after the word "from" in the first sentence and added the phrase "and from streets that

front residential property contiguous to the site" after the word "property" in the first sentence.

#### 631.00 LIGHT INDUSTRIAL DISTRICT M-1

631.01 <u>Intent of Light Industrial District M-1</u>. The intent of the M-1 Industrial District is to accommodate industrial uses that provide desirable employment, are consistent with the goal of maintaining environmental quality, and are not likely to create

offensive noise, vibrations, dust, heat, smoke, odor, safety hazard, glare or other objectionable influence to adjacent zoning districts. The uses permitted in the M-1 District are those light industries that manufacture, process, store, and distribute goods and materials and are, in general, dependent upon previously prepared materials refined elsewhere, and must be conducted within a completely enclosed structure, unless hereinafter excepted.

- 631.02 <u>Permitted Uses</u>. Within the M-1 Light Industrial District the following uses are permitted as of right:
  - Assembly of electrical appliances, electronic instruments and devices, and radios, and the manufacture of small parts, such as coils, condensers, transformers, and crystal holders;
  - Manufacture, compounding, processing, packaging or treatment of such products as bakery goods, candy, cosmetics, dairy products, perfumes, perfumed toilet soap, toiletries and food products when conducted within a completely enclosed building;
  - Manufacture, compounding, assembling or treatment of articles of merchandise from the following previously prepared materials: bone, cellophane, canvas, cloth, cork, feathers, felt, fiber, fur, glass, hair, horn, leather, paper, plastic, precious or semi precious metals or stones, shell, straw, textiles, tobacco, wood (excluding sawmills), yam, and paint, not including a boiling process;
  - Manufacture of pottery and figurines or other similar ceramic products, using only previously pulverized clay and kilns fired only by electricity or gas;
  - Manufacture of musical instruments, toys, novelties, and rubber and metal stamps;
  - Carpenter, electrical, plumbing, heating, appliance, bicycle, watch and shoe repair, residential painting, publishing, lithographing, cabinet, furniture, upholstery, gunsmith, or similar shops;
  - 631.02-7 Public service or storage buildings;
  - 631.02-8 Public utilities;
  - 631.02-9 Public utility transmission systems;
  - 631.02-10 Animal hospital, kennels;

	631.02-11	Building material sales yards, plumbing supplies storage;
	631.02-12	Coal and wood yards, lumber yards, feed and seed stores;
	631.02-13	Contractor's equipment storage yards or plants, or rental of equipment commonly used by contractors;
	631.02-14	Wholesale businesses, storage warehouses;
	631.02-15	Laboratory operation, including pharmaceutical, medical, dental, research, or development.
631.03	•	The following uses are permitted in an M-1 zoning district after the Special Use Permit in accordance with § 501.02:
	631.03-1	Mining operations;
	631.03-2	Manufacture, compounding, processing, packaging, or treatment of drugs or pharmaceuticals;
	631.03-3	Poultry hatcheries;
	631.03-4	Automobile reconditioning, including such activities as painting, upholstering, and body and fender work;
	631.03-5	Manufacture of meat or other food or beverage products not enumerated in § 631.01-2;
	631.03-6	Feed mills;
	631.03-7	Any use permitted in B-1 Business District as of right or by Special Use Permit;
	631.03-8	Storage trailer;
	631.03-9	Light fixtures exceeding 16 feet in height.
631.04	Accessory Use district:	es. The following accessory uses are permitted in the M-1 zoning
	631.04-1	Living quarters in the main building of persons employed on the premises;

- 631.04-2 Private parking garage;
- Temporary buildings for uses incidental to construction work, such buildings shall be removed upon completion or abandonment of the construction work:
- Parking under the regulations of § 703;
- 631.04-5 Lighting fixtures under 16 feet or under in height.
- Submission of development plan. Before a building permit shall be issued or construction commenced on any permitted use in the M-1 district, or a permit issued for a new use, the plans, in sufficient detail to show the operations and processes, shall be submitted to the Zoning Administrator for study. The Administrator may refer these plans to the Planning Commission for recommendation, and may require modifications to comply with the provisions of this ordinance. The Administrator shall act on any application received within thirty (30) days after receiving the application. If formal notice in writing is given to the applicant, the Administrator may extend the time for action for a thirty (30) day period. Failure on the part of the Administrator to act on the application within the approved time limit shall be deemed to constitute approval of the application. The Administrator shall ensure that the proposed development complies with all aspects of this Ordinance.
- 631.06 <u>Lot Area Requirements</u>. Lots in the M-1 zoning district shall be no smaller than one-half acre.
- 631.07 <u>Setback</u>. When the use is adjacent to a residential district, a setback of up to 100 feet is required. When the use is not adjacent to a residential district, no setback is required.
- 631.08 <u>Yard Requirements</u>. The following yard requirements shall be applicable in the M-1 district:
  - 631.08-1 Rear yard. All lots located in the M-1 district must have a rear yard of 25 feet.
  - 631.08-2 <u>Side yard</u>. Property located in the M-1 district must have a minimum side yard of 25 feet.
- 631.09 <u>Height Requirements</u>. Buildings may be erected up to 45 feet in height from grade.

- 631.10 <u>Accessory Structure Requirements</u>. Any accessory structure exceeding one story shall be at least 20 feet from any lot line.
- Residential Zone Buffer. All property lines in the M-1 zoning district that abut a residential district shall be appropriately screened, fenced, walled, or enclosed with a suitable enclosure of a minimum height of 4 feet, unless natural vegetation or wooded areas are used as buffer strips. All buffer strips must be at least 4 feet in height.

# 632.00 GENERAL INDUSTRIAL DISTRICT M-2

- Intent of General Industrial M-2 District. The intent of the M-2 Industrial District is to accommodate, in a manner consistent with the goal of maintaining environmental quality, the more intensive industrial and commercial uses that provide desirable employment. To the extent that these uses should be located near an adequate labor supply, the regulations herein are designed to protect adjacent residential and other districts from adverse effects.
- 632.02 <u>Permitted Uses.</u> Within the General Industrial District M 2 the following uses are permitted as of right:
  - Any use permitted as of right in the M-1 zoning district;
  - Automobile assembling, painting, upholstering, repairing, rebuilding, reconditioning, body and fender work, truck repairing or overhauling, tire retreading or recapping or battery manufacture;
  - 632.02-3 Blacksmith shop, welding or machine shop;
  - 632.02-4 Boat building;
  - 632.02-5 Monumental stone works;
  - 632.02-6 Sawmills and planing mills;
  - 632.02-7 Brick manufacture;
  - 632.02-8 Poultry hatcheries.
- 632.03 <u>Special Uses</u>. The following uses are permitted in an M-2 zoning district after the issuance of a Special Use Permit in accordance with § 501.02:
  - 632.03-1 Mining operations;

	032.03-2	drugs or pharmaceuticals;
	632.03-3	Manufacture or production of aluminum;
	632.03-4	Airports;
	632.03-5	Truck terminals;
	632.03-6	Sand and gravel operations;
	632.03-7	Petroleum storage;
	632.03-8	Junk yards and automobile graveyards;
	632.03-9	Manufacture, production, or processing of asphalt;
	632.03-10	Manufacture of meat or other food or beverage products not enumerated in § 631.02-2.
632.04	Accessory Us district:	es. The following accessory uses are permitted in the M-2 zoning
	632.04-1	Living quarters in the main building of persons employed on the premises;
	632.04-2	Private parking garage;
	632.04-3	Temporary buildings for uses incidental to construction work, such buildings shall be removed upon completion or abandonment of the construction work;
	632.04-4	Parking under the regulations of § 703.
632.05	construction of issued for a reprocesses, she Administrator recommendate this ordinance thirty (30) day	dependent Plan. Before a building permit shall be issued or commenced on any permitted use in the M-2 district, or a permit new use, the plans, in sufficient detail to show the operations and all be submitted to the Zoning Administrator for study. The may refer these plans to the Planning Commission for ion, and may require modifications to comply with the provisions of e. The Administrator shall act on any application received within a safter receiving the application. If formal notice in writing is given int, the Administrator may extend the time for action for a thirty (30)

632.03-2

Manufacture, compounding, processing, packaging, or treatment of

day period. Failure on the part of the Administrator to act on the application within the approved time limit shall be deemed to constitute approval of the application. The Administrator shall ensure that the proposed development complies with all aspects of this Ordinance.

- 632.06 <u>Lot Area Requirements</u>. Lots in the M-2 zoning district shall be no smaller than one-half acre.
- 632.07 <u>Setback</u>. When the use is adjacent to a residential district, a setback of up to 200 feet is required. When the use is not adjacent to a residential district, no setback is required.
- 632.08 <u>Yard Requirements</u>. The following yard requirements shall be applicable in the M-2 district:
  - 632.08-1 Rear yard. All lots located in the M-2 district must have a rear yard of 25 feet.
  - 632.08-2 <u>Side yard</u>. Property located in the M-2 district must have a minimum side yard of 25 feet.
- 632.09 <u>Height Requirements</u>. Buildings may be erected up to 45 feet in height from grade.
- 632.10 <u>Accessory Structure Requirements</u>. Any accessory structure exceeding one story shall be at least 20 feet from any lot line.
- Residential Zone Buffer. All property lines in the M-1 zoning district that abut a residential district shall be appropriately screened, fenced, walled, or enclosed with a suitable enclosure of a minimum height of 6 feet, unless natural vegetation or wooded areas are used as buffer strips. All buffer strips must be at least 6 feet in height.

## 640.00 FLOOD HAZARD DISTRICTS

#### 641.00 FLOOD HAZARD OVERLAY DISTRICT FH- I

Purpose of Flood Hazard District. The purpose of these provisions is to prevent the loss of life and property, the creation of health and safety hazards, the disruption of commerce and governmental services, the extraordinary and unnecessary expenditure of public funds for flood protection and relief, and the impairment of the tax base by:

- Regulating uses, activities, and development which, acting alone or in combination with other existing or future uses, activities, and development, will cause unacceptable increases in flood heights, velocities and frequencies;
- Restricting or prohibiting certain uses, activities, and development from locating within areas subject to flooding;
- Requiring all those uses, activities, and developments that do occur in flood-prone areas to be protected and/or flood proofed against flooding and flood damage;
- Protecting individuals from buying lands and structures which are unsuited for intended purposes because of flood hazards.
- 641.02 <u>Applicability</u>. These provisions shall apply to all lands within the jurisdiction of the Town of Broadway and identified as being in the 100-year floodplain by the Federal Insurance Administration.
- Compliance and Liability. No land shall hereafter be developed and no structure shall be located, relocated, constructed, reconstructed, enlarged, or structurally altered except in full compliance with the terms and provisions of this Ordinance and any other applicable ordinances and regulations which apply to uses within the jurisdiction of this Ordinance. The degree of flood protection sought by the provisions of this ordinance is considered reasonable for regulatory purposes and is based on acceptable engineering methods of study. Larger floods may occur on rare occasions. Flood heights may be increased by man-made or natural causes, such as ice jams and bridge openings restricted by debris. This ordinance does not imply that districts outside the floodplain district, or that land uses permitted within such district will be free from flooding or flood damages. This ordinance shall not create liability on the part of the Town of Broadway or any officer or employee thereof for any flood damages that result from reliance on this ordinance or any administrative decision lawfully made thereunder.
- Abrogation and Greater Restrictions. The provisions of this Section 641 shall prevail over other provisions of this Zoning Ordinance that are less restrictive that the provisions of this Section 641. However, the remaining provisions shall remain in full force and effect to the extent that such provisions are more restrictive than the provisions of this Section 641.
- 641.05 <u>Definitions</u>. For the purposes of this Section 641 only, the following definitions shall apply:

- Base Flood/One Hundred (100) Year Flood. A flood that, on the average, is likely to occur once every 100 years (i.e. that has a one (1) percent chance of occurring each year, although the flood may occur in any year);
- 641.05-2 <u>Base Flood Elevation (BFE)</u>. The Federal Emergency Management Agency designated 100 year water surface elevation;
- Development. Any man-made change to improved or unimproved real estate, including, but not limited to, buildings or other structures, mining, dredging, filling, grading, paving, excavation or drilling operations or storage of equipment or materials;
- 641.05-4 Flood. A general and temporary condition of partial or complete inundation of normally dry land areas from: (1) the overflow of inland or tidal waters; (2) the unusual and rapid accumulation or runoff of surface waters from any source; (3) mudslides (i.e. mudflows) which are proximately caused by flooding as defined herein and are akin to a river of liquid and flowing mud on the surfaces of normally dry land areas, as when earth is carried by a current of water and deposited along the path of the current. (b) The collapse or subsidence of land along the shore of a lake or other body of water as a result of erosion or undermining caused by waves or currents of water exceeding anticipated cyclical levels or suddenly caused by an unusually high water level in a natural body of water, accompanied by a severe storm, or by an unanticipated force of nature, such as a flash flood or by some similarly unusual and unforeseeable event which results in flooding as defined herein;
- 641.05-5 <u>Floodplain</u>. (a) A relatively flat or lowland area adjoining a river, stream or watercourse which is subject to partial or complete inundation; (b) An area subject to the unusual and rapid accumulation or runoff of surface water from any source;
- 641.05-6 <u>Flood-Prone Area.</u> Any land area susceptible to being inundated by water from any source;
- 641.05-7 <u>Floodway</u>. The designated area of the floodplain required to carry and discharge floodwaters of a given magnitude. For the purposes of this Section 641, the floodway shall be capable of accommodating a flood of the one hundred (100) year magnitude;

Historic Structure. Any structure that is (a) Listed individually in the National Register of Historic Places (a listing maintained by the Department of Interior) or preliminarily determined by the Secretary of the Interior as meeting the requirements for individual listing on the National Register; (b) Certified or preliminarily determined by the Secretary of the Interior as contributing to the historical significance of a registered historic district or a district preliminarily determined by the Secretary to qualify as a registered historic district; (c) Individually listed on a state inventory of historic places in states with historic preservation programs which

Individually listed on a local inventory of historic places in communities with historic preservation programs that have been certified either by an approved state program as determined by the Secretary of the Interior or directly by the Secretary of the Interior

have been approved by the Secretary of the Interior; or (d)

in states without approved programs;

Manufactured Home. A structure subject to Federal Regulations, which is transportable in one or more sections; is eight body feet or more in width and forty body feet or more in length in the traveling mode, or is 320 or more square feet when erected on site; is built on a permanent chassis; is designed to be used as a single family dwelling, with or without a permanent foundation when connected to the required facilities; and includes the plumbing, heating, air conditioning, and electrical systems contained in the structure;

New Construction. For the purposes of determining insurance rates, structures for which the "start of construction" commenced on or after the effective date of an initial FIRM or after December 31, 1974, whichever is later, and includes any subsequent improvements to such structures. For floodplain management purposes, "new construction" means structures for which the "start of construction" commenced on or after the effective date of a floodplain management regulation adopted by a community and includes any subsequent improvements to such structures;

Recreational Vehicle. A vehicle which is (a) built on a single chassis; (b) 400 square feet or less when measured at the largest horizontal projection; (c) designed to be self-propelled or permanently towable by a light duty truck; and (d) designed primarily not for use as a permanent dwelling but as temporary living quarters for recreational camping, travel, or seasonal use;

641.05-12 Start of Construction. The date the building permit was issued, provided the actual start of construction, repair, reconstruction, rehabilitation, addition, placement, or other improvement was within 180 days of the permit date. The actual start means either the first placement of permanent construction of a structure on a site, such as the pouring of slab or footings, the installation of piles, the construction of columns, or any work beyond the stage of excavation; or the placement of a manufactured home on a Permanent construction does not include land preparation, such as clearing, grading and filling; nor does it include the installation of streets and/or walkways; nor does it include excavation for a basement, footings, piers, or foundations or the erection of temporary forms; nor does it include the installation on the property of accessory buildings, such as garages or sheds not occupied as dwelling units or not part of the main For a substantial improvement, the actual start of structure. construction means the first alteration on any wall, ceiling, floor, or other structural part of a building, whether or not the alteration affects the external dimensions of the building;

641.05-13 <u>Substantial Damage</u>. Damage of any origin sustained by a structure whereby the cost of restoring the structure to its before damaged condition would equal or exceed fifty (50) percent of the market value of the structure before the damage occurred;

Substantial Improvement. Any reconstruction, rehabilitation, addition, or other improvement of a structure, the cost of which equals or exceeds fifty (50) percent of the market value of the structure before the "start of construction" of the improvement. This term includes structures which have incurred "substantial damage" regardless of the actual repair work performed. The term does not, however, include either: (1) any project for improvement of a structure to correct existing violations of state or local health, sanitary, or safety code specifications which have been identified by the local code enforcement official and which are the minimum necessary to assure safe living conditions; or (2) any alteration of a "historic structure", provided that the alteration will not preclude the structures continued designation as a "historic structure."

## 641.06 <u>Description of Districts.</u>

641.06-1 <u>Basis of Districts</u>. The FH-1 Flood Hazard Districts shall include areas subject to inundation by waters of the one hundred (100)-year flood, and shall be divided into a Floodway District and

a Flood-Fringe District. The basis for the delineation of these districts shall be the Flood Insurance Study (FIS) and accompanying maps for the Town of Broadway prepared by the Federal Emergency Management Agency, Federal Insurance Administration, dated February 2, 2008 as amended.

- Floodway District. The Floodway District is delineated, for purposes of this ordinance, using the criterion that certain areas within the floodplain must be capable of carrying the waters of the one hundred (100)-year flood without increasing the water surface elevation of that flood more than one (1) foot at any point. The areas included in this District are specifically defined in Table 2 of the above-referenced Flood Insurance Study and shown on the accompanying Flood Insurance Rate Map.
- 641.06-3 <u>Flood-Fringe District</u>. The Flood-Fringe District shall be that area of the one hundred (100)-year floodplain not included in the Floodway District. The basis for the outermost boundary of the District shall be the one hundred (100)-year flood elevations contained in the flood profiles of the above referenced Flood Insurance Study and as shown on the accompanying Flood Insurance Rate Map.

## 641.07 <u>Overlay Concept.</u>

- The Flood Hazard Districts described above shall be overlays to the existing underlying districts as shown on the Official Zoning Ordinance Map, and as such, the provisions for the Flood Hazard Districts shall serve as a supplement to the underlying district provisions.
- Any conflict between the provisions or requirements of the Flood Hazard Districts and those of any underlying district, the more restrictive provisions and/or those pertaining to the Flood Hazard districts shall apply.
- 641.08 Official Zoning Map. The boundaries of the Flood Hazard Districts are established as shown on the Flood Insurance Rate Map which is declared to be a part of this Ordinance and which shall be kept on file at the Zoning Administrator's office.
- 641.09 <u>District Boundary Changes</u>. The delineation of any of the Flood Hazard Districts may be revised where natural or man-made changes have occurred and/or where more detailed studies have been conducted or undertaken by the U.S. Army Corps

of Engineers or other qualified agency, or an individual documents the need for such change. However, prior to any such change, approval must be obtained from the Federal Insurance Administration and, further, such change must comply with the procedures for amendment of the Zoning Map under Virginia law.

- 641.10 <u>Interpretation of District Boundaries</u>. Initial interpretations of the boundaries of the Flood Hazard Districts shall be made by the Zoning Administrator. Should a dispute arise concerning the boundaries of any of the Districts, an appeal shall lie to the Board of Zoning Appeals as provided by Virginia law. The person questioning or contesting the location of the district boundary shall be given a reasonable opportunity to present his case to the Board and to submit his own technical evidence if he so desires.
- 641.11 District Provisions. All uses, activities, and development occurring within any Flood Hazard District shall be undertaken only upon the issuance of a Zoning Permit. Such development shall be undertaken only in strict compliance with the provisions of the Ordinance and with all other applicable codes and ordinances such as the Virginia Uniform Statewide Code, and the Code of Federal Regulations. Prior to the issuance of any such permit, the Zoning Administrator shall require all applications to include evidence of compliance with all applicable state and federal laws. Under no circumstances shall any use, activity, and/or development adversely affect the capacity of the channels or flood ways of any watercourse, drainage ditch, or any other drainage facility or system. Prior to any proposed alteration or relocation of any channels or of any watercourse, stream, etc., within this jurisdiction a permit shall be obtained from the U. S. Corps of Engineers, the Virginia State Water Control Board, and the Virginia Marine Resources Commission (a joint permit application is available from any of these organizations). Furthermore, notification of the proposal shall be given by the applicant to all affected adjacent jurisdictions, the Division of Soil and Water Conservation (Department of Conservation and Recreation), and the Federal Insurance Administration. Flood carrying capacity within an altered or relocated portion of any watercourse must be maintained.
- Applications for Zoning Permit. All applications for a Zoning Permit in a Flood Hazard District and all building permits issued for the floodplain shall be reviewed to assure that the following information is incorporated. A record of all such information shall be recorded by the Zoning Administrator including a record of such certificates which includes the specific elevation (in relation to sea level) to which such structures are flood proofed.
  - For structures to be elevated, the elevation of the lowest floor (including basement);

- For structures to be flood proofed (non-residential only), the elevation to which the structure will be flood proofed;
- The elevation of the one hundred (100)-year flood;
- Topographic information showing existing and proposed ground elevations;
- Information and certifications demonstrating that all new structures and substantial improvements will be constructed with materials resistant to flood damage;
- Information and certifications demonstrating that all new structures and substantial improvements will be constructed by methods and practices that minimize flood damages;
- 641.12-7 Information and certifications demonstrating that all such proposals are consistent with the need to minimize flood damages within the flood-prone area;
- All new subdivision proposals and other proposed developments greater than fifty (50) lots or five (5) acres, whichever is the lesser, shall include within such proposals base flood elevation data;
- Information and certifications demonstrating that for all new 641.12-9 construction and substantial improvements, fully enclosed areas below the lowest floor that are usable solely for parking of vehicles, building access or storage in an area other than a basement and which are subject to flooding shall be designed to automatically equalize hydrostatic flood forces on exterior walls by allowing for the entry and exit of floodwaters. Designs for meeting this requirement must either be certified by a registered professional engineer or architect or meet or exceed the following minimum criteria. A minimum of two openings having a total net area of not less than one square inch for every square foot of enclosed area subject to flooding shall be provided. The bottom of all openings shall be no higher than one foot above grade. Openings may be equipped with screens louvers, valves, or other coverings or devices provided that they permit the automatic entry and exit of floodwaters;
- A registered professional engineer or architect shall develop and/or review structural design, specifications, and plans for the construction and shall certify that the design and methods of

construction are in accordance with acceptable standards of practice.

- Manufactured Homes. In any Flood Hazard District manufactured homes that are placed or substantially improved on sites shall be elevated on a permanent foundation such that the lowest floor of the manufactured home is elevated to or above the base flood elevation and be securely anchored to an adequately anchored foundation system to resist floatation collapse and lateral movement of the structure resulting from hydrodynamic and hydrostatic loads.
- Recreational Vehicles. Recreational vehicles placed on sites in any Flood Hazard District shall either (a) be on the site for fewer than 180 consecutive days, be fully licensed and ready for highway use; or (b) meet the permit requirements for placement and the elevation and anchoring requirements for manufactured homes as specified above. A recreational vehicle is ready for highway use if it is on its wheels or jacking system, is attached to the site only by quick disconnect type utilities and security devices, and has no permanently attached additions.
- Floodway District. In the Floodway District no encroachments, including fill, new construction, substantial improvements, or other development shall be permitted unless it has been demonstrated through hydrologic and hydraulic analyses performed in accordance with standard engineering practice that the proposed encroachment would not result in any increase in the one hundred (100)-year flood elevation.
- Permitted Uses in Floodway District. In the Floodway District, the following uses and activities are permitted as of right provided that they are in compliance with the provisions of the underlying area and are not prohibited by any other ordinance, and provided that they do not require structures, fill, or storage of materials and equipment:
  - Agricultural uses, such as general farming, pasture, grazing, outdoor plant nurseries, horticulture, truck farming, forestry, sod farming, and wild crop harvesting;
  - Public and private recreational uses and activities such as parks, day camps, picnic grounds, golf courses, boat launching and swimming areas, horseback riding and hiking trails, wildlife and nature preserves, game farms, fish hatcheries, trap and skeet game ranges, and hunting and fishing areas;
  - 641.16-3 Individual septic drain fields serving on site sewage disposal systems for single-family homes, not including septic tanks, pumps, or pretreatment facilities, provided that (a) no other site for

a septic drain field can be found on the lot outside the floodplain district that meets the requirements of the Commonwealth health department's sewage handling and disposal regulations; and (b) the septic drain field is designed and installed to meet the design criteria of the Town Code.

- Special Uses. The following uses are permitted in a FH-1 zoning district after the issuance of a Special Use Permit in accordance with § 501.02 and the provisions of this Section 641, and provided further that that are in compliance with the regulations of the underlying district:
  - 641.17-1 Structures accessory to the use and activities in Section A above;
  - 641.17-2 Utilities and public facilities and improvements such as railroads, streets, bridges, transmission lines, pipelines, water and sewage treatment plants, and other similar or related uses;
  - Water-related uses and activities such as marinas, docks, wharfs, piers, etc.;
  - Extraction of sand, gravel, and other materials (where no increase in level of flooding or velocity is caused thereby);
  - 41.17-5 Temporary uses such as circuses, carnivals, and similar activities;
  - Storage of materials and equipment provided that they are not buoyant, flammable or explosive, and are not subject to major damage by flooding, or provided that such material and equipment is firmly anchored to prevent flotation or movement, and/or can be readily removed from the area within the time available after flood warning;
  - Other similar uses and activities provided they cause no increase in flood heights and/or velocities. All uses, activities, and structural development shall be undertaken in strict compliance with the flood-proofing provisions contained in all other applicable codes and ordinances;
  - Other similar uses, encroachments, and activities that would result in an increase in base flood elevations within the adopted regulatory floodway, provided that an application for a conditional FIRM and floodway revision is completed and fulfills the requirement for such revisions.

- Accessory Uses. The following accessory uses are permitted as of right in a Floodway Portion of a FH-1 Flood Hazard District, but are the only accessory uses so permitted; and provided, further, that no such accessory use shall contain an accessory structure without the issuance of a Special Use Permit in accordance with § 641.16-1:
  - Accessory residential uses such as yard areas, gardens, play areas, and pervious loading areas;
  - Accessory industrial and commercial uses, such as yard areas, pervious parking and loading areas, airport landing strips, etc.
- 641.19 <u>Prohibited Uses.</u> The following uses and activities are prohibited from any Floodway District:
  - 641.19-1 All dwellings and all manufactured homes;
  - 641.19-2 All businesses and industries;
  - Any use where people congregate in permanent structures, e.g., churches, schools, institutional uses, etc;
  - 641.19-4 Any agricultural buildings housing animals;
  - 641.19-5 Filling of land, except for restoration or reclamation purposes;
  - 641.19-6 Individual onsite waste disposal systems (septic tanks, etc.).
- Flood-Fringe. In the Flood-Fringe portion of the Flood Hazard District, the development and/or use of land shall be permitted in accordance with the regulations of the underlying district provided that all such uses, activities, and/or development shall be undertaken in strict compliance with the flood proofing and related provisions contained in the Virginia Uniform Statewide Building Code and all other applicable codes and ordinances, including the applicable portions of this Section 641.
- 641.21 <u>Design Criteria for Utilities and Facilities</u>. The following design criteria shall apply to any of the following facilities in any Flood Hazard District:
  - 641.21-1 <u>Sanitary Sewer Facilities</u>. All new or replacement sanitary sewer facilities shall be designed to minimize or eliminate infiltration of flood waters into the systems and discharges from the systems into the flood waters. In addition, they should be located and

- constructed to minimize or eliminate flood damage and impairment.
- Water Facilities. All new or replacement water facilities shall be designed to minimize or eliminate infiltration of flood waters into the system and be located and constructed to minimize or eliminate flood damages.
- Drainage Facilities. All storm drainage facilities shall be designed to convey the flow of surface waters without damage to persons or property. The systems shall ensure drainage away from buildings and onsite waste disposal sites. Drainage plans shall be consistent with local and regional drainage plans. The facilities shall be designed to prevent the discharge of excess runoff onto adjacent properties. The Zoning Administrator may require a primarily underground system to accommodate frequent floods and a secondary surface system to accommodate larger, less frequent floods.
- 641.21-4 <u>Utilities</u>. All utilities, such as gas lines, electrical and telephone systems being placed in a Flood Hazard District should be located, elevated (where possible), and constructed to minimize the chance of impairment during a flooding occurrence.
- 641.21-5 <u>Streets and Sidewalks</u>. Streets and sidewalks should be designed to minimize their potential for increasing and aggravating the levels of flood flow. Drainage openings shall be required to sufficiently discharge flood flows without unduly increasing flood heights.
- Special Use Permits and Variances: Additional Factors to be Satisfied. In passing upon applications for Special Use Permits in the Flood Hazard District, the Town Council shall consider the following factors. In considering the issuance of a variance, the Board of Zoning Appeals shall also consider the following factors. These factors are in addition to the relevant factors and procedures specified in other sections of the zoning ordinance and Virginia law:
  - The danger to life and property due to increased flood heights or velocities caused by encroachments. No Special Use Permit or variance shall permit any proposed use, development, or activity within any Floodway District that will cause any increase in the one hundred (100) -year flood elevation;

641.22-2	The danger that materials may be swept on to other lands or downstream to the injury of others;	
641.22-3	The proposed water supply and sanitation systems and the ability of these systems to prevent disease, contamination, and unsanitary conditions;	
641.22-4	The susceptibility of the proposed facility and its contents to flood damage and the effect of such damage on the individual owners;	
641.22-5	The importance of the services provided by the proposed facility to the community;	
641.22-6	The requirements of the facility for a waterfront location;	
641.22-7	The availability of alternative locations not subject to flooding for the proposed use;	
641.22-8	The compatibility of the proposed use with existing development and development anticipated in the foreseeable future;	
641.22-9	The relationship of the proposed use to the comprehensive plan and floodplain management program for the area;	
641.22-10	The safety of access by ordinary and emergency vehicles to the property in time of flood;	
641.22-11	The expected heights, velocity, duration, rate of rise, and sediment transport of the flood waters expected at the site;	
641.22-12	Whether such Special Use Permit or variance is necessary to preserve the historic character and design of a historic structure, when the proposed repair or rehabilitation of such structure will not preclude the structure's continued designation as a historic structure and the variance is the minimum necessary to preserve the historic character and design of the structure.	

Special Procedures of Special Use Permits and Variances. The Town Council and Board of Zoning Appeals may refer any application and accompanying documentation pertaining to any request for a Special Use Permit or variance to any engineer or other qualified person or agency for technical assistance in evaluating the proposed project in relation to flood heights and velocities, and the adequacy of the plans for protection and other related matters. The expenses

thereof shall be chargeable to the applicant. Special Use Permits and variances in the Flood Hazard District shall be issued only after the factfinder has determined that the granting of such will not result in: (a) unacceptable or prohibited increases in flood heights; (b) additional threats to public safety; (c) extraordinary public expense; (d) nuisances; (e) fraud or victimization of the public; or (f) conflict with local laws or ordinances (other than the law or ordinance from which any variance is being sought). A variance shall be issued only after the Board of Zoning Appeals has determined that a variance will be the minimum required to provide relief from any exceptional hardship to the applicant. Upon the issuance of a variance, the Zoning Administrator shall notify the applicant thereof, in writing, that the issuance of the variance to construct a structure below the one hundred (100)-year flood elevation: (a) increases the risks to life and property; and (b) will result in increased premium rates for flood insurance as much as \$25 for \$100 of insurance. A record shall be maintained of the above notification as well as all variance actions, including justification for the issuance of the variances. Any variances which are issued shall be noted in the annual or biennial report submitted to the Federal Insurance Administrator.

- Existing Structures in Floodplain Districts. A structure or use of a structure or premises which lawfully existed before the enactment of these provisions, but which is not in conformity with these provisions, may be continued subject to the following conditions and the other conditions of this Ordinance:
  - Existing structures in the Flood Hazard District shall not be expanded or enlarged unless it has been demonstrated through hydrologic and hydraulic analyses performed in accordance with standard engineering practice that the proposed expansion would not result in any increase in the one hundred (100)-year flood elevation.
  - Any modifications, alteration, repair, reconstruction, or improvement of any kind to a structure and/or use, located in any floodplain area shall be elevated and/or flood proofed to the greatest extent possible and shall be constructed according to the Virginia Uniform Statewide Building Code.
  - The modification, alteration, repair, reconstruction, or improvement of any kind to a structure and/or use, regardless of its locations in a floodplain area, shall be undertaken only in full compliance with the provisions of this ordinance and the Virginia Uniform Statewide Building Code. The entire structure must be brought to current code.

Violations and penalties. In addition to any other penalty prescribed by this zoning ordinance, all other actions are hereby reserved, including an action in equity for the proper enforcement of this Section 641. The imposition of a fine or penalty for any violation of, or noncompliance with, this article shall not excuse the violation or noncompliance to permit it to continue; and all such persons shall be required to correct or remedy such violations or noncompliances within a reasonable time. Any structure constructed, reconstructed, enlarged, altered, or relocated in noncompliance with this article may be declared by the Town Council to be a public nuisance and abatable as such. Flood insurance may be withheld from structures constructed in violation of this Section 641.

# ARTICLE 7 USE REGULATIONS

#### 701.00 AREA REGULATIONS

- 701.01 <u>Additional Regulations Regarding Yard and Setback Requirements</u>. The yard and setback requirements contained in Article 6 are subject to the following additional regulations:
  - 701.01-1 <u>Signs</u>. Except as expressly provided in § 705, yard and setback requirements shall not apply to signs permitted under § 705 or excluded from the operation of § 705 by the terms therein.
  - Party Walls. Minimum side yard requirements of this Ordinance shall not be applicable to a property line upon which rests the common or party wall of any building (including a townhouse). For the purposes of determining compliance with lot area and lot width requirements (other than the specific requirements for townhouses contained in Section 708), the property lines upon which rests the common or party wall of any building shall be ignored.
  - 701.01-3 <u>Encroachments Permitted into Front Yards</u>. The following structures shall be permitted to encroach into any yard, including front yards, provided applicable sight distance and fire safety requirements are met and maintained:
    - (a) Fences pursuant to section 709.00;
    - (b) Ground level terraces, patios, decks, and porches, which do not include a permanently roofed-over terrace or porch, porch except as permitted in the R-3 and R-5 Districts and which do not project into a required yard for a distance exceeding ten (10) feet;
    - (c) Awnings or canopies, provided they do not project more than eight feet from the existing building face;
    - (d) Bay windows, chimneys, and overhanging eaves or gutters projecting no more than three feet into the yard, and in the case of overhanging having a vertical surface whose area is more than twenty five (25) percent of the area obtained by multiplying the mean height of the structure by the length of the structure along the yard which is violated;

- (e) Arbors and trellises not exceeding ten feet in height, provided that such structures do not reduce the width of the yard to less than three feet;
- (f) Flag poles that do not exceed 25 feet in height;
- (g) Recreational equipment;
- (h) Non-enclosed steps;
- (i) Landscape features and planting boxes.

**Ordinance of December 6, 2011** added the phrase "porch except as permitted in the R-3 and R-5 Districts" after the word "porch" in Section 701.01-3(b).

**Ordinance of July 7, 2015** added the phrases "(including a townhouse)" and "(other than the specifric requirements for townhouses contained in Section 708)" to § 701.01-2.

- Accessory Structures and Yards. Accessory structures may encroach into the side or rear yards of any lot, but no accessory structure other than structures described in § 701.01-3 shall encroach into the front yard of a lot. Accessory structures shall not be located nearer to the lot lines than permitted under the terms of Article 6 for the zoning district in which the accessory structure is located. In the event that this Section 701 requires a greater distance between an accessory structure and a lot line than the provisions of Article 6 require, or otherwise imposes a more restrictive requirement on an accessory structure that encroaches into a yard, the more restrictive provisions of this Section 701 shall apply.
- 701.01-5 <u>Decks</u>. Decks not on the ground level are permitted to encroach into rear and side yards only. All decks must be no closer than 10 feet to a property line.
- 701.01-6 Compliance with Average Setback. Minimum setback requirements of this Ordinance (included modified corner setbacks under § 701.03 herein) for yards facing streets shall not apply to any lot where the average setback on developed lots within the same block and zoning district and fronting on the same street is less than the minimum. In such cases, the setback on such lot may be less than the required setback but not less that the average of the existing setbacks on the existing developed lots.
- 701.02 <u>Special Provisions for Corner Lots in Residential Districts.</u> In residential districts, the following rules shall govern corner lots:

- 701.02-1 Of the two (2) sides of a corner lot the front shall be deemed to be the shorter of the two sides fronting on streets.
- The side yard on the side facing the side street shall be twenty-five (25) feet or more for both main and accessory buildings and structures, subject to encroachments permitted under other terms of this Ordinance. Notwithstanding the provisions of § 701.06, no accessory structure shall be located in the side yard of a corner lot on the side facing the street in a residential district.
- For subdivisions platted after the enactment of this Ordinance, each corner lot shall have a minimum lot width equal to twenty-five (25) feet greater than the minimum lot width requirement for other lots within the applicable residential zoning district.

## 701.03 Conditional Height Regulation.

- The height limits applicable to any dwelling may be increased to forty-five (45) feet or three stories, whichever is less, provided that each side yard is the sum of twenty (20) feet plus one (1) foot for each foot or portion thereof by which the building height exceeds thirty-five (35) feet.
- A public or semi-public building such as a school, church, library, or general hospital may be erected to a height of sixty (60) feet from grade provided that required front, side, and rear yards shall be increased one (1) foot for each foot or portion thereof by which the building height exceeds thirty-five (35) feet.
- Church spires, belfries, cupolas, monuments, water, silo, barns, towers, chimneys, flues, flag poles, television and radio antenna are exempt, except for those structures exceeding two hundred (200) feet in height where written approval from the Federal Aviation Administration is necessary. Parapet walls may be up to four (4) feet above the height of the building on which the walls rest.
- 701.04 <u>Existing Lots Shall Abut on a Developed Street</u>. Each lot (except lots for townhouses) shall abut on a developed public street. Development on said existing lot shall not be construed as creating an obligation upon the Town to pay for grading, paving, or for sidewalks, sewers, curb and guttering improvements, or construction.

# 702.00 ADDITIONAL BUILDINGS ON A SINGLE LOT

- 702.01 <u>Additional Dwellings</u>. Additional dwellings on a single lot may be permitted provided that such additional dwellings are arranged in such a manner so that if the lot or parcel of land is ever subdivided to place each dwelling on a separate lot no substandard lots or violations of this Ordinance would be created.
- 702.02 <u>Temporary Buildings</u>. Temporary buildings used in conjunction with construction work only may be permitted in any district but shall be removed immediately upon completion or abandonment of construction.
- Additional Rules for Accessory Structures. Where an accessory building is attached to the main building, a substantial part of one wall of the accessory building shall be an integral part of the main buildings or such accessory building shall be attached to the main building in a substantial manner by a roof, and therefore such attached accessory building shall be considered part of the main building and shall comply in all respects with the requirements applicable to the main building. All accessory structures must be on the same lot at the main building.

#### 703.00 OFF-STREET PARKING

- 703.01 Requirement for Off-Street Parking. Off-street automobile storage or parking space shall be provided on every lot on which any permitted or special use is established in accordance with this Ordinance.
- 703.02 <u>General Requirements.</u> For the purpose of this Ordinance, the following general requirements are specified:
  - Definitions. The term "off-street parking space" shall mean a space at least ten (10) feet wide and twenty (20) feet in length with a minimum net area of two hundred (200) square feet, excluding area for egress and ingress and maneuvering of vehicles. A minimum of twenty-two (22) feet between parking rows back-to-back shall be clear for maneuvering into and out of parking spaces.
  - 703.02-2 <u>To Be Located on Same Lot</u>. Parking spaces for all dwellings shall be located on the same lot with the main buildings to be served.
  - 703.02-3 Permissible Off-Lot Locations. If an off-street parking space cannot be reasonably provided on the same lot on which the main use is conducted, such space may be provided on other off-street

property, provided such space lies within six hundred (600) feet of the property line of such main use.

- Multiple Uses. The required number of parking spaces for any number of separate uses may be combined in one lot, but the required space assigned to one use may not be assigned to another use at the same time.
- 703.02-5 <u>Change in Use</u>. Area reserved for off-street parking in accordance with the requirements of this Ordinance shall not be reduced in the area, encroached upon or changed to any other use unless the use which it serves is discontinued or modified.
- 703.02-6 Existing Nonconformities. Off-street parking existing at the effective date of this Ordinance in connection with the operation of an existing building shall not be reduced to an amount less than hereinafter required for a similar new building. Existing off-street parking which is provided in an amount less than the requirements stated hereinafter shall not be further reduced. The violation of the parking requirements contained herein shall not itself render any existing structure that those parking areas serve a nonconforming structure or use.
- 703.03 <u>Site Requirements.</u> All off-street parking shall be laid out, constructed, and maintained in accordance with the following requirements:
  - Any off street parking areas (whether required under this Ordinance, or whether in excess of those required hereunder), together with any driveways, shall be surfaced with an asphaltic, concrete, or other properly bound pavement so as to provide a durable and dustless surface.

**Ordinance of October 3, 2023** amended this section to extend its application to all parking and driveway spaces. The former paragraph only applied to parking areas of five (5) spaces or more.

- Adequate lighting shall be provided if off-street parking spaces are to be used at night. The lighting shall be arranged and installed to minimize glare on property in a residential district. Excluded from this requirement are private driveways and parking lots which have five (5) or less parking spaces.
- 703.03-3 Off-street facilities shall be drained to eliminate standing water and prevent damage to abutting property and/or public streets and

alleys and surfaced with erosion-resistant material in accordance with applicable specifications. Off-street parking areas shall be maintained in a clean, orderly, and dust-free condition at the expense of the owner or lessee and not used for the sale, repair, dismantling, or servicing of any vehicles, equipment, materials, or supplies.

- Off-street parking spaces in commercial and industrial districts, and in lots in residential districts with more than five (5) spaces, shall be clearly marked by painted lines, curbs, or other means to indicate individual spaces. Signs or markers shall be used as necessary to ensure efficient traffic operation.
- Access to the parking lot from adjacent streets shall be provided in accordance with the following requirements or as required by the Virginia Department of Transportation.
  - (a) Access shall be provided by means of not more than two (2) driveways for the first 120 feet of frontage along any one (1) street and shall have not more than one (1) additional driveway for each additional 150 feet of street frontage.
  - (b) Entrances or exits shall have access widths along the edge of the street pavement of not more than forty (40) feet as measured parallel to the street, and shall be located not closer than twenty (20) feet to street intersections nor ten (10) feet to adjoining property lines.
  - (c) Access driveways shall not be closer than twenty (20) feet from adjacent driveways at any point from the edge of the pavement to the right-of-way line.
- Parking Space Requirements for All Districts. In all districts, there shall be provided off-street automobile storage or parking space with vehicular access to a street or alley, and shall be equal in area to at least the minimum requirement for the specific land use set forth.

# 703.04-1 Dwellings:

Land Use		Parking Requirements		
(a)	One (1) and two (2) families	Two (2) spaces for each dwelling unit.		
(b)	Multi-family, townhouses	Two (2) spaces per dwelling unit.		
(c)	Hotels	One (1) space for each bedroom plus one (1) additional space for each two (2) employees.		
(d)	Manufactured home parks/subdivisions	Two (2) spaces per manufactured home.		
(e)	Travel trailer parks	One (1) space for each travel trailer, motor home, or camper.		
(f)	Boarding and rooming houses, dormitories	One (1) space for each bedroom.		
(g)	Bed and Breakfasts and Short Term Rentals	Two (2) spaces for the proprietor. One (1) for each two (2) employees plus another if there is an odd number of employees. One (1) for each guest room. Provided, however, that the Zoning Administrator may reduce the required number of spaces by two (2) during any period in which the Zoning Administrator determines that the proprietor is not residing at the location, and may subsequently require such spaces if the Zoning Administrator determines that the proprietor is residing at the location.		
(h)	Group Housing Project for Older Persons	One and one-half (1/2) spaces per dwelling unit.		

**Ordinance of May 2, 2023** removed the references to motels, inns, and bed and breakfasts from Section 703.04-1(c), added a reference to Short Term Rentals in Section 703.04-1(g), and added the final sentence to the "Parking Requirements" of Section 703.04-1(g).

# 703.04-2 Public Assembly:

Land Use	Parking Requirements
(a) Repealed by Ordinance of December 6, 2011	

(b)	Private clubs, lodges and fraternal or sororal buildings not providing overnight accommodations	One (1) space for each ten (10) seats or six hundred (600) square feet, whichever is greater.	
(c)	Theaters, churches and other places of worship, auditoriums, coliseums, stadiums, and similar places of assembly	One (1) space for each ten (10) seats.	
(d)	Schools, including kindergartens, playschools, and day care centers	One (1) space for each four (4) seats in assembly hall, or one (1) space for each employee, including teachers and administrators, plus five (5) spaces per classroom for high schools and colleges, whichever is greater.	
(e)	Skating rinks, dance halls, exhibition halls, pool rooms, and other places of amusement or assembly without fixed seating arrangements	One (1) space for each two hundred (200) square feet of floor area.	
(f)	Bowling alleys	Two (2) spaces for each alley.	
(g)	Libraries, museums	One (1) space for each five hundred (500) square feet of gross floor area.	

**Former Section 703.04-2(a)** provided the following parking regulation for "churches and religious assemblies": "One (1) space for each ten (10) seats in the main auditorium or sanctuary."

Ordinance of December 6, 2011 added the phrase "ten (10) seats

or" after the word "each" in Section 703.04-2(b), and added the phrase "whichever is greater" at the end of the same section. The same ordinance added the phrase "churches and other places of worship," after the word "theaters" in Section 703.04-2(c).

## 703.04-3 Health Facilities:

Land Use		Parking Requirements	
(a)	Hospitals and similar uses	One (1) space for each two (2) beds, plus one (1) space for each staff doctor, plus one (1) space for each four (4) employees, on the maximum working shift.	
(b)	Kennels and animal hospitals	A net parking area equal to thirty (30) percent of the total enclosed or covered area.	
(c)	Medical, dental, and health offices and clinics	At least ten (10) spaces. Three (3) additional parking spaces shall be furnished for each doctor and dentist having office in such clinics in excess of three (3) doctors or dentists plus one (1) space per each two (2) staff and employees.	

(d)	Homes for adults and similar uses	One (1) space for each four (4) beds plus one (1) space for every three (3) employees.

# 703.04-4 Businesses:

Land Use		Parking Requirements		
(a)	Automobile repair establishments	One (1) space for each three hundred (300) square feet, with a minimum of ten (10) spaces.		
(b)	Food stores	One (1) space for each two hundred (200) square feet of floor area designated for retail sales only.		
(c)	Restaurants, including bars, cafes, taverns, night clubs, lunch counters, and all similar dining and/or drinking establishments	One (1) space for each four (4) seats provided for patron use, plus one (1) space per employee on average shift.		
(d)	Office buildings, including banks, business, commercial and professional offices and buildings but not including medical, dental, and health offices and clinics	One (1) space for each three hundred (300) square feet of ground floor area, plus one (1) space for each five hundred (500) square feet of upper floor space.		
(e)	General business, commercial or personal service establishments catering to the retail trade	One (1) space for each three hundred (300) square feet of floor area designated for retail sales.		
(f)	Governmental Offices	One (1) space for each three hundred (300) square feet of ground floor area, plus one (1) space for each five hundred (500) square feet of upper floor area, and one (1) space for each governmental vehicle.		
(g)	Shopping Centers	One (1) space per two hundred (200) square feet of retail sales area.		
(h)	Furniture stores	One (1) space for each one thousand (1,000) square feet of gross floor area.		
(i)	Public utilities such as telephone exchanges and substations, radio and television stations, and electric power and gas substations	One (1) space for each employee on the maximum shift plus a parking area equal to twenty-five (25) percent of the gross floor area.		
(j)	Mortuaries and funeral parlors	Five (5) spaces per parlor unit or chapel unit, or one (1) space per four (4) seats, whichever is greater.		

(k)	Auction House	One (1) space per one hundred (100) square feet of building space that is open to the public.
-----	---------------	---

Ordinance of December 6, 2011 amended Section 703.04-4(e) to chance the phrase "two hundred (200)" to "three hundred (300)".

703.04-6

703.04-5 Industries:

Land Use		Parking Requirements	
(a)	Manufacturing and industrial establishments not catering to the retail trade	One (1) space for each three (3) employees on the maximum working shift, plus one (1) space for each company vehicle or manufactured equipment operating from the premises.	
(b)	Wholesale establishments	One (1) space for every fifty (50) square feet of customer service area, plus two (2) spaces for each three (3) employees on the maximum working shift, plus one (1) space for each company vehicle operating from the premises.	

Downtown Business Overlay District. foregoing provisions relating to parking, any business located within the Downtown Business Overlay District shall be exempt from the provisions of this Ordinance relating to parking. The Downtown Business Overlay District shall consist of all property north of Broadway Avenue, east of the Southern Railroad right of way, south of Lee Street, and west of Central and High Streets. Any business within the above described Overlay District shall restrict the parking of its employees to privately or Town owned areas. However, to the extent that a business within the Downtown Business Overlay District actually provides off-street parking or

> driveways, even though this Ordinance does not require them to do so, such off-street parking or driveways shall comply with the

Ordinance of October 3, 2023 added the final sentence of this seciton.

provisions of Section 703.03.

703.05 Off-Street Loading and Unloading Space. Off-street loading and unloading spaces shall be provided as hereinafter required by this Ordinance.

> 703.05-1 Floor area over ten thousand (10,000) square feet. There shall be provided for each hospital, hotel, commercial, or industrial

Notwithstanding the

building, or similar use requiring the receipt or distribution of materials or merchandise, and having a floor area of more than ten thousand (10,000) square feet, at least one off-street loading space for each ten thousand (10,000) square feet of floor space or fraction thereof. Such space shall be so located as not to hinder the free movement of pedestrians and vehicles over a sidewalk, road, or alley.

- Floor area less than ten thousand (10,000) square feet. There shall be provided for each hospital, hotel, commercial, or industrial building requiring receipt or distribution of materials or merchandise and having a floor area of less than ten thousand (10,000) square feet, sufficient off-street loading space (not necessarily a full space if shared by an adjacent establishment) so located as not to hinder the free movement of pedestrians and vehicles over a sidewalk, road, or alley.
- 703.05-3 <u>Size</u>. Size of off-street loading spaces shall have minimum dimensions of fourteen (14) feet in height, twelve (12) feet in width, and fifty (50) feet in length.
- 703.05-4 <u>Connection to Road or Alley.</u> Each required off-street loading space shall have direct access to a road or alley or have a driveway which offers satisfactory ingress and egress for trucks.
- 703.05-5 Permanent Reservation. Area reserved for off-street loading in accordance with the requirements of this Ordinance shall not be reduced in area or changed to any other use unless the use which is served is discontinued or modified, except where equivalent loading space is first provided that meets the requirements herein.
- Paving. Off street loading and unloading spaces required under this Section, or used on a regular basis for such purpose but not required under this Section, shall be surfaced with an asphaltic, concrete, or other properly bound pavement so as to provide a durable and dustless surface.

Ordinance of October 3, 2023 added Section 703.05-6.

- 703.06. Shared Parking. For uses located on contiguous but separate sites, the number of required parking spaces may be reduced in accordance with the following provisions:
  - (a) The uses are contiguous uses.

- (b) Parking areas of the respective uses are connected by safe and convenient pedestrian access, as well as by automobile access.
- (c) A shared parking agreement is submitted and approved by the Zoning Administrator. The agreement will be binding on the current and future property owners as long as the permitted uses remain substantially the same, and shall be in a form suitable for recordation among the land records of Rockingham County.
- (d) Reductions in required parking may be approved by the Zoning Administrator, at the request of the applicant, in accordance with the following calculation provided by the applicant:
  - i. The total number of parking spaces required for each land use is determined in accordance with Sec. 703.04.
  - ii. Determine the number of spaces needed by each use for each of four time periods, by multiplying the parking required for each use by the corresponding percentage of that use for that time period. The time periods are: (1) weekdays 8:00 a.m. to 6:00 p.m.; (2) weekdays 6:00 p.m. to 11:00 p.m., (3) weekend days 8:00 a.m. to 6:00 p.m., and (4) weekend days 6:00 p.m. to 11:00 p.m.
  - iii. Calculate the total number of spaces needed for all uses for each time period.
  - iv. The time period with the highest number of parking spaces required for the sum of all uses shall be the number of parking spaces required.

Ordinance of December 6, 2011 added this section.

#### 704.00 HOME OCCUPATIONS

- 704.01 <u>Purpose</u>. The purposes of the home occupation supplemental regulations and performance standards of this section are to:
  - 704.01-1 Establish criteria for the operation of home occupations in dwelling units;

- Permit and regulate the conduct of home occupations as accessory use in a dwelling unit, whether owner or renter-occupied;
- 704.01-3 Ensure that such home occupations are compatible with, and do not have a deleterious effect on, adjacent and nearby residential properties and uses;
- Ensure that public and private services such as streets, sewer, water, or utility systems are not burdened by the home occupation to the extent that usage exceeds that normally associated with residential use;
- Allow residents of the community to use their residences as places to enhance or fulfill personal economic goals under certain specified standards, conditions, and criteria;
- To Enable the fair and consistent enforcement of these home occupation regulations; and
- 704.01-7 Promote and protect the public health, safety, and general welfare.
- Applicability. Regulations of this section shall apply to all home occupations initiated or established after the date of the enactment of this section. No home occupation may be initiated or established after the date of the enactment of this section except in conformance with the regulations and performance standards set forth in this section. Existing home occupations may continue as a nonconforming use of a structure under the regulations in effect the day before the enactment of this section, and subject to the regulations of Section 506.06.
- Permitted Districts; Permits. Home occupations shall be permitted only in residential zoning districts, as an accessory use to a dwelling, and only in accordance with the regulations contained herein. No person shall operate a home occupation without first obtaining either (a) a permit therefore from the Zoning Administrator, who shall issue such a permit when the requirements of this Section 704 are met, or (b) a Special Use Permit in accordance with Section 501.02 of this Ordinance. All such permits shall expire at the earlier of (a) when the applicant ceases to occupy the premises for which the home occupation permit or the conditional use permit was issued; or (b) when such applicant ceases to conduct the home occupation for a period of six (6) months.
- 704.04 <u>Exclusions from Operation of Section</u>. Notwithstanding any other provision of this Ordinance, the following uses shall be specifically excluded from the definition of home occupation, and no home occupation permit shall issue for such uses: motor vehicle repair or service; appliance repair, machine shop,

welding shop, escort service, medical services, tattoo parlors, body piercing establishments, bookstore, movie theater or arcade, or storage yard. Such uses shall be permitted only where otherwise authorized in this Ordinance. Notwithstanding any other provision in this Section 704, tutoring services and art or music lessons provided on a part-time basis by the occupant of a dwelling shall not be considered home occupations subject to this chapter, and no permit shall be required to engage in such activities.

General Standards for all Home Occupations. Any home occupation shall be subordinate, secondary, and incidental in both character and scale to the use of a dwelling unit for residential purposes. Home occupations shall not compromise the residential character of an area, shall not generate traffic that is conspicuous or at a nonresidential level, shall not visually call unusual attention to the home, and shall not generate noise of a nonresidential level. There shall be no limit to the number of home occupations permitted per dwelling unit, provided that any home occupation, including a personal service home occupation, shall be subject to all of the following standards:

- 704.05-1 The home occupation shall be conducted entirely within the interior of the principal residential structure or within an accessory structure located on the same lot.
- There shall be no change in the outside residential appearance of any structure for the purpose of the establishment of the home occupation.
- 704.05-3 There shall be no addition of parking spaces to accommodate the home occupation.
- Home occupations shall not occupy more than a cumulative total of two hundred fifty (250) square feet of the finished floor area of any dwelling unit or accessory structure in which the home occupations are located.
- 704.05-5 There shall be no outdoor storage of goods, products, equipment, or other materials associated with the home occupation.
- There shall be no display of goods, merchandise, or products visible from the street or any adjoining property.
- No equipment or processes not normally associated with a dwelling unit or which cannot be accommodated on existing utility or standard electrical services shall be permitted.

- No commercial motor vehicle shall be used, parked, or stored on the site in connection with the home occupation.
- 704.05-9 Mechanized equipment shall be used only in a completely enclosed building.
- No equipment or process which generates dust, odors, noise of a nonresidential level, vibration, or electrical interference or fluctuation that is detectable beyond the property line or through common walls shall be used in a home occupation.
- 704.05-11 Deliveries related to a home occupation shall be limited to the United States Postal Service, parcel delivery services, and messenger services. The home occupation or personal service home occupation shall not involve the commercial delivery by tractor trailer of materials or products to or from the premises.
- A home occupation shall comply with all local, state, or federal regulations pertinent to the activity pursued, and the requirements of or authorization granted by this section shall not be construed as an exemption from such regulations.
- Notwithstanding any provision of Section 705, no sign may be placed on the property advertising the home occupation.
- Home Occupations Not Personal Service Home Occupations. No Special Use Permit shall be necessary to conduct a home operation that is not a personal service home occupation, but the applicant seeking to engage in such a home occupation shall obtain a home occupation permit from the Zoning Administrator. The Zoning Administrator may revoke such a permit when, after notice to the holder thereof, he finds that the holder has materially violated the applicable terms of this Section 704. Home occupations that are not personal service home occupations shall be subject to the following requirements:
  - No more than a total of one (1) person other than family members residing on the premises shall be engaged in or employed by such home occupations, regardless of the number of home occupations associated with the dwelling unit.
  - No customer or client shall travel to the dwelling in connection with the home occupation.
- 704.07 <u>Personal Service Home Occupations</u>. A Special Use Permit issued in accordance with Section 501.02 of this Ordinance is necessary to conduct a personal service

home occupation. Such personal service home occupations shall be subject to the following regulations, and to such other conditions as the Town Council may prescribe when issuing the permit:

- No persons other than family members residing on the premises shall be engaged in or employed by such personal service home occupation.
- Not more than one (1) separate entrance or exit to the residence or accessory structure solely for the purpose of the personal service home occupation shall be permitted. The creation of any such separate entrance shall not be permitted on the front facade of the residential dwelling.
- The number of clients or customers who may visit the personal service home occupation at any one (1) time may be established by the Town Council as a condition of the approval of a Special Use Permit.

#### **705.00 SIGNS**

- Intent; Exclusion. The purpose of the following sign requirements is to promote and protect the public health, welfare, and safety by regulating existing and proposed outdoor advertising, and outdoor signs of all types. It is intended to protect property values, create a more attractive economic and business climate, and enhance and protect the scenic and natural beauty of the Town. It is further intended to reduce sign or advertising distraction and obstructions that may be caused by signs overhanging or projecting over public right-of-way, provide more open space, curb the deterioration of the natural environment, and enhance community development. Any sign of one (1) square foot or less in sign area is excluded entirely from the operation of this Section 705.
- General Requirements. Signs shall be a permitted accessory use in all zoning districts, but shall in every case comply with the requirements of this Section 705. Except as provided herein no outdoor advertising sign or advertising structure shall be erected without a Sign Permit that permits such sign. The following general requirements shall be applicable to all signs:
  - 705.02-1 For the purpose of computing sign area only one side of a "V-Type" or double-faced sign shall be considered.
  - No sign or sign structure shall be located in such a manner as to materially impede the view of any road intersection; or in such a

manner as to materially impede the view of the intersection of a road with a railroad grade crossing, or be located as to impede the safe flow of traffic.

- No sign shall be erected, relocated, or maintained so as to prevent free ingress to or egress from any door, window, or fire escape.
- No portion of any sign structure except official road markers, (which shall adhere to the applicable state and local laws) shall be less than eight (8) feet above the level of an adjacent sidewalk or other pedestrian thoroughfare or less than sixteen (16) feet above the level of an adjacent public driveway, alley or street.
- All signs shall comply with the applicable requirements of the Building Code.
- 705.02-6 In the event any sign that requires a Sign Permit is to be relocated, the owner of said sign shall obtain a new Sign Permit.
- All signs that are a structure may be erected up to a height of thirty-five (35) feet from grade, and no higher.
- All signs coming within the jurisdiction of State and Federal Laws along Interstate Highway and Federal-aid primary highway systems shall conform to said laws in lieu of any other sign regulations in this Ordinance.
- All signs in existence at the time of the passage of this Ordinance, which do not conform to this Ordinance, shall be classified as nonconforming, but may be continued providing they are property maintained during the life of such advertisement or advertising structure.
- All signs shall be placed so that they do not obstruct signs lawfully placed on adjacent properties.
- No business shall maintain more than two permanent signs at any one location, regardless whether such signs are wall-mounted, freestanding, or handing. A business may maintain one additional sign for each secondary entrance, provided that such sign is located near such secondary entrance. For shopping centers consisting of five (5) or more separate businesses and having a continuous street frontage of two hundred feet (200') or more, one of the two

permanent signs shall be located on the identification sign that identifies the businesses in the shopping center.

- 705.02-12 <u>Sign Permits</u>. Except as otherwise provided herein, a sign permit shall be required before the erection, alteration, or relation of a sign. The following rules are applicable:
  - (a) Each application for such permit shall be accompanied by plans showing the area of the sign, the size, character, and design proposed, the method of illumination, if any, the exact location proposed for such sign, the method of fastening such sign, the name and address of the sign owner and of the sign erector.
  - (b) The fee for a sign permit shall be as shown on the schedule of fees that accompanies this Ordinance.
  - (c) A sign permit shall become null and void if the work for which the permit was issued has not been completed within a period of six (6) months after the date of the issuance of the permit.
  - (d) A permit shall not be required for the repainting of a sign without changing the sign's wording, composition, or color; for minor nonstructural repairs to a sign; or per changing the advertising copy or message on any approved painted or printed sign or on a theater marquee and similarly approved signs which are specifically designed for the use of replaceable copy.
  - (e) An application for a sign permit may be combined with an application for a zoning permit if the applicant includes the information necessary for a sign permit. In such event only the zoning permit fee shall be due, and the zoning permit issued shall note the signs permitted.
- 705.03 Permissible Signs in All Districts. The following signs are allowed in all districts and do not require any permit, but shall be subject to the general requirements of §§ 705.02-1 through 705.02-12:
  - 705.03-1 Signs advertising sale, rental, or lease of the land or building upon which signs are located, provided that:

- (a) In residential districts, real estate signs shall not be in excess of six (6) square feet, and there shall be no more than one (1) such sign on any single lot.
- (b) In business districts, there shall be no sign in excess of thirty-two (32) square feet and no more than three (3) such signs on any single lot.
- (c) In industrial districts, there shall be no sign in excess of ninety (90) square feet and no more than three (3) such signs on any single lot.
- Directional signs for parks and playgrounds and other permitted nonresidential uses, provided that such signs shall not exceed four (4) square feet in area, shall be within one (1) mile of the use, and shall not be illuminated;
- Professional name plates not exceeding two (2) square feet in area; such signs to be non-illuminated.
- One sign or bulletin board indicating the name of the institution or civic association not exceeding ten (10) square feet in area on premises of public or semi-public facilities.
- 705.03-5 Signs located on the premises relating to active construction projects.
- 705.03-6 Memorial signs or tablets, including names of buildings and date of erection when cut into masonry, bronze, or other materials.
- 705.03-7 Traffic or other public signs or notices posted or erected by or at the direction of a governmental agency.
- Customary signs, in conjunction with residential usage, including mailbox lettering, names of residents, house number, names of farms and estates, and other similar usage not exceeding ten (10) square feet.
- One subdivision identification sign at the main entrance or entrances to the subdivision provided that such sign shall not exceed one hundred (100) square feet, may be illuminated, and shall make no reference to the sale or lease of the lots or houses located within said identified subdivision.

- 705.03-10 Temporary signs, including political advertisements, that are:
  - (a) Not in place for longer than sixty (60) days;
  - (b) Are removed by sign owner within five (5) days after the date of the event or activity to which the sign makes reference, or if he fails to do so, by the Zoning Administrator at the owner's expense, five (5) days following registered notification of the owner;
  - (c) No temporary sign shall exceed thirty (30) square feet per sign in area;
  - (d) All portable and vehicular signs shall adhere to the requirements set forth herein except for those signs not exceeding forty (40) square feet that are attached to vehicles which are used primarily for other purposes than displaying an outdoor advertising sign.
- 705.03-11 Informational signs of a public or quasi-public nature identifying or locating civic, educational or cultural purposes, and signs drawing attention to public parking lots, rest rooms, or to other public convenience relating to such places or activities. Such signs shall not exceed an area of six (6) square feet, shall not be illuminated, shall contain no advertising matter, and shall be set back not less than five (5) feet from the fronting highway.
- 705.03-12 Official notices or signs posted or displayed by or under the direction of any public or court officer in the performance of his official or directed duties, or by trustees under deeds of trust, deeds of assignment or other similar instruments. Such signs shall not exceed an area of two (2) square feet, shall not be illuminated and shall contain no advertising matter, other than that which may be required by law.
- Trespassing signs, signs indicating the private nature of a road, driveway, or premises, and signs controlling hunting and fishing on the premises, provided that the area of any such sign shall not exceed four (4) square feet.
- 705.03-14 Signs advertising yard or garage sales.
- 705.03-15 Temporary exterior banner signs applying to season or other brief activity including special sales, "grand opening," "under new management," or "going out of business" events to be held within

the premises. An exterior banner sign may be displayed for fourteen (14) days not more than four (4) times per calendar year. A total maximum of fifty (50) square feet shall be permitted for banner signs, with no more than two (2) signs being displayed at any one time. Banner signs must be attached flush to the building.

- 705.04 <u>Signs as Permitted Uses</u>. The following signs are permitted after the issuance of an appropriate sign permit identifying the sign:
  - 705.04-1 <u>Residential Districts</u>. Within any residential district the following signs are permitted:
    - (a) One (1) sign for each subdivision relating to the sale of property within said subdivision, provided that such sign shall be within said subdivision, shall not exceed thirty (30) square feet per sign area, shall not be illuminated, shall be maintained at subdivider's expense, and shall be removed by subdivider when eighty (80) percent of the lots in said subdivision are sold.
    - (b) Where multi-family dwellings are a permitted use, one (1) sign for identifying multi-family dwellings of more than six (6) units, provided that such sign shall be located only on the premises of the multi-family dwellings, shall not exceed nine (9) square feet in area, shall indicate nothing other than the name and/or address of the premises, and the name of the management, and may be illuminated only by indirect illumination.
  - Business Districts. The following signs are permitted within a commercial district, shall be erected or displayed only upon such walls of a building as face a street, alley, or parking area, or as freestanding signs upon the lot, and are subject to the following provisions as to size and location. Such signs shall also be subject to other applicable provisions of this Ordinance, including § 705.02-12.
    - (a) <u>One-Story Buildings</u>. The total area of all signs facing a street, alley, or parking area shall not exceed two (2) square feet for each foot of building width facing such street, alley, or parking area.
    - (b) <u>First Floor Businesses in Multi-Story Buildings</u>. The total area of all signs facing a street, alley, or parking area shall not exceed two (2) square feet for each foot of building

- width facing such street, alley, or parking area; and further provided that all such signs shall be kept within a height of twenty (20) feet above the sidewalk.
- (c) <u>Upper Stories of Multi-Story Buildings Containing One (1) or More Businesses Above the First Floor</u>. The total area of all signs facing a street, alley, or parking area on any wall above the twenty (20) foot height specified in § 705.04-2(b) above shall not exceed forty (40) square feet or one-fortieth (1/40) of the area of that wall above such twenty (20) foot height, whichever is greater.
- (d) Multi-Story Buildings Occupied by One (1) Business Only. Where entire buildings over one (1) story in height are occupied by one (1) business, a total sign area of one hundred (100) square feet facing any street, alley, or parking areas, or of one-fortieth (1/40) of the wall area facing such street alley, or parking area, whichever is greater, may be substituted for the allowable sign areas specified in (b) and (c) above, and in such case, the sign may be located without regard to the twenty (20) foot height provisions contained in (b) above.
- (e) <u>Signs Hung on Marquees</u>. No sign shall be hung on a marquee, canopy, or portico if said sign shall extend beyond the established setback line. The area of any such sign shall be included in determining the total area of signs erected or displayed.
- (f) Projection and Height of Signs. A sign may be erected or displayed flat against a wall or at an angle thereto, but no sign shall project beyond the established setback line. The bottom of a sign that is six (6) square feet in size and that is erected flat against a wall shall not be less than eight (8) feet above the sidewalk, alley, or parking area. The bottom of a sign projecting from a wall shall not be less than ten (10) feet above a walkway or parking area, nor less than fourteen (14) feet above an alley.
- (g) <u>Freestanding Signs</u>. Freestanding signs upon a lot may be erected or displayed only where drive-in service or parking is provided. Such signs shall be no closer than ten (10) feet of any building or any side lot line, and not encroach on any setback line. No more than two (2) such freestanding signs shall be permitted on any lot. No signs other than

those indicated on the Sign Permit shall be attached to a freestanding sign. Freestanding signs shall not be erected more than thirty (30) feet above grade and shall not exceed one hundred (100) square feet in area. Where signs are erected as freestanding signs upon the lot, the total area of all signs permitted by this section shall be two (2) square feet for each foot of lot frontage.

- (h) Identification Signs for Shopping Centers. Identification signs for shopping centers consisting of five (5) or more separate businesses and having a continuous street frontage of at least two hundred (200) feet shall be permitted, and the area of such signs shall not be included in the total area of signs otherwise permitted in this section for the separate businesses. The total area of such identification signs for any shopping center shall not exceed one (1) square foot for each foot of street frontage, nor shall the total area of such signs facing any street, alley, or parking area exceed one hundred fifty (150) square feet. This identification sign shall be one of the two signs permitted under § 705.02-12 for all business identified therein, and the other sign shall not be a freestanding sign.
- (i) Advertising Theatre Acts, etc. Signs advertising the acts or features to be given in a movie theatre or theatres may be displayed on permanent frames erected on theatre buildings in accordance with the provision of this section as to size and location; provided that the bottom of any such frame erected flat against a wall may be less than eight (8) feet above the sidewalk, alley, or parking area; provided further that when the area of any such frame facing a street, alley, or parking area does not exceed twenty-four (24) square feet and the area of any such frames facing such street, alley, or parking area does not exceed forty-eight (48) square feet, the area of the signs displayed thereon shall not be included in determining the total area of signs erected or displayed.
- (j) <u>Window Signs</u>. For any window sign, the maximum area shall not exceed 20 percent of the total window area in which that sign is to be displayed or six square feet, whichever is less. Such signs shall be located no less than five feet from grade in order to create good pedestrian visibility.

- (k) <u>Wall Signs</u>. Wall signs shall be permitted.
- (l) <u>Hanging Signs</u>. Hanging signs shall be mounted perpendicular to the facade of the building and shall project no more than four feet from the building and no closer than one foot to the back of the curb. Such signs shall be no less than eight feet in height from grade to the lowest part of the sign. There shall be no more than one projecting sign per business. Hanging signs shall not exceed six square feet in area.
- (m) <u>Individual Letter Signs</u>. Signs made up of letters only that are attached directly to the building are individual letter signs. Individual letter signs (and the type) shall be no larger than 12 inches in height and shall not exceed the height of the building.
- (n) Portable Signs. A portable sign is a non-illuminated or illuminated self-contained sign, not permanently attached to the ground, including those signs commonly known as "trailer signs" or "spider signs" but not including signs painted or otherwise inscribed on a self-propelled vehicle. For purposes of this Ordinance, portable signs are freestanding signs and shall be considered in the total allowable square footage for freestanding signs and must meet the setback regulations set forth herein for freestanding signs.
- (o) Roof Signs. Roof signs, not exceeding a total area of one hundred (100) square feet, may be erected or displayed; provided that the area of any roof sign shall be included in the total area of signs permitted herein. Roof top signs or roof top sign structures shall not extend more than twenty (20) feet above the roof line. Roof top signs or sign structures shall not extend beyond or overhang any exterior wall of the building upon which secured.
- 705.04-3 <u>Industrial Districts</u>. Any sign permitted in a commercial district shall be permitted in an industrial district also. The allowable area of any such sign shall be one and one-half times the allowable area of such sign if such sign were in a commercial district.
- 705.05 <u>Temporary Window Signs</u>. A temporary window sign is a sign describing a brief commercial activity. A temporary window sign is permissible in a commercial district without a sign permit. The size of a temporary window sign shall not be

included in the aggregated signage permitted in this section. If a temporary window sign becomes work, tattered, or illegible, it must be removed upon five days' written notice by the Zoning Administrator. The sign must in any even be removed within five (5) days of the conclusion of the advertised event.

705.06 <u>Signs as Special Use</u>. The following signs may be permitted in conjunction with an approved Special Use:

- Signs related to special use. Except as hereinafter provided, within any residential district, signs otherwise illegal but relating to buildings and uses permitted under a valid Special Use Permit are permitted, provided that they shall not exceed twenty (20) square feet per sign area, shall indicate nothing other than the activity engaged in, the name of the owner, firm, organization, or agency, and the hours of activity, shall be limited to two (2) signs per use, shall not encroach on any yard or setback, and, if stipulated by the Town Council as part of the Special Use Permit, shall be only indirectly illuminated. In business and industrial districts, signs relating to buildings and uses permitted under a valid Special Use Permit shall be permitted, provided that all requirements of sign area and character for permitted signs are met.
- Directional signs related to special use. Within any residential district, directional signs for uses and buildings permitted under a valid Special Use Permit are permitted, provided that they shall not exceed four (4) square feet per sign area; shall be within one (1) mile of the use; and shall not be illuminated.
- 705.07 <u>Signs Prohibited in all Districts</u>. The following signs are prohibited in all districts:
  - Any sign erected or painted upon a fence, tree, fire escape, or utility pole.
  - Any sign which uses the work "Stop" or "Danger" prominently displayed or which is a copy or imitation of official traffic control signs.
  - Any sign which contains flashing or intermittent illumination, or scrolling text.
  - Any sign which is manufactured and is designed to and effectively does distract the attention of passing motorists on any highways by loud and blatant noises, movable objects or flashing lights.

705.07-5 Billboards or general advertising signs unrelated to the use of the property.

# 705.08 <u>Maintenance and Removal of Signs</u>:

- 705.08-1 All signs and sign structures shall be kept in repair and in proper state of preservation. All signs must adhere to the provisions of the Building Code.
- Signs which are no longer functional, or are abandoned, shall be repaired, removed, or relocated at the owner's expense in compliance with the provisions of this Ordinance within thirty (30) days following dysfunction.
- After a reasonable attempt to notify the owner of the premises, the Zoning Administrator may order or cause the removal of any nonconforming sign that has been abandoned. The Zoning Administrator may presume that such sign has been abandoned if the business that such sign advertises has been closed for at least two (2) years.

## 706.00 MANUFACTURED HOMES

Any manufactured home placed in the Town of Broadway after the date of enactment or amendment of this Ordinance, shall meet the following requirements:

- No manufactured home shall be placed in any zoning district that does not specifically allow manufactured home parks or manufactured home subdivisions as a permitted use. The allowance of "single family dwellings" as a permitted use shall not be deemed to allow the placement of manufactured homes.
- All manufactured homes shall meet the plumbing requirements and the electrical wiring and connection, construction, blocking, and anchoring requirements of the Virginia State Building Code and shall display the seal of a testing laboratory approved by the Commonwealth of Virginia.
- All manufactured homes shall be completely skirted; such that no part of the undercarriage shall be visible to a casual observer, in

accordance with methods and materials approved by the Building Official.

- All manufactured homes shall be supplied with public water and wastewater disposal or such individual service evidenced by permits from the Health Department.
- All required off-street parking area shall be paved or graveled and have unobstructed access to either a public or private street.
- 706.02 <u>Timing of Occupancy</u>. In any manufactured home park, a minimum of three (3) spaces shall be completed and ready for occupancy before the first occupancy is permitted.
- 706.03 <u>Inspections.</u> All manufactured home parks for which permits are granted under this section will be subject to periodic inspection to ensure that the parks are meeting the requirements of this Ordinance.
- 706.04 <u>Manufactured Home Accessory Structures</u>. All manufactured home accessory structures erected or constructed after the date of enactment or amendment of this Ordinance must meet the following requirements:
  - 706.04-1 All manufactured home accessory structures must meet the plumbing, electrical connection, wiring, construction, and other applicable requirements of the Building Code.
  - Except in the case of an awning, ramada, or other shade structure, where a manufactured home accessory structure is attached to the manufactured home unit, a substantial part of one wall of the accessory structure shall be flush with part of the manufactured home unit, or such accessory structure shall be attached to the manufactured home unit in a substantial manner by means of a roof. All manufactured home accessory structures, whether attached or detached, shall be designed and constructed as free standing structures. No detached manufactured home accessory structure, except ramadas, shall be erected closer than seven and a half (7½) feet to a manufactured home.
  - Manufactured home accessory structures, except ramadas, shall not exceed the height of the manufactured home.
  - No manufactured home accessory structure shall be erected or constructed on any manufactured home lot or stand except as an accessory to a manufactured home.

- 706.05 <u>Manufactured Home Park Application and Site Plan.</u> Applicants for manufactured home parks shall meet the following special requirements:
  - 706.05-1 Site plans shall be legibly drawn at a scale consistent with its purpose.
  - 706.05-2 The following information shall be required of site plans:
    - (a) The date of the site plan, the name of the surveyor and the number of sheets comprising the site plan;
    - (b) The scale and the north designation;
    - (c) The name and signature of the owner, and the name of the proposed park; said name shall not closely approximate that of any existing manufactured home park or subdivision in the Town of Broadway;
    - (d) A vicinity map showing the location and area of the proposed park;
    - (e) The boundary lines, area, and dimensions of the proposed park, with the locations of property line monuments shown;
    - (f) The names of all adjoining property owners;
    - (g) Proposed layout, including interior streets with dimensions and such typical street cross sections and centerline profiles as may be required in evaluating the street layout; water, sewer, drainage, and utility lines, facilities and connections, with dimensions shown; location and type of solid waste collection facilities; interior monuments and lot lines, dimensions, and areas of manufactured home lots, common open space and recreation areas, common parking areas, and other common areas; locations and dimensions of parking manufactured home stands and spaces, management offices, laundry facilities, recreation buildings, and other permanent structures; location and nature of fire-fighting facilities, including hydrants, fire extinguishers, and other fire fighting equipment; location of fuel storage facilities and structure of high flammability; and location and dimensions of landscaping amenities, including street lights, sidewalks, planted areas, significant natural features to be retained, and fencing and screening.

The site plan shall be accompanied by a narrative statement describing how the standards and requirements set forth herein are to be met; a statement from the Health Official certifying approval of the proposed site plan; and where appropriate, statements from the Highway Engineer certifying approval of the street and drainage; water and sewer, or utility system layouts by the owner/operator.

706.06 <u>Manufactured Home Park Design Standards</u>. The following standards shall be applicable to manufactured home parks:

- 706.06-1 Streets. An internal street system shall be provided to furnish convenient access to manufactured home stands and other facilities in the park, shall be designed such that connection to existing drainage and utility systems is convenient, and shall meet the following requirements:
  - (a) All internal streets shall be permanently paved with a durable, dust-proof, hard surface. Minimum pavement widths shall be twenty-four (24) feet for streets providing access to forty (40) or more manufactured home stands, and eighteen (18) feet for streets providing access to less than forty (40) manufactured home stands. Widths shall be measured from curb face to curb face. Curbs and gutters shall be provided and built to Town standards;
  - (b) Dead end streets shall be limited in length to 600 feet, shall be provided with cul-de-sacs with turning areas of not less than forty (40) feet in radius;
  - (c) Streets shall be adapted to the topography, shall follow the contours of the land as nearly as possible, and shall have safe grades and alignments; and
  - (d) Driveway entrances to manufactured home parks from any public street or road shall conform to the current construction standards of the Department of Highways and Transportation.
- Vehicle Parking. Off-street parking shall be provided for the use of occupants at the minimum ratio provided in § 703 for each manufactured home. Each off-street parking space shall be paved or graveled and have unobstructed access to either a public or private street. On street parking is prohibited unless the paved street on which the manufactured home fronts is expanded to accommodate additional parking lanes or parking bays.

- 706.08 <u>Lighting</u>. All streets and walkways within the manufactured home development shall be lighted.
- 706.09 <u>Disposition of Garbage and Rubbish</u>. It shall be the responsibility of the manufactured home park to collect or cause to provide for adequate collection containers for park residents.
- 706.10 <u>Installation of Storage Tanks</u>. Gasoline, liquefied petroleum, gas, or oil storage tanks shall be so installed as to comply with all County, State, and Federal fire prevention and protection regulations.
- Record of Tenants for Manufactured Home Parks. The operator of a manufactured home park shall keep an accurate register of all tenants occupying manufactured homes located in the park. The register shall show the name and permanent residence address of the owner and occupants of any manufactured home located in the park, the make and registration of any manufactured home, the time and date of arrival and departure, and such other information as might be necessary to provide information about the occupants of the manufactured home. These records shall be open to the law enforcement officers and public officials whose duties necessitate acquisition of the information contained in the register. The register record for each occupant registered shall not be destroyed for a period of three (3) years following the date of departure of the registrant from the park.
- Occupancy Required. No manufactured home or accessory structure shall be occupied in any manufactured home park until a certificate of occupancy shall have been issued by the county building inspector to the effect that the manufactured home park or the portion thereof for which such certificate is requested is in compliance with all applicable provisions of this chapter and the Uniform Building Code. Such certificate shall not be issued until after the same has been approved by the health department, zoning administrator, and other agencies concerned.

## 707.00 PARKING OR STORAGE

- 707.01 <u>Abandoned Vehicles</u>. No abandoned vehicles are permitted within the Town of Broadway.
- 507.02 Storage of Vehicles. In residential and commercial zones, the parking or storage of small cargo or utility trailers, recreational vehicles and similar equipment, including, but not limited to, boats, boat trailers, motor homes, and tent trailers, shall be permitted only subject to the following rules:

- No such vehicle shall be stored in any required yard or setback, except that such vehicles may be stored in a driveway or parking area with an all weather surface, even within such yard or setback.
- In residential districts, or on commercial lots adjacent to residential districts, such vehicles or equipment shall not be used for living, housekeeping or business purposes when parked or stored on the lot for more than fourteen (14) consecutive days or for more than twenty-one (21) days per calendar year. In other commercial districts, such vehicles shall not be used for living, housekeeping, or business purposes when parked or stored on the lot for more than ninety (90) days in a calendar year.
- 707.02-3 Wheels or other transporting devices shall not be removed except for necessary repairs or seasonal storage.
- A maximum of three such vehicles shall be permitted for each lot.
- No such vehicle may be parked or stored on any public right-ofway, except that such vehicles may be so parked or stored for a period not exceeding 72 hours while in process of loading or unloading.
- Any person may apply for a special use permit to allow any parking or storage not permitted by this Section 707, and such parking or storage is a use permitted by special use permit in all residential and commercial districts.
- 707.03 <u>Movable Storage Boxes</u>. Movable storage boxes, also known as portable ondemand storage units, may be placed temporarily on a property in a residential zone for loading or unloading. The following rules are applicable:
  - 707.03-1 <u>Setbacks and Yards</u>. Such units shall not be placed in a front yard area, except on a driveway. Such units shall, at all times, stored at least 20 feet from a front property line and at least 5 feet from a side or rear property line.
  - 707.03-2 <u>Temporary Placement</u>. For the purposes of this Section 707.03, temporary placement shall mean no more than 14 consecutive days at a time, and with at least one year between successive placements. Not more than one unit shall be placed on a residential property at a time and if multiple units are used for

sequential loading or unloading, the fourteen-day limit shall apply to all cumulatively.

707.03-3

Natural Disasters. The above restrictions notwithstanding, when the principal structure on the property has been made uninhabitable as a result of a natural disaster for which a local state of emergency declaration has been issued or a fire or other damaging event beyond the control of the owner, one or more movable storage boxes may be used for on-site storage purposes exceeding 14 days while the principal building is undergoing reconstruction/repair. The authorization for such use shall be dependent on issuance of a building permit for the reconstruction/repair of the principal residence and shall expire upon issuance of a certificate of occupancy for the principal structure or 12 months from the date of the event that damaged the structure, whichever occurs first. For good cause shown and to recognize extenuating circumstances, the Administrator may extend administratively authorization for as much as an additional twelve-month period or until a certificate of occupancy is issued, whichever occurs first.

## 708.00 SPECIAL REGULATIONS FOR TOWNHOUSES

708.01 Purpose and Intent.

- A. It is the purpose and intent of this article to accommodate various types of unique and innovative housing forms which are usually based upon a concept of reducing the area of individually owned lots, and grouping such lots together to more efficiently utilize the total space within the subdivision by creating common open spaces, scenic and recreational areas, and other spaces which would compensate for the reduction of land area contained within the individually owned lot. The following standards have been developed and apply to a variety of such attached and semi-detached single-family housing forms including but not limited to townhouses or any other similar designation that meets the definition of "townhouse" under these Land Development Regulations. Except as varied in this section 708, all other provisions contained in the town zoning code (including specific provisions applicable to the zoning district in which a townshouse development is located) are applicable to these types of housing forms unless specifically noted.
- B. Specifically, this chapter is designed to allow the grouping of separately owned one-family dwelling units into a group of townhouses in such a manner as to make efficient, economical and aesthetically pleasing use of land so restricted that the same will be continually well maintained in order to preserve the health,

welfare, safety and convenience of the surrounding neighborhood and insure a reasonable amount of open space and architectural variety.

**Ordinance of July 7, 2015,** added this section and renumbered former § 708.01 to § 702.02-1.

708.02	Regulations	Regarding	Construction a	and Siting of	Townhouses.

- 708.02-1 <u>Groupings</u>. No more than six (6) and no less than four (4) townhouses shall be included in any townhouse grouping.
- 708.02-2 <u>Fire Walls</u>. Attached dwellings shall be separated by a noncombustible party wall to the roof line, with a fire resistance of not less than two hours duration.
- 708.02-3 <u>Dwelling Unit Access.</u> No two townhouse dwelling units shall be served by the same interior or exterior stairway or by the same exterior door.
- 708.02-4 <u>Height</u>. The maximum height for any townhouse shall not exceed that allowed in the district in which the development is located.
- 708.02-5 <u>Lot Area for Each Townhouse Unit</u>. No townhouse lot shall contain an area of less than two thousand (2,000) square feet and a minimum lot and building width of not less than twenty feet
- 708.02-6 <u>Hydrants</u>. Townhouse developments shall provide fire hydrants on or adjacent to the area to be developed as required by the County fire department or the County Building Code.
- Special Regulations for Accessory Buildings. In addition to any other requirements for accessory buildings imposed in these Land Development Regulations, no accessory building in a townhouse development shall be erected in any side or front yard. An accessory building may be located in a rear yard and shall occupy not over twenty-five percent of the rear yard area and shall be set back at least ten feet from any alley or rear service line.
- Tos.02-8 <u>Landscaping</u>. Notwithstanding any other provision in these Land Development Regulations, in townhouse developments no less than seventy-five percent of all yards on either public or grandfathered private streets shall be landscaped with pervious surfaces.

Walls. Each one-dwelling unit shall be attached by not more than two party walls. Where units are offset from one another and a common party wall is used, the wall may be placed equidistant on each side of the lot line not exceeding the length of the offset.

**Ordinance of July 7, 2015** recodified former § 708.01 as new § 708.02-1, changing the maximum permitted number of units in a

building from ten (10) to six (6); recodified former  $\S$  708.02 to new  $\S$  708.02-2, and added  $\S\S$  708.02-3 through -9.

708.03

Streets. A well-defined system shall be designed to allow free movement throughout the development while discouraging excessive speeds, and shall structure the development in clearly defined clusters and/or groups of townhouses. Where dwelling units are located on residential service streets or courts, the street system must be designed to discourage all traffic except that of owner/occupants, their guests, and their services. The right-of-way width of public streets serving a group of townhouses and the improvements therein shall conform to all applicable town standards and requirements for such streets. Each townhouse shall front on a dedicated public street constructed to the standards generally applicable to streets in the Town. No townhouse shall be constructed so as to provide direct vehicular ingress or egress from a particular townhouse lot to any controlled access highway, major thoroughfare, arterial, or collector street. Further, notwithstanding any standard generally applicable to streets in the Town, streets in a Townhouse development shall have:

- 708.03-1 Concrete curb and gutter on both sides of street easement.
- 708.03-2 Sidewalk four (4) feet in width on at least one (1) side of the easement, constructed on concrete, brick, stone, gravel, or some other material of reasonable durability and safeness.

**Ordinance of December 6, 2011** added the phrase "except as provided in Sec. 905.02-1." to the end of the first sentence of Section 703.03.

types of streets used to access a townhouse, repealed former § 703.03-3, which provided a minimum radius for cul-de-sacs.

Ordinance of July 7, 2015 rewrote the language of the initial paragraph to prohibit private access easements, and to limit the

## 708.04 Common Areas.

708.04-1

Maintenance of Common Areas. Common areas shall be maintained by and be the sole responsibility of the developer-owner of the townhouse development until such time as the developer-owner conveys such common area to homeowner's association whose members shall be all the individual owners of the townhouses in the townhouse development. Said land shall be conveyed to and be held by said homeowner's association solely for recreational and parking purposes of the owners of the individual townhouse lots in the development. In the event of such

conveyance by the developer-owner to a homeowner's association, deed restrictions and covenants shall provide, among other things, that any assessments, charges for cost of maintenance of such common areas shall constitute a pro-rata lien upon the individual townhouse lots. Such assessments shall be a lien superior to all other liens save and except tax liens and mortgage liens, provided the mortgage liens are first liens against the property encumbered thereby, subject only to tax liens, and secure indebtedness which are amortized in monthly or quarter-annual payments over a period of not less than ten years. Other methods may be acceptable if the same positively provide for the proper and continuous payment of taxes and maintenance without expense to the general taxpayers. The instrument incorporating such provisions shall be approved by the Town Council and shall be recorded in the public records of the County. Maintenance of townhouse exteriors, lawns, refuse handling, taxes, lighting, and drainage shall be provided in a similar manner so as to discharge any responsibility from the Town of Broadway. In the event that such association or a similar entity fails to maintain the common area in accordance with the landscaping plan approved by the Town Council as part of the Special Use Permit or the R-5 site plan approval, or should the entity fail to maintain the common area in a reasonable condition and state of repair, the determination of such failure to be made by the zoning adminstrator, then and in that event the Town Council, may at its option through its own agents or by independent contractor, enter the common area for purposes of maintenance thereof, and assess such charges, together with an additional charge of twenty-five percent of the costs for management fees, to such association. Such charges shall constitute a lien upon each and every lot in the project.

708.04-2 Common Area Open Space. A minimum of thirty percent (30%) of the townhouse project site shall be provided as common open space. At least fifty percent (50%) of such space shall be unencumbered with any structure or off-street parking and shall be landscaped and well maintained with grass, trees and shrubbery. The remaining space may be used for other common purposes,

including recreational or maintenance purposes.

**Ordinance of July 7, 2015** renumbered former § 708.04 as new § 708.04-1, added provisions to that section to give homeowners association liens the same status as tax liens, provide for Town

Council approval of the instrument requiring liens for maintenance, and added  $\ 708.04\text{--}2,\ providing}$  for open space requirements.

708.05 <u>Facades</u>. Facades of dwelling units in a townhouse development shall be varied by changed front yards of not less than two (2) feet and variations in materials or

design, so that no more than two (2) abutting units will have the same front yard depth or essentially the same architectural treatment of facades and roof lines.

**Ordinance of July 7, 2015**, amended this section to provide that not more than two (2) abutting units will have the same front yard depth, rather than four (4) as under former § 708.05.

708.06 <u>Utilities</u>. Any utility entering a unit must do so on the property of that unit. All townhouses must be connected to public water and sewer lines, and all electrical and telephone lines in a townhouse development site shall be placed underground. Property and adequate access for firefighting purposes and access to service areas to provide garbage and waste collection, and for other necessary services, shall be provided.

**Ordinance of July 7, 2015** added the second and third sentences of this section and changed the section title to "Utilities".

- <u>Lighting</u>. Lighting of buildings, accessways, and parking areas shall be provided for the safety and convenience of all residents but it shall be so arranged as not to reflect toward public streets or cause disturbance to building occupants or surrounding landowners.
- 708.08 <u>Storage of Trash and Rubbish</u>. Exterior storage areas shall be provided for trash and rubbish. Such storage areas shall be screened on at least three sides and contain vermin-proof containers.
- 708.09 <u>Parking</u>. Section 613.14 of the Land Development Regulations shall not apply to townhouse developments located in an R-3 zoning district.

Ordinance of July 7, 2015 added this section.

#### **709.00 FENCES**

- 709.01 <u>Material</u>. No fragile, readily flammable material such as paper, cloth or canvas shall constitute a part of any fence, nor shall any such material be employed as an adjunct or supplement to any fence.
- 709.02 <u>Corner Lots.</u> In the case of corner lots in residential districts, there shall be no planting, fence or obstruction to vision more than three (3) feet high, less than twenty (20) feet from the intersection of two street lines.
- 709.03 <u>Height Limit: Residential and Commercial.</u> Fences shall not exceed a height of six (6) feet on any other size, as measured from the top most point thereof to the ground or surface, along the center line of the fence, in a commercial or residential zone. Notwithstanding the foregoing, in any residential district, fences

parallel to a street right of way abutting the front yard of a lot shall not exceed four (4) feet.

- 709.04 <u>Height Limit: Industrial and Public Buildings</u>. Fences surrounding industrial sites, public playgrounds, institutions, or schools may not exceed a height of fourteen (14) feet.
- No Setback Applicable; Not Accessory Structure. A fence is not an accessory structure. Notwithstanding any other provision of this Ordinance, a fence may be located along or parallel to a property line without regard to any setback or yard regulation in this Ordinance, whether applicable to accessory structures or otherwise.

## 710.00 TELECOMMUNICATIONS TOWERS

- Purposes. It is the purpose of this section to: (a) provide adequate sites for the provision of telecommunications services with minimum negative impact on the citizens and resources of the Town; (b) encourage the location of Telecommunications Towers in non-residential areas and minimize the total number of Telecommunications Towers in the Town; (c) encourage strongly the joint use of new and existing Telecommunications Towers and the use of Alternative Telecommunications Structures; (d) to encourage the configuration of such Towers in a way that minimizes the burdens created by them; and (e) to treat providers of functionally equivalent services in a reasonably like manner.
- 710.02 <u>Definitions</u>. For purposes of this Section 710, the following terms shall carry the meaning assigned in this section:
  - 710.02-1 <u>Antenna</u>. A structure or device used to collect or radiate electromagnetic waves.
  - 710.02-2 <u>Telecommunications Antenna</u>. Any exterior apparatus designed for telephonic, radio, or television communications through the sending, or receiving or both of electromagnetic waves.
  - 710.02-3 <u>Telecommunications Tower</u>. Any structure that is designed and constructed primarily for the purpose of supporting and including one or more Telecommunications Antenna or microwave dishes, including but not limited to, self-supporting lattice towers, guyed towers, man-made trees, monopole towers, telephone, radio and television transmission towers.

- 710.02-4 <u>Alternative Telecommunications Structure.</u> Buildings, clock towers, bells, steeples, electric poles, utility poles, water storage tanks, and other similar alternative design mounting structures that are used for the purpose of supporting and obscuring the presence of antennas.
- 710.02-5 <u>Height</u>. When referring to a Telecommunications Tower or similar structure, the distance measured from ground level to the highest point on the Telecommunications Tower, even if the highest point is an antenna.
- Applicability. This Section 710 shall not govern any television reception antennae under twenty-five (25) feet in height and used exclusively for non-commercial purposes, or any amateur radio antenna under seventy-five (75) feet in height, or its installation, if it is owned and operated by a federal licensed amateur radio station operator and used exclusively for non-commercial purposes. This Section 710 shall apply in lieu of other setback or height requirements normally applicable to accessory structures.
- Special Use Permit Required. No Telecommunications Tower or Antenna shall be located within the Town without the issuance of a Special Use Permit. A telecommunications tower shall be a use expressly permitted by Special Use Permit in all zoning districts. In approving or denying Special Use Permit applications for Telecommunications Towers or Antennas, the Council will consider the character of the neighborhood, conformity with the comprehensive plan, the guidelines of this Ordinance, the purposes of this Ordinance, the public's needs for the facility, and any other issues bearing on the propriety of the application. The Zoning Administrator may require the submission of whatever information he reasonably deems necessary to enable the Town Council to evaluate such applications.
- 710.05 <u>Siting Requirements</u>. The following rules shall be applicable to Telecommunications Towers and Antennae:
  - A new Telecommunications Tower shall not be permitted unless the applicant demonstrates to the reasonable satisfaction of the Council that existing Telecommunications Towers or Alternative Telecommunications Structures cannot accommodate the applicant's proposed antenna.
  - A Telecommunications Tower shall not be artificially lighted, unless required by the Federal Aviation Administration (FAA) or other applicable authority.

- No signs, letters, symbols, or trademarks, or advertising of any type may be placed on the Telecommunications Towers or Antenna, other than as required by Federal Communications Commissioner (FCC) of Regulations or other applicable law.
- All Telecommunications Towers and Antennas must meet or exceed current standards and regulations of the FAA, the FCC and any other agency or the federal or state government with the authority to regulate such facilities.
- 710.05-5 Telecommunications Towers shall either maintain a galvanized steel finish or, subject to any applicable standards of the FAA, be painted, so as to reduce visual obtrusiveness. Dish antennas will be of a neutral, nonreflective color with no logos.
- At the Telecommunications Tower, the design of the buildings and related structures used in conjunction with telecommunications facilities shall, to the extent possible, use materials, colors, textures, screening, and landscaping that will blend the telecommunications facilities to the natural setting and the built environment.
- 710.05-7 If an antenna is installed on an Alternative Telecommunications Structure, the antenna and supporting electrical and mechanical equipment must be of a color that is identical to, or closely compatible with, the color of the alternative telecommunications structure so as to make the antenna and related equipment as visually unobtrusive as possible.
- 710.06 Setbacks. Telecommunications Towers must be set back a distance equal to three hundred percent (300%) of the height of the Telecommunications Tower from any off-site structures used for human habitation. Any Antenna mounted on an Alternative Telecommunications Structure need not comply with the setback requirements of this section if its height is no more than one hundred ten percent (110%) of the height of the structure on which it is mounted.
- 710.07 <u>Co-location</u>. All Telecommunication Towers shall be designed and built to accommodate a minimum of three (3) or more Telecommunications Antennae. The tower must available for use by other telecommunications service providers on a reasonable and nondiscriminatory basis, and the owner thereof must certify so to the Zoning Administrator from time to time upon request. All Telecommunications Tower applications shall include a complete and accurate inventory and map of the applicant's and other known existing and proposed Telecommunications Towers and other structures on which a

Telecommunications Antenna could be located or co-located within five (5) miles of the proposed Telecommunications Tower.

Abandonment. Any Telecommunications Antenna or Telecommunications Tower that is not operated for a continuous period of twelve (12) months shall be considered abandoned, and its owner shall remove it within sixty (60) days notice from the Town. The owner of the land upon which the Tower sits and the owner of the tower shall be jointly and severally liable for the expenses of such removal. Removal includes the removal of the Antennas, Telecommunications Tower, and associated facilities, fence footers, underground cables, and support buildings.

- 710.09 <u>Security Requirements</u>. The following requirements shall govern the securing of Telecommunications Antennae and Telecommunications Towers:
  - 710.09-1 Telecommunications Antennae or Towers, that include buildings or added structures, shall be enclosed by security fencing not less than six feet in height and shall also be equipped with an appropriate anti-climbing device unless the Town Council determines, in issuing its Special Use Permit, that one is not required.
  - 710.09-2 If a Telecommunications Tower or Antenna is mounted on an alternative telecommunications structure, the security fencing shall not be required unless the Town Council specifies so in a condition to its Special Use Permit.
  - 710.09-3 Monopole towers and other single-pole structures, standing alone, shall be secured by anti-climbing devices.

#### 711.00 NONCOMMERICAL RADIO OR TELEVISION ANTENNAE OR DISHES

- Antenna structures including guy wires for radio, television, and other noncommercial communication purposes, including freestanding satellite dishes, which do not exceed twenty-five feet in height (or, in the case of amateur radio antennae, do not exceed seventy-five feet in height), are permitted notwithstanding the requirements of Section 710, and subject to the following rules:
  - A freestanding antenna covered under this Section 711 is an accessory structure, and all regulations applicable to accessory structures shall apply, except that a freestanding antenna may be located as close to a principal structure as the owner of the antenna wishes, and except for height regulations applicable to accessory

structures. Guy wires shall not be permitted in the front setback areas.

- 711.01-2 The measurement of height for the purposes of this section shall include the antenna, any ancillary antennae, and any support structure.
- 711.01-3 Reasonable and customary engineering practices shall be followed in the placement of all antennae.
- 711.01-4 If it is not possible to receive an adequate satellite signal except through a placement of a freestanding satellite dish or dish antennae that violates the provisions of this ordinance, the party desiring to place the dish may apply for a Special Use Permit to permit such placement, describing in such application the location of the desired placement of the satellite antenna or dish. The Town Council shall consider such application in accordance with the procedures of Section 501.02. Such placement is an authorized Special Use for all zoning districts.

# 712.00 SPECIAL RULES FOR BED AND BREAKFAST ESTABLISHMENTS AND INNS

- 712.01 <u>Structure</u>. A bed and breakfast establishment must appear to be an ordinary dwelling.
- 712.02 <u>Sign.</u> Notwithstanding the provisions of Section 705, a bed and breakfast or an inn in a residential district shall display no other than a single sign no larger in surface area three (3) square feet.
- Meals; Duration of Stay. Guests in a bed and breakfast establishment or an inn may be provided with morning meals only and may lodge no more than fourteen (14) consecutive days. If guests are provided with more than morning meals, the establishment shall be considered a restaurant in addition to being an inn or bed and breakfast, and shall be permissible only if restaurants are a legal use in the district in which the establishment is located.

#### 713.00 ADULT USES

713.01 Findings. The Town Council finds that:

- 713.01-1 <u>Statutory Authorization</u>. The Town is empowered by Va. Code §§ 15.2-2280(1) and 2281 to regulate the use of land within its boundaries in order to promote the creation of a convenient, attractive, and harmonious community.
- 713.01-2 <u>Effect of Adult Businesses</u>. The Town Council is aware of and has taken into consideration numerous scholarly studies showing the secondary effects of adult businesses on the community, which effects include increased crime in the vicinity of such businesses and decreased property values in such areas.
- 713.01-3 <u>Goals</u>. The Town Council has determined that the goals of Va. Code § 15.2-2280(1) will be furthered by the regulation of adult businesses as contemplated herein, consistent both with the Town's responsibilities to its citizens and with its responsibilities under the Fourteenth Amendment to the Constitution of the United States as a political subdivision of the Commonwealth of Virginia.
- Adult Uses in Districts. In any district in which a use is otherwise permitted, if such use constitutes an "adult use," as defined herein, the minimum requirements and standards set out in this Section 713 shall apply to such use, in addition to the other standards of this Ordinance. Where this Section 713 imposes a conflicting but more restrictive standard than the other provisions of this Ordinance, this Section 713 shall prevail.
- 713.03 <u>Definitions</u>. In this Section 713, unless the context otherwise requires, the following words and terms are defined as set out herein:
  - 713.03-1 *Adult bookstore*: An establishment that devotes more than fifteen (15) percent of the total floor area utilized for the display of books and periodicals to the display and sale of the following:
    - (a) Books, magazines, periodicals or other printed matter, or photographs, films, motion pictures, video cassettes, slides, tapes, records or other forms of visual or audio representations which are characterized by an emphasis upon the depiction or description of "specified sexual activities" or "specified anatomical areas"; or
    - (b) Instruments, devices or paraphernalia which are designed for use in connection with "specified sexual activities."

An adult bookstore does not include an establishment that sells books or periodicals as an incidental or accessory part of its principal stock-in-trade and does not devote more than fifteen (15) percent of the total floor area of the establishment to the sale of books and periodicals.

- 713.03-2 *Adult use*: Any adult book store, adult motion picture theatre, adult mini-motion picture theatre, adult motion picture arcade, or adult drive-in theatre, as defined herein.
- Adult motion picture theatre: An establishment, with a capacity of fifty (50) or more persons, where, for any form of consideration, films, motion pictures, video cassettes, slides or similar photographic reproductions are shown; and in which a substantial portion of the total presentation time is devoted to the showing of material which is distinguished or characterized by an emphasis upon the depiction or description of "specified sexual activities" or "specified anatomical areas" for observation by patrons.
- Adult mini-motion picture theatre: An establishment, with a capacity of more than five (5) but less than fifty (50) persons, where, for any form of consideration, films, motions pictures, video cassettes, slides or similar photographic reproductions are shown, and in which a substantial portion of the total presentation time is devoted to the showing of material which is distinguished or characterized by an emphasis upon the depiction or description of "specified sexual activities" or "specified anatomical areas" for observation by patrons.
- Adult motion picture arcade: Any place to which the public is permitted or invited wherein coin or slug-operated or electronically, electrically or mechanically controlled still or motion picture machines, projectors, or other image producing devices are maintained to show images to five (5) or fewer persons per machine at any one (1) time, and where the images so displayed are distinguished or characterized by an emphasis on depicting or describing "specified sexual activities" or "specified anatomical areas."
- Adult drive-in-theatre: An open lot or part thereof, with appurtenant facilities, devoted primarily to the presentation of motion pictures, films, theatrical productions and other forms of visual productions, for any form of consideration, to persons in motor vehicles or on outdoor seats, and presenting material distinguished or characterized by an emphasis on matter depicting, describing or relating to "specific sexual activities" or "specified anatomical areas" for observation by patrons.

#### 713.03-7 *Specified anatomical areas*:

- (a) Less than completely and opaquely covered human genitals, pubic region, buttock, or female breast below a point immediately above the top of the areola; and
- (b) Human male genitals in a discernibly turgid state, even if completely and opaquely covered.

#### 713.03-8 *Specified sexual activities*:

- (a) Human genitals in a state of sexual stimulation or arousal;
- (b) Acts of human masturbation, sexual intercourse or sodomy; and
- (c) Fondling or other erotic touching of human genitals, pubic region, buttock or female breast.
- 713.04 <u>Requirements and Standards for Adult Uses</u>. The following rules shall be applicable to adult uses:
  - No adult use may be established within one thousand (1,000) feet of any other adult use in any zoning district.
  - No adult use may be established within five hundred (500) feet of a residentially zoned district, or a school, educational institution, church, public park, playground, playfield or day care center.
  - No adult use shall be established without a Special Use Permit. Such use shall be an authorized Special Use in any district where a similar commercial establishment not constituting an adult use would be permitted.
  - The "establishment" of an adult use as referred to herein shall include the opening of such business as a new business, the relocation of such business, the enlargement of such business in either scope or area, or the conversion, in whole or part, of an existing business to any adult use.
- Measurement of Distance. All distances specified in this Section 713 shall be measured from the property line of one (1) use to the property line of another, at the shortest point. The distance between an adult use and a residentially zoned district shall be measured from the property line of the use to the nearest point of the boundary line of the residentially zoned district.

#### 714.00 ZERO LOT LINE DEVELOPMENTS

- General Provisions. In any district in which a Zero Lot Line Development is a Special Use, the following provisions shall apply to lots constituting a Zero Lot Line Development, in addition to the normal regulations applicable under the other provisions of this Ordinance. To the extent of any conflict between another Section of this Ordinance and this Section 714, this Section 714 shall prevail.
- Building Setbacks. In any lot that is part of a Zero Lot Line Development, a dwelling unit may be placed on one interior side property line, giving it one zero side setback. The setback on the other size property line shall be twice the setback normally required in the zoning district in which the lot is situated. These side setback reductions shall not apply to side setbacks adjacent to lots that are not part of the Zero Lot Line Development, or to side setbacks adjacent to streets; in both instances, the setbacks applicable to the lot shall apply.
- 714.03 <u>Accessory Structures</u>. Accessory buildings and structures shall conform to the setbacks set or required for the primary dwelling unit.
- Privacy. In order to maintain privacy, no windows, doors, air conditioning units, or any other types of openings in the walls along a zero lot line shall be allowed except where such openings do not allow for visibility into the side yard of the adjacent lot, such as a clerestory skylight or opaque window.
- Eaves. Eaves along a zero lot line may project a maximum of 18 inches over the adjacent property line.
- Maintenance and Drainage Easements. A perpetual easement related to maintenance, eaves and drainage of at least five feet shall be provided on the lot adjacent to the zero lot line property line which, with the exception of walls and/or fences, shall be kept clear of structures. Such easement shall further provide that building wall along the zero lot line shall be maintained in its original color and treatment unless otherwise agreed to in writing by the two affected lot owners. Before any zero lot line lot is first conveyed after the construction of a zero lot line dwelling, the Zoning Administrator shall review and approve a substantially final form of such easement and shall ensure that such easement complies with the requirements of this Section 714.06.
- Platting Requirements. In connection with the application for a Special Use Permit for a Zero Lot Line Development, the applicant shall submit a plat showing the approximate proposed location of all buildings situated on a zero lot line lot, and the location of the easements described in Section 714.06.

714.08

<u>Special Rules for Special Use Permits</u>. To the extent that a Special Use Permit is required to create a Zero Lot Line Development, Section 804.10 shall not apply. Any such Special Use Permit shall expire in accordance with its terms, in accordance with with other law, upon a change in location of the improvements on the property, or upon the construction of additional buildings (not accessory structures) on the property.

Ordinance adopted April 6, 2010, added this section in its entirety.

# ARTICLE 8 ADMINISTRATION

#### 801.00 ZONING ADMINISTRATOR

Appointment and Removal of Zoning Administrator; Vacancies; Inability to Act. The Town Council shall appoint the Zoning Administrator, who shall serve at the pleasure of the Town Council. The Town Council may determine the Zoning Administrator's compensation directly, or may permit the Town Manager to determine such compensation. The Zoning Administrator may simultaneously hold another office in the Town government. When the office of Zoning Administrator is vacant, the Town Manager shall serve as Zoning Administrator. When the office of Zoning Administrator is not vacant, but the Zoning Administrator is absent from the Town Office or otherwise unable to act, the Town Manager may exercise all the powers and discharge all the functions of the Zoning Administrator (including, without limitation, those functions of the Zoning Administrator in his capacity as Subdivision Administrator under Section 912.01 of these Land Development Regulations).

Ordnance of February 7, 2023 rewrote this section to clarify that the Town Council appoints the Zoning Administrator and can remove him, and generally made provisions for a person other than the Town Manager to serve as Zoning Administrator. The amended section also provides the Town Manager with authority to act when the Zoning Administrator is absent or unable to act.

801.02 Powers and Duties Relating to Zoning. The Zoning Administrator is authorized and empowered on behalf of and in the name of the Broadway Town Council to administer and enforce the provisions set forth herein to the full extent permitted by Virginia law. Such authority includes, but is not limited to, receiving applications, inspecting premises, issuing Zoning Permits and Certificates of Occupancy for uses and structures which are in conformance with the provisions of this Zoning Ordinance; ordering, in writing, the remedy for any condition found in violation of this Zoning Ordinance; and the bringing of legal actions, including injunction, abatement, or other appropriate action or proceeding, to insure compliance with this Zoning Ordinance. The Zoning Administrator does not have the authority to take final action on applications or matters involving variances or modifications nor on Special Use Permits.

801.03 Engineering Report. Whenever any provision of this Zoning Ordinance requires or permits the submission of a site plan, plat, or sketch of any type, the Zoning Administrator may, at his option, submit such site plan, plat, or sketch to a professional engineer of his choice, retained by the Town, to obtain review of the plat by such engineer and such engineer's advice thereon. The proponent of the

site plan, plat, or sketch shall bear all fees associated with such engineer's review, regardless whether the request land development action is ultimately approved.

- Warrants. After reasonably but unsuccessfully attempting to reach agreement with a property owner or tenant of property that the Zoning Administrator reasonably wishes to inspect for the purpose of determining such property's compliance with any provision of this Ordinance, the Zoning Administrator may seek a warrant authorizing such an inspection from any magistrate with due jurisdiction, from the General District Court of Rockingham County, or from the Circuit Court of Rockingham County.
- 801.05 <u>Copies of Ordinance</u>. The Zoning Administrator shall maintain a copy of this Ordinance in his office.

#### 802.00 ZONING AND BUILDING PERMIT PROCEDURES

- 802.01 <u>Requirement for Permit</u>. No buildings or structures shall be constructed, repaired, reconstructed, enlarged, or altered, until after the Zoning Administrator has issued a Zoning Permit.
- Application. An application for a Zoning Permit for a permitted use shall be made in duplicate and accompanied by two (2) copies of an acceptable site plan with such reasonable information shown thereon as shall be required by the Zoning Administrator. Such site plan shall include, as a minimum, the following: lot dimensions with property line monuments located thereon; location and size of existing and proposed structures; yard dimensions and the use of structures; easements (private and public); water courses; fences; street names and street right-of-way lines; the proposed use of the site; and such other information regarding abutting property as directly affects the application.
- 802.03 <u>Fee.</u> Each application for a Zoning Permit shall be accompanied by payment of the fee indicated on the fee schedule of this Ordinance.
- Issuance of Permit. If the proposed building or use is in conformity with the provisions of this Ordinance and all other applicable law, the Zoning Administrator shall sign and return one (1) copy of the application to the applicant and shall issue a Zoning Permit. The permit shall indicate whether it is based on a use as of right, a Special Use Permit, or a variance. The Zoning Administrator shall retain the application and one (1) copy of the site plan for his records. If the application and site plan submitted does not conform to the requirements set forth in this Ordinance, the Zoning Administrator shall not issue a Zoning Permit, but shall return one (1) copy of the site plan to the applicant along with a signed refusal in writing. Such refusal shall state the reasons for refusal and shall cite the

portions of this Ordinance with which the submitted plan does not comply. The Zoning Administrator shall retain one copy of the site plan and one (1) copy of the refusal.

- 802.05 <u>Delinquent Taxes</u>. The Zoning Administrator shall not act on any application for a Zoning Permit if Town property taxes on the property that is the subject of the Zoning Permit application have not been paid.
- 802.06 <u>Lapse</u>. Any Zoning Permit shall automatically expire six (6) months from the date of issuance if the person to which the permit has issued has not clearly demonstrated that the permit is being exercised for the purpose for which it was issued, or if the work so authorized is suspended or discontinued for a period of one (1) year.

#### 803.00 CERTIFICATES OF OCCUPANCY

- 803.01 <u>Certificate of Occupancy Required.</u> A Certificate of Occupancy shall be required in advance of occupancy or use of a building hereafter erected; a building hereafter altered so as to affect height, or the side, front, or rear yard dimensions, or a change of type of occupancy or use of any building or premises.
- 803.02 <u>Issuance of Certificate of Occupancy</u>. The Rockingham County Department of Building shall sign and issue a Certificate of Occupancy, as stated on the application for such certificate and signed thereto by the owner or his appointed agent, is found to conform to the applicable provisions set forth herein and if the building, as finally constructed, complies with the sketch or plan submitted for the Zoning Permit.
- 803.03 <u>Denial of Certificate of Occupancy</u>. A Certificate of Occupancy shall not be issued unless the proposed use of a building or land conforms to the applicable provision set forth herein.

#### 804.00 SPECIAL USE PERMITS

Application Procedures for Special Uses. Any party in interest may apply for a Special Use Permit to allow any use that this Ordinance specifically authorizes as a Special Use in a particular zoning district. No other Special Uses are permissible. Applications for a Special Use Permit for a Special Use shall be submitted to the Zoning Administrator, who shall refer the application to the Planning Commission for a recommendation to the Broadway Town Council. Applications for Special Use Permits must contain the same information required

by an application for a Zoning Permit, along with any other information that the Zoning Administrator may reasonably require.

- Fee. Each application for a Special Use Permit shall be accompanied by payment of a fee as set forth in the fee schedule of this Ordinance. In addition, upon subsequent invoice, the applicant shall be liable for the cost of any public notice that Virginia law requires in connection with the consideration of the Special Use Permit.
- Planning Commission Consideration. The application shall be sent to the Planning Commission for review and recommendation, including the reporting of any conditions that the Planning Commission believes should attend a Special Use Permit. If the Commission fails to submit a report within a sixty (60) day period, the Town Council shall consider the application without the Planning Commission's recommendation.
- Town Council Consideration. The Town Council reserves to itself (and not to the 804.04 Board of Zoning Appeals) the right to consider and adjudicate applications for Special Use Permits. Before adjudicating such application, the Town Council shall hold any public hearing that Virginia law requires. No Special Use Permit shall issue unless the Town Council finds that the proposed special use is in accordance with the Comprehensive Plan, is in harmony with the general purposes and intent of this Ordinance and the other policies of the Town, and does not adversely affect the character of the zoning district, the conservation of property values, or the health and safety of residents and workers in the general neighborhood. The Town Council shall also consider the current and future need for the proposed use in the Town, whether the proposes Special Use is of such size, or so located and laid out in relation to access streets, that vehicular and pedestrian traffic to and from such use creates undue congestion or hazards prejudicial to the general neighborhood, and may consider any other factor that Virginia law permits. To the full extent that Virginia law permits, the Town Council may stipulate such additional requirements as are necessary to protect the public interest. The Town Council may require the applicant to furnish a performance bond in an amount sufficient for and conditioned upon the fulfilling of any and all conditions and requirements stipulated by the Town Council.
- 804.05 <u>Approval</u>. If the Town Council approves the application for a Special Use Permit, the Zoning Administrator shall issue a Zoning Permit, indicating the conditional nature of the use.
- 804.06 <u>Disapproval</u>. If the Town Council disapproves the application for a Special Use Permit, the Town Council shall cause the Zoning Administrator to inform the applicant of the decision in writing within thirty (30) days from the date of the disapproval, stating the reasons for disapproval. The Zoning Administrator shall

retain one (1) copy of the site plan and one (1) copy of the refusal, and shall keep them as a public record.

- 804.07 <u>Repeated Applications</u>. A property owner, or his appointed agent, shall not initiate a substantially similar application for a Special Use Permit relating to a rejected application more often than once every twelve (12) months.
- 804.08 <u>Lapse of Permit</u>. A Special Use Permit must be put into effect six (6) months after the date the permit is issued, unless otherwise provided in the permit itself. Expiration of a Special Use Permit shall occur with the discontinuance or suspension for a period of one (1) year of an activity that it authorized.
- 804.09 <u>Renewal of Permit.</u> Renewal of a Special Use Permit does not require a public hearing unless the original conditions in the permit are changed.
- 804.10 <u>Change in Ownership.</u> Upon a change in beneficial ownership, any Special Use Permit for the property shall expire. Provided, however, that this provision shall not apply to a Special Use Permit for a townhouse development.

Ordinance of July 7, 2015 added the last sentence of § 804.10

#### 805.00 AMENDMENT OF ORDINANCE

- 805.01 <u>Procedures for Amendment</u>. This Ordinance may be amended by any procedure permitted under Virginia law for the amendment or Zoning or Subdivision ordinances.
- 805.02 Special Procedures for Citizen-Initiated Amendment. Applications for amendments initiated by any person, firm, or corporation owning the subject property shall be submitted in writing to the Zoning Administrator and shall be accompanied by two (2) copies of an acceptable site plan, where applicable, of the proposed amendment with such reasonable information shown thereon as shall be required by the Zoning Administrator. Where site plans are required, they shall show, as a minimum, the following: lot dimensions with property line monuments located thereon; location and size of existing and proposed structures, yard dimension and the use of structures, easements (private and public) water courses, fences; street names and street right-of-way lines, and such other information regarding abutting property as directly affects the application. Proposals for amendments not initiated by either the Planning Commission or the Town Council shall be accompanied by payment of a fee as set forth in the fee schedule to this Ordinance.
- 805.03 <u>Withdrawal of Citizen-Initiated Amendment</u>. Any petition for an amendment may be withdrawn prior to action thereon by the Town Council at the discretion

of the person, firm or corporation initiating such a request, upon written notice to the Zoning Administrator.

- 805.04 <u>Repeated Applications</u>. No more than one citizen application for any amendment affecting a specific parcel of land or specific changes to the text of the regulations may be initiated during any single twelve (12) month period.
- Failure of Planning Commission to Submit Recommendation. The failure of the Planning Commission to submit a recommendation to the Town Council regarding a proposed amendment of this Ordinance or the zoning map within sixty (60) days after the first Planning Commission meeting following the referral of the same to the Planning Commission shall be deemed an approval by the Planning Commission of the proposed amendment.

#### 806.00 CONDITIONAL ZONING

- 806.01 <u>Intent.</u> The intent of this section is to provide for conditional zoning as part of the amendment of the zoning district map to the full extent permissible for the Town under Virginia law.
- Proffer of Conditions. An owner may proffer reasonable conditions, in addition to the regulations established elsewhere in this Ordinance, as part of an amendment to the zoning district map. The proffered conditions shall be in writing and shall be made prior to the public hearing before the Town Council, except that conditions not materially affecting the proposal may be proffered after the public hearing. The conditions shall be subject to all requirements and limitations imposed by Virginia law.
- 806.03 <u>Enforcement</u>. The Zoning Administrator shall have full authority to enforce all duly adopted conditions to the maximum extent permissible under Virginia law.

#### 807.00 BOARD OF ZONING APPEALS

- 807.01 <u>Establishment; Membership</u>. There is hereby established a Joint Board of Zoning Appeals for the Town of Broadway and the Town of Timberville. Such Board shall consist of five (5) members appointed as Virginia law requires. The persons who are members of the Board on the date of the enactment of this Ordinance shall continue in office for the terms to which they have been appointed.
- 807.02 <u>Compensation</u>. Members of the Board may receive compensation from funds appropriated for that purpose by the Town Council.

- 807.03 <u>Rules of Procedure</u>. Said Board is authorized to adopt rules that do not conflict with or vary the provisions of this Ordinance or other ordinances of the Town of Broadway or general laws of the Commonwealth for the conduct of its affairs.
- 807.04 Officers. Said Board shall elect a Chairman and a Vice Chairman from its own membership and a Secretary who may or may not be a member of the Board. Such officers shall serve annual terms as such and may succeed themselves.
- 807.05 <u>Conflicts.</u> Any member of said Board shall be disqualified to act upon a matter before said Board with respect to property in which the member has an interest.
- 807.06 <u>Meetings</u>. The meetings of said Board shall be held at the call of the Chairman and at such other times as said Board may determine.
- 807.07 <u>Quorum</u>. A quorum shall be at least three (3) members.
- Powers of Board. The Board of Zoning Appeals shall have all powers required or permitted under Virginia law, including the power to issue variances to the provisions of the Zoning Ordinance (but not the Subdivision Ordinance) under the standards and conditions that Virginia law prescribes, to hear and adjudicate appeals from the actions of the Zoning Administrator or other Town zoning officials that Virginia law authorizes, to hear appeals regarding the interpretation of the boundaries of the districts on the Zoning Map as Virginia law authorizes, and all other powers expressly permitted to Boards of Zoning Appeals under Virginia law, excepting specifically the power to issue and revoke Special Use Permits or special exceptions, which power is reserved to the Town Council.
- 807.09 <u>Procedures for Application for Variance</u>. An application for a variance shall be made to the Zoning Administrator in accordance with the following procedures:
  - 807.09-1 Submission of Application. The application shall be accompanied by an acceptable site plan with such reasonable information shown thereon as may be required by the Zoning Administrator. Such site plan shall include, as a minimum, the following: lot dimensions with property line monuments located thereon; location and size of existing and proposed structures; yard dimensions and the use of structures; easements (private and public); water courses; fences; road names and road right-of-way lines; and such other information regarding abutting property as directly affects the application. The Zoning Administrator shall transmit the application and accompanying maps, plans, or other information promptly to the Secretary of the Board. The Zoning Administrator shall also transmit a copy of the application and materials to the Planning Commission which may send a recommendation to the

Board or appear at the hearing at which the Board considers the application.

- Posting and Land. The Zoning Administrator shall cause to have posted in a conspicuous place on the property in question one or more signs, each of which shall not be less than six (6) square feet in area, shall contain information as to the proposed change and the date and time of the public hearing, and the cost of each shall be paid by the applicant prior to the public hearing. These signs shall be posted at least fifteen (15) days prior to the public hearing.
- 807.09-3 <u>Action</u>. Before adjudicating the application, the Board shall hold any public hearing that Virginia law requires on the application. The Board shall adjudicate the application within ninety (90) days from the date the Zoning Administrator receives the application unless the Board shall extend such deadline in its sole discretion.
- 807.09-4 <u>Limitation of Repetitive Applications</u>. A property owner or his appointed agent shall not initiate a substantially similar application for a variance more often than once every twelve (12) months without specific approval of the Board.
- 807.09-5 <u>Withdrawal of Application</u>. An application for a variance may be withdrawn prior to action thereon at the written request of the person, firm, or corporation initiating such application. Such request shall be delivered to the Zoning Administrator.
- 807.09-6 <u>Fee.</u> Each application for a variance shall be accompanied by payment of a fee as set forth in the fee schedule to this Ordinance. In addition, the applicant shall be required, upon invoice, to bear the cost of the publication of any notice required under Virginia law. All fees are nonrefundable on the withdrawal of an application.
- 807.10 <u>Procedures for Appeals</u>. In any situation where an appeal lies to the Board of Zoning Appeals, the procedures for considering such appeal shall be the ones provided by Virginia law. In addition, the following rules are applicable:
  - 807.10-1 <u>Board May Issue Zoning Permit</u>. In exercising the powers granted the Board under Virginia law to adjudicate appeals, the said Board may, in conformity with the provisions of this Ordinance, reverse or affirm, wholly or in part, or may modify the order, requirements, decisions, or determination of the Zoning Administrator, and to that end shall have all the powers of the

Zoning Administrator and may issue or direct the issuance of a Zoning Permit.

- 807.10-2 <u>Withdrawal of Application</u>. Any application for appeal before the Board may be withdrawn prior to action hereon by said Board at the discretion of the person, firm, or corporation initiating such a request upon written notice to the Secretary of said Board and the Zoning Administrator.
- 807.10-3 Fee. Each application for an appeal shall be accompanied by payment of a fee as set forth on the fee schedule to this Ordinance. In addition, the applicant shall be required, upon invoice, to bear the cost of the publication of any notice required under Virginia law. All fees are nonrefundable on the withdrawal of an application.
- Appeal Period. Any appeal from the decision of the Zoning Administration to the Board of Zoning Appeals shall be taken no later than thirty (30) days from the Zoning Administrator's written notification of his decision and the appeal rights of the recipient. Notwithstanding the foregoing, in the event of the Zoning Administrator's determination regarding a violation concerning temporary or seasonal commercial uses, parking of commercial trucks in residential zoning districts, or similar short-term, recurring violations, such appeal shall be filed within ten (10) days of such decision.
- 807.11 <u>Procedures for Application for Interpretation of Zoning Map</u>. In any proceeding before the Board of Zoning Appeals for the resolution of uncertainty regarding the boundary lines of districts on the zoning map, the following procedures shall be applicable.
  - Submission of Application. The application shall be accompanied by an acceptable site plan with such reasonable information shown thereon as may be required by the Zoning Administrator. Such site plan shall include, as a minimum, the following: lot dimensions with property line monuments located thereon, location and size of existing and proposed structures, yard dimensions and the use of structures, easements (private and public), water courses, fences, road names and road right-of-way lines, and such other information regarding abutting property as directly affects the application. The Zoning Administrator shall transmit the application and accompanying maps, plans, or other information promptly to the Secretary of the Board. The Zoning Administrator

shall also transmit a copy of the application and materials to the Planning Commission which may send a recommendation to the Board or appear at the hearing at which the Board considers the application.

- Action. Before adjudicating the application, the Board shall hold any public hearing that Virginia law requires on the application. The Board shall adjudicate the application within ninety (90) days from the date the Zoning Administrator receives the application unless the Board shall extend such deadline in its sole discretion.
- Withdrawal of Application. An application for an interpretation of the zoning map may be withdrawn prior to action thereon at the written request of the person, firm, or corporation initiating such application. Such request shall be delivered to the Zoning Administrator.
- 807.11-4 <u>Fee.</u> Each application shall be accompanied by payment of a fee as set forth in the fee schedule to this Ordinance. In addition, the applicant shall be required, upon invoice, to bear the cost of the publication of any notice required under Virginia law. All fees are nonrefundable on the withdrawal of an application.

#### 808.00 TRANSITIONAL ZONING

Areas unclassified by the zoning district map and areas newly added to the jurisdiction of the Town by annexation or other means shall be construed as being temporarily with the C-1 district until otherwise designated by action of the Town Council in accordance with the provisions of Virginia law.

#### 809.00 ENFORCEMENT OF ORDINANCE

- Officials to Follow Ordinance. All departments, officials, and public employees of the Town of Broadway which are vested with the duty or authority to issue permits or licenses shall conform to the provisions of this Ordinance. They shall issue permits for uses, buildings, or purposes, and shall approve subdivisions, only when they are in harmony with the provisions of these regulations. Any such permit or approval, if issued in conflict with the provisions of these regulations shall be null and void.
- 809.02 <u>Citizen Complaints.</u> Whenever a violation of these regulations occurs, or is alleged to have occurred, any person may file a written complaint. Such

complaint stating fully the causes and basis thereof shall be filed with the Zoning Administrator. He shall record properly such complaint, immediately investigate, and take any action thereon the he deems to be in the best interests of the Town or that the Town Council shall order.

Violations of Ordinance. Any person, firm, or corporation, whether as principal 809.03 agent, employee, or otherwise, violating, causing, or permitting the violation of any of the provisions of the Zoning Ordinance shall be guilty of a misdemeanor and, upon conviction thereof, may be fined up to one thousand dollars (\$1,000.00). If, at the time of conviction thereof, the violation has not been remedied, and a court of competent jurisdiction orders the remedy of such violation within a specified time frame, any person, firm, or corporation, whether as principal agent, employee, or otherwise, who fails to obey such order shall be guilty of a further misdemeanor and upon due conviction thereof shall be punishable by a fine of up to One Thousand Dollars (\$1,000.00). Such person, firm, or corporation shall be guilty of an additional misdemeanor for each ten (10) day period thereafter that such violation continues, and upon due conviction thereof shall be punishable by a fine of up to One Thousand Five Hundred Dollars (\$1,500.00). This section shall not limit any civil or other remedies available to the Town at law or in equity for the violation of this Ordinance.

#### 810.00 LEGAL STATUS PROVISIONS

- 810.01 <u>Conflict with Other Laws</u>. Wherever the requirements or regulations of this Ordinance are at variance with the requirements of any other lawfully adopted statutes, rules, regulations, or ordinances, the most restrictive or that imposing the higher standards, shall govern.
- Severability. Each phrase, sentence, paragraph, section, or other provision of this Ordinance is severable from all other such phrases, sentences, paragraphs, sections, and provisions. Should any phrase, sentence, paragraph, section or provision of these regulations be declared by the Courts to be unconstitutional, *ultra vires*, or otherwise invalid, such declaration shall not affect any other portion or provision of these regulations.
- 810.03 Repeal of Prior Land Development Regulations. This Ordinance is a comprehensive enactment of all of the resolutions and ordinances of the Town of Broadway relating to zoning and subdivision regulations. All prior ordinances affecting zoning and subdivision regulations are hereby repealed.
- 810.04 <u>Effective Date</u>. These regulations shall take effect and be in force from and after the date the Town Council approves them.

810.05

<u>Table of Contents; Typographical Errors</u>. The Town Attorney is authorized to create and maintain a table of contents for these Land Development Regulations, and to publish the same as part of the codification thereof. The Town Attorney is further authorized to correct obvious typographical errors in these Land Development Regulations when such correction involves no change in the substance thereof. In no event shall any such correction change the meaning or effect of these Land Development Regulations. Any such corrections shall be made by delivering a notice of such correction to the Town Clerk.

Ordinance of December 6, 2011 added this section.

#### 811.00 ATTENDANCE OF APPLICANTS AT MEETINGS

- Attendance. The applicant or proponent of any land development action, or an authorized representative of such applicant or proponent, shall appear at and attend (a) all public hearings, if any are required under this Ordinance or under Virginia law, and (b) the consideration of such land development action by a public body of the Town.
- Failure to Attend. In the event such proponent, applicant, or representative fails to so appear and attend, the public body may deny, dismiss, or defeat such land development action without prejudice to the right of such proponent or applicant to refile or reapply for such land development action; or, the public body may adjourn the public hearing or defer the consideration of such action until the next regularly scheduled meeting of the public body; provided, however, that the public body shall not adjourn the public hearing or defer consideration of any land development action if deferral would result in the approval of the land development action by virtue of inaction by the public body within the time provided by law for the approval or denial of such land development action.
- Notice of Deferral. If the public body defers a land development action or adjourns a public hearing in connection with one under this Section 811, it shall cause written notice of the time and place of its next regularly scheduled meeting to be mailed, first class, postage prepaid, to the proponent or applicant at the address indicated on the application form, not less than ten (10) days in advance of the date of such meeting, and the applicant, proponent, or his representative shall appear and attend at the time and place designated in such notice. In the event the applicant, proponent, or his representative fails to appear and attend the reconvened public hearing or deferred consideration of such land development action, the application shall be denied and dismissed by the public body, and the subject matter thereof shall not be considered by the public body for a period of six (6) months from the date of such denial and dismissal.

- Notice. The Town Manager shall append a statement setting forth the provisions of this section to all application forms to which this section is applicable.
- 811.05 <u>Definitions</u>. For the purposes of this Section 811, the term "public body" means the Town Council and the Planning Commission. The term "land development action" means a request for a rezoning or change in zone boundaries, a request for a variance, an application for a special use permit, a request for a subdivision, and a request for any vacation of or encroachment upon any public right of way

# TOWN OF BROADWAY, VIRGINIA LAND DEVELOPMENT REGULATIONS

## TABLE OF CONTENTS

Revision date appears in the lower right hand corner of each article.

ARTICL	<u> LE I: AUTHORITY AND ENACTMENT</u>	
101.00	AUTHORITY TO ESTABLISH ZONING	1
102.00	AUTHORITY TO ESTABLISH SUBDIVISION REGULATIONS	
103.00	ENACTMENT.	
103.00		
ARTICI	LE II: PURPOSES OF THE REGULATIONS	
201.00	PURPOSES	1
202.00	INTERPRETATION OF ORDINANCE	
203.00	NON-EXCLUSIONARY INTENT	2
ARTICI	LE III: DEFINITIONS OF TERMS USED IN THIS ORDINANCE	
AKIICI	EIII. DEFINITIONS OF TERMS USED IN THIS ORDINANCE	
301.00	GENERAL	1
302.00	SPECIFIC DEFINITIONS	1
303.00	CONTIGUOUSNESS	
ADTECT		
ARTICI	LE IV: ESTABLISHMENT OF DISTRICTS	
401.00	DIVISION OF THE TOWN OF BROADWAY INTO DISTRICTS	1
402.00	INCORPORATION OF THE ZONING MAP	
403.00	MAP AMENDMENT	
404.00	REPLACEMENT OF THE OFFICIAL ZONING MAP	
405.00	RULES FOR DETERMINING BOUNDARIES	
ARTICI	LE V: APPLICATION OF ZONING REGULATIONS	
501.00	USES	1
501.00	Permitted Uses	
501.01	Special Uses	
501.02	Accessory Use	
501.03	Ordinance Exclusive	
502.00	BUILDINGS	

503.00	LOTS AND YARDS	2
504.00	GARDENING	2
505.00	PERMITS ISSUED PRIOR TO ADOPTION OF ORDINANCE	2
506.00	EXISTING NON-CONFORMING LOTS, BUILDINGS, AND USES	2
506.01	Intent	
506.02	Lots of Record.	
506.02-1	Two or More Lots Not Zoned R-1	3
506.02-2	Two or More Lots Zoned R-1	3
506.02-3	Single Nonconforming Lots	3
506.03	Nonconforming Structures	3
506.03-1	Nonconforming Unsafe Structures	4
506.03-2	Enlargement of Nonconforming Structures	4
506.03-3	Installation of Plumbing in Nonconforming Structure	4
506.03-4	Movement of Nonconforming Structures	
506.03-5	Damage or Destruction of Nonconforming Structure by Natural Disaster or Act of	
506.02.6	God	
506.03-6	Damage or Destruction of Nonconforming Structure other than by Natural Disaster Act of God	
506.04	Nonconforming Uses of Land	
506.04-1	Extension of Nonconforming Use	
506.04-1	Return to Nonconforming Use Prohibited	
506.04-2	Relocation of nonconforming Use on Same Lot	
506.04-3	Cessation of Nonconforming Use	
506.04-4	Additional Structures in Support of Nonconforming Use	
506.04-5	Nonconforming Annexed Agricultural Land	
506.05	Nonconforming Uses of Structure	
506.05-1	Expansion of Structure	
506.05-1	Expansion Within Structure	
506.05-2	Discontinuance of Use	
506.05-4	Return to Nonconforming Use Prohibited	
506.05-4	Change from Nonconforming Use of Structure to Another Nonconforming Use	
506.05-3	Nonconforming Manufactured Homes	
506.07	Condominium Conversions	
507.00	PRIVATE ACCESS EASEMENTS	
307.00	TRIVATE ACCESS EASEMENTS	0
<u>ARTICL</u>	E VI: USES IN DISTRICTS	
601.00	DIVISION OF THE TOWN OF BROADWAY INTO DISTRICTS	1
601.00	Intent of the Conservation District C-1	
601.01	Permitted Uses	
611.00	LOW DENSITY RESIDENTIAL DISTRICT R-1	
611.00		
611.01	Intent of the Low Density Residential District R-1  Permitted Uses	
611.02		
011.03	Special Uses.	3

611.04	Accessory Uses	3
611.05	Lot Area Requirements	4
611.06	Lot Width Requirements	4
611.07	Setback Requirements	4
611.08	Yard Requirements	4
611.08-1	Rear Yard	4
611.08-2	Side Yard	4
611.09	Height Requirements	4
611.10	Accessory Structure Requirements	4
612.00	RESIDENTIAL DISTRICT R-2	5
612.01	Intent of Residential District R-2	5
612.02	Permitted Uses	5
612.03	Special Uses	5
612.04	Accessory Uses	6
612.05	Lot Area Requirements	7
612.06	Lot Width Requirements	
612.07	Setback Requirements	7
612.08	Yard Requirements	7
612.08-1	Rear Yard	7
612.08-2	Side Yard	7
612.09	Height Requirements	7
612.10	Accessory Structure Requirements	
613.00	MEDIUM DENSITY RESIDENTIAL DISTRICT R-3	
613.01	Intent of Medium Density Residential District R-3	8
613.02	Permitted Uses	
613.03	Special Uses	9
613.04	Accessory Uses	10
613.05	Lot Area Requirements	10
613.05-1	Area	10
613.05-2	Maximum Density	10
613.06	Lot Width Requirements	10
613.07	Setback Requirements	11
613.08	Yard Requirements	11
613.08-1	Rear Yard	11
613.08-2	Side Yard	11
613.09	Height Requirements	11
613.10	Accessory Structure Requirements	11
613.11	Street Design	11
613.12	Street Lighting	11
613.13	Pedestrian Circulation	11
613.14	Parking	12
614.00	RESIDENTIAL DISTRICT R-4	
614.01	Intent of the Residential District R-4	
614.02	Permitted Uses	12
614.03	Special Uses	
614.04	Accessory Uses	

614.05	Lot Area Requirements	13
614.06	Lot Width Requirements	13
614.07	Setback Requirements	14
614.08	Yard Requirements	14
614.08-1	Rear Yard	14
614.08-2	Side Yard	14
614.09	Height Requirements	14
614.10	Accessory Structure Requirements	14
614.11	Density Requirement	14
615.00	PLANNED UNIT DEVELOPMENT DISTRICT R-5	15
615.01	Intent of Planned Development District R-5	15
615.02	Permitted Uses	15
615.03	Other Uses	15
615.04	Uses Permitted by Approval	15
615.05	Qualifying Requirements	16
615.05-1	Ownership Requirements	16
615.05-2	Availability of Public Utilities	16
615.05-3	Land Suitability	16
615.06	Site Design Requirements	16
615.06-1	Maximum Density	16
615.06-2	Common Open Space	16
615.06-3	Functional Relationships	
615.06-4	Lot Design	17
615.06-5	Street Design	17
615.06-6	Street Names and Signs	17
615.06-7	Street Lighting	18
615.06-8	Pedestrian Circulation	18
615.06-9	Parking	18
615.06-10	Water and Sewer	18
615.06-11	Community Facilities	19
615.06-12	Fire Hydrants	19
615.06-13	Drainage	19
615.06-14	Floodways	19
615.06-15	Easements	19
615.06-16	Grading	19
615.06-17	Natural Amenities	19
615.06-18	Landscaping and Screening	20
615.07	Data to Accompany Application	
615.08	Application	21
615.09	Processing Fee	21
615.10	Preliminary Plan Approval	
615.10-1	By Planning Commission	
615.10-2	By Town Council	
615.11	Status of Approval	22
615.12	Final Plan Application	
615.13	Phasing Plan	

615.14	Compliance with Preliminary Plan	23
615.15	Final Plan Approval	23
615.16	Subdivision Plat Requirements	23
615.17	Required Improvements	23
615.18	Monuments	23
615.19	Plans and Specifications of Physical Improvements	23
615.20	Maintenance of Common Property	
615.21	Advertising and Sale	
615.22	Changes in Final Development	24
615.23	Development Schedule and Review	25
615.24	Repealed by Ordinance of December 6, 2011	25
621.00	GENERAL BUSINESS DISTRICT B-1	25
621.01	Intent of the General Business District B-1	25
621.02	Permitted Uses	26
621.03	Special Uses	29
621.04	Accessory Uses	30
621.05	Lot Area Requirements	30
621.06	Lot Width and Setback Requirements	31
621.07	Yard Requirements	31
621.07-1	Rear yard	31
621.07-2	Side yard	31
621.08	Height Requirements	31
621.09	Accessory Structure Requirements	
621.10	Additional Parking Requirements	
622.00	PLANNED BUSINESS DISTRICT B-2	
622.01	Intent of the Planned Business District B-2	31
622.02	Permitted Uses	32
622.03	Special Uses	33
622.04	Procedures for Consideration	33
622.05	Various Regulations	33
622.06	Setback Requirements	
631.00	LIGHT INDUSTRIAL DISTRICT M-1	
631.01	Intent of Light Industrial District M-1	34
631.02	Permitted Uses	34
631.03	Special Uses	35
631.04	Accessory Uses	
631.05	Submission of development plan	36
631.06	Lot Area Requirements	
631.07	Setback	
631.08	Yard Requirements	36
631.09	Height Requirements	
631.10	Accessory Structure Requirements	
631.11	Residential Zone Buffer	
632.00	GENERAL INDUSTRIAL DISTRICT M-2	
632.01	Intent of General Industrial M-2 District	37
632.02	Permitted Uses	

632.03	Special Uses	38
632.04	Accessory Uses	38
632.05	Submission of Development Plan	39
632.06	Lot Area Requirements	39
632.07	Setback	39
632.08	Yard Requirements	39
632.08-1	Rear yard	39
632.08-2	Side yard	39
632.09	Height Requirements	39
632.10	Accessory Structure Requirements	39
632.11	Residential Zone Buffer	
640.00	FLOOD HAZARD DISTRICTS	40
641.00	FLOOD HAZARD OVERLAY DISTRICT FH-1	40
641.01	Purpose of Flood Hazard District	40
641.02	Applicability	
641.03	Compliance and Liability	40
641.04	Abrogation and Greater Restrictions	
641.05	Definitions	
641.05-1	Base Flood/One Hundred (100) Year Flood	41
641.05-2	Base Flood Elevation (BFE)	
641.05-3	Development	
641.05-4	Flood	41
641.05-5	Floodplain	42
641.05-6	Flood-Prone Area	
641.05-7	Floodway	42
641.05-8	Historic Structure	42
641.05-9	Manufactured Home	42
641.05-10	New Construction	42
641.05-11	Recreational Vehicle	43
641.05-12	Start of Construction	43
641.05-13	Substantial Damage	43
	Substantial Improvement	
	Description of Districts	
641.06-1	Basis of Districts	44
641.06-2	Floodway District	44
641.06-3	Flood-Fringe District	
641.07	Overlay Concept	44
641.08	Official Zoning Map	
641.09	District Boundary Changes	45
641.10	Interpretation of District Boundaries	45
641.11	District Provisions	
641.12	Applications for Zoning Permit	
641.13	Manufactured Homes	
641.14	Recreational Vehicles	47
641.15	Floodway District	47
641.16	Permitted Uses in Floodway District	

641.17	Special Uses	48
641.18	Accessory Uses	49
641.19	Prohibited Uses	49
641.20	Flood-Fringe	50
641.21	Design Criteria for Utilities and Facilities	
641.21-1	Sanitary Sewer Facilities	
641.21-2	Water Facilities	
641.21-3	Drainage Facilities	50
641.21-4	Utilities	
641.21-5	Streets and Sidewalks	51
641.22	Special Use Permits and Variances: Additional Factors to be Satisfied	51
641.23	Special Procedures of Special Use Permits and Variances	52
641.24	Existing Structures in Floodplain Districts	
641.25	Violations and penalties	53
ADTICI	E VIII. LICE DECLII ATIONG	
AKTICL	E VII: USE REGULATIONS	
701.00	AREA REGULATIONS	1
701.00	Additional Regulations Regarding Yard and Setback Requirements	
701.01	Signs	
701.01-1	Party Walls	
701.01-3	Encroachments Permitted into Front Yards	
701.01 3	Accessory Structures and Yards	
701.01 4	Decks	
701.01-6	Compliance with Average Setback	
701.01 0	Special Provisions for Corner Lots in Residential Districts	2
701.02	Conditional Height Regulation	
701.03	Existing Lots Shall Abut on a Developed Street	
702.00	ADDITIONAL BUILDINGS ON A SINGLE LOT	
702.01	Additional Dwellings	
702.02	Temporary Buildings	
702.03	Additional Rules for Accessory Structures	
703.00	OFF-STREET PARKING	
703.01	Requirement for Off-Street Parking	
703.02	General Requirements	
703.02-1	Requirement for Off-Street Parking	
703.02-2	To Be Located on Same Lot	
703.02-3	Permissible Off-Lot Locations.	
703.02-4	Multiple Uses	
703.02-5	Change in Use	
703.02-6	Existing Nonconformities	
703.02 0	Site Requirements	
703.04	Parking Space Requirements for All Districts	
703.04-1	Dwellings	
. 55.51	- ··o-	

703.04-2	Public Assembly	7
703.04-3	Health Facilities	8
703.04-4	Businesses	8
703.04-5	Industries	9
703.04-6	Downtown Business Overlay District	10
703.05	Off-Street Loading and Unloading Space	10
703.05-1	Floor area over ten thousand (10,000) square feet	
703.05-2	Floor area less than ten thousand (10,000) square feet	
703.05-3	Size	
703.05-4	Connection to Road or Alley	11
703.05-5	Permanent Reservation	11
703.06	Shared Parking	11
704.00	HOME OCCUPATIONS	12
704.01	Purpose	12
704.02	Applicability	13
704.03	Permitted Districts; Permits	13
704.04	Exclusions from Operation of Section	13
704.05	General Standards for all Home Occupations	
704.06	Home Occupations Not Personal Service Home Occupations	15
704.07	Personal Service Home Occupations	
705.00	SIGNS	
705.01	Intent; Exclusion	16
705.02	General Requirements	
705.03	Permissible Signs in All Districts	
705.04	Signs as Permitted Uses	
705.04-1	Residential Districts	
705.04-2	Business Districts	21
705.04-3	Industrial Districts	24
705.05	Temporary Window Signs	24
705.06	Signs as Special Use	
705.06-1	Signs related to special use	
705.06-2	Directional signs related to special use	
705.07	Signs Prohibited in all Districts	
705.08	Maintenance and Removal of Signs	26
706.00	MANUFACTURED HOMES	
706.01	Requirements	26
706.02	Timing of Occupancy	
706.03	Inspections	
706.04	Manufactured Home Accessory Structures	
706.05	Manufactures Home Park Application and Site Plan	
706.06	Manufactured Home Park Design Standards	
706.06-1	Streets	
706.07	Vehicle Parking	
706.08	Lighting	
706.09	Disposition of Garbage and Rubbish	
706.10	Installation of Storage Tanks	

706.11	Record of Tenants for Manufactured Home Parks	30
706.12	Certificate of Use and Occupancy Required	30
707.00	PARKING OR STORAGE	30
707.01	Abandoned Vehicles	30
707.02	Storage of Vehicles	30
707.03	Movable Storage Boxes	31
707.03-1	Setbacks and Yards	
707.03-2	Temporary Placement	31
707.03-3	Natural Disasters	
708.00	SPECIAL REGULATIONS FOR TOWNHOUSES	32
708.01	Groupings	32
708.02	Fire Walls	32
708.03	Streets	32
708.04	Common Areas	33
708.05	Facades	33
708.06	Location of Utilities	33
708.07	Lighting	33
708.08	Storage of Trash and Rubbish	33
709.00	FENCES	33
709.01	Material	33
709.02	Corner Lots	34
709.03	Height Limit: Residential and Commercial	34
709.04	Height Limit: Industrial and Public Buildings	
709.05	No Setback Applicable; Not Accessory Structure	34
710.00	TELECOMMUNICATIONS TOWERS	
710.01	Purposes	34
710.02	Definitions	34
710.02-1	Antenna	34
710.02-2	Telecommunications Antenna	34
710.02-3	Telecommunications Tower	35
710.02-4	Alternative Telecommunications Structure	35
710.02-5	Height	35
710.03	Applicability	35
710.04	Special Use Permit Required	35
710.05	Siting Requirements	
710.06	Setbacks	36
710.07	Co-location	37
710.08	Abandonment	37
710.09	Security Requirements	37
711.00	NONCOMMERCIAL RADIO OR TELEVISION ANTENNAE OR DISHES	37
711.01	Rules	37
712.00	SPECIAL RULES FOR BED AND BREAKFAST ESTABLISHMENTS	
	AND INNS	38
712.01	Structure	38
712.02	Sign	38
712.03	Meals: Duration of Stay	

713.00	ADULT USES	39
713.01	Findings	39
713.01-1	Statutory Authorization	
713.01-2	Effect of Adult Businesses	39
713.01-3	Goals	39
713.02	Adult Uses in Districts	
713.03	Definitions	
713.04	Requirements and Standards for Adult Uses	
713.05	Measurement of Distance	
714.00	ZERO LOT LINE DEVELOPMENTS	
714.01	General Provisions	42
714.02	Building Setbacks	42
714.03	Accessory Structures	
714.04	Privacy	42
714.05	Eaves	43
714.06	Maintenance and Drainage Easements	43
714.07	Platting Requirements	43
714.08	Special Rules for Special Use Permits	43
ARTICL	E VIII: ADMINISTRATION	
001.00	ZONING A DAMNIGED A TOD	1
801.00	ZONING ADMINISTRATOR	
801.01	Appointment	
801.02	Powers and Duties Relating to Zoning	
801.03	Engineering Report	
801.04	Warrants	
801.05	Copies of Ordinance	
802.00	ZONING AND BUILDING PERMIT PROCEDURES	
802.01	Requirement for Permit	
802.02	Application	
802.03	Fee	
802.04	Issuance of Permit	
802.05	Delinquent Taxes	
802.06	Lapse	
803.00	CERTIFICATES OF OCCUPANCY	
803.01 803.02	Certificate of Occupancy Required	
803.02	Issuance of Certificate of Occupancy	
804.00	Denial of Certificate of Occupancy	
804.00		
804.01	Application Procedures for Special Uses	
804.02	FeePlanning Commission Consideration	4 1
804.03	Town Council Consideration	
804.04		
804.05	Approval	
004.00	Disapproval	

804.07	Repeated Applications	5
804.08	Lapse of Permit	5
804.09	Renewal of Permit	5
804.10	Change in Ownership	
805.00	AMENDMENT OF ORDINANCE	5
805.01	Procedures for Amendment	5
805.02	Special Procedures for Citizen-Initiated Amendment	5
805.03	Withdrawal of Citizen-Initiated Amendment	6
805.04	Repeated Applications	6
805.05	Failure of Planning Commission to Submit Recommendation	6
806.00	CONDITIONAL ZONING	
806.01	Intent	6
806.02	Proffer of Conditions	6
806.03	Enforcement	6
807.00	BOARD OF ZONING APPEALS	6
807.01	Establishment; Membership	6
807.02	Compensation	7
807.03	Rules of Procedure	7
807.04	Officers	7
807.05	Conflicts	7
807.06	Meetings	7
807.07	Quorum	7
807.08	Powers of Board	7
807.09	Procedures for Application for Variance	7
807.09-1	Submission of Application	
807.09-2	Posting and Land	8
807.09-3	Action.	8
807.09-4	Limitation of Repetitive Applications	8
807.09-5	Withdrawal of Application	
807.09-6	Fee	8
807.10	Procedures for Appeals	8
807.10-1	Board May Issue Zoning Permit	
807.10-2	Withdrawal of Application	9
807.10-3	Fee	
807.10-4	Appeal Period	9
807.11	Procedures for Application for Interpretation of Zoning Map	9
807.11-1	Submission of Application	9
807.11-2	Action	10
807.11-3	Withdrawal of Application	10
807.11-4	Fee	
808.00	TRANSITIONAL ZONING	10
808.01	Unclassified areas	10
809.00	ENCORCEMENT OF ORDINANCE	
809.01	Officials to follow Ordinance	
809.02	Citizen Complaints	
809.03	Violations of Ordinance	

810.00	LEGAL STATUS PROVISIONS	11
810.01	Conflict with Other Laws	11
810.02	Severability	11
810.03	Repeal of Prior Land Development Regulations	12
810.04	Effective Date	
810.05	Table of Contents; Typographical Errors	
811.00	ATTENDANCE OF APPLICANTS AT MEETINGS	
811.01	Attendance	
811.02	Failure to Attend	
811.03	Notice of Deferral	
811.04	Notice	
811.05	Definitions	
<u>ARTICL</u>	E IX: SUBDIVISION REGULATIONS	
901.00	SUBDIVISION REQUIREMENTS	1
902.00	GENERAL PROVISIONS	
902.01	Ordinance Mandatory	1
902.02	Approval Required	
902.03	Sales Prohibited Without Compliance	1
902.04	Penalty for Illegal Conveyance	1
902.05	Recordation	
902.06	Improvements	2
902.07	Necessary Changes	
902.08	Private Contracts	
902.09	Payment of Pro Rata Share of Certain Facilities	2
903.00	LOTS TO WHICH ORDINANCE APPLICABLE; APPROVAL PROCESS	
903.01	General Applicability	
903.02	Administrative Approval	
903.03	Planning Commission and Council Approval	
903.04	Partition Suits	
904.00	PROVISIONS REGARDING LOTS	4
904.01	Lot Size and Width	
904.02	Lot Shape	4
904.03	Remnants	5
905.00	PROVISIONS REGARDING STREETS AND RELATED PUBLIC IMPROVEMENTS	5
905.01	Street Orientation	5
905.01-1	Lot Shall Abut on a Street	5
905.01-2	Blocks	5
905.01-3	Street Alignment and Layout	5
905.01-4	Alleys	
905.01-5	Reserve Strips	
905.01-6	Cul-De-Sacs	6
905.01-7	Intersections	6

905.01-8	Curve Radii	6
905.02	Street Design	7
905.02-1	Street Width	7
905.02-2	Construction and Surfacing	7
905.02-3	Curbs and Gutters	7
905.02-4	Street Names	7
905.03	Monuments	8
905.04	Related Required Improvements	8
905.04-1	Sidewalks	8
905.04-2	Street Identification Signs	8
905.04-3	Lighting	8
905.04-4	Railroad Crossing	
906.00	UTILITY AND STORMWATER REQUIREMENTS	9
906.01	Plans and Specifications for Utility Fixtures and Systems to be Submitted for	
	Approval	9
906.02	Public Water and/or Sewer Required	9
906.03	Fire Protection	
906.04	Flood Control	9
906.05	Storm Drainage	9
906.05-1	Existing Storm Sewers	9
906.05-2	New Storm Sewers	10
906.05-3	Design of Storm Drainage Facilities	10
906.05-4	Watercourses	
906.05-5	Street Design	10
906.05-6	Certification of Required Improvements	10
906.05-7	Maintenance of Systems, etc	10
906.05-8	Stormwater and Sewer Systems	10
906.06	Easements	11
907.00	RELATION TO EROSION AND SEDIMENT CONTROL LAWS	11
907.01	Incorporation of Rockingham County Ordinance	11
907.02	Plan	11
908.00	PRELIMINARY SKETCHES AND PLATS	11
908.01	Preliminary Sketch	11
908.02	Preliminary Plat	11
908.03	Procedure for Consideration	13
908.04	Time Limits for Consideration	13
908.05	Amount of Bond	13
908.06	No Guarantee	13
908.07	Six Months Limit	13
908.08	Review by Engineer	13
909.00	REQUIREMENTS FOR FINAL PLATS	14
909.01	Characteristics of Final Plats	
909.02	Requirements	14
910.00	CONSIDERATION OF FINAL PLATS	15
910.01	Action on Plat	15
910.02	Time Limit	16

910.03	Required Surety	16
910.04	Additional Bond for Highway	16
910.05	Recordation	
910.06	Endorsement	17
911.00	VACATION OF PLAT OR BOUNDARY LINE	17
911.01	Administrative Approval	17
911.02	Vacation of Plat	17
912.00	ADMINISTRATION	17
912.01	Subdivision Administrator	17
912.02	Variances to Subdivision Ordinance	17
912.03	Delinquent Taxes	18

12000823

# TOWN OF BROADWAY, VIRGINIA LAND DEVELOPMENT REGULATIONS

## TABLE OF CURRENT ARTICLES

Revision date appears in the lower right hand corner of each article.

Title		Revision Date
Ι	AUTHORITY AND ENACTMENT	December 6, 2011
II	PURPOSES OF THE REGULATIONS	December 6, 2011
III	DEFINITIONS OF TERMS USED IN THIS ORDINANCE	May 2, 2023
IV	ESTABLISHMENT OF DISTRICTS	December 6, 2011
V	APPLICATION OF ZONING REGULATIONS	December 6, 2011
VI	USES IN DISTRICTS	October 3, 2023
VII	USE REGULATIONS	October 3, 2023
VIII	ADMINISTRATION	February 7, 2023
IX	SUBDIVISION REGULATIONS	July 7, 2015

# TOWN OF BROADWAY, VIRGINIA SCHEDULE OF TAXES, FEES, AND RATES

(Note: All taxes and fees are annual unless otherwise noted or the context of the Code clearly requires otherwise.)

This rate schedule is part of the Town Code and is designed to provide easy reference to the various tax rates enacted by the Town Council. It is authorized by Town Code § 50.10.

#### **Property Taxes**

Tax or Classification	Code Section	<u>Rate</u>
Real Estate	§ 51.10	\$0.10 per \$100 of assessed value
Automobiles and trucks (regardless of weight); motorcycles, mopeds, all- terrain vehicles, and off-road motorcycles; and boats (regardless of weight)	§ 52.10	\$0.61 per \$100 of assessed value
Manufactured Homes	§ 52.10	\$0.07 per \$100 of assessed value
Tools and Machinery	§ 53.01	\$0.40 per \$100 of assessed value

### **Business License Taxes or Fees**

Tax or Classification	Code Section	<u>Rate</u>
Business License Fee (applicable only if license tax would be less than \$25)	§§ 54.16, 54.35	\$25
Business License Tax – Retail Merchants	§ 54.16	\$0.08 per \$100 of gross receipts less than \$1,500,000 \$0.05 per \$100 of gross receipts more than \$1,500,000
Business License Tax – Wholesale Merchants	§ 54.16	\$0.03 per \$100 of gross receipts less than \$1,500,000 \$0.025 per \$100 of gross receipts more than \$1,500,000
Business License Tax – Contractors	§ 54.16	\$0.08 per \$100 of gross receipts less than \$1,500,000 \$0.05 per \$100 of gross receipts more than \$1,500,000
Business License Tax – Repair, Personal, Business and Other Services	§ 54.16	\$0.08 per \$100 of gross receipts less than \$1,500,000 \$0.05 per \$100 of gross receipts more than \$1,500,000
Business License Tax Financial, Real Estate and Professional Services	§ 54.16	\$0.20 per \$100 of gross receipts
Flat Business License Tax – Fortune Tellers	§ 54.31(a)	\$1,000
Flat Business License Tax Building or Savings and Loan Associations and Loan Companies	§ 54.31(b)	\$50
Business License Tax – Operators of Coin Machines	§ 54.32	\$0.08 per \$100 of gross receipts

Tax or Classification	Code Section	Rate
Flat Business License Tax – Resident Photographers with no Definite Place of Business	§ 54.33	\$30
Flat Business License Tax – Circuses and Carnivals	§ 54.34	\$100 per performance
Business License Tax – Telephone companies	§ 54.38	one-half of 1% of gross receipts

### Alcoholic Beverage License Fees

Tax or Classification	<b>Code Section</b>	<u>Rate</u>
Alcoholic Beverage License Tax – Distillers	§ 57.21	5,000 gallons or less, exempt; 5,000 gallons – 36,00 gallons, \$750; 36,000 gallons or more, \$1000
Alcoholic Beverage License Tax – Fruit Distillers	§ 57.21	\$1,500
Alcoholic Beverage License Tax – Bed and Breakfast	§ 57.21	\$40
Alcoholic Beverage License Tax – Museums	§ 57.21	\$10
Alcoholic Beverage License Tax – Tastings	§ 57.21	\$5 per event
Alcoholic Beverage License Tax – Equine Events	§ 57.21	\$10
Alcoholic Beverage License Tax – Day Spas	§ 57.21	\$20
Alcoholic Beverage License Tax – Motor Car Sporting Event Facility	§ 57.21	\$10
Alcoholic Beverage License Tax – Meal Assembly Kitchen	§ 57.21	\$20
Alcoholic Beverage License Tax – Annual Arts Venue Event	§ 57.21	\$20
Alcoholic Beverage License Tax – Art Instruction Studio	§ 57.21	\$20
Alcoholic Beverage License Tax – Brewery	§ 57.21	500 barrels or less, \$250; more than 500 barrels, \$1,000
Alcoholic Beverage License Tax – Beer Bottler	§ 57.21	\$500

Tax or Classification	Code Section	<u>Rate</u>
Alcoholic Beverage License Tax – Wholesale Beer	§ 57.21	\$75 (credit against BPOL tax)
Alcoholic Beverage License Tax – Retail beer in restaurant, hotel or, club, or retail off- premises	§ 57.21	\$25
Alcoholic Beverage License Tax – Beer shipper	§ 57.21	\$10
Alcoholic Beverage License Tax – Winery	§ 57.21	\$50
Alcoholic Beverage License Tax – Wholesale Wine	§ 57.21	\$50 (credit against BPOL tax)
Alcoholic Beverage License Tax – Farm Winery	§ 57.21	\$50
Alcoholic Beverage License Tax – Wine Shipper	§ 57.21	\$10
Alcoholic Beverage License Tax Retail beer and wine in restaurant, hotel or, club, or retail off-premises	§ 57.21	\$37.50
Alcoholic Beverage License Tax – Hospital, Wine and Beer	§ 57.21	\$10
Alcoholic Beverage License Tax – Wine and beer, Banquets (temporary)	§ 57.21	one day: \$5; multi-day, \$20
Alcoholic Beverage License Tax – Wine and beer, gourmet brewing shop	§ 57.21	\$150
Alcoholic Beverage License Tax – Wine and Beer Shipper	§ 57.21	\$10

Tax or Classification	Code Section	<u>Rate</u>
Alcoholic Beverage License Tax – Wine and beer, banquet facility (annual)	§ 57.21	\$15
Alcoholic Beverage License Tax – Wine and beer, gourmet oyster house	§ 57.21	\$37.50
Alcoholic Beverage License Tax – Mixed beverages, restaurant	§ 57.21	Up to 100 people, \$200; 100- 150 people, \$350; more than 150 people, \$500
Alcoholic Beverage License Tax – Mixed beverages, private, non-profit club	§ 57.21	\$350
Alcoholic Beverage License Tax – Mixed beverages, caterer	§ 57.21	\$500
Alcoholic Beverage License Tax – Mixed Beverages, Special Events	§ 57.21	\$10 per day
Alcoholic Beverage License Tax – Mixed Beverages, Club Events	§ 57.21	\$10 per day
Alcoholic Beverage License Tax – Mixed Beverages, Motor Sports Events	§ 57.21	\$300
Alcoholic Beverage License Tax – Mixed Beverages, Banquets	§ 57.21	\$75
Alcoholic Beverage License Tax – Limited Mixed Beverages, Restaurant	§ 57.21	Up to 100 people, \$100; 100- 150 people, \$250; more than 150 people, \$400
Alcoholic Beverage License Tax –Mixed Beverages, Motor Sports Facility (annual)	§ 57.21	\$300

Tax or Classification	Code Section	<u>Rate</u>
Alcoholic Beverage License Tax –Mixed Beverages, Performing Arts Facility (annual)	§ 57.21	\$300

### <u>Utilities Taxes and Fees</u>

These taxes are monthly.

Tax or Classification	Code Section	<u>Rate</u>
Consumption Tax – Electricity	§ 57.34	under 2,500 kWh, \$0.00038 per kWh; between 2,500 kWh and 50,000 kWh, \$0.00024 per kWh; over 50,000 kWh, \$0.00018 per kWh
Consumption Tax – Natural Gas	§ 57.34	\$0.004 per CCF
Consumer Utility Tax – Electricity	§ 57.32	Residential Consumers: \$1.05 plus the rate of \$0.011261 on each kWh delivered monthly to residential consumers by a service provider not to exceed \$1.50 monthly.  Commercial and Industrial  Consumers: \$1.72 plus the rate of \$0.010791 on each kWh delivered monthly to such consumers, not to exceed \$15.00 monthly.
Consumer Utility Tax – Natural Gas	§ 57.33	Residential Consumers: \$1.50 per month  Commercial and Industrial Consumers: \$3.50 plus the rate of \$0.045 on each CCF delivered monthly, not to exceed \$15.00 monthly
Public Right of Way Permit Fee	§ 135.09	\$100

### Other Miscellaneous Taxes and Fees

Tax or Classification	Code Section	<u>Rate</u>
Meals Tax	§ 55.02	4.5%
Cigarettes	§ 57.13	\$0.0125 per cigarette
Bank Franchise Tax	§ 57.42	80% of state rate per \$100 of capital
Transient Occupancy Tax	§ 58.02	7%

### Permit Application Fees

<u>Permit</u>	Code Section	<u>Amount</u>
Permit for an Itinerant Merchant, Peddler, or Solicitor	§ 152.03	\$500
Short Term Rental Registration	§ 151.03	\$25

### **Utility-Related Fees**

<u>Fee</u>	Code Section	<u>Amount</u>
Water Connection	§ 132.09	\$6,500, including the meter
Sewer Connection	§ 131.11, 131.40	\$6,100

### **Utility Rates**

<u>Fee</u>	Code Section	<u>Amount</u>
Water Use	§ 132.01(b)	(per two months)
		0-3,000 gal - \$23.52
		3,001-100,000 gal – add'l \$5.61 per 1,000 gal
		100,001-400,000 gal – add'1 \$5.71 per 1,000 gal
		Above 400,000 gal – add'l \$6.38 per 1,000 gal
Sewer Use	§ 131.40	(per two months)
		0-3,000 gal - \$17.81
		3,001-100,000 gal – add'1 \$4.21 per 1,000 gal
		100,001-400,000 gal – add'1 \$4.28 per 1,000 gal
		Above 400,00 gal – add'l \$4.53 per 1,000 gal
Solid Waste Disposal	§ 134.12(b)	\$36.00 (per two months) \$6.00 (additional toter)
D :	66 121 40 122 00	, ,
Deposits	§§ 131.40, 132.09	\$200 for renters; no deposit for owners
Reconnection Fee	§ 135.04	\$50

# AN ORDINANCE TO AMEND, RE-ENACT, AND RECODIFY THE GENERAL ORDINANCES OF THE TOWN OF BROADWAY, VIRIGNIA, AND REPEALING THE TOWN CODE OF THE TOWN OF BROADWAY, VIRGINIA (2003)

WHEREAS the Town Council (the "Council") of the Town of Broadway, Virginia (the "Town") is empowered under Virginia Code § 15.2-1433 to enact ordinances, to codify such ordinances into an organized town code, and to amend, re-enact, and re-codify such ordinances at its pleasure; and

WHEREAS the Town Code of the Town of Broadway was last recodified and re-enacted in 2003; and

WHEREAS the Council has determined that it is now appropriate to amend, revise, re-enact, and recodify ordinances of general applicability in the Town.

NOW, THEREFORE, BE IT ENACTED BY THE TOWN COUNCIL OF THE TOWN OF BROADWAY, VIRGINIA, to wit:

### TITLE I: GENERAL PROVISIONS

Chapter

10. GENERAL PROVISIONS

#### **CHAPTER 10: GENERAL PROVISIONS**

#### § 10.01 TITLE AND NATURE

- (a) This Code shall be known as the "Town Code of the Town of Broadway, Virginia (2023)".
- (b) All ordinances of a permanent and general nature (except ordinances relating to zoning or subdivision, which are contained in the Land Development Regulations), as revised, codified, rearranged, renumbered, and consolidated into component codes, titles, chapters, and sections, shall be known and designated as the "Code," for which designation "code of ordinances," or "codified ordinances" may be substituted. Code title, chapter, and section headings do not constitute any part of the law as contained in the Code.
- (c) All references to codes, titles, chapters, and sections are to the components of the Code unless otherwise specified. Sections of this Code may refer to or cite other sections of this Code by the designation"\\$" followed by the number, such as "\\$ 10.01", and with or without the preceding phrase "Town Code". Headings and captions used in this Code other than the title, chapter, and section numbers are employed for reference purposes only and shall not be deemed a part of the text of any section.
- (d) Whenever in one section reference is made to another section hereof, that reference shall extend and apply to the section referred to as subsequently amended, revised, recodified, or renumbered unless the subject matter is changed or materially altered by the amendment or revision.

#### § 10.02 RULES OF INTERPRETATION

- (a) Unless otherwise provided herein, or by law or implication required, the same rules of construction, definition, and application shall govern the interpretation of this Code as those governing the interpretation of state law.
- (b) The construction of this Code shall be moreover by the following rules, unless that construction is plainly repugnant to the intent of the Council or of the context of the section being interpreted:
- (i) When a statute, Code provision or ordinance requires an act to be done which, by law, an agent, designee, assistant, or deputy as well may do as the principal, that requirement shall be satisfied by the performance of the act by an authorized agent or deputy.
- (ii) Words denoting the masculine gender shall be deemed to include the feminine and neuter genders; words in the singular shall include the plural, and words in the plural shall include the singular; the use of a verb in the present tense shall include the future, if applicable.

- (iii) A general term following specific enumeration of terms is not to be limited to the class enumerated unless expressly so limited.
- (c) This Code shall be interpreted in harmony with the Town's Land Development Regulations as one, unified legal code for the Town.

#### § 10.03 APPLICATION TO FUTURE ORDINANCES

All provisions of this Title I shall apply to ordinances hereafter adopted which amend or supplement this code unless otherwise specifically provided.

#### **§ 10.04 CAPTIONS**

Headings and captions used in this Code other than section numbers are employed for reference purposes only and shall not be deemed a part of the text of any section.

#### § 10.05 GENERAL DEFINITIONS

- (a) Words and phrases shall be taken in their plain, or ordinary and usual sense. However, technical words and phrases having a peculiar and appropriate meaning in law shall be understood according to their technical import.
- (b) For the purpose of this Code, the following definitions shall apply unless the context clearly indicates or requires a different meaning.

*Code* means this code as modified by amendment, revision, and adoption of new titles, chapters, or sections, but does not include the Land Development Regulations.

Council or Town Council means the Town Council of the Town.

Land Development Regulations means the Town's Land Development Regulations, adopted originally on December 6, 2011, and subsequently amended, which consist of the Town's Zoning Ordinance and the Town's Subdivision Ordinance.

Law means any constitution, charter, statute, ordinance, judicial decision, or regulation, whether local, state, or federal.

May means that the act referred to is permissive.

Month or calendar month means a calendar month, regardless of the number of days in such month.

Oath includes an affirmation in all cases in which, by law, an affirmation may be substituted for an oath, and in those cases the words swear and sworn shall be equivalent to the

words *affirm* and *affirmed*. All terms shall mean a pledge taken by the person and administered by any individual authorized under state law.

Officer, Office, Employee, Commission, or Department mean an officer, office, employee, commission, or department of the Town unless the context clearly requires otherwise

*Person* means a natural person, individual, person, persons, firm, corporation, partnership, trustee or other fiduciary, limited liability company, lessee, or receiver. Whenever used in any clause requiring an act or proscribing an omission, the terms *person* or *whoever* as applied to any unincorporated entity shall mean the partners or members thereof, and as applied to corporations or limited liability company, the officers or agents thereof.

Preceding or following mean next before or next after, respectively.

Shall means that the act referred to is mandatory.

Signature includes a mark when the person cannot write, and also includes an electronic signature in any instance that state law permits.

State means the Commonwealth of Virginia.

*Town* means the Town of Broadway, Virginia.

Written means any representation of words, letters, or figures, whether by printing or otherwise, and includes an e-mail, word processing or PDF document, or other electronic collection of words, letters, or figures.

*Year* means a calendar year, unless otherwise expressed.

(c) A defined term in this Code may be capitalized for clarity, but the lack of capitalization is not to be taken to mean that the Town Council intends a different definition.

#### § 10.06 SEVERABILITY

If any provision of this Code as now or later amended or its application to any person or circumstance is held invalid, the invalidity does not affect other provisions that can be given effect without the invalid provision or application.

# § 10.07 AUTHORITY OF TOWN ATTORNEY TO RENUMBER, CORRECT ERRORS, ETC.

(a) The Town Attorney is hereby authorized and empowered, without action of the Town Council, to (i) create a table of contents for this code; (b) to renumber, reorganize, recodify, or reclassify any provision in this Code, without altering the substance thereof, to provide for the better or more concise organization of the Code, or to prevent duplication of section numbers; and

- (c) to correct obvious scriviner's errors in this Code or in any other enactment of the Town Council, in order to implement the manifest intention of the Council.
- (b) The Town Attorney shall exercise his powers under paragraph (a) by transmitting a letter to the Town Clerk detailing the changes being made and his reasons for the same.

#### § 10.08 REPEAL OF PRIOR CODE

This Code, from and after its effective date, shall contain all of the provisions of a general nature pertaining to the subjects herein enumerated and embraced. All prior ordinances pertaining to the subjects treated by this Code shall be deemed repealed from and after the effective date of this Code. The Town Code of the Town of Broadway adopted on August 5, 2003, is hereby repealed. Nothing in this Code shall affect the validity of the Town's Land Development Regulations, which are not repealed, or of the Town's existing uncodified guidelines for implementing the Public-Private Education Facilities and Infrastructure Act of 2002 (presently codified at Virginia Code § 56-575.1 *et seq.*).

#### § 10.09 SAVING OF PRIOR UNCODIFIED ORDINANCES

All ordinances of a temporary or special nature and all other ordinances pertaining to subjects not embraced in this Code shall remain in full force and effect unless herein repealed expressly or by necessary implication.

#### § 10.10 EFFECTIVE DATES

- (a) Future ordinances shall take effect upon the Council's adoption of such ordinances unless the Council specifies a different effective date.
- (b) This Code shall take effect on the first day of the calendar month following its adoption.

#### § 10.11 TRANSITION PROVISIONS

- (a) All offenses committed under laws in force prior to the effective date of this Code shall be prosecuted and remain punishable as provided by those laws. This Code does not affect any rights or liabilities accrued, penalties incurred, or proceedings begun prior to the effective date of this Code. Such liabilities, proceedings and rights are continued; punishments, penalties, or forfeitures shall be enforced and imposed as if this Code had not been enacted.
- (b) In connection both the adoption of this Code, and the future amendment hereof, no suit, proceedings, right, fine, forfeiture, or penalty instituted, created, given, secured, or accrued under any ordinance previous to its repeal shall in any way be affected, released, or discharged, but may be prosecuted, enjoyed, and recovered as fully as if the ordinance had continued in force unless it is otherwise expressly provided.

(c) When any ordinance repealing a former ordinance, clause, or provision shall be itself repealed, the repeal shall not be construed to revive the former ordinance, clause, or provision, unless it is expressly provided.

#### § 10.12 REFERENCES TO OTHER LAW

When this Code references any other law, all future amendments to such law, together with any recodifications or changes in section number, are adopted by reference in this Code as if they had been in existence at the time this code was adopted, unless this Code expresses a clear intention to the contrary.

#### § 10.13 GENERAL PROVISIONS REGARDING PENALTIES

- (a) Whenever any section of this Code criminalizes any act or omission, and this Code does not provide a penalty for such act or omission, the person engaging in such act or omission shall be guilty of the lesser of (i) a Class 1 misdemeanor, as that term is defined in State law, or (ii) the maximum penalty permissible under state law.
- (b) The imposition of any penalty pursuant to this or any other provision of this Code, shall not prohibit the Town from seeking equitable relief in any court of competent jurisdiction, to enjoin the violation of any provision of this Code or other ordinance of the locality.
- (c) Each day any violation of this Code or any other ordinance of the Town shall continue shall constitute a separate offense, except where otherwise provided.
- (d) In the event this Code provides for a penalty for an act or omission that impermissibly exceeds the penalty imposed under state law, then the penalty imposed under this Code shall be the maximum penalty imposed under state law.

#### § 10.14 GENERAL PROVISIONS REGARDING NOTICES

- (a) The provisions of this section shall apply whenever the Town is required under this Code or other law (other than the Land Development Regulations) to provide notice to someone, and this Code or other law does not provide procedures for giving such notice.
- (b) The Town may give notice by any one or more of the following methods, and such notices shall be effective at the earlier of the time indicated below, or the time indicated in paragraph (c):
- (i) By personal delivery to the person affected, whether such delivery occurs through a sheriff, a Town police officer, another agent or employee of the Town, or a private process server; and such notice shall be effective when delivered;
- (ii) In the event that such notice concerns an act or omission in connection with particular property that the Town knows is unoccupied or not owner-occupied, or when the Town

does not know whether the property is unoccupied or owner occupied, by both (A) posting on the front door to such property, and (B) providing notice to the owner of such property at the owner's last known address; and such notice shall be effective as of the latest date that both such posting and such other notice have been accomplished;

- (iii) In the event that such notice concerns an act or omission in connection with particular property that the Town knows is owner-occupied, by posting on the front door to such property;
- (iv) In cases where the notice does not concern an act or omission in connection with any particular property, by posting on the front door of the last known address of the person affected, and such notice shall be effective when posted;
- (v) By deposit with a reputable overnight delivery service or overnight U.S. Mail, addressed to the last known address of the person affected; and such notice shall be effective one business day after such deposit; or
- (vi) By certified mail, return receipt requested, to the last known address of the person affected; and such notice shall be effective five (5) days after such notice is sent.
- (c) Notices to be provided under the Land Development Regulations shall be governed by the provisions thereon, or by state law, as applicable, and not by this section.
- (d) The Town may determine a recipient's last known address by any reasonable method, including without limitation the following methods: (i) consultation with the records of Rockingham County officials, such as (without limitation) the Commissioner of the Revenue, the Treasurer, or the Rockingham County Circuit Court; and (ii) consultation with the Town's utility, tax, or other records.

#### § 10.15 NOTICE ACTUALLY RECEIVED

Whenever this Code requires notice, notice that a recipient has actually received is valid notice to such person, effective when so received, even if not given by the method prescribed.

#### § 10.16 NOTICE TO PERSONS NOT NATURAL PERSONS

Whenever this Code requires notice to a person that is not a natural person, notice to such a person may be delivered to any registered agent for such person, or to any official who is an authorized recipient of service to such person under state law related to service upon persons who are natural persons.

#### TITLE III: ADMINISTRATION

### Chapter

- 30. GOVERNING BODY; ELECTED OFFICERS
- 31. UNELECTED OFFICERS AND EMPLOYEES
- 32. DEPARTMENTS, BOARDS AND COMMISSIONS
- 33. ELECTIONS
- 34. FINANCE
- 35. PROCUREMENT ACT PROVISIONS

#### **CHAPTER 30: GOVERNING BODY; ELECTED OFFICERS**

#### § 30.01 TERM OF VICE-MAYOR

At the Organizational Meeting of the Town Council each year, the Town Council shall elect one of its members to serve as the Town's Vice-Mayor. The Vice Mayor shall serve until the following annual Organizational Meeting of the Town Council, or until he is no longer a member of the Town Council, whichever comes first. The failure of the Town Council to elect a Vice-Mayor during its organizational meeting shall not affect the validity of the term or office of a Vice-Mayor elected at another meeting.

#### § 30.02 MEETINGS OF TOWN COUNCIL

- (a) <u>Organizational Meeting</u>. The Town Council shall hold its Organizational Meeting each year on the first Tuesday in January, at 7:00 PM, at the Town Council Chambers. If the date of the Organizational Meeting falls on a state holiday, the Organizational Meeting shall be held on the next succeeding Tuesday, unless the Town Council by motion or resolution shall set a different date. Notwithstanding the foregoing, the Mayor of the Town may, by proclamation, fix a different date within the month of January for the organizational meeting. Such proclamation shall be posted at the Town Office, and copies posted on the Town's website.
- (b) Other Regular Meetings. Pursuant to its charter power to provide, by ordinance, for the time of the meetings of the Town Council, the Town Council provides as follows: Regular meetings of the Town Council shall take place on the first Tuesday of every month, and on the Thursday preceding such Tuesday, in each instance at 7:00 PM, at the Town Council Chambers. In the event that the first Tuesday of a month is a state holiday or is the day before a state holiday, or in the event that a quorum cannot be obtained on such date, such meeting shall take place on the following Tuesday without any action by the Town Council. The Council may, by motion, set a different date for any regular meeting, and shall post notice of such changed meeting on the Town's website and on the bulletin board outside the Town Office.
- (c) <u>Special Meetings</u>. Special meetings of the Town Council may be held in accordance with the procedures prescribed in the Town Charter.

#### § 30.03 SALARIES OF TOWN COUNCIL MEMEBERS

Each Town Council member shall receive a salary of \$250 per month, except the Mayor, who shall receive a salary of \$300/month. A member of the Town Council who serves as a representative to the Broadway Hometown Partnership or to the Central Shenandoah Planning District Commission shall receive an additional salary of \$50 per month.

# § 30.04 COUNCIL REPRESENTATIVES AND DEPARTMENT LIAISONS; COMMITTEES

- (a) The Mayor shall designate one of the Town Council members to serve as its representative on each of the Planning Commission, the Broadway Hometown Partnership, and the Central Shenandoah Planning District Commission.
- (b) The Mayor may appoint a Member of the Town Council to serve as the liaison between Town staff and the Town Council for a particular subject or policy area.
- (c) The Town Council may form such standing or ad hoc committees as it may deem expedient to carry out its business. The Mayor shall appoint the members of all such committees.

#### § 30.05 PUBLIC HEARINGS; PUBLIC COMMENT

- (a) The Town Council and the Planning Commission shall hold public hearings (a) whenever required by state or federal law; (b) whenever specified by motion or other action of the Town Council or Planning Commission, respectively; or (c) when directed by the Town Manager, the Mayor, or the Chairman of the Planning Commission to facilitate business on an agenda for a particular meeting of the Town Council or Planning Commission. The Town Manager and the Town Attorney shall cause any necessary advertisements for such public hearings to appear in the *Daily News-Record* or other publication appropriate under Virginia law.
- (b) The presiding officer at any public hearing or public comment period may set reasonable limits on the time allotted to any speaker from the general public. Such limits shall be enforced with reasonable uniformity between speakers, and without reference to the viewpoint of the speaker. The presiding officer may allow the applicant or principal proponent of the proposed action to speak at greater length than commenters from the general public, in order to allow presentation of the proposed action, or to respond to comments from members of the public.

#### § 30.06 POLICY REGARDING REMOTE PARTICIPATION IN MEETINGS

- (a) A Town Council member may participate in a meeting through electronic communication means from a remote location, including a location that is not open to the public, only as described in this section.
- (b) Remote participation is permissible in the event a Town Council member is unable to attend the meeting in person because of (a) a personal matter; or (b) a temporary or permanent disability or other medical condition prevents the Town Council member from attending; or (c) a member of the Town Council's family has a medical condition that requires

the member to provide care for such family member, thereby preventing the Town Council member's physical attendance.

- (c) In the event that the Town Council member seeks to participate remotely on account of a personal matter, then on or before the day of a meeting, the member shall notify the Town Manager that the member is unable to attend the meeting due to a personal matter. The member must identify with specificity the nature of the personal matter. At the beginning of the meeting, the Town Council must then vote to approve or disapprove the member's remote participation.
- (d) In the event that the Town Council member seeks to participate remotely on account of the disability or medical circumstances described above, then on or before the day of a meeting, the member shall notify the Town Manager that the member is unable to attend the meeting due to such circumstances, and shall specify whether such circumstances relate to the member's own disability or medical condition, or the medical condition of a family member. At the beginning of the meeting, the Town Council must then vote to approve or disapprove the member's remote participation.
- (e) In the event that the Town Council votes to approve the remote participation, the minutes of the meeting shall include: (a) the fact of the member's remote participation; (b) the grounds for remote participation as provided to the Town Manager (including the specific nature of the personal matters that prevent participation in person, if the reason for the remote participation if a personal matter); and (c) the physical location from which the member is participating remotely (which location need not be open to the public).
- (f) The Town Council shall not disapprove remote participation unless such participation would violate this policy or other law. In the event of such disapproval, the minutes shall record with specificity the violations that members of the Town Council believed would have occurred in connection with the remote participation.
- (g) Notwithstanding the remote participation of one or more members pursuant to this policy, a quorum of the Town Council must nevertheless be assembled physically, in person, at the Town Council chambers or other central meeting location.
- (h) This policy shall be applied strictly and uniformly, without exception, to the entire membership of the Town Council and without regard to the identity of the member requesting remote participation or the matters that will be considered or voted on at the meeting.
- (i) A member may participate remotely on account of a personal matter no more often than three (3) times in any calendar year. There is no limit to the number of times that a member may participate remotely due to a temporary or permanent disability or other medical condition, or the medical condition of a family member as described above.

- (j) The Town Manager shall ensure that the member participating remotely can hear all comments made in the meeting room, and that other members of the Town Council and the public can hear comments that the member participating remotely makes.
- (k) If the Town Council goes into closed session, the member participating remotely shall ensure that no third party is able to hear or otherwise observe the closed meeting.
- (l) This section shall not apply to instances in which other law permits the Town Council as a whole to meet by electronic communication means, even if some Town Council members are present in person at a public meeting location during such meetings.

#### CHAPTER 31: UNELECTED OFFICERS AND EMPLOYEES

#### SUBCHAPTER A – UNELECTED TOWN OFFICERS

#### § 31.10 TOWN CLERK; DUTIES

There is hereby established the office of Clerk of the Town Council of the Town of Broadway. The Clerk shall perform the duties prescribed by the Town's Charter or by general state law, as well as any other duties assigned by the Town Council or the Town Manager. The Clerk shall serve an indefinite term at the pleasure of the Town Council, and shall have all the powers described in the Town Charter or other law, but shall report to the Town Manager for the day-to-day conduct of his duties.

#### § 31.11 TOWN TREASURER/DIRECTOR OF FINANCE; DUTIES

There is hereby established the office of Treasurer of the Town of Broadway. The Treasurer shall perform the duties prescribed by the Town's Charter or by general state law, as well as any other duties assigned by the Town Council or the Town Manager. The Treasurer shall be the officer responsible for the administration of the Town's financial affairs, subject to the direction and control of the Town Manager and the Town Council. The Treasurer shall serve an indefinite term at the pleasure of the Town Council, and shall have all the powers described in the Town Charter or other law, but shall report to the Town Manager for the day-to-day conduct of his duties. At the direction on the Town Manager, the Treasurer may also have the title of Director of Finance, and in his capacity as Director of Finance, shall have such duties as the Town Manager may prescribe.

#### § 31.12 TOWN MANAGER; DUTIES

(a) There is hereby established the office of Town Manager of the Town of Broadway. The Town Manager shall perform the duties prescribed by the Town's Charter or by general state law, as well as any other duties assigned by the Town Council. The Town Manager shall be the administrative head of the Town government, with all the powers permitted to such administrative head by the Town Charter or Virginia law, except as may be otherwise provided herein. To the extent permitted by law, the Town Manager shall have the executive powers of the Mayor to take care that the laws be faithfully execution, subject to the supervision and authority of the Mayor. The Town Manager shall have such other powers as the Town Council may prescribe, or as may be prescribed by general law. The Town Manager shall serve an indefinite term at the pleasure of the Town Council. Except as may be otherwise provided, the Town Manager shall have the power to make and to execute all agreements, contracts and documents on behalf of the Town.

- (b) The Town Manager may delegate any power or duty that he has to any other Town officer or employee, whether or not this Code specifically refers to the "Town Manager or his designee". The Town Manager may revoke such delegation at any time. The Town Manager may supervise such other officer or employee in the exercise of any such power or duty.
- (c) If this Code or other law charges the Town Manager with the responsibility for any decision, and the Town Manager believes that the Town Council should, instead, take such decision, then the Town Manager may refrain from making any such decision and may refer such decision to the Town Council for action.
  - (d) The Town Manager shall serve as the Town's FOIA Officer.
- (e) The Town Manager may appoint such deputy or assistant Town Managers as the Council may provide by ordinance, and may make delegations to such deputies or assistants under subparagraph (b) above.

Ordinance of April 2, 2024 amended section (e) to provide that deputy or assistant town manager positions must be created by ordinance

#### § 31.13 TOWN ATTORNEY

There is hereby established the office of Town Attorney of the Town of Broadway. The Town Attorney shall have responsibility for the management, charge, and control of the Town's legal affairs, subject to the direction and supervision of the Town Council and the Town Manager. The Town Attorney or his assistant or deputy shall attend all meetings of the Town Council unless excused from doing so by the Mayor or Town Manager. The Town Attorney shall receive for his services a compensation agreed to among the Attorney and the Town Council, which compensation may be on an hourly basis or on another basis, or a combination thereof. The Town Attorney may use other attorneys in his firm to conduct the Town's legal work if he judges that doing so will result in greater efficiency, provided that the Town Manager does not object to such use. The Town Council may also from time to time designate one or more Assistant Town Attorneys, or may retain special counsel for particular legal services such as a bond issue. The Town Attorney and any Assistant Town Attorneys are hereby authorized, at the specific request of the Town Manager, and with the approval of the Commonwealth's Attorney if required under state law, to prosecute criminal cases charging violation of Town ordinances, or misdemeanors alleged to have occurred within the limits of the Town. The Town Attorney, any Assistant Town Attorneys, and any other attorneys retained under this section are the attorneys of the municipal corporation constituting the Town, and are not the attorneys of the Town Manager in his individual capacity, or of any other officer or employee of the Town in their individual capacities, absent a separate written engagement between such individual and such attorney. Such an engagement is only permissible in accordance with the ethics rules of the Virginia state bar and further provided that the Town waives any conflict that might arise from such an engagement.

#### § 31.14 CHIEF OF POLICE

There is hereby established the office of Chief of Police of the Town of Broadway. The Chief of Police shall be the chief law enforcement officer of the Town, shall have general supervision over the police force of the Town, subject to the direction of the Town Manager and the Mayor, and shall have such other duties as provided in the Town's charter, general law, or the direction of the Town Manager. The Chief of Police shall not be required to provide a bond.

#### § 31.15 BONDS

The following officers shall be required to post bond upon the assumption of their positions, in an amount to be determined by the Town's insurance brokers: Town Manager, Town Treasurer, Town Clerk. In addition, the Town Manager may require any other officer or employee to post a bond if he determines such employee will have access to the Town's funds.

#### § 31.16 SALARIES FOR OFFICERS

The salaries or other compensation of the officers appointed under this Subchapter A shall be as stated in the budget adopted by the Town each year, as the Town Council provides by separate ordinance, or as the Town has otherwise agreed.

# § 31.17 MAYOR'S AUTHORITY REGARDING ADMINISTRATIVE LEAVE FOR CERTAIN PERSONNEL

The Mayor is authorized, in his discretion, to place any Town employee described in Sections 31.10, 31.11, 31.12, or 31.14 on temporary paid administrative leave. The administrative leave shall be effective immediately upon the Mayor's determination to invoke it, and shall last through the earliest of (a) such date as the Mayor shall determine; (b) the date of the next regular monthly meeting of the Town Council; or (c) the date that the Town Council votes to discontinue such temporary paid administrative leave at a meeting other than its regular monthly meeting. Subject to other provisions of law, the Town Council may review such temporary paid administrative leave at any meeting, and may vote to continue, discontinue, or modify such temporary paid administrative leave at any meeting. If such temporary paid administrative leave is still in effect at the next regular monthly meeting of the Town Council after it begins, the Clerk shall place a discussion regarding such temporary paid administrative leave on the agenda for the meeting. This Section shall not affect the authority of the Town Council, on its own initiative, to place any such employee on administrative leave of any duration, paid or unpaid, for any reason that applicable law permits. The provisions of this Section apply notwithstanding any contrary provisions in the Town's personnel manual. The procedures contained the Town's personnel manual applicable to suspensions do not apply to an action by the Mayor or Town Council under this Section. This Section does not affect the authority delegated to the Town Manager to issue suspensions to Town employees under the Town's personnel manual.

Ordinance of March 18, 2024 added this section.

#### § 31.18 DEPUTY TOWN MANAGER

There is hereby established the office of Deputy Town Manager. The Deputy Town Manager shall undertake such responsibilities, and exercise such authority, as the Town Manager may delegate to the Deputy Town Manager, or as the Council may provide. In the absence or disability of the Town Manager, or during a vacancy in the office of Town Manager, the Deputy Town Manager may exercise all the authority of the Town Manager as provided in this Code or under general law.

Ordinance of April 2, 2024 added this section.

#### SUBCHAPTER B – OTHER TOWN EMPLOYEES

#### § 31.20 OTHER EMPLOYEES

The Town Council has the responsibility for hiring and appointing the officers or employees described in §§ 31.10 through 31.14, may terminate such officers at its pleasure, and reserves to itself the power to terminate such officers or employees, who shall serve at the pleasure of the Town Council. The Town Council delegates the responsibility for hiring all other employees of the Town to the Town Manager, subject to the Town Council's appropriations for their salaries. The Town Manager may also terminate the employment of any such employee for any reason that does not violate law when, in his judgment, such termination promotes the best interests of the Town.

#### § 31.21 BACKGROUND CHECKS

The Town Manager shall obtain background checks on each employee of the Town before making an offer of employment to such person. He shall evaluate the results of such background check in determining whether to make an offer to such potential employee, or, if responsibility for such appointment rests in the Town Council, he shall make such information available to the Town Council.

#### § 32.22 BONUSES FOR EMPLOYEES

The town manager is authorized to establish bonus programs to address a variety of needs, including recruitment, retention, holiday bonuses, and performance, for any employee of the Town (except the Town Manager). Bonuses may be provided to employees in accordance with adopted personnel rules and regulations. Holiday bonuses may be provided from year to year in accordance with a schedule that the Town Manager may create in consultation with the Town Council or its Personnel Committee. A "bonus" shall be defined as a lump-sum payment to an employee that is not part of the base salary.

#### **CHAPTER 32: DEPARTMENTS, BOARDS, AND COMMISSIONS**

#### SUBCHAPTER A – POLICE DEPARTMENT

#### § 32.10 ESTABLISHMENT OF POLICE DEPARTMENT

There is hereby established a Town of Broadway Police Department. The Police Department shall be the official law enforcement agency of the Town, and shall have the responsibility of keeping good peace and order within the Town consistently with state law, Town ordinances, and the directions of the Town Manager, and shall have all authority provided to police departments of towns under Virginia law. The Police Department shall operate under the supervision of the Chief of Police, subject to the ultimate authority of the Town Manager and the Mayor. The Chief of Police shall provide for the smooth organization and operation of the Police Department.

#### § 32.11 EMPLOYMENT OF OFF DUTY POLICE OFFICERS

A Town police officer not on duty may accept employment or independent contractor work that occasionally requires him to use his police powers in the course of such employment. The Chief of Police is authorized to promulgate, in writing, reasonable rules and regulations applicable to such off duty employment or contract work.

# § 32.12 REIMBURSEMENT OF TOWN FOR EXPENSES INCURRED IN RESPONDING TO TERRORISM HOAX INCIDENT

Any person who is convicted of a violation of subsections B or C of Va. Code § 18.2-46.6, when such violation the a proximate cause of any incident resulting in an appropriate emergency response, shall be liable at the time of sentencing or in a separate civil action to the Town or to any volunteer rescue squad within the Town, or both, which may provide such emergency response for the reasonable expense thereof, in an amount not to exceed \$1,000 in the aggregate for a particular incident occurring in the Town. If the Police Chief determines that it is in the Town's best interests to attempt to collect such amounts, the Town Treasurer shall bill a flat fee of \$250 or a minute-by-minute accounting of the actual costs incurred to the Town. As used in this section, "appropriate emergency response" includes all costs of providing law-enforcement, fire-fighting, rescue, and emergency medical services.

# § 32.13 REIMBURSEMENT OF TOWN FOR EXPENSES INCURRED FOR METHAMPHETAMINE LAB CLEANUP COSTS

Any person who is convicted of an offense for manufacture of methamphetamine pursuant to Virginia Code § 18.2-248 or 18.2-248.03 shall be liable at the time of sentencing or in a separate civil action to the Town or to any other law-enforcement entity for the expense in cleaning up

any methamphetamine lab related to the conviction. The amount charged shall not exceed the actual expenses associated with cleanup, removal, or repair of the affected property or the replacement cost of personal protective equipment used. If the Police Chief determines that it is in the Town's best interests to attempt to collect such amounts, the Town Treasurer shall bill such person for all such amounts.

# § 32.14 REIMBURSEMENT OF EXPENSES INCURRED IN RESPONDING TO DUI AND OTHER TRAFFIC INCIDENTS

Any person convicted of violating any of the following provisions shall, at the time of sentencing or in a separate civil action, be liable to the Town or to any responding volunteer fire or rescue squad, or both, for restitution of reasonable expenses incurred by the locality for responding law enforcement, firefighting, rescue and emergency services, including those incurred by the volunteer fire or rescue squad, or by any combination of the foregoing, when providing an appropriate emergency response to any accident or incident related to such violation. A person convicted of violating any of the following provisions shall, at the time of sentencing or in a separate civil action, be liable to the Town or to any responding volunteer fire or rescue squad, or both, for restitution of reasonable expenses incurred by the Town when issuing any related arrest warrant or summons, including the expenses incurred by the by any volunteer fire or rescue squad, or by any combination of the foregoing:

- A. The provisions of Virginia Code §§ 18.2-36.1, 18.2-51.4, 18.2-266, 18.2-266.1, 29.1-738, 29.1-738.02, or 46.2-341.24, when such operation of a motor vehicle, engine, train or watercraft while so impaired is the proximate cause of the accident or incident;
- B. The provisions of Article 7 (§ 46.2-852 et seq.) of Chapter 8 of Title 46.2 of the Code of Virginia relating to reckless driving, when such reckless driving is the proximate cause of the accident or incident;
- C The provisions of Article 1 (§ 46.2-300 et seq.) of Chapter 3 of Title 46.2 of the Code of Virginia relating to driving without a license or driving with a suspended or revoked license; and
- D. The provisions of Va. Code § 46.2-894 relating to improperly leaving the scene of an accident.

Personal liability under this section for reasonable expenses of an appropriate emergency response pursuant shall not exceed \$1,000 in the aggregate for a particular accident, arrest, or incident occurring in such locality. In determining the "reasonable expenses," a locality may bill a flat fee of \$350 or a minute-by-minute accounting of the actual costs incurred. If the Police Chief determines that it is in the Town's best interests to attempt to collect such amounts, the Town Treasurer shall bill such amounts. As used in this section, "appropriate emergency

response" includes all costs of providing law-enforcement, firefighting, rescue, and emergency medical services.

#### § 32.15 TRESPASS VIOLATIONS

The owner, lessee, custodian, or persons lawfully in charge of real property (as such term is defined in state law) may designate the Town's Police Department as a "person lawfully in charge of the property" for the purpose of forbidding another to go or remain upon such lands, buildings, or premises as specified in the designation. Any such designation shall be in writing and shall be on file with the Police Department. The Police Chief may establish policies to govern such designations and procedures for the submission thereof. The Police Chief may reject any such designation for any reason.

# § 32.16 ASSESSMENT FOR MAINTENANCE OF ELECTRONIC SUMMONS SYSTEM

There is hereby imposed and assessed by the Town, in accordance with Section 17.1-279.1 of the Code of Virginia, 1950, as amended, or any successor statute, an additional sum of five dollars (\$5.00) as part of the costs in each criminal and traffic case prosecuted on a town warrant or summons in either the circuit court, general district court, or juvenile and domestic relations district court. The assessment shall be collected by the Clerk of the court in which the warrant or summons is filed and remitted to the Town treasurer. The Town shall employ such funds to defray the hardware, software and other equipment costs associated with implementation and maintenance of the Town's electronic summons system.

#### SUBCHAPTER B – PLANNING COMMISSION

#### § 32.20 ESTABLISHMENT OF PLANNING COMMISSION

There is hereby established a local planning commission for the Town of Broadway, under the name and title "Planning Commission of the Town of Broadway".

#### § 32.21 MEMBERSHIP; TERM

(a) The Planning Commission shall consist of five (5) members, all of whom shall be residents of the Town, and one-half of whom shall be owners of real property. One Member of the Planning Commission shall be a member of the Town Council (the "Council Representative").

(b) Each member of the Planning Commission shall serve a term of four (4) years, except that the term of the Council Representative shall expire at the expiration of his term on the Council or upon the selection of a successor to the Council Representative at the first meeting of the Town Council in any calendar year, whichever occurs first.

#### § 32.22 POWERS OF PLANNING COMMISSION

The Planning Commission shall have all the powers delegated or permitted to be delegated to Planning Commissions under Virginia law.

#### § 32.23 SALARIES OF PLANNING COMMISSION

Each member of the Planning Commission shall receive a salary of \$30 per month, except that the Council Representative shall receive a salary of \$50 per month in addition to any salary received as a member of the Town Council.

#### SUBCHAPTER C - INDUSTRIAL DEVELOPMENT AUTORITY

#### § 32.30 ESTABLISHMENT OF AUTHORITY

There is hereby established a political subdivision of the Commonwealth of Virginia with such public and corporate powers as are set forth in the Industrial Development and Revenue Bond Act (Chapter 49, Title 15.2 of the Code of Virginia 1950, as amended), including such powers as may hereafter be granted from time to time by the General Assembly of Virginia.

#### § 32.31 NAME OF AUTHORITY

The name of the political subdivision of the Commonwealth of Virginia shall be the Industrial Development Authority of the Town of Broadway, Virginia (the "<u>Authority</u>").

#### § 32.32 LEGISLATIVE INTENT REGARDING CODIFICATION

The provisions of §§ 32.30 and 32.31 are declarative of existing law, and are intended to codify, reenact as positive law, and reaffirm the Town Council's Ordinance of February 6, 2001, which created the Authority. The Authority continues in existence as it did before the enactment of this Subchapter, and the existing officers and directors of the Authority continue in office. This codification shall not affect any existing obligation of the Authority.

#### SUBCHAPTER D – FIRE AND RESCUE

#### § 32.41 DESIGNATION OF ORGANIZATIONS; DEFINITION

The Broadway Volunteer Fire Department, Incorporated, is designated as the Fire Department of the Town. The Broadway Emergency Squad, Inc., is designated as the Emergency Medical Service provider of the Town. Such two entities shall be deemed "first responders" within the Town and are recognized as part of the Town's safety program. Such entities are referred to in this Chapter as the "Units".

#### § 32.42 LIMITATIONS

- (a) The Town reserves such other powers as the Code of Virginia may grant it with respect to the Units.
- (b) Nothing in this chapter shall create a command structure allowing the Town to control the Units. To the extent that any governmental control is appropriate, the Town recognizes Rockingham County's unique responsibility to maintain a coordinated fire and rescue system throughout Rockingham County.
  - (c) This chapter shall have no effect outside boundaries of the Town.
- (d) Nothing in this chapter shall be construed to create any duty for public support of the Units.

#### § 32.43 REQUIREMENT TO PROVIDE FIREFIGHTING SERVICE

By accepting the designation in this Chapter, the Broadway Volunteer Fire Department, Incorporated agrees to provide firefighting service to areas within the political boundaries of the Town.

#### **CHAPTER 33: ELECTIONS**

### § 33.01 POLLING PLACE

The Town shall consist of a single precinct, and the polling place for such precinct shall be the Broadway Fire Department.

#### **CHAPTER 34: FINANCE**

#### § 34.01 ESTABLISHMENT OF FUNDS

The Town Treasurer shall be charged with maintaining the following funds in order to provide for the sound and efficient financial administration of the Town: a General Fund, a Water Fund, and a Sewer Fund. This section shall not be construed to require separate depository or investment accounts for the assets of each fund.

#### § 34.02 AUTHORITY TO SIGN CHECKS

The Mayor, Treasurer, Deputy Town Manager, and Town Manager are authorized signatories on the Town's accounts. Every check shall be only for a duly authorized and appropriated disbursement of the Town's funds. Each check must be signed by two authorized signatories; provided, however, in exigent circumstances, the Mayor by written executive order may suspend the requirement for two signatures so that a single Town official named above may sign checks. Any such executive order shall detail the exigent circumstances that necessitate suspending the two-signature requirement, and shall be effective for no more than one calendar month (but may be renewed by additional executive order). The Clerk shall provide the Council with a copy of any such executive order.

**Ordinance of April 2, 2024** added the Deputy Town Manager as an authorized signatory, and expanded the authority of the Mayor to authorize a single signatory in exigent circumstances by removing the requirement that it be related to a State-recognized state of emergency, but Council notification of any suspension of the two-signature requirement.

#### § 34.03 PENALTY FOR RETURNED CHECKS

Any persons who utters, publishes or passes any check, draft, or order for payment of taxes or any other sums due to the Town, which is subsequently returned for insufficient funds or because there is no account or the account has been closed, or because such check, draft, or order was returned because of a stop-payment order placed in bad faith on the check, draft, or order by the drawer, shall, in addition to the amount otherwise owed to the Town, shall be assessed an additional fee of Fifty Dollars (\$50.00).

#### **CHAPTER 35: PUBLIC PROCUREMENT**

#### SUBCHAPTER A: GENERAL PROVISIONS

#### § 35.10 **PURPOSE**

The purpose of this chapter is to enunciate the Town's policies pertaining to governmental procurement from nongovernmental sources, to encourage competition among vendors and contractors, to provide for the fair an equitable treatment of all persons involved in public purchasing by this Town, to maximize the purchasing value of public funds in procurement so that high quality goods and services may be obtained at the lowest possible price, and to increase public confidence in procurement practices by providing safeguards for maintaining a procurement system of quality and integrity.

#### § 35.11 APPLICATION

This chapter shall govern the procurement of goods and services by the Town. To the maximum degree possible under state law, this Title and the regulations issued hereunder shall supersede the provisions of the Virginia Public Procurement Act. Those portions of the Virginia Public Procurement Act that apply to all localities, notwithstanding the adoption of an alternate procurement ordinance, shall continue to apply to the Town and shall be read in conjunction with this chapter, which supplements such provisions of state law.

#### § 35.12 **DEFINITIONS**

For the purposes of this chapter, the following words and phrases shall have the meanings respectively ascribed to them by this section:

- (a) Act. Virginia Public Procurement Act, as amended from time to time, but presently codified at Title 2.2, chapter 43 of the Code of Virginia.
- (b) *Business*. Any corporation, general or limited partnership, limited liability company, limited liability partnership, individual, sole proprietorship, joint stock company, joint venture or any other non-governmental legal entity.
- (c) Construction. Building, improving or demolishing any structure, building, or highway, and any draining, dredging, excavation, grading or similar work upon real property, but not including maintenance or repairs.
- (d) *Contract*. All types of Town agreements, oral or written, regardless of what they may be called, for the procurement of goods, services, insurance or construction.
- (e) *Contractor*. Any person having a contract with the Town or a using agency thereof.

- (f) *Emergency*. An emergency shall exist when, in the opinion of a using agency, a breakdown in an essential service occurs or under any other circumstances when supplies are needed for immediate use in work which may vitally affect the safety, health or welfare of the public.
- (g) *Employee*. An individual drawing a salary, wages or other compensation from the Town whether elected or not, or any noncompensated individual performing personal services for the Town or any entity established by the Town.
- (h) *Insurance*. A contract whereby, for a stipulated consideration, one party undertakes to compensate the other for loss or provide a defense to any claim.
- (i) *Invitation for bids*. All documents, whether attached or incorporated by reference, utilized for soliciting sealed bids in a competitive sealed bidding process, formal or informal.
- (j) *Nonresponsible*. A bidder or responder to a request for proposal is nonresponsible when the Town Manager determines that it has overestimated its capacities and lacks the ability to supply the goods or to perform the services that the Town seeks to procure in a particular procurement proceeding.
- (k) *Person.* Any business, association, individual, union, committee, club, other organization or group of individuals.
- (l) Request for proposals. All documents, including those attached or incorporated by reference, used to solicit proposals in a competitive negotiation process, formal or informal.
- (m) *Using agency*. Any department, division, office, council, or other unit in the Town government requiring goods, services, insurance or construction as provided for in this chapter.

Except to the extent the definition is altered herein, any other terms used in this Chapter and not otherwise defined, but which are defined in the definitions section of the Act, shall have the same meaning as in the definitions section of the Act.

#### SUBCHAPTER B: PURCHASING SYSTEM

#### § 35.20 ESTABLISHMENT OF PURCHASING SYSTEM

There is hereby created a purchasing system to operate under the direction and supervision of the Town Manager.

#### § 35.21 AUTHORITY AND DUTIES OF TOWN MANAGER

The Town Manager shall serve as the principal public purchasing official for the Town and shall be responsible for the procurement of goods, services, insurance and construction in accordance with this Title, as well as the management and disposal of goods. In his capacity as such, the Town Manger shall have the following duties and powers:

- (a) *Duties*. In accordance with this chapter, the Town Manager shall:
  - (1) Purchase or supervise the purchasing of all goods, services, insurance and construction needed by the Town;
  - (2) Sell, trade or otherwise dispose of surplus goods belonging to the Town;
  - (3) Establish and maintain programs for specifications development, contract administration, inspection and acceptance, in cooperation with the public agencies using the goods, services and construction;
  - (4) Maintain a current file of source of goods, services, insurance and construction to be known as a "bidder list" in which vendors can request to be included; and
  - (5) Perform such other functions and duties as may be necessary for the effective operation of the purchasing system.
- (b) Receiving and tabulating bids for construction: The Town Manager shall advertise, receive, open, and tabulate any and all bids for construction in accordance with the requirements and conditions set forth in the invitation to bid.
- (c) Award of contracts:
  - (1) The Town Manager shall award all contracts on behalf of the Town Council except for:
    - (a) Contracts for construction which are expected to exceed \$200,000.00; and
    - (b) Contracts where the Town Manager, in his sole discretion, has determined that an award by the Town Council is appropriate.
  - (2) The Town Council shall award all contracts not awarded by the Town Manager.
- (d) Except as otherwise provided herein, the Town Manager have the power to take any action, on behalf of the Town, that a "public body" is permitted to take under the Virginia Public Procurement Act.

#### § 35.22 DELEGATION OF AUTHORITY BY TOWN MANAGER

The Town Manager may delegate authority to purchase certain goods, services or construction items to other Town officials, provided adequate accountability to the Manager is maintained.

#### § 35.23 PURCHASES FOR FIRE DEPARTMENT AND RESCUE SQUADS

The Town Manager shall have authority to sell supplies, materials and equipment to volunteer fire departments and rescue squads within the Town at the same cost as the cost of such supplies, materials and equipment to the Town.

#### § 35.24 UNAUTHORIZED PURCHASES

Except as otherwise provided in this chapter, no elected or appointed official, or any employee, shall purchase or contract for any goods, services, insurance or construction within the purview of this Title other than by and through the Town Manager, and any purchase order or contract made contrary to the provisions hereof is not approved and the Town shall not be bound thereby. Any person responsible for such purchase shall be held personally liable for such purchase, and if already paid for out of Town funds, the amount may be recovered in the name of the Town in an appropriate action instituted therefor.

#### § 35.25 REGULATIONS

The Town Manager may promulgate reasonable regulations, not inconsistent with this chapter or with state law, for the implementation of the provisions of this chapter and the establishment of procurement procedures hereunder, including procedures involving competitive sealed bidding and competitive negotiation, substantially as those terms are used in the Act, and procedures for small purchases where permitted by the Act.

#### § 35.26 AUTHORITY FOR PROPERTY DISPOSAL

The Town Manager shall sell, transfer, trade or otherwise dispose of all materials, supplies, equipment or other personal property of the Town which has become obsolete and/or unusable by the using agency for which it was purchased and is unusable for public use.

#### § 35.27 SALE OF SURPLUS PROPERTY

All sales of property pursuant to this chapter shall be sold on the basis of competitive bids, public auction or firm price offered to all persons wishing to participate in the sale. The Town Manager shall use whichever method he believes will raise the highest revenue for the Town.

#### SUBCHAPTER C: SOURCE SELECTION, DEBARMENT, AND REMEDIES

# § 35.30 EXCEPTIONS TO REQUIREMENT FOR COMPETITIVE PROCUREMENT

The following exceptions apply to the requirements of this chapter:

- (a) The Town may enter into contracts without competition for the purchase of goods or services which are performed or produced by *bona fide* nonprofit organizations.
- (b) The Town may enter into contracts for (i) legal services, (ii) expert witnesses, and (iii) other services associated with litigation or regulatory proceedings, in each instance without competitive sealed bidding or competitive negotiation.
- (c) The Town may enter into contracts for special police work without competitive sealed bidding or competitive negotiation.
- (d) The Town may award single or term contracts not expected to exceed the Small Purchase Limit without competitive sealed bidding or competitive negotiation under alternate written procedures, which the Town Manager may adopt generally, or may adopt in advance of a particular procurement for such procurement. Such procedures shall require obtaining quotes from at least three offerors whenever practical. No small purchase procedure for construction projects shall waive compliance with the Statewide Building Code. The Small Purchase Limit is
  - (i) \$200,000, for goods and services other than professional services and transportation-related construction;
  - (ii) \$80,000, for professional services; and
  - (iii) \$25,000, for transportation-related construction.
- (e) The Town may enter into contracts for purchases of legal publications without competitive sealed bidding or competitive negotiation.
- (f) If an emergency occurs during office hours, the using agency shall immediately notify the Town Manager, who shall either purchase directly or authorize the purchase of the needed goods. If any emergency occurs during a time when the Town Manager is unavailable, the using agency may purchase any goods or services needed to meet such existing emergency; provided, that the head of the using agency shall send as soon as practicable to the Town Manager a requisition and a copy of the delivery receipt together with written explanation of the circumstances of the emergency.

(g) This Title shall not apply to financing transactions entered into by the Industrial Development of the Authority of the Town of Broadway, Virginia, entered into at the request of a "conduit borrower" as that term is defined in Title 26 of the Code of Federal Regulations.

#### § 35.31 **DEBARMENT**

The Town Manager may, in the public interest, debar a prospective contractor for any of the causes in paragraph (a) of this section, using the procedures in paragraph (c) of this section. The existence of a cause for debarment under such paragraph (a), however, does not necessarily require that the contractor be debarred; the seriousness of the contractor's acts or omissions and any mitigating factors should be considered in making any debarment decision.

- (a) Causes: The Town Manager may debar a prospective contractor for any of the following causes:
  - (1) Conviction of or civil judgment for:
    - (a) commission of fraud or a criminal offense in connection with (i) obtaining, (ii) attempting to obtain, or (iii) performing a public contract or subcontract;
    - (b) violation of federal of state antitrust statutes relating to the submission of offers:
    - (c) commission of embezzlement, theft, forgery, bribery, falsification or destruction of records, making false statements or receiving stolen property; or
    - (d) commission of any other offense indicating a lack of business integrity or business honesty that seriously and directly affects the present responsibility of a government contractor or subcontractor.
  - (2) Violation of the terms of a government contract or subcontract so serious as to justify debarment, such as (a) a willful failure to perform in accordance with the terms of one or more contract, or (b) a history of failure to perform, or of unsatisfactory performance of, one or more contracts, or (b) a history of failure to perform, or of unsatisfactory performance of, one or more contracts.
  - (3) Any other cause of so serious or compelling a nature that it affects the present responsibility of a government contractor or subcontractor.

- (b) Responsibility of Manager. The Town Manager shall promptly investigate any allegations of improper contractor conduct prior to initiating debarment procedures.
- (c) *Procedures:* The following procedures governing the debarment decision-making process are designed to be as informal as practicable, consistent with principles of fundamental fairness.
  - (1) Debarment shall be initiated by advising the prospective contractor by certified mail, return receipt requested, that debarment is being considered. Such notice shall include the reasons for the proposed debarment in terms sufficient to put the contractor on notice of the conduct or transaction upon which it is based.
  - (2) Debarment shall be for a period commensurate with the seriousness of the cause, as determined by the Town Manager in his sole discretion.
  - (3) The prospective contractor may submit to the Town Manager, within thirty days after receipt of notice, in person, in writing or through a representative, information and argument to the proposed debarment, including any additional specific information that raises a genuine dispute over the material fact. If the proposed debarment is based upon a cause other than those specified in paragraph (1) of subsection (a), an informal hearing allowing the examination and cross-examination of witnesses shall be provided if so requested by the prospective contractor. In such cases, the Town Manager shall conduct the hearing and shall render his decision within fifteen days thereafter, or within fifteen days after receipt of written information and argument if no hearing is requested or required to be held.
  - (4) The decision of the Town Manager services shall be final unless the prospective contractor appeals by invoking administrative procedures provided for by this chapter, or in the alternative by instituting legal action as provided in the Act.

#### § 35.32 AWARDS

Unless cancelled or rejected, a responsive bid from the lowest responsible bidder in a competitive sealed bidding process, which does not exceed available funds, shall be accepted as submitted.

Appeal of awards and decisions to award: Any bidder or offeror who desires to appeal an award or a decision to award shall first file a written protest with the Town Manager or his designee, specifying the basis for the protest and the relief sought, no later than ten days after the award or the announcement of the decision to award, whichever occurs first. The Town Manager or his designee shall render a written decision within ten days of receipt of the written protest stating

the reasons for the decision. This decision shall be final and binding unless any bidder or offeror appeals according to the administrative appeals procedure established by this Title.

- (a) Determinations of nonresponsibility: Nonresponsible bidders shall be notified of the determination of nonresponsibility in writing by certified mail, return receipt requested. A determination of nonresponsibility is subject to administrative appeal according to the administrative appeals procedure established by this chapter.
- (b) Waiver of irregularities: The Town Manager may accept late bids, waive informalities in bids, or reject all bids when such action is necessary to protect the interests of the Town.

## § 35.33 NEGOTIATION WITH LOWEST RESPONSIBLE BIDDER

If the bid from the lowest responsible bidder exceeds available funds, the Town may negotiate with the apparent low bidder to obtain a contract price within available funds. Such negotiations shall be conducted in accordance with the following procedures:

- (a) The Town Manager shall advise the lowest responsible bidder, in writing, that the proposed purchase exceeds available funds. He shall further suggest a reduction in scope for the proposed purchase, and invite the lowest responsible bidder to amend its bid proposal based upon the proposed reduction in scope.
- (b) Repetitive informal discussions with the lowest responsible bidder for purposes of obtaining a contract within available funds shall be permissible.
- (c) The lowest responsible bidder shall submit an addendum to its bid, which addendum shall include: the change in scope for the proposed purchase, the reduction in price, and the new contract value.
- (d) If the proposed addendum is acceptable to the Town, the Town may award a contract within funds available to the lowest responsible bidder based upon the amended bid proposal.
- (e) If the Town and the lowest responsible bidder cannot negotiate a contract within available funds, all bids shall be rejected.

## § 35.34 ADMINISTRATIVE APPEALS

The following procedures shall govern appeals under this chapter:

- (a) Any bidder or offeror may appeal:
  - an award or a decision to award within ten days of the date of the adjudication of the initial protest pursuant to § 35.32 of this chapter;

- (2) a decision to refuse to allow withdrawal of bids within thirty days of the refusal;
- (3) a decision of disqualification within thirty days of the receipt of the determination of disqualification; or
- (4) a decision of debarment within thirty days of the date of the initial appeal pursuant to § 35.31(c)(3) of this chapter; or
- (5) a determination of non responsibility within ten days of the receipt of the determination of non responsibility notice pursuant to § 35.32(a) of this chapter.
- (b) Any appeal permitted by this section shall be in accordance with the following administrative procedures:
  - (1) Any aggrieved person shall file a letter of appeal with the Town Manager. The letter of appeal shall specify the basis for the appeal, the relief sought, and whether the bidder or offeror wishes to have a hearing with respect to the appeal. If the appeal concerns an award or a decision to award, the letter of appeal shall also contain a statement that it has been mailed or otherwise delivered to all other bidders or offerors. Upon request, the Town Manager shall provide the names and addresses of all other bidders or offerors to the appealing bidder or offeror. No appeal will be allowed if the letter of appeal is untimely filed or does not contain the information required by this subparagraph. No person who has unsuccessfully protested or appealed a decision to award may protest or appeal the subsequent award of the same contract.
  - (2) Any bidder or offeror notified of the appeal may:
    - (a) file a written statement setting forth its position with the Town Manager, whether or not the letter of appeal requests a hearing. Such written statement also shall specify whether the bidder or offeror wishes to have a hearing. Such written statement shall be filed within eight days of the date of the letter of appeal.
    - (b) participate in any hearing requested by the appealing party or by any other bidder or offeror.
  - (3) If no hearing is requested, the Town Manager shall render a written decision within ten days of receipt of the letter of appeal or any written statement filed as provided in subparagraph (2) above, whichever is later. The written decision shall be sent to the appealing party and any other

- bidder or offeror who submitted a written statement with respect to the appeal.
- (4) If a hearing is requested, it shall be held within 30 days of receipt of the letter of appeal, and a final decision shall be rendered within 30 days after the hearing. During the hearing, all parties to the process shall have the opportunity to present pertinent information and to cross-examine adverse witnesses. The hearing shall be an informal administrative proceeding rather than a judicial-type trial, and shall be conducted by a disinterested person designated by the Town Manager.
- (5) The findings of fact shall be final and conclusive and shall not be set aside unless they are fraudulent or arbitrary or capricious, or so grossly erroneous as to imply bad faith. No determination on an issue of law shall be final if appropriate legal action is instituted in a timely manner.
- (6) Any party to the administrative procedure shall be entitled to institute judicial review if such action is brought within 30 days of receipt of the written decision.
- (c) Notwithstanding anything to the contrary contained in this section, no administrative protest or appeal shall lie to:
  - (1) a decision to award or an award made by the Town Council;
  - (2) a recommendation made by any person, selection committee, or other entity, to the Town Council concerning the award of a contract.
  - (3) any decision by the Town Manager or the Town Council to reject all bids.
- (d) No person shall be entitled to initiate a civil action for review of any determination in this Chapter without having previously exhausted its administrative remedies under this Chapter. In enacting this section § 35.34, the Town Council specifically intends to establish an alternative procedure for procurement appeals rather than using the procedures contained in the Act.

## § 35.35 AUTHORITY FOR SCORING SYSTEM

Notwithstanding any other provision of this chapter, and, to the full extent permissible under state law, notwithstanding any definition of "competitive sealed bidding" or "competitive negotiation" under state law, in any competitive sealed bidding process or competitive negotiation, the Town Manager is authorized to use a "scoring system" under which he will evaluate bids or proposals based on cost and other permissible factors stated in the request for proposal or invitation to bid. A reference in this ordinance or regulations issued thereunder to the "lowest responsible bidder" shall mean, in the context of a procurement process that involves

a scoring system, the responsible bidder with the highest score. The Town Manager shall provide an explanation of the scoring system to be used in the invitation to bid or the request for proposal.

# § 35.36 RELATIONSHIP TO OTHER PROCUREMENT LAW

The provisions of this Chapter do not inhibit or repeal any uncodified procedures, policies, or guidelines that the Town Council may have adopted, or may adopt, to implement (a) the Public-Private Education Facilities and Infrastructure Act of 2002 (the "PPEA", presently codified at Virginia Code § 56-575.1 *et seq.*) or (b) the provisions of Virginia law that permit design-build contracts or construction management at risk contacts. To the extent that such procedures, policies, or guidelines reference the procedures or principles of the Virginia Public Procurement Act (including, without limitation, competitive sealed bidding or competitive negotiation), the applicable provisions of this of this Chapter supplement, <u>mutatis mutandis</u>, the PPEA and such other law.

# **TITLE V: TAXATION**

# Chapter

- 50. GENERAL PROVISIONS
- 51. REAL ESTATE
- 52. PERSONAL PROPERTY
- 53. MACHINERY AND TOOLS
- 54. BUSINESSES AND PROFESSIONS
- 55. MEALS TAX
- 56. RESERVED
- 57. OTHER TAXES
- 58. TRANSIENT OCCUPANCY TAX

## **CHAPTER 50: GENERAL PROVISIONS**

## § 50.01 APPLICIBILITY

The provisions of this chapter apply to all Town taxes assessed under this Title V unless a certain tax or taxes are specifically excluded, and, to the fullest extent permitted by state law, shall apply to any other amounts due to the Town and treated like taxes under applicable state or local law.

# § 50.02 FAILURE-TO-PAY PENALTY; FAILURE TO FILE RETURN; COLLECTION COSTS

- (a) Unless otherwise stated, the penalty for failure to pay any tax or installment under this title by the date due shall be 5% of the tax past due, and shall be assessed on the day after the first installment is due.
- (b) The penalty for failure to pay the tax on meals or transient occupancies by the due date thereof shall be 10% of the tax for the first month or portion thereof past due, and five percent for each month or portion thereof thereafter, up to a maximum of 25% of the tax collected but not remitted.
- (c) The penalty for failure to pay the tax of cigarettes by the due date thereof shall be 10% for each month or portion thereof the tax is not paid, together with interest in the amount of 0.75% each month the tax is not paid.
- (d) In any case where in a penalty is otherwise due, the minimum penalty shall be \$10.00. Notwithstanding any other provision herein, in no case shall the penalty exceed the amount of the tax assessable. Any penalty, when so assessed, shall become a part of the tax due and may be collected in the same manner.
- (e) In addition to any other amount owed in connection with a delinquent tax, the taxpayer shall pay the actual costs of collection thereof in the maximum amount of 20% of the amount due but not paid. Such costs shall become part of the tax due and may be collected in the same manner. Such costs shall include, without limitation, the reasonable costs incurred by the Town in paying the Town Attorney to assist with such collection, at the Town Attorney's normal hourly rate. The Treasurer may waive the provisions of this paragraph in his reasonable discretion.
- (f) If any person required to file any information return in connection with a tax does not do so by the due date thereof, in addition to the other interest and penalties for failure to pay provided in this title, such person shall be liable for an additional penalty of 10% of the tax due, or \$10.00, whichever is greater. Such penalty shall be an addition to the tax due and may be collected in the same manner.

## § 50.03 INTEREST

Interest shall commence the first day of the calendar month following the day any taxes under this title are due to be filed or paid, at the rate of 10% per annum. Where required by state law, the Town shall pay interest to a taxpayer who has been overcharged at the same rate.

# § 50.04 EXCEPTIONS TO PENALTY AND INTEREST

The penalties described in § 50.02 and the interest due and payable under § 50.03 shall not apply in the following circumstances:

- (a) If the failure to file or pay was not the fault of the taxpayer, or was the fault in whole or in part of the Treasurer or the Rockingham County Commissioner of the Revenue. The determination of fault shall be made by the Treasurer in his reasonable discretion, using any presumptions regarding physical or mental impairment provided in state law. In any such event, the taxes shall be paid within 30 days of the date otherwise due, or, in the case of a fiduciary for an individual, within 120 days after the due qualification thereof.
- (b) If the assessment of any tax (other than an excise tax or a business license tax) is made fewer than two weeks before the due date thereof, in which case the tax shall be paid within two weeks of such an assessment.

## § 50.05 EXTENSIONS OF TIME TO PAY

The time to pay a tax may be extended in the following circumstances:

- (a) The Treasurer may grant a reasonable extension of time (not to exceed 90 days) the payment of real estate, personal property, merchants' capital, and the business, license tax, whenever good cause exists. The Treasurer shall keep a record of any such extension granted.
- (b) In the case of real estate taxes on a primary residence, and in the case of personal property taxes on a qualifying vehicle (as defined under applicable law), the due date for taxes shall be extended automatically during the time in which the taxpayer is a member of the armed services of the United States who is deployed outside the United States, and for a period of 90 days following the end of such deployment.

# § 50.06 MANNER OF CREDITING PAYMENT

The Treasurer shall credit any payment of local levies first to penalties, interest and the principal of the longest overdue tax bill (not subject to a defense of the applicable statute of limitations) within the levy.

# § 50.07 OFFERS IN COMPROMISE

With the consent of the Town Manager, the Treasurer may compromise and settle the amount due and payable to the Town on any tax, or debt collected as if a tax, when the Treasurer determines that the collection of the entire amount due is in substantial doubt and the best interests of the Town shall be served by such a compromise.

# § 58.08 RELEASE OF CERTAIN DELINQUENT TAX LIENS

The Treasurer is authorized (but is not required) to release the lien for delinquent taxes, in order to facilitate conveyance of the property thereby encumbered, if:

- (a) The purchaser is unrelated by blood or marriage to the owner;
- (b) The purchaser has no business association with the owner;
- (c) The purchaser owes no delinquent real estate taxes for any property; and
- (d) The property, including land and improvements, is valued at less than \$50,000.

No such release shall operate as a discharge of the personal liability of the owner of the property so encumbered at the time the liens were imposed.

## § 50.09 REGULATIONS

The Treasurer is empowered to promulgate reasonable regulations to interpret the provisions of this Title 5 or to provide for the orderly administration and collection of the taxes to be paid hereunder. A copy of all such regulations shall be available in the Treasurer's office.

## § 50.10 RATE SCHEDULE

The rates of several of the various taxes and fees enacted under this Title or other provisions of the Town Code may appear on a rate schedule appended to the Town Code for ease of reference thereof. Such rate schedule shall be an integral part of the Town Code and the Town Council has enacted it into positive law along with the rest of the Town Code.

# § 50.11 PREVIOUSLY GRANTED EXEMPTIONS

Any exemption from real property tax previously granted by the Town Council before the enactment of this Code shall remain in effect after the enactment of this Code until the Town Council modifies or repeals it as may be permitted by Virginia law. No ordinance granting such an exemption is repealed by the enactment of this Title.

Editor's Note – The real and personal property of the "John Kline Homestead" was exempted from property taxation by designation by Ordinance of September 6, 2011. This section is not set out in the Town Code, which contains only ordinances of general applicability, but attention is called to this ordinance by this reference. A copy of the ordinance is available from the Town Clerk.

# § 50.12 FAILURE TO FILE REQUIRED RETURN; FILING OF FALSE RETURN; FAILURE TO COLLECT OR REMIT TRUST TAX

- (a) A person executing an application, report or return under this Title, whether individually or as an agent, certifies by his signature that the information on the application is true and correct to the best of his knowledge and belief. No oath shall be necessary on any such application, but the person's signature shall be sufficient. No such person shall intentionally subscribe any such application, report, or return which he does not believe to be true and correct as to every material matter.
- (b) No person who is required to collect, account for and pay over any transient occupancy, food and beverage, cigarette taxes authorized under this Title, shall willfully fail to collect or truthfully account for and pay over such tax, or willfully evade or attempt to evade any such tax or the payment thereof. All such taxes are held in trust for the Town from the time of their collection to the time such taxes are remitted to the Town.
- (c) No person required by this Title or regulations made under authority hereof to make a return, keep any records or supply any information, for the purpose of the computation, assessment or collection of any Town tax, shall willfully fail to make such returns, keep such records or supply such information, at the time or times required by this Title or such regulations.
  - (d) Any person who violates this section shall be guilty of
- (i) a Class 3 misdemeanor while the tax owed is less than \$1,000 and the tax is not a transient occupancy, food and beverage, or cigarette tax; and upon due conviction thereof shall be fined not more than \$500; or
- (ii) a Class 1 misdemeanor, when the tax owed is more than \$1,000, or when the tax owed is a transient occupancy, food and beverage, or cigarette tax; and upon due conviction thereof shall be incarcerated for a period not to exceed twelve (12) months or fined not more than \$2,500, or both.

Such penalties shall be in addition to any other civil penalties due to the Town under this Title or other law.

(e) This section applies to a corporate or partnership officer of an entity that is liable for the taxes described in this section. The phrase "corporate or partnership officer" has the same meaning as in state law.

## **CHAPTER 51: REAL ESTATE**

## SUBCHAPTER A: GENERAL PROVISIONS

# § 51.10 LEVY OF TAX

There is hereby imposed an annual tax on all tracts of land, lots, and improvements thereon not exempt from taxation under applicable law, in a millage rate set by the Council from time to time and appearing on the rate schedule to the Town Code. Such tax shall be for general Town purposes.

# § 51.11 **DUE DATE**

Such taxes on real estate for each year shall be due and payable during the year for which the same are assessed on or before December 5.

# § 51.12 LIABILITY FOR TAX

All taxes annually imposed on real estate shall be payable by the persons who on January 1 in each and every year are the respective owners of record thereof.

# § 51.13 ASSESSMENT OF PROPERTY FOR TAX

For the purposes of collecting real property tax under this chapter, the Town shall compute the tax based on the assessed value of real property within the Town as finally determined by the appropriate officials of Rockingham County.

## SUBCHAPTER B: EXEMPTION FOR ELDERLY AND DISABLED

## § 55.20 EXEMPTION; OWNERSHIP

The Treasurer shall exempt from the tax real property owned and occupied as the sole dwelling house of a natural person or persons

- (a) who are either at least 65 years of age as of December 31st of the year preceding the year of assessment; or
- (b) who are permanently and totally disabled as defined under applicable state law as of December 31 of the year preceding the year of assessment

upon the due application therefor. The Treasurer shall grant such applications only upon the restrictions and conditions hereinafter established.

For the purposes of this subchapter, the term "own" shall include (i) the ownership of a legal life estate in real property; (ii) the ownership of real property by a revocable inter vivos trust over which the taxpayer has a power of revocation; and (iii) the ownership of a present possessory equitable interest in real property.

The exemption provided hereunder shall extend to the lot on which is situate the dwelling house of the person eligible for the exemption, provided that the exemption shall be limited to 10 acres of property and the improvements thereon.

## § 55.21 INCOME LIMITATION

In order to qualify for such an exemption, the total combined income received from all sources during the preceding calendar by owners of the dwelling who use it as their principal residence, and others who live in the dwelling (except for persons living in the dwelling and providing bona fide caregiving services to the owner whether such relatives are compensated or not), shall not exceed the greater of \$50,000, or the income limits based upon family size for the Town's metropolitan statistical area, annually published by the Department of Housing and Urban Development for qualifying for federal housing assistance pursuant to § 235 of the National Housing Act (12 U.S.C. § 1715z). The Treasurer shall exclude from his calculation any amount up to \$10,000 of income of each relative who is not the spouse of an owner living in the dwelling.

## § 55.22 ASSET LIMITATION

In order to qualify for such an exemption, the net combined financial worth, including the present value of all equitable interests, as of December 31 of the immediately preceding calendar year, of the owners, and of the spouse of any owner, excluding the value of the dwelling and the

land upon which it is situated (not exceeding 10 acres) shall not exceed \$200,000. Such value shall also exclude furniture, household appliances and other items typically used in a home.

## § 55.23 PRORATION OF EXEMPTION

When a taxpayer who is otherwise eligible for an exemption under this chapter owns real property in a form of joint tenancy with someone other than his spouse, and the other owners would not be eligible for an exemption if they owned such property alone, then the exemption shall be reduced *pro rata* in proportion to the ownership interest of the persons not eligible.

# § 55.24 APPLICATION FOR EXEMPTION

- (a) The person claiming such exemption shall file annually with the Treasurer on forms supplied by the Town, a sworn affidavit setting forth:
  - (1) The names of the persons occupying such real estate; and
  - (2) That the total combined net worth, including equitable interests and the combined income from all sources, of the persons specified do not exceed the limits prescribed herein.
- (b) If such person is under 65 years of age such form shall have attached thereto a certification by the Social Security Administration, the Department of Veterans Affairs or the Railroad Retirement Board, or if such person is not eligible for certification by any of these agencies, a sworn affidavit by two medical doctors who are either licensed to practice medicine in the Commonwealth or are military officers on active duty who practice medicine with the United States Armed Forces, to the effect that the person is permanently and totally disabled, as defined in applicable state law; however, a certification pursuant to 42 U.S.C. 423 (d) by the Social Security Administration so long as the person remains eligible for such social security benefits shall be deemed to satisfy such definition. The affidavit of at least one of the doctors shall be based upon a physical examination of the person by such doctor. The affidavit of one of the doctors may be based upon medical information contained in the records of the Civil Service Commission which is relevant to the standards for determining permanent and total disability under applicable state law.
- (c) Such application with affidavit shall be filed no later than the first day of September of each year, except that the Treasurer may, in his reasonable discretion, accept later filings from owners otherwise meeting the provisions of this section who are unable to file by the first day of September due to personal or family illness or other ascertainable hardships, and from owners eligible to file for the first time, provided that such late acceptance will not disrupt the Town's tax administration.

- (d) In lieu of filing a new application each year, an individual with an effective exemption based on an application submitted and approved may file, for the succeeding two tax years, a sworn statement with the Treasurer that no circumstance has arisen since the date of the application that materially affects such individual's eligibility for an exemption.
- (e) The taxpayer shall notify the Treasurer if, after filing an application for exemption, circumstances arise that cause the taxpayer unexpectedly to exceed the income or net worth limitations described herein. In such an event that Treasurer shall revoke the exemption for the year of such change in circumstances.

# § 55.25 POWER OF TREASURER

The Treasurer may make further inquiry of persons seeking such exemption as may be reasonably necessary in determining the qualifications therefor. The Treasurer may require production of certified tax returns to establish the financial worth or income of any applicant. The Treasurer may require that such further inquiries shall be answered under oath.

# § 55.26 REGULATIONS

The Treasurer may promulgate reasonable rules and regulations (including forms) not inconsistent with the provisions of this chapter or state law to provide for the effective administration of this chapter.

# § 55.27 CHANGE IN CIRCUMSTANCES

An individual who does not qualify for an exemption under this subchapter based upon the previous year's income or financial worth limitations may nonetheless qualify for an exemption for the instant year if he files an affidavit with the Treasurer that clearly shows a substantial change in circumstances, not volitional, that will result in income or financial levels that are within the limitations of this subchapter.

## § 55.58 COUNTY EXEMPTIONS

In lieu of granting an exemption based upon the other procedures contained in this subchapter, the Treasurer may accept evidence that Rockingham County has granted the exemption described herein for elderly or disabled individuals, and shall exempt any property that the Treasurer is satisfied that Rockingham County has exempted.

SUBCHAPTER C: OTHER EXEMPTIONS FROM REAL PROPERTY TAX

## § 55.30 EXEMPTION FOR DISABLED VETERANS

The Treasurer shall exempt from real property tax any real property for which the Treasurer is notified that a valid Rockingham County exemption exists based on the owner's status as a disabled veteran under applicable state and Rockingham County law.

# § 55.31 EXEMPTION FOR CERTAIN WIDOWS OR WIDOWERS OF MEMBERS OF THE ARMED FORCES

The Treasurer shall exempt from real property tax any real property for which the Treasurer is notified that a valid Rockingham County exemption exists based on the owner's status as widow or widower of an individual killed in action while in the service of the armed forces of the United States under applicable state and Rockingham County law.

## § 55.32 LEGISLATIVE INTENT

The Town designates the Commissioner of the Revenue of Rockingham County as the official responsible to administer the exemptions contained in this subchapter to the fullest extent permissible by law, the intention of the Town Council being that an exemption from Town real estate taxes shall exist in every case in which a valid Rockingham County exemption described in this subchapter exists, in order to obviate the need for the Town to maintain a separate administrative system for such exemptions.

## **CHAPTER 52: PERSONAL PROPERTY TAX**

## SUBCHAPTER A: TAX GENERALLY

# § 52.10 LEVY OF TAX

There is hereby levied an annual tax on all tangible personal property within the Town that falls into the following categories:

- (a) Automobiles and trucks (regardless of weight); motorcycles, mopeds, all-terrain vehicles, and off-road motorcycles; and boats (regardless of weight); and
  - (b) manufactured homes.

Such tax shall be at a millage rate set by the Council from time to time and appearing on the rate schedule to the Town Code. Such tax shall be for general Town purposes. The millage rate need not be the same from category to category to the fullest extent permissible under state law. The Town does not tax any other tangible personal property except as described in Chapter 53 herein.

# § 52.11 TAX DAY

The date for determining the situs of personal property for the purposes of assessing tax under this Chapter shall be January 1. Situs of property shall be determined under the rules prescribed in state law.

#### § 52.12 **DUE DATE**

Such taxes on tangible personal property for each year shall be due and payable during the year for which the same are assessed on or before December 5.

## § 52.13 ASSESSMENT OF PROPERTY FOR TAX

For the purposes of collecting tangible personal property tax under this chapter, the Town shall compute the tax based on the assessed value of property sited within the Town as finally determined by the appropriate officials of Rockingham County.

## SUBCHAPTER B – APPLICATION OF PERSONAL PROPERTY TAX RELIEF

# § 52.21 PURPOSE; DEFINITIONS; RELATION TO OTHER ORDINANCES

- (a) The purpose of this subchapter is to provide for the implementation of the changes to the Personal Property Tax Relief Act of 1998 (PPTRA) effected by legislation adopted during the 2004 Special Session I and the 2005 Regular Session of the General Assembly of Virginia, and regularly readopted during budget cycles since then. Such changes require the Town to establish how it will allocate personal property tax relief funds transferred to the Town by the Commonwealth.
- (b) Terms used in this subchapter that have defined meanings set forth in PPTRA shall have the same meanings as set forth therein.
- (c) To the extent that the provisions of this subchapter conflict with any prior Ordinance or provision of the Town Code, this subchapter shall control, the overriding intent of the Town Council being to implement an ordinance that will entitle the Town to receive personal property tax relief funds from the Commonwealth.

# § 52.22 METHOD OF COMPUTING AND REFLECTING TAX RELIEF

- (a) For tax years commencing in 2006, and for each tax year thereafter, the Town adopts the provisions of Item 503.E of the Acts of Assembly 2005, Chapter 951, and subsequently enacted annual provisions in the state budget that are substantially similar, providing for the computation of tax relief as a specific dollar amount to be offset against the total taxes that would otherwise be due but for PPTRA and the reporting of such specific dollar relief on the tax bill.
- (b) The Town Council shall annually by motion or resolution set the percentage of tax relief at such a level that it is anticipated fully to exhaust PPTRA relief funds provided to the Town of Broadway by the Commonwealth.
- (c) Personal property tax bills shall set forth on their face the specific dollar amount of relief credited with respect to each qualifying vehicle, together with an explanation of the general manner in which relief is allocated.

## § 52.23 ALLOCATION OF RELIEF AMONG TAXPAYERS

- (a) Allocation of PPTRA relief shall be provided in accordance with the general provisions of this subchapter as implemented by the specific provisions of the Town of Broadway's annual budget relating to PPTRA relief.
- (b) Relief shall be allocated in such as manner as to eliminate personal property taxation of each qualifying vehicle with an assessed value of \$1,000 or less.

- (c) Relief shall be in any event allocated in such a manner that when a vehicle is leased by a member of the United States Military on active duty, or a spouse thereof, or both, under a contract requiring the tenant thereof to pay personal property tax, and such vehicle would otherwise be exempt from tax under other applicable law, the tax based on the first \$20,000 in value of such vehicle shall be fully eliminated.
- (d) Relief with respect to qualifying vehicles with assessed values of more than \$1,000 shall be provided at a percentage, annually fixed and applied to the first \$20,000 in value of each such qualifying vehicle, that is estimated fully to use all available state PPTRA relief, taking into account the exemptions described in paragraphs (b) and (c). The percentage shall be established annually as a part of the adopted budget for the Town of Broadway.
- (e) Paragraph (d) of the Ordinance of September 1, 2006, which set out certain provisions regarding certain personal property tax bills that were delinquent on that date, is not repealed by the enactment of this title.

Editor's Note – The ordinance of April 4, 2006, which originally enacted provisions substantially similar to this subchapter, also contained in paragraph (d) thereof certain transitional provisions applicable to personal property tax bills delinquent on September 1, 2006. This provision, being of a temporary or transitional nature, is not set out in the Code. A copy is available from the Town Clerk.

## **CHAPTER 53: TOOLS AND MACHINERY TAX**

# § 53.01 LEVY OF TAX

There is hereby levied an annual tax on all tools and machinery within the Town that is segregated for local taxation under state law. Such tax shall be at a millage rate set by the Council from time to time and appearing on the rate schedule to the Town Code. Such tax shall be for general Town purposes.

# § 53.02 TAX DAY

The date for determining the situs of machinery and tools for the purposes of assessing tax under this Chapter shall be January 1. Situs of property shall be determined under the rules prescribed in state law.

## § 53.03 **DUE DATE**

Such taxes on machinery and tools for each year shall be due and payable during the year for which the same are assessed on or before December 5.

# § 53.04 ASSESSMENT OF PROPERTY FOR TAX

For the purposes of collecting machinery and tools tax under this chapter, the Town shall compute the tax based on the assessed value of property sited within the Town as finally determined by the appropriate officials of Rockingham County.

## **CHAPTER 54: BUSINESSES AND PROFESSIONS**

## SUBCHAPTER A – IMPOSITION OF LICENSE TAX GENERALLY

# § 54.10 LICENSE REQUIREMENT; ISSUANCE OF LICENSE

- (a) Every person shall apply for a license for each business or profession when engaging in a business in the Town if such person is not otherwise exempt under state law and if
  - (i) the person has a definite place of business in this jurisdiction;
- (ii) there is no definite place of business anywhere and the person resides in the Town (including a direct seller whose sales exceed \$4,000 per year); or
- (iii) there is no definite place of business in this jurisdiction but the person operates amusement machines or is classified as an itinerant merchant, peddler (including a peddler at wholesale), carnival, circus, contractor (subject to the provisions of this chapter relating to contractors), or public service corporation.
- (b) A separate license shall be required for each definite place of business and for each business. A person engaged in two or more businesses or professions carried on at the same place of business may elect to obtain one license for all such businesses and professions if all of the following criteria are satisfied:
- (i) each business or profession is subject to licensure at the location and has satisfied any requirements imposed by state law or other provisions of this Code;
- (ii) all of the businesses or professions are subject to the same tax rate, or, if subject to different tax rates, the licensee agrees to be taxed on all businesses and professions at the highest rate; and
- (iii) the taxpayer agrees to supply such information as the Treasurer may require concerning the nature of the several businesses and their gross receipts.
- (c) No license shall issue except upon payment of any gross receipts tax, flat tax, or flat fee imposed by this chapter.
- (d) No license shall issue unless the Treasurer is satisfied that the applicant has paid all delinquent Town business license, real estate, personal property, meals, or other taxes due under this title, or other debts owed the Town that have the same status as unpaid taxes under applicable law.
- (e) The Treasurer shall promulgate forms upon which an applicant may apply for a license.

- (f) This chapter shall be interpreted in conjunction with the provisions of Virginia law that enable localities to impose the tax described herein. Any terms not defined herein or in the general provisions of the Town Code shall have the meaning attributed to them in the provisions of state law that authorize the Town to impose the tax described herein.
- (g) Any person who does not obtain a license required by this Chapter, and who does so with the specific intent of evading the any tax imposed under this title, shall be guilty of a Class 1 misdemeanor and upon due conviction thereof shall be incarcerated for a period not to exceed twelve (12) months and fined not more than \$2,500, or both.
- (h) Any person who knowingly does not obtain a license required by this Chapter shall be guilty of a Class 3 misdemeanor and upon due conviction thereof shall be fined not more than \$500. This offense is a lesser included offense to the offense described in paragraph (g). Each day of the offense described in this paragraph (h) shall be a separate offense.

# § 54.11 PAYMENT OF TAX; PENALTY; INTEREST

- (a) <u>Timing</u>. Each person required to obtain a license shall apply for a license
- (i) prior to beginning business, if he was not subject to licensure the Town on or before January 1 of the license year, or
- (ii) no later than March 1 of the license year if he had been issued a license for the preceding year.
- (b) <u>Payment</u>. The tax shall be paid on or before March 1, or in the case of a taxpayer beginning business, no more than 30 days after beginning business. Provided, however, that any dealer in motor vehicles who itemizes the tax and passes it along to his customer as a separate item on the invoice of such customer shall pay the tax on a quarterly basis each calendar quarter, such tax being due on the fifteenth day of the first month following the conclusion of such quarter.
- (c) Extension of Time to Pay. In exercising his power to extend the date for the payment of the tax under § 50.05(a), the Treasurer may condition the extension upon the timely payment of a reasonable estimate of the appropriate tax; the tax is then subject to adjustment to the correct tax at the end of the extension, together with interest from the due date until the date paid and, if the estimate submitted with the extension is found to be unreasonable under the circumstances, with a penalty of 10 percent of the portion paid after the due date.
  - (d) <u>Penalty</u>.

- (i) The Treasurer shall impose a penalty of 10 percent of the tax upon the failure to file an application or the failure to pay the tax by the appropriate due date. Only the late filing penalty shall be imposed if both the application and payment are late; however, the Treasurer shall impose both penalties if he determines that the taxpayer has a history of noncompliance.
- (ii) In the case of an assessment of additional tax made by the Treasurer after the initial filing, if the application and, if applicable, the return were made in good faith and the understatement of the tax was not due to any fraud, reckless or intentional disregard of the law by the taxpayer, there shall be no late payment penalty assessed with the additional tax. If such additional assessment of tax by is not paid within 30 days, the treasurer may impose a 10 percent late payment penalty.
- If the failure to file or pay was not the fault of the taxpayer, the penalties (iii) shall not be imposed, or if imposed, shall be abated by Treasurer. In order to demonstrate lack of fault, the taxpayer must show that he Acted Responsibly and that the failure was due to Events Beyond His Control. For the purposes of this paragraph, Acted Responsibly means that: (i) the taxpayer exercised the level of reasonable care that a prudent person would exercise under the circumstances in determining the filing obligations for the business and (ii) the taxpayer undertook significant steps to avoid or mitigate the failure, such as requesting appropriate extensions (where applicable), attempting to prevent a foreseeable impediment, acting to remove an impediment once it occurred, and promptly rectifying a failure once the impediment was removed or the failure discovered. Events Beyond the Taxpayer's Control include, but are not limited to, the unavailability of records due to fire or other casualty; the unavoidable absence (e.g., due to death or serious illness) of the person with the sole responsibility for tax compliance; or the taxpayer's reasonable reliance in good faith upon erroneous written information from a Town official who was aware of the relevant facts relating to the taxpayer's business when he provided the erroneous information.
- (iv) This paragraph (d) shall control the assessment of penalties for failure to obtain the license and pay the tax required by this Chapter, and shall override the provisions of § 50.02(a). The provisions of § 50.02(d) and (e) continue to apply.
- (e) <u>Interest</u>. Notwithstanding the provisions of §§ 50.03 and 50.04, interest at an annual rate of 10 percent shall be charged on the late payment of the tax from the due date until the date paid without regard to fault or other reason for the late payment. Whenever an assessment of additional or omitted tax is found to be erroneous, all interest and any penalties charged and collected on the amount of the assessment found to be erroneous shall be refunded together with interest at an annual rate of 10 percent on the refund from the date of payment or the due date, whichever is later. Interest shall be paid on the refund of any such tax from the date of payment or due date, whichever is later, whether attributable to an amended return or other reason. No interest shall accrue on an adjustment of estimated tax liability to actual liability at the conclusion of a base year. No interest shall be paid on a refund or charged on a late payment, provided the refund or the late payment is made not more than 30 days from the date of the

payment that created the refund or the due date of the tax, whichever is later, but all such interest shall be waived.

# § 54.12 SITUS OF GROSS RECEIPTS

- (a) General Rule. The gross receipts included in the taxable measure shall be only those gross receipts attributed to the exercise of a privilege subject to licensure at a definite place of business within the Town. In the case of activities conducted outside of a definite place of business, such as during a visit to a customer location, the gross receipts shall be attributed to the definite place of business from which such activities are initiated, directed, or controlled. The situs of gross receipts for different classifications of business shall be attributed to one or more definite places of business or offices as follows:
- (i) Except as otherwise provided herein, the gross receipts of a contractor shall be attributed to the definite place of business at which his services are performed, or if his services are not performed at any definite place of business, then the definite place of business from which his services are directed or controlled.
- (ii) The gross receipts of a retailer or wholesaler shall be attributed to the definite place of business at which sales solicitation activities occur, or if sales solicitation activities do not occur at any definite place of business, then the definite place of business from which sales solicitation activities are directed or controlled; however, a wholesaler or distribution house subject to a license tax measured by purchases shall determine the situs of its purchases by the definite place of business at which or from which deliveries of the purchased goods, wares and merchandise are made to customers. Any wholesaler who is subject to license tax in two or more localities and who is subject to multiple taxation because the localities use different measures, may apply to the Department of Taxation for a determination as to the proper measure of purchases and gross receipts subject to license tax in each locality;
- (iii) The gross receipts of a business renting tangible personal property shall be attributed to the definite place of business from which the tangible personal property is rented or, if the property is not rented from any definite place of business, then to the definite place of business at which the rental of such property is managed; and
- (iv) The gross receipts from the performance of services shall be attributed to the definite place of business at which the services are performed or, if not performed at any definite place of business, then to the definite place of business from which the services are directed or controlled.
- (b) <u>Special Rule for Contractors</u>. When a contractor has paid any local license tax required by the county, city or town in which his principal office and any branch office or offices may be located, no further license or license tax shall be required by the Town. Notwithstanding the foregoing, when the amount of business done by any such contractor in the Town exceeds the sum of \$25,000 in any year, such contractor shall obtain a local license from the Town, and the

amount of business done in such other county, city or town in which a license tax is paid may be deducted by the contractor from the gross revenue reported to the county, city or town in which the principal office or any branch office of the contractor is located. Reciprocally, the Town shall allow such a deduction from gross revenue when the contractor maintains a definite place of business within the Town and a license tax is paid to another locality in the Commonwealth. Any contractor conducting business in the Town for less than thirty days without a definite place of business in any county, city or town of the Commonwealth, but who will do more than \$25,000 of business within the Town in a license year, shall be subject to the license fee or license tax imposed on contractors by the Town for the gross receipts derived from activities within the Town. Any contractor conducting such work for more than thirty days has a definite place of business within the Town, and shall be taxed on that basis.

- (c) Apportionment. If the licensee has more than one definite place of business and it is impractical or impossible to determine to which definite place of business gross receipts should be attributed under the general rule, the gross receipts of the business shall be apportioned between the definite places of businesses on the basis of payroll. Gross receipts shall not be apportioned to a definite place of business unless some activities under the applicable general rule occurred at, or were controlled from, such definite place of business. Gross receipts attributable to a definite place of business in another jurisdiction shall not be attributed to the Town solely because the other jurisdiction does not impose a tax on the gross receipts attributable to the definite place of business in such other jurisdiction.
- (d) Agreements. The Treasurer may enter into agreements with any other political subdivision of Virginia concerning the manner in which gross receipts shall be apportioned among definite places of business. However, the sum of the gross receipts apportioned by the agreement shall not exceed the total gross receipts attributable to all of the definite places of business affected by the agreement. Upon being notified by a taxpayer that its method of attributing gross receipts is fundamentally inconsistent with the method of one or more political subdivisions in which the taxpayer is licensed to engage in business and that the difference has, or is likely to, result in taxes on more than 100 percent of its gross receipts from all locations in the affected jurisdictions, the Treasurer shall make a good faith effort to reach an apportionment agreement with the other political subdivisions involved. If an agreement cannot be reached, either the Treasurer or taxpayer may seek an advisory opinion from the Department of Taxation pursuant to state law, and notice of the request shall be given to the other party. Notwithstanding the provisions of state law, when a taxpayer has demonstrated to a court that two or more political subdivisions of Virginia have assessed taxes on gross receipts that may create a double assessment within the meaning of state law, the court shall enter such orders pending resolution of the litigation as may be necessary to ensure that the taxpayer is not required to pay multiple assessments even though it is not then known which assessment is correct and which is erroneous.

## § 54.13 LIMITATIONS AND EXTENSIONS

- (a) Extension. Where, before the expiration of the time prescribed for the assessment of any license tax imposed pursuant to this chapter, both the Treasurer and the taxpayer have consented in writing to its assessment after such time, the tax may be assessed at any time prior to the expiration of the period agreed upon. The period so agreed upon may be extended by subsequent agreements in writing made before the expiration of the period previously agreed upon.
- (b) <u>Fraud</u>. Notwithstanding any limitation period prescribed by state law, the Treasurer shall assess the local license tax omitted because of fraud or failure to apply for a license for the current license year and the six preceding license years.
- (c) <u>Tolling of Limitation Period</u>. The period for collecting any local license tax shall not expire prior to the latest of
  - (i) the period specified in state law for the assessment of tax;
- (ii) two years after the date of assessment if the period for assessment has been extended pursuant to this § 54.13;
- (iii) two years after the final determination of an appeal for which collection has been stayed pursuant to subchapter B of this chapter; or
- (iv) two years after the final decision in a court application for review of an assessment pursuant to state law for which collection has been stayed.
- (d) <u>State law exemptions</u>. No license shall be required of any person exempt from license taxes and fees under the provisions of state law.

# § 54.14 STATEMENTS ON LICENSE APPLICATIONS; PENALTY

A person executing an application for a license, whether individually or as an agent, certifies by his signature that the information on the application is true and correct to the best of his knowledge and belief. No such person shall make a materially false or misleading statement on a license application. Any person who makes such a statement is guilty of a Class 1 misdemeanor and upon due conviction thereof shall be incarcerated for a period not to exceed twelve (12) months and fined not more than \$2,500, or both.

# § 54.15 DEFINITION OF GROSS RECEIPTS

Gross receipts shall in every case be defined in such a manner, and shall be subject to such exclusions and deductions, as provided by state law.

## **§ 54.16 RATE OF TAX**

The tax imposed by this chapter shall be at a millage rate set by the Town Council and described on the rate schedule attached to this Code, except in situations in which the Town Council levies a flat license tax in lieu of the tax on gross receipts, or in such cases where the Town Council levies a license fee in lieu of any tax. Such flat license taxes and the license fee shall also appear on the rate schedule attached to this Code, or in the appropriate section of this Code.

## SUBCHAPTER B: PROCUEDURES FOR DISPUTED LICENSE TAXES

# § 54.20 ADMINISTRATIVE APPEALS BEFORE TREASURER

(a) <u>Definitions</u>. For purposes of this subchapter:

"Amount in dispute," when used with respect to taxes due or assessed, means the amount specifically identified in the administrative appeal or application for judicial review as disputed by the party filing such appeal or application.

"Appealable event" means an increase in the assessment of a local license tax payable by a taxpayer, the denial of a refund, or the assessment of a local license tax where none previously was assessed, arising out of the Treasurer's (i) examination of records, financial statements, books of account, or other information for the purpose of determining the correctness of an assessment; (ii) determination regarding the rate or classification applicable to the licensable business; (iii) assessment of a local license tax when no return has been filed by the taxpayer; or (iv) denial of an application for correction of erroneous assessment attendant to the filing of an amended application for license. An appealable event shall include a taxpayer's appeal of the classification applicable to a business, including whether the business properly falls within a business license subclassification established by the Town, regardless of whether the taxpayer's appeal is in conjunction with an assessment, examination, audit, or any other action taken by the Town.

"Frivolous" means a finding, based on specific facts, that the party asserting the appeal is unlikely to prevail upon the merits because the appeal is (i) not well grounded in fact; (ii) not warranted by existing law or a good faith argument for the extension, modification, or reversal of existing law; (iii) interposed for an improper purpose, such as to harass, to cause unnecessary delay in the payment of tax or a refund, or to create needless cost from the litigation; or (iv) otherwise frivolous.

"Jeopardized by delay" means a finding, based upon specific facts, that a taxpayer designs to (i) depart quickly from Rockingham County; (ii) remove his property therefrom; (iii) conceal himself or his property therein; or (iv) do any other act tending to prejudice, or to render wholly or partially ineffectual, proceedings to collect the tax for the period in question.

- (b) Filing and contents of administrative appeal. Any person assessed with a local license tax as a result of an appealable event as defined in this subchapter may file an administrative appeal of the assessment within one year from the last day of the tax year for which such assessment is made, or within one year from the date of the appealable event, whichever is later, with the Treasurer, and with a copy to the Town Attorney. The appeal must be filed in good faith and sufficiently identify the taxpayer, the tax periods covered by the challenged assessments, the amount in dispute, the remedy sought, each alleged error in the assessment, the grounds upon which the taxpayer relies, and any other facts relevant to the taxpayer's contention. The Treasurer shall hold a conference with the taxpayer if requested by the taxpayer or sua sponte, or require submission of additional information and documents, an audit or further audit, or other evidence deemed necessary for a proper and equitable determination of the appeal, and may have the assistance of the Town Attorney in so doing. The assessment placed at issue in the appeal shall be deemed prima facie correct. The Treasurer shall undertake a full review of the taxpayer's claims and issue a written determination to the taxpayer setting forth the facts and arguments in support of his decision.
- (c) Administrative Appeal of Classification. The taxpayer may at any time also file an administrative appeal of the classification applicable to the taxpayer's business, including whether the business properly falls within a business license subclassification established by the Town. However, the appeal of the classification of the business shall not apply to any license year for which the Tax Commissioner has previously issued a final determination relating to any license fee or license tax imposed upon the taxpayer's business for the year. In addition, any appeal of the classification of a business shall in no way affect or change any limitations period prescribed by law for appealing an assessment.
- (d) Notice of right of appeal and procedures. Every assessment made by the Treasurer pursuant to an appealable event shall include or be accompanied by a written explanation of the taxpayer's right to file an administrative appeal and the specific procedures to be followed, the name and address to which the appeal should be directed, an explanation of the required content of the appeal, and the deadline for filing the appeal. For purposes of facilitating an administrative appeal of the classification applicable to a taxpayer's business, the Town shall maintain on its website the specific procedures to be followed with regard to such appeal and the name and address to which the appeal should be directed.
- (e) <u>Suspension of collection activity during appeal</u>. Provided a timely and complete administrative appeal is filed, collection activity with respect to the amount in dispute relating to any assessment by the Treasurer shall be suspended until a final determination is issued thereon, unless the Treasurer (i) determines that collection would be jeopardized by delay as defined in this section; (ii) determines that the taxpayer has not responded to a request for relevant information after a reasonable time; or (iii) determines that the appeal is frivolous as defined in this section. Interest shall accrue in accordance with §54.11(e), but no further penalty shall be imposed while collection action is suspended.

(f) Procedure in event of nondecision. Any taxpayer whose administrative appeal pursuant to the provisions of this § 54.20 has been pending for more than one year without the issuance of a final determination may, upon not less than 30 days' written notice to the Treasurer, with a copy to the Town Attorney, elect to treat the appeal as denied and appeal the assessment or classification of the taxpayer's business to the Tax Commissioner in accordance with the provisions of § 54.21. The Tax Commissioner shall not consider an appeal filed pursuant to the provisions of this subsection if he finds that the absence of a final determination on the part of the Treasurer was caused by the willful failure or refusal of the taxpayer to provide information requested and reasonably needed by the Treasurer to make his determination.

# § 54.21 APPEALS TO TAX COMMISSIONER

- (a) Any person assessed with a local license tax as a result of a determination or that has received a determination with regard to the person's appeal of the license classification or subclassification applicable to the person's business, upon an administrative appeal to Treasurer pursuant to § 54.20, that is adverse to the position asserted by the taxpayer in such appeal, may appeal such assessment or determination to the Tax Commissioner within 90 days of the date of the determination by the Treasurer. The appeal shall be in such form as the Tax Commissioner may prescribe and the taxpayer shall serve a copy of the appeal upon the Treasurer and the Town Attorney. The Tax Commissioner shall permit the Treasurer to participate in the proceedings, shall permit the Town Attorney to represent the Treasurer, and shall issue a determination to the taxpayer within 90 days of receipt of the taxpayer's application, unless the taxpayer and the Treasurer are notified that a longer period will be required. The appeal shall proceed in the same manner as an application pursuant to Va. Code § 58.1-1821, and the Tax Commissioner pursuant to Va. Code § 58.1-1822 may issue an order correcting such assessment or correcting the license classification or subclassification of the business and the related license tax or fee liability.
- (b) Suspension of collection activity during appeal. On receipt of a notice of intent to file an appeal to the Tax Commissioner under § 54.21(a), collection activity with respect to the amount in dispute relating to any assessment shall be suspended until a final determination is issued by the Tax Commissioner, unless the Treasurer (i) determines that collection would be jeopardized by delay as defined in this section; (ii) determines, or is by the Tax Commissioner, that the taxpayer has not responded to a request for relevant information after a reasonable time; or (iii) determines, or is advised by the Tax Comissioner, that the appeal is frivolous as defined in this section. Interest shall accrue in accordance with the provisions of § 54.11(e), but no further penalty shall be imposed while collection action is suspended. The requirement that collection activity be suspended shall cease unless an appeal pursuant to § 54.21(a) is filed and served on the necessary parties within 30 days of the service of notice of intent to file such appeal, or within 90 days of the determination being appealed.
- (c) <u>Implementation of determination of Tax Commissioner</u>. Promptly upon receipt of the final determination of the Tax Commissioner with respect to an appeal pursuant to §54.21(a), the Treasurer shall take those steps necessary to calculate the amount of tax owed by or refund

due to the taxpayer consistent with the Tax Commissioner's determination and shall provide that information to the taxpayer. The following provisions shall apply:

- (i) If the determination of the Tax Commissioner sets forth a specific amount of tax due, the Treasurer shall issue a bill to the taxpayer for such amount due, together with interest accrued and penalty, if any is authorized by this section, within 30 days of the date of the determination of the Tax Commissioner.
- (ii) If the determination of the Tax Commissioner sets forth a specific amount of refund due, the Treasurer shall issue a payment to the taxpayer for such amount due, together with interest accrued pursuant to this section, within 30 days of the date of the determination of the Tax Commissioner.
- (iii) If the determination of the Tax Commissioner does not set forth a specific amount of tax due, or otherwise requires the Treasurer to undertake a new or revised assessment that will result in an obligation to pay a tax that has not previously been paid in full, the Treasurer shall promptly commence the steps necessary to undertake such new or revised assessment, and provide the same to the taxpayer within 60 days of the date of the determination of the Tax Commissioner, or within 60 days after receipt from the taxpayer of any additional information requested or reasonably required under the determination of the Tax Commissioner, whichever is later. The Treasurer shall issue a bill to the taxpayer for the amount due, together with interest accrued and penalty, if any is authorized by this section, within 30 days of the date of the new assessment.
- (iv) If the determination of the Tax Commissioner does not set forth a specific amount of refund due, or otherwise requires the Treasurer to undertake a new or revised assessment that will result in an obligation on the part of the Town to make a refund of taxes previously paid, the Treasurer shall promptly commence the steps necessary to undertake such new or revised assessment or to determine the amount of refund due in the case of a correction to the license classification or subclassification of the business, and provide the same to the taxpayer within 60 days of the date of the determination of the Tax Commissioner, or within 60 days after receipt from the taxpayer of any additional information requested or reasonably required under the determination of the Tax Commissioner, whichever is later. The Treasurer shall issue a refund to the taxpayer for the amount of tax due, together with interest accrued, within 30 days of the date of the new assessment or determination of the amount of the refund.

# § 54.22 JUDICIAL REVIEW OF DETERMINATION OF TAX COMMISSIONER

(a) <u>Judicial Review</u>. Following the issuance of a final determination of the Tax Commissioner pursuant to § 54.21(a), the taxpayer or Treasurer may apply to the Rockingham County Circuit Court for judicial review of the determination, or any part thereof, pursuant to Va. Code § 58.1-3984. In any such proceeding for judicial review of a determination of the Tax Commissioner, the burden shall be on the party challenging the determination of the Tax Commissioner, or any part thereof, to show that the ruling of the Tax Commissioner is erroneous

with respect to the part challenged. Neither the Tax Commissioner nor the Department of Taxation shall be made a party to an application to correct an assessment merely because the Tax Commissioner has ruled on it.

- (b) <u>Suspension of payment of disputed amount of tax due upon taxpayer's notice of</u> intent to initiate judicial review.
- (i) On receipt of a notice of intent to file an application for judicial review, pursuant to Va. Code § 58.1-3984, of a determination of the Tax Commissioner pursuant to §54.21(a), and upon payment of the amount of the tax relating to any assessment that is not in dispute together with any penalty and interest then due with respect to such undisputed portion of the tax, the Treasurer shall further suspend collection activity while the court retains jurisdiction unless the court, upon appropriate motion after notice and an opportunity to be heard, determines that (i) the taxpayer's application for judicial review is frivolous, as defined in this section; (ii) collection would be jeopardized by delay, as defined in this section; or (iii) suspension of collection would cause substantial economic hardship to the Town. For purposes of determining whether substantial economic hardship to the locality would arise from a suspension of collection activity, the court shall consider the cumulative effect of then-pending appeals filed within the locality by different taxpayers that allege common claims or theories of relief.
- (ii) Upon a determination that the appeal is frivolous, that collection may be jeopardized by delay, or that suspension of collection would result in substantial economic hardship to the Town, the court may require the taxpayer to pay the amount in dispute or a portion thereof, or to provide surety for payment of the amount in dispute in a form acceptable to the court.
- (iii) No suspension of collection activity shall be required if the application for judicial review fails to identify with particularity the amount in dispute or the application does not relate to any assessment by the Treasurer.
- (iv) The requirement that collection activity be suspended shall cease unless an application for judicial review pursuant to Va. Code § 58.1-3984 is filed and served on the necessary parties within 30 days of the service of the notice of intent to file such application.
- (iv) The suspension of collection activity required by this § 54.22(b) shall not be applicable to any appeal of a local license tax that is initiated by the direct filing of an action pursuant to Va. Code § 58.1-3984 without prior exhaustion of the appeals provided by this subchapter.
- (c) <u>Suspension of payment of disputed amount of refund due upon locality's notice of intent to initiate judicial review.</u>
- (i) Payment of any refund determined to be due pursuant to the determination of the Tax Commissioner of an appeal pursuant to § 54.21(a) shall be suspended if the Town

serves upon the taxpayer, within 60 days of the date of the determination of the Tax Commissioner, a notice of intent to file an application for judicial review of the Tax Commissioner's determination pursuant to Va. Code § 58.1-3984 and pays the amount of the refund not in dispute, including tax and accrued interest. Payment of such refund shall remain suspended while the court retains jurisdiction unless the court, upon appropriate motion after notice and an opportunity to be heard, determines that the Town's application for judicial review is frivolous, as defined in this section.

- (ii) No suspension of refund activity shall be permitted if the Town's application for judicial review fails to identify with particularity the amount in dispute.
- (iii) The suspension of the obligation to make a refund shall cease unless an application for judicial review pursuant to Va. Code § 58.1-3984 is filed and served on the necessary parties within 30 days of the service of the notice of intent to file such application.
- (d) Accrual of interest on unpaid amount of tax. Interest shall accrue in accordance with the provisions of §54.11(e) , but no further penalty shall be imposed while collection action is suspended.

## § 54.23 PRIVATE LETTER RULINGS

Any taxpayer or authorized representative of a taxpayer may request a written ruling regarding the application of a local license tax to a specific situation from Town Attorney. Any person requesting such a ruling must provide all facts relevant to the situation placed at issue and may present a rationale for the basis of an interpretation of the law most favorable to the taxpayer. In addition, the taxpayer or authorized representative may request a written ruling with regard to the classification applicable to the taxpayer's business, including whether the business properly falls within a business license subclassification established by the Town. Any misrepresentation or change in the applicable law or the factual situation as presented in the ruling request shall invalidate any such ruling issued. A written ruling may be revoked or amended prospectively if (i) there is a change in the law, a court decision, or the guidelines issued by the Department of Taxation upon which the ruling was based or (ii) the Town Attorney the taxpayer of a change in the policy or interpretation upon which the ruling was based. However, any person who acts on a written ruling which later becomes invalid shall be deemed to have acted in good faith during the period in which such ruling was in effect.

## § 54.24 RECORD KEEPING AND AUDITS

Every person who is assessable with a local license tax shall keep sufficient records to enable the Treasurer to verify the correctness of the tax paid for the license years assessable and to enable the Treasurer to ascertain what is the correct amount of tax that was assessable for each of those years. All such records, books of accounts and other information shall be open to inspection and examination by the Treasurer in order to allow the assessor to establish whether a particular receipt is directly attributable to the taxable privilege exercised within the Town. The Treasurer

shall provide the taxpayer with the option to conduct the audit in the taxpayer's local business office, if the records are maintained there. In the event the records are maintained outside the Town, copies of the appropriate books and records shall be sent to the assessor's office upon demand.

# SUBCHAPTER C: SPECIAL RULES

## § 54.30 LIMITATION ON RATES AND AMOUNTS; STATE LAW EXEMPTIONS

- (a) In the event that state law limits the maximum rate or amount of license tax otherwise assessable to any taxpayer under this chapter, and the tax to be charged hereunder exceeds the maximum rate or amount allowed, the tax shall be reduced for that taxpayer to the extent necessary to cause the tax to comply with state law.
- (b) In the event that state law exempts a particular business, person, or activity from taxation, the Town shall not impose a license fee or license tax on such person.

# § 54.31 FLAT TAXES ON CERTAIN BUSINESSES

Any person engaged in the following activities is engaged in a business or profession within the meaning of this chapter, and shall be subject to a flat business license tax as shown on the schedule attached to the Town Code:

- (a) Any person who shall pretend to tell fortunes, to act as a clairvoyant, or to practice palmistry or phrenology; and
  - (b) Building or savings and loan associations and loan companies.

# § 54.32 OPERATORS OF COIN MACHINES

- (a) <u>License required</u>. Any amusement operator who operates more than three coin amusement machines within the Town is engaged in a business within the Town with a definite place of business in the location of such machines, all within the meaning of this chapter, and shall pay the license tax on gross receipts from such machines at the same rate as if such receipts were retail sales, subject to any dollar or other limitation imposed by state law. The Town does not require stickers to evidence payment of such tax.
- (b) <u>Penalty</u>. In addition to the other penalties and interest provided herein, any coin amusement machine operated in violation of this chapter shall be subject to civil forfeiture to the Town.

# § 54.33 PHOTOGRAPHERS

A photographer who receives compensation for taking photographs, is a resident of the Town, but has no definite place of business anywhere, is required to obtain a business license pursuant to § 54.10(a)(ii). Such photographer shall pay a flat license tax as shown on the schedule attached to the Town Code.

## § 54.34 CARNIVALS

A carnival, circus, or speedway conducted within the limits of the Town is engaged in a business within the Town with a definite place of business in the location of such carnival, circus, or speedway, or with no definite place of business but which gives a performance within the Town, all within the meaning of this chapter, and unless otherwise exempt shall obtain a license therefor and shall pay a flat license tax per performance in the amount shown on the schedule attached to the Town Code.

# § 54.35 EXEMPTION WHEN TAX BELOW \$25; FEE ON CERTAIN BUSINESSES

If the gross receipts of a business are below the threshold that would cause the license tax otherwise due under this chapter to be less than \$25, then

- (a) such business shall be exempt from the license tax on its gross receipts; and
  - (b) in lieu of such a license tax, the person shall pay a license fee of \$25.

In enacting this section, the Town is exercising is power to set a threshold below which the license tax shall not apply, and is setting such threshold for each classification at an amount of gross receipts that would otherwise result in a license tax of \$25. In addition, in that the businesses to which this section applies are not subject to a tax on gross receipts, the Town is exercising its power to impose a \$25 license fee on such businesses.

## § 54.36 GASOLINE SALES

Gasoline sales constitute a separate subclassification of business for the purposes of this chapter. The Town exempts gasoline sales from any license tax or license fee.

## § 54.37 ALCOHOLIC BEVERAGES TAX CREDIT

Any beer wholesaler or wholesale wine licensee paying a tax under this Chapter 54 shall be allowed a credit against such tax for any amounts paid to the Town as an alcoholic beverage license tax under § 57.21 et squitur.

## § 54.38 TELEPHONE COMPANIES

In lieu of any other license tax imposed under this chapter, all persons engaged in the business of providing telephonic communications shall obtain a license each year and shall pay a tax equal to one-half of one percent of the gross receipts (as defined in this chapter) during the preceding year from local telephone exchange service, but excluding long distance telephone calls.

## **CHAPTER 55: MEALS TAX**

# § 55.01 DEFINITIONS

The following words and phrases, when used in this ordinance, shall have, for the purposes of this chapter, the following respective meanings except where the context clearly indicates a different meaning:

- (a) <u>Cater.</u> The furnishing of food, beverages, or both on the premises of another, for compensation.
- (b) Food. All food, beverages or both, including alcoholic beverages, purchased in or from a food establishment, whether prepared in such food establishment or not, and whether consumed on the premises or not, and without regard to the manner, time or place of service.
- (c) Food establishment. Any place in or from which food or food products are prepared, packaged, sold or distributed in the Town, including but not limited to, any restaurant, dining room, grill, coffee shop, cafeteria, café, snack bar, lunch counter, convenience store, grocery store, movie theater; delicatessen, confectionery, bakery, eating house, eatery, drugstore, ice cream/yogurt shop, lunch wagon or truck, pushcart or other mobile facility from which food is sold, public or private club, resort, bar, lounge, or other similar establishment, public or private, and shall include private property outside of and contiguous to a building or structure operated as a food establishment at which food or food products are sold for immediate consumption.
- (d) Meal. Meal shall mean any prepared food or drink which is (i) (A) offered or held out for sale by a food establishment for the purpose of being consumed by any person to satisfy the appetite immediately and (B) ready for immediate consumption; or (ii) sandwiches, salad bar items sold from a salad bar, prepackaged single-serving salads consisting primarily of an assortment of vegetables, and nonfactory sealed beverages. All such food and beverage, unless otherwise specifically exempted or excluded herein shall be included, whether intended to be consumed on the seller's premises or elsewhere, whether designated as breakfast, lunch, snack, dinner, supper or by some other name, and without regard to the manner, time or place of service.

## § 55.02 IMPOSITION OF TAX

There is hereby imposed and levied by the Town on each person a tax at the rate described in the rate schedule to the Town Code, on the amount paid for meals purchased from any food

establishment, whether prepared in such food establishment or not, and whether consumed on the premises or not.

# § 55.03 COLLECTION OF TAX BY SELLER

Every person receiving any payment for food with respect to which a tax is levied hereunder shall collect and remit the amount of the tax imposed by this ordinance from the person on whom the same is levied or from the person paying for such food at the time payment for such food is made; provided, however, no blind person operating a vending stand or other business enterprise under the jurisdiction of the Department for the Visually Handicapped and located on property acquired and used by the United States for any military or naval purpose shall be required to collect or remit such taxes. All tax collection shall be deemed to be held in trust for the Town.

# § 55.04 EXEMPTIONS; LIMITS ON APPLICATION

- (a) The tax imposed under this chapter shall not be levied on factory-prepackaged candy, gum, nuts and other items of essentially the same nature served for on or off-premises consumption.
- (b) The tax imposed under this chapter shall not be levied on the following items when a food establishment sells them exclusively for off-premises consumption:
  - (1) Donuts, crackers, chips, cookies and other factory-prepackaged items of essentially the same nature sold primarily as snacks.
  - (2) Food sold in bulk. For the purposes of this provision, a bulk sale shall mean the sale of any item that would exceed the normal, customary and usual portion sold for on premises consumption (e.g. a whole cake, a gallon of ice cream). Any food or beverage that is catered or delivered by a food establishment for off-premises consumption is not "sold in bulk" for the purposes of this paragraph, and is subject to tax.
  - (3) Alcoholic and non-alcoholic beverages sold in factory sealed containers.
  - (4) Any food or food product purchased with food coupons issued by the United States Department of Agriculture under the Food Stamp Program or drafts issued through the Virginia Special Supplemental Food Program for Women, Infants, and Children.

- (c) A grocery store, supermarket or convenience store shall not be subject to the tax except for any portion or section therein designated for the sale of prepared food and beverages constituting meals under this Chapter.
- (d) The tax imposed hereunder shall not be levied on the following purchases of food and beverages, whether consumed on premises or off-premises:
  - (1) Food and beverages furnished by food establishments to employees as part of their compensation when no charge is made to the employee.
  - (2) Food and beverages sold by day care centers, public or private elementary or secondary schools or food sold by any college or university to its students or employees.
  - (3) Food and beverages for use or consumption and which are paid for directly by the Commonwealth, any political subdivision of the Commonwealth or the United States.
  - (4) Food and beverages furnished by a hospital, medical clinic, convalescent home, nursing home, home for the aged, infirm, handicapped, battered women, narcotic addicts or alcoholics, or other extended care facility to patients or residents thereof and the spouses and children of such persons.
  - (5) Food and beverages furnished by a public or private non-profit charitable organization or establishment or a private establishment that contracts with the appropriate agency of the Commonwealth to offer meals at concession prices to elderly, infirm, blind, handicapped or needy persons in their homes or at central locations.
  - (6) Food and beverages sold on an occasional basis, not exceeding 12 times per calendar year, by a non-profit educational, charitable or benevolent organization, church, or religious body as a fundraising activity, the gross proceeds of which are to be used by such organization exclusively for non-profit educational, charitable, benevolent or religious purposes.
  - (7) Food and beverages sold through vending machines.

# § 55.05 GRATUITIES AND SERVICE CHARGES

Where a purchaser provides a gratuity for an employee of a seller, and the amount of the gratuity is wholly in the discretion of the purchaser, the gratuity is not subject to the tax imposed by this chapter, whether paid in cash to the employee or added to the bill and charged to the purchaser's account, provided in the latter case, the full amount of the gratuity is turned over to the employee by the seller. An amount or percent, whether designated as a gratuity, tip or service charge, that

is added to the price of the food and beverages by the seller, and required to be paid by the purchaser, as a part of the selling price of the food and beverages, is subject to the tax imposed by this chapter to the extent that such gratuity, tip, or service charge (a) exceeds twenty percent (20%) of the price of the food and beverages or (b) is not remitted by the seller to the employee providing such services.

# § 55.06 REPORT OF TAXES COLLECTED; REMITTANCE; PRESERVATION OF RECORDS.

- (a) It shall be the duty of every person required by this chapter to pay to the Town the taxes imposed by this chapter to make a report thereof setting forth such information as the Treasurer may prescribe and require, including in the aggregate all purchases taxable under this chapter, the amount charged the purchaser for each such purchase, the date thereof, the taxes collected thereon and the amount of tax required to be collected by this chapter.
- (b) Reports and tax payments for the months of January, February and March shall be made on or before the following April 20th. Reports and tax payments for the months of April, May, and June shall be made on or before the following July 20th. Reports and tax payments for the months of July, August, and September shall be made on or before the following October 20th. Reports and tax payments for the months of October, November, and December shall be made on or before the following January 20th.
- (c) Further, every such person shall maintain records supporting the reports required by paragraph (a) of this section. Such records shall be kept and preserved for a period of five (5) years. The Treasurer or his duly authorized agents shall have the power to examine such records at reasonable times and without unreasonable interference with the business of such person, for the purpose of administering and enforcing the provisions of this ordinance, and to make transcripts of all or any parts thereof.
- (d) A remitter of the tax imposed by this Chapter shall be entitled to offset against the tax liability due to the Town a dealer administrative discount in accordance with the following chart:

Balance of Tax	Percentage Discount
\$0 - \$62,500	0.03
\$62,501 - \$208,000	0.0225
\$208,000 and up	0.15

Such discount shall be available only to a remitter who pays such tax on or before the due date thereof, and who, as of such due date, has no other outstanding past due tax liability to the Town.

### § 55.07 LEGISLATIVE INTENT

The intent of the Town Council is to levy the tax on meals served in or by restaurant, fast-food, catering, and similar establishments, as permitted under Virginia law, as opposed to normal grocery items that are not meals. This Chapter shall be liberally construed to effectuate the Town Council's intent.

# **CHAPTER 56: RESERVED**

### **CHAPTER 57: OTHER TAXES AND FEES**

### SUBCHAPTER A: EXCISE TAX ON CIGARETTES

# § 57.11 ENACTMENT OF TAX

There is hereby imposed an excise tax on the sale of cigarettes to provide revenue for the general fund of the Town as set forth hereafter in this Chapter.

### § 57.12 **DEFNINITIONS**

Except where the context clearly indicates a different meaning, the following words and phrases shall, for purposes of this subchapter, have the meanings respectively ascribed to them in this section, whether or not capitalized:

- (a) <u>Cigarette</u> means any roll of any size or shape for smoking, whether filtered or unfiltered, with or without a mouthpiece, made wholly or partly of cut, shredded or crimped tobacco or other plant or substitute for tobacco, whether flavored, adulterated or mixed with another ingredient, if the wrapper or cover is made of any material other than leaf tobacco or homogenized leaf tobacco, regardless of whether the roll is labeled or sold as a cigarette or by any other name.
- (b) <u>Dealer</u> means every manufacturer, manufacturer's representative, selfwholesaler, wholesaler, public warehouser or other person who supplies a seller within the Town with cigarettes.
- (c) <u>Package</u> means any container, regardless of the material used in its construction, in which separate cigarettes are placed. Packages are those containers of cigarettes from which individual cigarettes are ordinarily taken when they are consumed by their ultimate user.
- (d) <u>Person</u> means any individual, firm, unincorporated association, company, corporation, limited liability company, joint stock company, group, agency, syndicate, trust or trustee, receiver, fiduciary, partnership and conservator. The word "Person" as applied to a partnership, unincorporated association or other joint venture means the partners or members thereof, and as applied to a corporation shall include all the officers.
- (e) <u>Sale</u> means every act or transaction, regardless of the method or means employed, including the use of vending machines and other mechanical devices, whereby title to any cigarettes shall be transferred from the seller to any other person within the town.
- (f) <u>Seller</u> means every person who transfers title to any cigarettes, or in whose place of business title to any cigarettes is transferred, within the Town, for any purpose other than resale.

(g) <u>Stamp</u> means a small gummed piece of paper or decal used to evidence payment of the tax authorized under this Chapter and to be affixed, manually or by machine, to every package of cigarettes sold within the Town.

# § 57.13 LEVY AND RATE

In addition to all other taxes of every kind now or hereafter imposed by law, there is hereby levied and imposed by the town an excise tax on the sale of cigarettes at a rate per cigarette sold within the Town that appears on the rate schedule to the Town Code. The tax shall be paid by the seller, if not previously paid, and collected in the manner and at the time provided for in this subchapter, provided that the tax shall be paid but once. The tax shall not apply to free distribution of sample cigarettes in packages containing five or fewer cigarettes.

**Ordinance of June 4, 2024** amended this Section to increase the rate of cigarette taxes from \$0.01 per cigarette (20 cents per pack) to \$0.0125 per cigarette, and moved the amount of the tax from this section into the tax rate table in the Town Code. It also make an uncodified transition provision for cigarette tax stamps purchased before the effective date of the transition.

### § 57.14 METHODS OF PAYMENT

Payment of the tax imposed by this Chapter shall be evidenced by the use of a stamp, purchased and affixed by the dealer or seller to every package of cigarettes to be sold within the Town. The tax shall be paid at the time the stamps are purchased from the treasurer. Every dealer and seller shall have the right to buy stamps and to affix the same to packages of cigarettes as provided in this Chapter.

### § 57.15 SALE OF STAMPS

The treasurer shall make available for sale stamps of such denominations and in such quantities as may be necessary for the payment of the taxes imposed by this Chapter.

# § 57.16 DUTIES OF DEALERS

- (a) Every dealer in cigarettes has a duty to purchase stamps necessary to pay the tax levied and imposed by this subchapter and to affix or cause to be affixed stamps of the prescribed monetary value to each package of cigarettes prior to delivering or furnishing such cigarettes to any seller within the Town.
- (b) Any dealer found to have sold, used, stored, received, purchased, or possessed more than 60 packages of cigarettes that is unable to furnish evidence of sufficient tax payments and stamp purchases to cover the cigarettes is presumed to be in violation of this Chapter unless the delivery memorandum described in section 57.17(c) has been timely filed.

### § 57.17 DUTIES OF SELLERS

- (a) Every seller is required to examine each package of cigarettes, prior to exposing the same for sale, and ascertain whether each package has the proper stamps affixed or imprinted as required by this Chapter. If unstamped or improperly stamped packages of cigarettes are discovered, the seller shall either:
  - (1) Purchase and affix stamps covering the tax imposed by this subchapter to such packages of cigarettes; or
  - (2) Immediately notify the dealer that provided the cigarettes and, upon such notification, the dealer shall either affix the stamps necessary or replace the packages with properly-stamped packages.
- (b) Any seller found to possess, prior to being offered for sale, more than 60 packages of cigarettes without stamps affixed that is not in the process of affixing such stamps or taking the action required under section 57.17(a) shall be presumed to be in violation of this subchapter. If such person received the cigarettes within the preceding 48 hours, and has not offered them for sale, the presumption shall not apply.
- (c) In the event any seller elects to purchase and affix stamps before offering cigarettes for sale, the dealer delivering and furnishing such cigarettes shall not also be required to purchase and affix stamps; provided that the dealer files a copy of the delivery memorandum with the treasurer on the day following delivery that provides the name and address of the seller and the quantity and type of cigarettes.

# § 57.18 MONTHLY REPORTS AND RECORDKEEPING

- (a) Every dealer is required to report monthly to the treasurer the following information:
  - (1) The quantity of stamped cigarettes sold or delivered to:
    - a. Each dealer:
    - b. Each seller;
    - c. Each separate person or place of business within the Town;
  - (2) The quantity of stamps on hand, both affixed and unaffixed on the first day of the previous calendar month and the quantity of stamps or stamped cigarettes received during that period;

- (3) Copies of any cigarette tax reports submitted to the Virginia Department of Taxation;
- (4) Such further information as the treasurer may require for the proper administration and enforcement of this subchapter for the determination of the exact number of cigarettes in the possession of each dealer or user.
- (b) All information and reporting required under this section shall be filed with the Treasurer between the first and 20th day after the close of each calendar month.
- (c) Upon request by the Treasurer, any seller shall provide in writing the name and address of the dealer who provides the seller with cigarettes.
- (d) Every dealer and seller in the Town shall maintain and keep complete records of the number of packages of cigarettes delivered or sold for three years and make all records available for examination by the treasurer, upon demand, at any reasonable time.

### § 57.19 ASSESSMENT OF UNPAID TAXES

- (a) When any person liable for the tax imposed by this subchapter is unable to furnish evidence of sufficient tax payments and stamp purchases to cover cigarettes which were sold, used, stored, received, purchased or possessed by him, a presumption arises that such cigarettes were received, sold, used, stored, purchased or possessed without the proper tax having been paid. The Treasurer shall assess the tax as due and unpaid.
  - (b) The presumption in § 57.19(a) shall not arise if:
    - (1) The person liable for the tax is a dealer and the delivery memorandum described in § 57.17(c) has been timely filed; or
    - (2) The cigarettes are in transit or are in the possession of distributors or public warehouses that have filed notice with the treasurer indicating that the cigarettes are temporarily within the town and will be sent to consignees or purchasers outside the town in the normal course of business.
- (c) Any person liable for the tax shall be notified by certified mail of the tax, penalty and interest assessed, all of which shall be due within ten days after notice has been received.

### § 57.110 DISPLAY OF STAMPS; SEIZURE

(a) Stamps shall be placed upon each package of cigarettes in a manner readily visible to the purchaser and shall be affixed so that their removal will require water or steam.

- (b) Any vending machine located within the Town containing cigarettes upon which the stamp has not been affixed or containing cigarettes placed in a manner that does not allow visual inspection for the stamp shall be presumed to contain untaxed cigarettes in violation of this subchapter. If a vending machine does not allow for visual inspection of the stamp, the Treasurer is authorized to direct the owner to open the machine to determine whether the cigarettes are stamped.
- (c) Any cigarettes, coin-operated vending machines, counterfeit stamps, or other property found in violation of this subchapter, other than motor vehicles, shall be declared contraband goods and may be seized by the Treasurer. In addition to any tax due, the dealer or other person liable for the tax possessing such untaxed cigarettes shall be subject to the civil and criminal penalties provided under this subchapter.
- (d) In lieu of seizure, the Treasurer may seal vending machines to prevent the continued illegal sale or removal of such cigarettes. The removal of such seal from a vending machine by any unauthorized person shall be a violation of this subchapter. Nothing in this subchapter prevents the seizure of any vending machine after it is sealed.
- (e) All cigarette vending machines shall be plainly marked with the name, address and telephone number of the owner.
- (f) Any seized cigarettes, vending machines or other property used in the furtherance of any illegal evasion of the tax may be disposed of by sale or other method deemed appropriate by the treasurer after all administrative appeals have been exhausted by the aggrieved party.
- (g) No credit from any sale of cigarettes, vending machines or other seized property shall be allowed toward any tax, penalties or interest assessed.

### § 57.111 NOTICES AND HEARINGS

- (a) Notice of sealing or seizure shall be sent to any party having a known property interest therein by certified mail within 24 hours of the sealing or seizure. When such parties are not known, notice shall be posted to the door or wall of the room or building that contained the sealed or seized property. All notices shall include procedures for appeal and the defenses available under this section.
- (b) Any person assessed by the Treasurer with a cigarette tax, penalty, and interest or any person whose cigarettes, vending machines and other property have been sealed or seized may appeal the action to the Town Manager. Appeals must be made in writing and received within ten days of the receipt of notice by the aggrieved party. The appeal must state the reasons why the action taken was in error and may request the correction of an assessment and the return of sealed or seized property. Within five days after receipt of the appeal, the Town Manager shall notify the aggrieved party by certified mail of a date and time for the informal presentation of evidence at a hearing to be held within 15 days of the date notification is mailed. A request for a

hearing will be denied if the assessed tax, penalties and interest have not been paid or if the request is untimely. The Town Manager shall notify the aggrieved party by certified mail within five days of the hearing stating whether the appeal has been granted or denied. The aggrieved party may appeal the decision of the Town Manager by filing a petition for review with the Circuit Court of Rockingham County within 30 days.

- (c) An appeal may be granted if the aggrieved party shows by a preponderance of the evidence that:
  - (1) The illegal sale or use of seized cigarettes, vending machines, or other property was not intentional;
  - (2) The seized cigarettes, vending machines, or other property was in the possession of a person other than the aggrieved without consent at the time of the seizure or sealing; or
  - (3) The aggrieved was authorized to possess the untaxed cigarettes.
- (d) When an appeal is granted, relief shall include a refund of the tax erroneously assessed with any interest and penalties paid thereon and the return of any cigarettes, vending machines or other property sealed or seized.

# § 57.112 REFUND FOR UNUSED OR DAMAGED STAMPS

- (a) Any person that acquires stamps from the Treasurer is entitled to a refund of the face value of the stamps if the person returns the stamps to the Treasurer with an affidavit stating that the stamps were acquired by such person but not used because the person ceased to be engaged in a business necessitating the use of the stamps, or because damage renders the stamps unusable.
- (b) Any refund for unused or damaged stamps may be made on vouchers approved by the Treasurer. The refunds shall be charged against the sums collected for the sale of stamps. Payment to the person requesting the refund shall be made within 30 days of the request.

# § 57.113 INSPECTION OF RECORDS AND DELEGATION OF DUTIES

The Treasurer is authorized to examine the books, records, invoices, papers and all cigarettes in and upon any premises where placed, stored, sold, offered for sale, or displayed for sale. The Treasurer is authorized to delegate any of the powers and duties set out in this subchapter to one or more deputies or assistants, except as may be prohibited by law.

### § 57.114 SPECIFIC VIOLATIONS ENUMERATED

It shall be a violation of this subchapter for any person:

- (a) To perform any act or fail to perform any act for the purpose of evading any tax imposed by this subchapter or for any dealer or seller, with intent to violate any provision of this subchapter, to fail or refuse to perform any of the duties imposed upon him under the provisions of this subchapter, or to fail or refuse to obey any lawful order issued under this subchapter.
- (b) To falsely or fraudulently make, forge, alter or counterfeit any stamp, invoice or reports, to procure or cause to be made, forged, altered or counterfeited any such stamp, or to knowingly and willfully alter, publish, pass or tender as true any false, altered, forged or counterfeited stamp or stamps.
- (c) To sell, offer for sale, authorize or approve the sale of any cigarette upon which the Town stamp has not been affixed.
- (d) To reuse or refill any package for which the tax has been paid with cigarettes for which the tax has not been paid.
- (e) To remove from any package any stamp with intent to reuse or cause the same to be reused.
- (f) To buy, sell, offer for sale or give away any used, removed, altered or restored stamps to any person.
- (g) To reuse any stamp which has theretofore been used for evidence of the payment of any tax prescribed by this subchapter.
  - (h) To sell or offer to sell any stamp provided for herein, except as to the Treasurer.
- (i) To possess, store, use, authorize or approve the possession, storage or use of cigarettes in quantities of more than 60 packages upon which stamps have not been affixed unless:
  - (1) The person is a seller in the process of affixing such stamps, taking the action required under § 57.17(a) or has been in possession of the cigarettes for less than 48 hours and has not offered them for sale; or
  - (2) The person is a dealer and the delivery memorandum described in § 57.17(c) has been timely filed.
- (j) To transport, authorize or approve the transportation of any cigarettes, in quantities of more than 60 packages into or within the Town upon which the stamp has not been affixed, if they are:

- (1) Not accompanied by a bill of lading or other document indicating the true name and address of the consignor or seller and the consignee or purchaser and the brands and quantity of cigarettes transported; or
- (2) Accompanied by a bill of lading or other document which is false or fraudulent in whole or part.

### § 57.115 VIOLATIONS

- (a) In addition to the other penalties for nonpayment of tax imposed under this Title 57, if any Seller engage in fraud or evasion of the tax imposed by this subchapter, there shall be imposed a civil penalty of 50% of the tax due.
- (b) Any person who fails to pay the tax imposed by this subchapter shall be guilty (i) of a Class 3 misdemeanor if the amount of tax not paid is \$1,000 or less, or (ii) of a Class 1 misdemeanor if the amount of tax not paid is more than \$1,000; and, in either instance, upon due conviction thereof, shall be subject the penalties prescribed in state law for such misdemeanors.

### § 57.116 REGULATIONS

The Treasurer may issue reasonable regulations, not arbitrary or capricious, that are necessary or convenient to the administration of this subchapter and the collection of the taxes imposed herein.

### SUBCHAPTER B – LICENSE TAXES ON ALCOHOLIC BEVERAGES

### § 57.21 LEVY OF LICENSE TAX

The Town Council hereby levies a license tax on the sale of alcoholic beverages, in the amounts shown on the schedule to the Town Code.

# § 57.22 LICENSE REQUIRED

No person shall sell an alcoholic beverage within the Town without having first obtained a license as provided in this subchapter. No license shall issue to a person who does not also hold a valid state alcoholic beverage license. A separate license shall be required for each place of business.

### § 57.23 PENALTY

In addition to the penalties provided otherwise in this title, any person who sells an alcoholic beverage without a license shall be guilty of a Class 3 misdemeanor, and upon due conviction thereof, shall be fined an amount not to exceed \$500. Any person who sells an alcoholic beverage without a license and with the specific intent of evading the license taxes imposed in

this subchapter shall be guilty of a Class 2 misdemeanor and upon due conviction thereof, shall be fined an amount not to exceed \$1,000 or imprisoned for not more than six months, or both.

### **SUBCHAPTER C – UTILITIES**

# § 57.31 **DEFINITIONS**

For the purposes of this subchapter:

- (a) <u>Consumer</u> means every person who, individually or through agents, employees, officers, representatives or permitees, makes a taxable purchase of electricity or natural gas services in the Town.
- (b) <u>Gas utility</u> means a public utility authorized to furnish natural gas service in Virginia.
- (c) <u>CCF</u> means the volume of gas at standard pressure and temperature in units of 100 cubic feet.
- (d) <u>Kilowatt hours (kWh) delivered</u> means 1000 watts of electricity delivered in a one-hour period by an electric provider to an actual consumer, except that in the case of eligible customer-generators (sometimes called cogenerators) as defined under state law, it means kWh supplied from the electric grid to such customer-generators, minus the kWh generated and fed back to the electric grid by such customer-generators.
  - (e) Person means any individual, corporation, company, association or other entity.
- (f) <u>Pipeline distribution company</u> means a person, other than a pipeline transmission company which transmits, by means of a pipeline, natural gas, manufactured gas or crude petroleum and the products or byproducts thereof to a purchaser for purposes of furnishing heat or light.
- (g) <u>Residential consumer</u> means the owner or tenant of property used primarily for residential purposes, including but not limited to, apartment houses and other multiple-family dwellings.
- (h) <u>Service provider</u> means a person who delivers electricity to a consumer or a gas utility or pipeline distribution company which delivers natural gas to a consumer.
- (i) <u>Used primarily</u> relates to the larger portion of the use for which electric or natural gas utility service is furnished.

### § 57.32 ELECTRIC UTILITY CONSUMER TAX

- (a) <u>Levy</u>. There is hereby imposed and levied a monthly tax on each purchase of electricity delivered to consumers by a service provider, classified as determined by such provider, at the rates that appear on the rate schedule to the Town Code.
- (b) <u>Exemptions.</u> The following consumers of electricity are exempt from the tax imposed by this section § 57.32:
  - (i) Any public safety agency as defined in state law.
  - (ii) Any church or religious body exempt from real property taxation;
- (iii) The United States of America, the Commonwealth of Virginia and the political subdivisions thereof, including the Town.
- (c) <u>Billing, collection and remittance of tax</u>. The service provider shall bill the electricity consumer tax to all users who are subject to the tax and to whom it delivers electricity and shall remit the same to this jurisdiction on a monthly basis. Such taxes shall be paid by the service provider to the Town in accordance with state law. If any consumer receives and pays for electricity but refuses to pay the tax imposed by this section, the service provider shall notify the Treasurer of the name and address of such consumer. If any consumer fails to pay a bill issued by a service provider, including the tax imposed by this section, the service provider must follow its normal collection procedures and upon collection of the bill or any part thereof must apportion the net amount collected between the charge for electric service and the tax and remit the tax portion to the Town. Any tax paid by the consumer to the service provider shall be deemed to be held in trust by such provider until remitted to the Town.
- (d) Computation of bills not on monthly basis. Bills shall be considered as monthly bills for the purposes of this paragraph if submitted 12 times per year of approximately one month each. Accordingly, the tax for a bi-monthly bill (approximately 60 days) shall be determined as follows: (i) the kWh will be divided by 2; (ii) a monthly tax will be calculated using the rates set forth above; (iii) the tax determined by (ii) shall be multiplied by 2; (iv) the tax in (iii) may not exceed twice the monthly "maximum tax".

### § 57.33 NATURAL GAS UTILITY CONSUMER TAX

- (a) <u>Levy</u>. There is hereby imposed and levied a monthly tax on each purchase of natural gas delivered to consumers by pipeline distribution companies and gas utilities classified by "class of consumers" as such term is defined in state law, at the rates that appear on the rate schedule to the Town Code.
- (b) <u>Exemptions</u>. The following consumers of natural gas shall be exempt from the tax imposed by this section § 57.33:

- (i) Any public safety agency as defined in state law;
- (ii) Any church or religious body exempt from real property taxation;
- (iii) The United States of America, the Commonwealth of Virginia and the political subdivisions thereof, including the Town.
- gas consumer tax to all users who are subject to the tax and to whom it delivers natural gas and shall remit the same to this jurisdiction on a monthly basis. Such taxes shall be paid by the service provider to the Town in accordance with state law. If any consumer receives and pays for natural gas billed but refuses to pay the tax imposed by this section, the service provider shall notify the Treasurer of the name and address of such consumer. If any consumer fails to pay a bill issued by a service provider, including the tax imposed by this section, the service provider must follow its normal collection procedures and upon collection of the bill or any part thereof must apportion the net amount collected between the charge for electric service and the tax and remit the tax portion to the Town. Any tax paid by the consumer to the service provider shall be deemed to be held in trust by such provider until remitted to the Town.
- (d) Computation of bills not on monthly basis. Bills shall be considered as monthly bills for the purposes of this ordinance if submitted 12 times per year of approximately one month each. Accordingly, the tax for a bi-monthly bill (approximately 60 days) shall be determined as follows: (i) the CCF will be divided by 2; (ii) a monthly tax will be calculated using the rates set forth above; (iii) the tax determined by (ii) shall be multiplied by 2; (iv) the tax in (iii) may not exceed twice the monthly "maximum tax".

# § 57.34 CONSUMPTION TAX; ELECTION TO RECEIVE PAYMENTS

The Town Council elects to receive payments of the local consumption tax on pipeline distribution companies, gas utilities, and consumers of electricity as though the Town had enacted a license tax on such pipeline distribution companies, gas suppliers gas utilities, and electric suppliers in an amount of one-half of one percent of the gross receipts thereof within the Town, all as defined in Chapter 54, and up to the limits that appear on the rate schedule to the Town Code.

Editor's Note – For additional information, see Va. Code § 15.2-2901(F).

### § 57.35 MAINTENANCE OF RECORDS

Each service provider shall keep complete records showing all purchases in the Town, which records shall show the price charged against each purchaser with respect to each purchase, the date thereof, the date of payment thereof and the amount of tax imposed under this subchapter. Such records shall be kept open for inspection by the duly authorized agents of the Town at

reasonable times, and the duly authorized agents of the Town shall have the right, power and authority to make such transcripts thereof during such times as they may deem reasonably necessary for the fair and efficient collection of the tax imposed under this subchapter.

#### SUBCHAPTER D – BANK FRANCHISE TAX

### § 57.41 **DEFININTIONS**

For the purposes of this subchapter, the terms "bank" and "net capital" shall have the same definitions as in state law relating to the state bank franchise tax.

# § 57.42 LEVY

The Town Council hereby levies upon each bank located within in the boundaries of the Town a tax on net capital equaling 80% of the state rate of taxation for each \$100 of net capital. If any bank has offices located in the Town and any other one or more political subdivisions, including cities, towns, and counties, the tax imposed under this subchapter shall be imposed only upon such proportion of the taxable value of net capital as the total deposits of the bank, or offices located in the Town, bears to the total deposits as of the end of the preceding year. For the purposes of this paragraph, offices located within the Town shall be deemed not within Rockingham County.

# § 57.43 FILING OF RETURNS; PAYMENT OF TAX

Each bank having its principal office located in the Town as of January 1 of each year shall prepare and file with the Treasurer, on or before March 1 of each year, a return in duplicate which shall set forth the tax on net capital as computed under this subchapter. The return shall be in the form prescribed by the Virginia Department of Taxation. The Treasurer shall certify a copy of the bank's return and schedules and transmit such certified copy to the Virginia Department of Taxation. Additionally, a copy of the real estate deduction schedules and the apportionment for branch offices shall be filed with the Treasurer for each bank having an office in the Town. The return shall set forth the tax on net capital owing the Town, computed as described in this subchapter. Each bank owing tax under this chapter shall pay such tax to the Town before June 1 of each year.

### **CHAPTER 58: TRANSIENT OCCUPANCY TAX**

## § 58.01 **DEFINITIONS**

The following words, terms and phrases, when used in this Chapter, shall have the meanings ascribed to them in this section, except where the context clearly indicates a different meaning:

Accommodation means any public or private hotel, inn, hostelry, short-term rental, motel or rooming, boarding or lodging house within the Town offering lodging as defined in this section, for compensation, to any transient as defined in this section.

*Hotel* means any structure or group of structures for rent or for hire that is primarily kept, used, maintained, advertised, or held out to the public as a place where sleeping accommodations are offered to transients in return for compensation.

Lodging means the rental of room or space to any transient for compensation, in an accommodation as defined in this section, by a lodging provider or lodging intermediary, or the occupancy of such room or space by such transient. If the charge for the accommodation made by any person to such transient includes any charge for meals, parking or other services not related to the occupancy of the room in addition to lodging or the use of such room or space, then such portion of such total charge as represents only room or space rental shall be distinctly set out and billed to such transient by such person as a separate item.

Lodging fee means the room charge less the discount room charge, if any, provided that the lodging fee shall not be less than zero dollars (\$0.00).

Lodging intermediary means any person other than a lodging provider that facilitates the sale of an accommodation, charges a room charge to the customer, and charges a lodging fee to the customer, which fee it retains as compensation for facilitating the sale. For purposes of this definition, "facilitates the sale" includes brokering, coordinating, or in any other way arranging for the purchase of the right to use accommodations via a transaction directly, including via one (1) or more payment processors, between a customer and a lodging provider.

Lodging provider means any person that furnishes accommodations to the general public for compensation. The term "furnishes" includes the sale of use or possession or the sale of the right to use or possess.

Short-term rental means any building, structure, or unit, on the same tax parcel, sharing the same mailing address, that is used, or is intended to be used, as a residence or home for one (1) or more persons available for rent or for hire to transients.

*Transient* means any person who, for a period of fewer than thirty (30) consecutive days, either at their own expense or at the expense of another, obtains lodging for which a charge is made at an accommodation, as defined in this section.

# § 58.02 IMPOSITION OF TAX; EXEMPTION

- (a) There is hereby imposed and levied upon every transient obtaining or occupying lodging within the Town, in addition to all other taxes and fees of every kind now imposed by law, a tax on the total price paid for the lodging by the transient, or on the transient's behalf. The tax is a percentage of such price, as shown on the rate schedule to the Town Code.
- (b) No tax shall be payable under this Chapter on any charge for lodging in, and during care or treatment in, any hospital, medical clinic, nursing or convalescent home, extended health care facility, sanatorium or sanitorium, home for the aged, infirmed, orphaned, disabled, or mentally retarded or other like facility; or in any educational institution.

# § 58.03 COLLECTION OF TAX LODGING PROVIDER OR LODGING INTERMEDIARY

- (a) For any lodging not facilitated by a lodging intermediary, the lodging provider shall collect the tax levied pursuant to this article from the transient, or from the person paying for the lodging, at the time that payment for the lodging is made.
- (b) For any lodging facilitated by a lodging intermediary, the lodging intermediary shall be deemed to have made the sale of the lodging and is responsible for collecting the tax levied for the lodging from the transient or the person paying for the lodging, at the time that payment for the lodging is made. When the lodging occurs at a hotel, the lodging intermediary shall remit the taxes on the lodging fee to the city and remit the remainder, if any, to the hotel, which shall directly remit said remaining tax to the city. When the lodging occurs at a short-term rental, as defined in this article, or any other accommodation that is not a hotel, the lodging intermediary is responsible for remitting the full amount of tax to the city.
- (b) The taxes collected by any person shall be deemed to be held in trust for the Town by the person required to collect them, until they have been remitted to the Town as provided in this Chapter.

### § 58.04 GRATUITIES AND SERVICE CHARGES

Where a transient occupant provides a gratuity for an employee of a transient occupancy provider, and the amount of the gratuity is wholly in the discretion of the occupant, the gratuity is not subject to the tax imposed by this chapter, whether paid in cash to the employee or added to the bill and charged to the purchaser's account, provided in the latter case, the full amount of the gratuity is turned over to the employee by the provider. An amount or percent, whether designated as a gratuity, tip or service charge, that is added to the price of the transient

occupancy by the provider, and required to be paid by the purchaser, as a part of the consideration for the transient occupancy, is subject to the tax imposed by this chapter to the extent that such gratuity, tip, or service charge (a) exceeds twenty percent (20%) of the consideration for the transient occupancy or (b) is not remitted by the seller to the employee providing such services.

# § 58.05 REPORT OF TAXES COLLECTED; REMITTANCE; PRESERVATION OF RECORDS; CIVIL PENALTIES FOR FAILURE TO FILE

- (a) It shall be the duty of every person that this Chapter requires to collect taxes, to pay to the Town the taxes imposed by this Chapter, and to make a report thereof setting forth such information as the Treasurer may prescribe and require, including in the aggregate all consideration taxable under this chapter, the amount charged the occupant for each such purchase, the date thereof, the taxes collected thereon and the amount of tax that this Chapter requires to be collected. If a person, including a lodging intermediary, is collecting taxes from, or on behalf of, multiple accommodations, the report shall also be sufficient to identify the lodging charges and tax owed on lodging at each individual accommodation, including the accommodation's address and, in cases where a lodging intermediary is responsible for collecting and remitting the taxes, information sufficient to identify the lodging provider connected to the accommodation.
- (b) Reports and tax payments for the months of January, February and March shall be made on or before the following April 20th. Reports and tax payments for the months of April, May, and June shall be made on or before the following July 20th. Reports and tax payments for the months of July, August, and September shall be made on or before the following October 20th. Reports and tax payments for the months of October, November, and December shall be made on or before the following January 20th. Lodging providers shall be required to file such reports with the Treasurer on the date described even in the event no tax is due and regardless of whether they collected the tax or if it was done on their behalf by a lodging intermediary.
- (c) Further, every such person shall maintain records supporting the reports required by paragraph (a) of this section, including records sufficient to determine for each stay at an accommodation the quantum of tax collected by a lodging provider or lodging intermediary. Such records shall be kept and preserved for a period of five (5) years. The Treasurer or his duly authorized agents shall have the power to examine such records at reasonable times and without unreasonable interference with the business of such person, for the purpose of administering and enforcing the provisions of this Chapter, and to make transcripts of all or any parts thereof. All records kept under this section should be sufficient to identify each individual accommodation from which the lodging charges were collected, and the tax owed for transient stays at that accommodation, including the accommodation's address and, where taxes were collected and remitted by a lodging intermediary, identifying the lodging provider connected with the accommodation on whose behalf the taxes were collected.

(d) Whenever any person required to collect and remit to the Town any tax that this Chapter imposes shall cease to operate or otherwise dispose of their business, such tax shall immediately become due and payable, and such person shall forthwith make a report and remittance thereof.

# TITLE VII: CRIMINAL CODE

# Chapter

- 71. GENERAL OFFENSES
- 72. MINORS CURFEW
- 73. USE OF PUBLIC PROPERTY
- 74. ANIMALS
- 75. NOISE
- 76. FIRE CODE

### **CHAPTER 71: GENERAL OFFENSES**

## § 71.01 INCORPORATION BY REFERENCE OF STATE CRIMINAL CODE

All of the provisions and requirements of the laws of the State of Virginia contained in Title 18.2 of the Code of Virginia, 1950, as amended and in force from time to time, except (a) those provisions and requirements the violation of which constitute a felony;(b) those provisions and requirements which by their nature can have no application to or within the municipality; and (c) to the extent the Town Council specifically rejects any amendment thereto or portion thereof, are adopted and incorporated in this Ordinance by reference and made applicable within the municipality. References to "the state" contained in the Code provisions and requirements adopted shall be deemed to refer to the Town of Broadway. The provisions and requirements are adopted, mutatis mutandis, and made a part of this Title as fully as though set forth at length. It shall be unlawful for any person, within the municipality, to violate or fail, neglect or refuse to comply with any provision of Title VII, the Criminal Code of the Town of Broadway, as hereby amended and revised by the applicable provisions of Title 18.2 of the Code of Virginia, 1950, as amended, which is adopted by this Ordinance. Except to the extent otherwise described in the Town Code, the penalty for each such violation shall be as prescribed in state law, but in no event shall the penalty imposed for the violation of any provision or requirement adopted exceed the penalty imposed for a similar offense under the incorporated provisions of the Code of Virginia, 1950, as amended. All future amendments to the sections of the Code of Virginia incorporated by reference under the provisions of this paragraph shall be included within the scope of this paragraph automatically upon their effective date, without any formal Town Council action.

### § 71.02 DISCHARGE OF WEAPONS

- (a) Unless exempt under other law, no person shall discharge a firearm within the Town.
- (b) No person shall shoot an arrow from a bow within the Town in such a manner that can reasonably be expected to result in the impact of the arrow on the property of another without the permission of the owner of such other property. For the purposes of this paragraph, "bow" includes all compound bows, crossbows, longbows and recurve bows having a peak draw weight of ten pounds or more. The term "bow" does not include bows which have a peak draw of less than ten pounds or which are designed or intended to be used principally as toys. The term "arrow" means a shaft-like projectile intended to be shot from a bow.
- (c) No person shall shoot a pneumatic gun within the Town except upon private property with the permission of the owner or other legal possessor thereof when conducted with reasonable care to prevent a projectile from crossing the bounds of the property. In such event, to avoid a violation of this paragraph, any operator of a pneumatic gun that is under the age of sixteen (16) must do so under the supervision of a parent, guardian, or other adult supervisor. For the purposes of this paragraph, "pneumatic gun" means any implement, designed as a gun, that will expel a BB or a pellet by action of pneumatic pressure. "Pneumatic gun" includes a

paintball gun that expels by action of pneumatic pressure plastic balls filled with paint for the purpose of marking the point of impact.

- (d) The provisions of this section § 71.02 shall not apply to a facility approved as a shooting range under other applicable law.
- (e) Any person who violates § 71.02(a) is guilty of a Class 1 misdemeanor and upon due conviction thereof, shall be imprisoned for a term not to exceed one year and shall be fined an amount not to exceed \$2,500, or both.
- (f) Any person who violates §§ 71.02(b) or (c) is guilty of a Class 3 misdemeanor and upon due conviction thereof, shall be fined an amount not to exceed \$500.

### § 71.03 WILLFUL DAMAGE OR DEFACEMENT OF PROPERTY

- (a) No person shall engage willful and malicious damage to or defacement of any public buildings, facilities or personal property or of any private buildings, facilities or personal property.
- (b) Any person who violates § 71.03(a) is guilty of a Class 1 misdemeanor and upon due conviction thereof, shall be imprisoned for a term not to exceed one year and shall be fined an amount not to exceed \$2,500, or both. The punishment for any such violation in which the defacement is (i) more than 20 feet off the ground, (ii) on a railroad or highway overpass, or (iii) committed for the benefit of, at the direction of, or in association with any criminal street gang, as that term is defined by state law, shall include a mandatory minimum fine of \$500.
- (c) In the event that a court orders community service for a defendant for the violation of § 71.03(a), under any provision of state law requiring or permitting such community service, then such community service, to the extent feasible, shall include the repair, restoration or replacement of any damage or defacement to property within the Town, and may include clean-up, beautification, landscaping or other appropriate community service within the Town. The Town Manager or his designee shall supervise the performance of any community service work required and report thereon to the court imposing such requirement.

## § 71.04 TRICK OR TREAT

It shall be unlawful for any person to appear on the street, highways, public homes, private homes or public places in the Town to make trick or treat visitation. This provision shall not apply to children 14 years of age and under on Halloween Night or upon such other night as the Town Council designates for Halloween trick or treating. Any person who violates this § 71.04 shall be guilty of a Class 4 misdemeanor and upon due conviction thereof shall be fined an amount not to exceed \$100.

# § 71.05 RIOT ACT

The Chief of Police shall have the power to regulate, restrict or prohibit any assembly of persons or the movement of persons or vehicles if there exists an imminent threat of any civil commotion or disturbance in the nature of a riot which constitutes a clear and present danger. In ordering such regulation, restriction, or prohibition, the Chief of Police shall give the persons so regulated, restricted, or prohibited notice, reasonable under the circumstances, of the conduct so regulated, restricted, or prohibited. Any person who does not obey such regulation, restriction, or prohibition shall be guilty of a Class 1 misdemeanor and, upon due conviction thereof, shall be punished by confinement in jail for not more than twelve months and a fine of not more than \$2,500, either or both.

### § 71.06 **ATTEMPTS**

Every person who attempts to commit an offense which is a misdemeanor shall be punishable by the same punishment prescribed for the offense the commission of which was the object of the attempt.

### **CHAPTER 72: MINORS CURFEW**

### § 72.01 DEFINITIONS

For the purpose of this Chapter, the following terms are defined:

- (a) "Minor" is any person under the age of eighteen (18) years.
- (b) "Parent" is any person having legal custody of a minor as a natural or adoptive parent, as a legal guardian of the minor's person, as a person who stands in *loco parents*, or as a person to whom legal custody has been given by order of court.
- (c) "Street" is a way or place open to the use of the public as a matter of right for purposes of vehicular travel, or in the case of a sidewalk, for pedestrian travel. The term "street" includes cartway or traffic lanes, the curb, the sidewalks, whether paved or unpaved, and any grass plots or other grounds found within the legal right-of-way of a street. The term "street" applies irrespective of what such street is formally named, whether alley, avenue, court, road or otherwise.

### § 72.02 CURFEW

It shall be unlawful for any minor to be or remain in or upon the streets within the Town at night between the hours of 11:00 pm. and 6:00 a.m.

### § 72.03 EXCEPTIONS

This Chapter shall not apply to a minor who is

- (a) accompanied by a parent of the minor;
- (b) accompanied by an adult authorized by a parent of the minor to take the parent's place in accompanying a minor for a designated period of time and purpose within a specified area;
- (c) returning home, by a direct route from (and within thirty minutes of the termination of) a school activity, an organized dance, a theatre or sporting event, or an activity of a religious or other voluntary association;
- (d) on the sidewalk of the place where the minor resides, or the sidewalk or either next door neighbor as long as the neighbor has not communicated an objection to a law enforcement officer;
- (e) on a direct route traveling to or from an emergency errand which shall include, but not be limited to, traveling to obtain the services of a doctor, hospital, fire station, rescue squad, or police department;

- (f) going to or returning home, by a direct route, from a place of employment at which the minor is employed during hours that necessitate the minor's travel on the streets during curfew hours;
- (g) in a motor vehicle with the consent of the minor's parent. This contemplates normal travel such as bona fide movement through the Town and travel beginning or ending in the Town.

### § 72.04 PARENTAL RESPONSIBILITY

It shall be unlawful for a parent having legal custody of a minor knowingly to permit or by inefficient control to allow, the minor to be or remain upon street of the Town under circumstances not constituting a exception to, or otherwise beyond the scope of, this Section. The term "knowingly" includes knowledge which a parent should reasonably be expected to have concerning the whereabouts of a minor in that parent's legal custody.

### § 72.05 PENALTY

Any person who violates §§ 72.02 or 72.04 shall be guilty of a Class 4 misdemeanor and upon due conviction thereof shall be fined an amount not to exceed \$100.

### **CHAPTER 73: USE OF PUBLIC PROPERTY**

### § 73.01 POSTING PROHIBITED

No person shall post or affix, or cause to be posted or affixed, any bills, flyers, or other material on or to any public property, including but not limited to light poles, walls, and buildings.

### § 73.02 OPENING PERMIT REQUIRED

It shall be unlawful for any person, other an authorized official of the Town, to make any opening or entrance in any street, alley, sidewalk, or public way of the Town unless a permit to make the opening has been obtained prior to commencement of the work.

### § 73.03 BARRIERS AROUND EXCAVATIONS

Any person engaged in or employing others in excavating or opening any street, sidewalk, alley, or other public way shall have the excavation or opening fully barricaded at all times to prevent injury to persons or animals.

### § 73.04 UNLOADING ON STREET OR SIDEWALK

No person shall unload any heavy material in the streets of the Town by throwing or letting the material fall upon the pavement of any street, alley, sidewalk, or other public way, without first placing some sufficient protection over the pavement.

### § 73.05 OBSTRUCTIONS

- (a) No person shall obstruct any street, alley, sidewalk, or other public way within the Town by erecting thereon any fence or building, or permitting any fence or building to remain thereon. Each day that any fence or building is permitted to remain upon the public way shall constitute a separate offense.
- (b) No owner, occupant, or person having the care of any building or lot of land, bordering on any street or sidewalk, shall permit such street or sidewalk to be encumbered with barrels, boxes, cans, articles, or substances of any kind, so as to interfere with the free and unobstructed use thereof.

# § 73.06 PENALTY

Any person who violates any provision of this Chapter (other than § 73.07) shall be guilty of a Class 3 misdemeanor and upon due conviction thereof shall be fined an amount not to exceed \$500, and, in addition, may be required to repair or restore any damaged property to its condition prior to the violation.

# § 73.07 PERMIT REQUIRED FOR DISTURBANCE OF PUBLIC RIGHTS OF WAY

- (a) No person shall enter into any repair, alteration, construction, or reconstruction of any type whatever, other than emergency repairs to or maintenance of public utility facilities, within the right-of-way lines of any public highway, without first having obtained a permit for such work from the Town Manager.
- (b) Such permit may require the notification of all emergency services likely to be affected by such repair, alteration, construction or reconstruction; the types of traffic control devices necessary to properly warn the motoring public and provide for reinspection by the Town from time to time and at the conclusion of such repair, alteration, construction, or reconstruction.
- (c) Any person holding such a permit shall be responsible for furnishing and maintaining the required traffic control devices in accord with the Virginia Manual of Uniform Traffic Control Devices for Streets and Highways.
- (d) Any person who violates this section shall be guilty of a Class 4 misdemeanor for the first offence and upon due conviction thereof shall be fined an amount not more than \$250. Upon the second and subsequent violations thereof, such person shall be likewise guilty of a Class 3 misdemeanor and shall be fined an amount not more than \$500.

### **CHAPTER 74: ANIMALS**

### SUBCHAPTER A – DANGEROUS AND VISCIOUS DOGS

### § 74.11 DANGEROUS DOGS

- (a) <u>Definitions</u>. For the purposes of this Chapter:
- (i) "Dangerous dog" means a canine or canine crossbreed that has bitten, attacked, or inflicted injury on a person or companion animal that is a dog or cat, or killed a companion animal that is a dog or cat. When a dog attacks or bites a companion animal that is a dog or cat, the attacking or biting dog shall not be deemed dangerous (i) if no serious physical injury as determined by a licensed veterinarian has occurred to the dog or cat as a result of the attack or bite; (ii) if both animals are owned by the same person; (iii) if such attack occurs on the property of the attacking or biting dog's owner or custodian; or (iv) for other good cause as determined by the court. No dog shall be found to be a dangerous dog as a result of biting, attacking, or inflicting injury on a dog or cat while engaged with an owner or custodian as part of lawful hunting or participating in an organized, lawful dog handling event. No dog that has bitten, attacked, or inflicted injury on a person shall be found to be a dangerous dog if the court determines, based on the totality of the evidence before it, that the dog is not dangerous or a threat to the community.
  - (ii) "Law-enforcement officer" includes a Town Police Officer.
- (iii) "Animal control officer" means a Rockingham County animal control officer.
- (b) Determination of Dangerousness. Any Town law-enforcement officer who has reason to believe that a canine or canine crossbreed within the Town is a dangerous dog shall apply to a magistrate serving Rockingham County for the issuance of a summons requiring the owner or custodian, if known, to appear before the General District Court of Rockingham County at a specified time. The summons shall advise the owner of the nature of the proceeding and the matters at issue. If a law-enforcement officer successfully makes an application for the issuance of a summons, he shall contact the local animal control officer and inform him of the location of the dog and the relevant facts pertaining to his belief that the dog is dangerous; or, he may confine the animal himself in the Town's own facilities until the animal can be surrendered to an animal control officer. The animal control officer shall confine the animal until such time as evidence shall be heard and a verdict rendered. If the animal control officer determines that the owner or custodian can confine the animal in a manner that protects the public safety, he may permit the owner or custodian to confine the animal until such time as evidence shall be heard and a verdict rendered. The court, through its contempt powers, may compel the owner, custodian, or harborer of the animal to produce the animal. If, after hearing the evidence, the court finds that the animal is a dangerous dog, the court shall order the animal's owner to comply with the provisions of this Section § 74.11. The court, upon finding the animal to be a dangerous dog, may order the owner, custodian, or harborer thereof to pay restitution for actual damages to

any person injured by the animal or whose companion animal was injured or killed by the animal. The court, in its discretion, may also order the owner to pay all reasonable expenses incurred in caring and providing for such dangerous dog from the time the animal is taken into custody until such time as the animal is disposed of or returned to the owner. The procedure for appeal and trial shall be the same as provided by law for misdemeanors. Trial by jury shall be provided under the procedures contained in state law. The Town shall be required to prove its case beyond a reasonable doubt.

- (c) <u>Limitations on Determination</u>. No canine or canine crossbreed shall be found to be a dangerous dog solely because it is a particular breed, nor is the ownership of a particular breed of canine or canine crossbreed prohibited. No animal shall be found to be a dangerous dog if the threat, injury, or damage was sustained by a person who was (i) committing, at the time, a crime upon the premises occupied by the animal's owner or custodian; (ii) committing, at the time, a willful trespass upon the premises occupied by the animal's owner or custodian; or (iii) provoking, tormenting, or physically abusing the animal, or can be shown to have repeatedly provoked, tormented, abused, or assaulted the animal at other times. No police dog that was engaged in the performance of its duties as such at the time of the acts complained of shall be found to be a dangerous dog. No animal that, at the time of the acts complained of, was responding to pain or injury, or was protecting itself, its kennel, its offspring, a person, or its owner's or custodian's property, shall be found to be a dangerous dog.
- (d) <u>Minors</u>. If the owner of an animal found to be a dangerous dog is a minor, the custodial parent or legal guardian shall be responsible for complying with all requirements of this section.
- (e) <u>Consequences After Determination</u>. The owner of any animal found to be a dangerous dog shall, within 45 days of such finding, obtain a dangerous dog registration certificate from the Town Treasurer for a fee of \$150, in addition to other fees that may be authorized by law. The Treasurer shall also provide the owner with a uniformly designed tag that identifies the animal as a dangerous dog. The owner shall affix the tag to the animal's collar and ensure that the animal wears the collar and tag at all times. By January 31 of each year, until such time as the dangerous dog is deceased, all certificates obtained pursuant to this paragraph shall be updated and renewed for a fee of \$85 and in the same manner as the initial certificate was obtained. The Treasurer shall post registration information on the Virginia Dangerous Dog Registry.
- (f) Conditions to Keeping Dangerous Dog. All dangerous dog registration certificates or renewals thereof required to be obtained under this Section § 74.11 shall only be issued to persons 18 years of age or older who present satisfactory evidence (i) of the animal's current rabies vaccination, if applicable; (ii) that the animal has been neutered or spayed; and (iii) that the animal is and will be confined in a proper enclosure or is and will be confined inside the owner's residence or is and will be muzzled and confined in the owner's fenced-in yard until the proper enclosure is constructed. In addition, owners who apply for certificates or renewals thereof under this section shall not be issued a certificate or renewal thereof unless they present satisfactory evidence that (a) their residence is and will continue to be posted with clearly visible

signs warning both minors and adults of the presence of a dangerous dog on the property and (b) the animal has been permanently identified by means of electronic implantation. All certificates or renewals thereof required to be obtained under this section shall only be issued to persons who present satisfactory evidence that the owner has liability insurance coverage, to the value of at least \$100,000, that covers animal bites. The owner may obtain and maintain a bond in surety, in lieu of liability insurance, to the value of at least \$100,000.

- dangerous dog shall be confined indoors or in a securely enclosed and locked structure of sufficient height and design to prevent its escape or direct contact with or entry by minors, adults, or other animals. While so confined within the structure, the animal shall be provided proper care according to state law. When off its owner's property, an animal found to be a dangerous dog shall be kept on a leash and muzzled in such a manner as not to cause injury to the animal or interfere with the animal's vision or respiration, but so as to prevent it from biting a person or another animal.
- (h) Notifications to Town and County. The owner shall cause the local animal control officer and the Town police department to be promptly notified of (i) the names, addresses, and telephone numbers of all owners; (ii) all of the means necessary to locate the owner and the dog at any time; (iii) any complaints or incidents of attack by the dog upon any person or cat or dog; (iv) any claims made or lawsuits brought as a result of any attack; (v) chip identification information; (vi) proof of insurance or surety bond; and (vii) the death of the dog.
- (i) <u>Escape</u>; <u>Relocation</u>. After an animal has been found to be a dangerous dog, the animal's owner shall immediately, upon learning of same, cause the local animal control officer and the Town police department to be notified if the animal (i) is loose or unconfined; (ii) bites a person or attacks another animal; or (iii) is sold, is given away, or dies. Any owner of a dangerous dog who relocates to a new address shall, within 10 days of relocating, provide written notice to the appropriate local animal control authority for the old address from which the animal has moved and the new address to which the animal has been moved, and to the Town police department. Any dangerous animal relocating into the Town shall provide the same notification.

### (i) Penalty.

- (I) Any owner or custodian of a canine or canine crossbreed or other animal is guilty of a Class 2 misdemeanor if the canine or canine crossbreed previously declared a dangerous dog pursuant to this Section § 74.11 or the parallel statute of another jurisdiction, or state law, when such declaration arose out of a separate and distinct incident, attacks and injures or kills a cat or dog that is a companion animal belonging to another person. Upon due conviction thereof, such person shall be fined an amount not to exceed \$1,000 or imprisoned for a term not to exceed six months, or both.
- (II) Any owner or custodian of a canine or canine crossbreed or other animal is guilty of a Class 1 misdemeanor if the canine or canine crossbreed previously declared

a dangerous dog pursuant to this Section § 74.11 or the parallel statute of another jurisdiction, when such declaration arose out of a separate and distinct incident, bites a human being or attacks a human being causing bodily injury. Upon due conviction thereof, such person shall be fined an amount not to exceed \$2,500 or imprisoned for a term not to exceed six months, or both.

- (III) The provisions of subsections (I) and (II) shall not apply to any animal that, at the time of the acts complained of, was responding to pain or injury, or was protecting itself, its kennel, its offspring, a person, or its owner's or custodian's property, or when the animal is a police dog that is engaged in the performance of its duties at the time of the attack.
- (IV) The owner of any animal that has been found to be a dangerous dog who willfully fails to comply with the requirements of this Section § 74.11 is guilty of a Class 1 misdemeanor. Upon due conviction thereof, such person shall be fined an amount not to exceed \$2,500 or imprisoned for a term not to exceed six months, or both.
- (V) Whenever an owner or custodian of an animal found to be a dangerous dog is charged with a violation of this section, the animal control officer or a Town police officer shall confine the dangerous dog until such time as evidence shall be heard and a verdict rendered. The court, through its contempt powers, may compel the owner, custodian, or harborer of the animal to produce the animal.
- (k) <u>Disposal of Animal After Conviction</u>. Upon conviction, the court may (i) order the dangerous dog to be disposed of by an animal control officer or (ii) grant the owner up to 45 days to comply with the requirements of this section, during which time the dangerous dog shall remain in the custody of the animal control officer until compliance has been verified. If the owner fails to achieve compliance within the time specified by the court, the court shall order the dangerous dog to be disposed of by the animal control officer. The court, in its discretion, may order the owner to pay all reasonable expenses incurred in caring and providing for such dangerous dog from the time the animal is taken into custody until such time that the animal is disposed of or returned to the owner.
- (l) <u>Fees</u>. All fees collected pursuant to this section, less the costs incurred by the animal control authority in producing and distributing the certificates and tags required by this section and fees due to the State Veterinarian for maintenance of the Virginia Dangerous Dog Registry, shall be paid into a special dedicated fund in the treasury of the Town for the purpose of paying the expenses of any training course required for animal control officers under state law, or may be remitted to Rockingham County for the expenses of the same.
- (m) <u>Section Parallel to Rockingham County Ordinance and State Law;</u> <u>Enforcement.</u> This Section § 74.11 does not displace the provisions of the Rockingham County animal control ordinance within the Town, or the provisions of state law that parallel this Section § 74.11. Violations of this Section § 74.11 may be enforced by both the Town under this Section § 74.11 and by Rockingham County under its animal control ordinance, or by the

Commonwealth under its laws, or by all three. Animal control officers have full authority to enforce this Section § 74.11 on the same basis as Town police officers.

## § 74.12 VICIOUS DOGS

- (a) <u>Definitions</u>. For the purposes of this Chapter, unless otherwise defined:
- (i) "Serious injury" means an injury having a reasonable potential to cause death or any injury other than a sprain or strain, including serious disfigurement, serious impairment of health, or serious impairment of bodily function and requiring significant medical attention.
- (ii) "Vicious dog" means a canine or canine crossbreed that has (i) killed a person, (ii) inflicted serious injury to a person, or (iii) continued to exhibit the behavior that resulted in a previous finding by a court or, on or before July 1, 2006, by an animal control officer of any jurisdiction as authorized by ordinance that it is a dangerous dog, provided that its owner has been given notice of that finding.
  - (iii) "Law-enforcement officer" includes a Town Police Officer.
- (iv) "Animal control officer" means a Rockingham County animal control officer.
- Determination of Viciousness. Any law-enforcement officer or animal control officer who has reason to believe that a canine or canine crossbreed within the Town is a vicious dog shall apply to a magistrate serving Rockingham County for the issuance of a summons requiring the owner or custodian, if known, to appear before the General District Court of Rockingham County at a specified time. The summons shall advise the owner of the nature of the proceeding and the matters at issue. If a law-enforcement officer successfully makes an application for the issuance of a summons, he shall contact the local animal control officer and inform him of the location of the dog and the relevant facts pertaining to his belief that the dog is vicious; or, he may confine the animal himself in the Town's own facilities until the animal can be surrendered to an animal control officer. The animal control officer shall confine the animal until such time as evidence shall be heard and a verdict rendered. The court, through its contempt powers, may compel the owner, custodian, or harborer of the animal to produce the animal. If, after hearing the evidence, the court finds that the animal is a vicious dog, the court shall order the animal euthanized. The court, upon finding the animal to be a vicious dog, may order the owner, custodian, or harborer thereof to pay restitution for actual damages to any person injured by the animal or to the estate of any person killed by the animal. The court, in its discretion, may also order the owner to pay all reasonable expenses incurred in caring and providing for such vicious dog from the time the animal is taken into custody until such time as the animal is disposed of. The procedure for appeal and trial shall be the same as provided by law for misdemeanors. Trial by jury shall be provided under the procedures contained in state law. The Town shall be required to prove its case beyond a reasonable doubt.

- (c) Exceptions. No canine or canine crossbreed shall be found to be a vicious dog solely because it is a particular breed, nor is the ownership of a particular breed of canine or canine crossbreed prohibited. No animal shall be found to be a vicious dog if the threat, injury, or damage was sustained by a person who was (i) committing, at the time, a crime upon the premises occupied by the animal's owner or custodian; (ii) committing, at the time, a willful trespass upon the premises occupied by the animal's owner or custodian; or (iii) provoking, tormenting, or physically abusing the animal, or can be shown to have repeatedly provoked, tormented, abused, or assaulted the animal at other times. No police dog that was engaged in the performance of its duties as such at the time of the acts complained of shall be found to be a vicious dog. No animal that, at the time of the acts complained of, was responding to pain or injury or was protecting itself, its kennel, its offspring, a person, or its owner's or custodian's property, shall be found to be a vicious dog.
- (d) <u>Penalty</u>. Any owner or custodian of a canine or canine crossbreed or other animal whose willful act or omission in the care, control, or containment of a canine, canine crossbreed, or other animal is so gross, wanton, and culpable as to show a reckless disregard for human life and is the proximate cause of such dog or other animal attacking and causing serious injury to any person is guilty of a Class 1 misdemeanor, and upon due conviction thereof, such person shall be fined an amount not to exceed \$2,500 or imprisoned for a term not to exceed six months, or both.
- (e) <u>Section Parallel to Rockingham County Ordinance and State Law; Enforcement.</u> This Section § 74.12 does not displace the provisions of the Rockingham County animal control ordinance within the Town, or the provisions of state law that parallel this Section § 74.12. Violations of this Section § 74.12 may be enforced by both the Town under this Section § 74.12 and by Rockingham County under its animal control ordinance, or by the Commonwealth under its laws, or by all three. Animal control officers have full authority to enforce this Section § 74.12 on the same basis as Town police officers.

### SUBCHAPTER B – OTHER PROVISIONS REGARDING ANIMALS

## § 74.21 DOGS RUNNING AT LARGE

No owner or other person having custody, possession, or control of a dog shall permit such dog to run at large, or to remain unconfined, unrestricted or not penned up. A dog shall be deemed to run at large while roaming, running or self-hunting off the property of its owner or custodian and not under its owner's or custodian's immediate control. Any person who violates this Section § 74.21 shall pay a civil penalty of \$150; provided, however, that if a dog violating this section causes damage to property or injury to person, in lieu of the civil penalty, the owner or custodian thereof shall be guilty of a Class 3 misdemeanor and upon due conviction thereof shall be fined an amount not to exceed \$500, in addition to any proceedings under Subchapter A of this Chapter. This Section § 74.21 may be enforced both by Town police officers and by appropriate Rockingham County officials.

### § 74.22 BARKING DOGS

- (a) <u>Same Constitute a Nuisance</u>. Except as otherwise provided in this Section § 74.22, any person who owns or who has possession of an animal within the Town, and who allows that animal to howl, bark, meow, or squawk in such a manner that creates noise that is plainly audible at least once a minute for ten (10) consecutive minutes (1) inside the confines of the dwelling unit, house or apartment of another or (2) at fifty (50) or more feet from the animal, is maintaining a public nuisance.
- (b) <u>Penalty</u>. Such person shall be guilty of a Class 4 misdemeanor and upon due conviction thereof shall be fined an amount that does not exceed \$100. In addition, such public nuisance may be abated by any remedy available under Title 9 of this Code or under other applicable law.
- (c) <u>Exception</u>. All dog kennels legally operating within the Town shall be exempt from this Section 74.22.
- (d) <u>Definition</u>. For purposes of this section, "plainly audible" has the same meaning as in Chapter 75 of this Code.

### § 74.23 LIVESTOCK AND FOWL

No person shall keep livestock (as defined in the Land Development Regulations) within the Town except as permitted in the Land Development Regulations. No person shall permit any livestock to run at large within the Town. Any person who violates this Section § 74.23 shall be guilty of a Class 4 misdemeanor and upon due conviction thereof shall be fined an amount that does not exceed \$100. Each day of keeping livestock in violation of this section shall be a separate offense.

# § 74.24 ADOPTION OF ROCKINGHAM COUNTY ANIMAL REGULATIONS BY REFERENCE

Except to the extent that the Town has adopted its own ordinances regarding a particular subject in this Code, and to the full extent permitted by state law, the Town adopts the animal ordinances of Rockingham County. Such ordinances apply within the Town, and shall be administered by appropriate Rockingham County officials such that Rockingham County officials shall issue all licenses required and receive all fees for such licenses. Such ordinances may be enforced both by Town police officers and by appropriate Rockingham County officials.

### **CHAPTER 75: NOISE**

### § 75.01 DECLARATION OF POLICY

At certain levels, noise can be detrimental to the health, welfare, safety and quality of life of inhabitants of the town, and in the public interest noise should be restricted. Moreover, certain social gatherings, using amplified sound, create excessive noise and related adverse effects for the citizens of this Town. It is, therefore, the policy of the Town to reduce, and eliminate where possible, excessive noise and related adverse conditions in the community, and to prohibit unnecessary, excessive, harmful and annoying noises from all sources as an exercise of the Town's police power. The Town exercises its power to prohibit both certain specified activities, as described in § 75.03, and to prohibit any other activity that causes noise above a certain level, as described in § 75.04.

# § 75.02 **DEFINITIONS**

The following words, terms and phrases, when used in this Chapter, shall have the meanings ascribed to them in this section, except where the context clearly indicates a different meaning:

Decibel (dB) means a unit for measuring the volume of a sound, equal to twenty (20) times the logarithm to the base ten (10) of the ratio of the pressure of the sound measured to the reference pressure, which is twenty (20) micropascals (twenty (20) micronewtons per square meter).

Dwelling unit means a building or portion thereof designed or intended to be occupied as living quarters by one (1) or more persons including permanent provisions for living, sleeping, eating, cooking and sanitation.

*Motor vehicle* means every vehicle defined as a motor vehicle by § 46.2-100 of the Code of Virginia (1950), as amended.

Owner means the person owning, controlling, or possessing land, premises, or personal property.

*Person* means any individual, partnership, corporation, limited liability company, association, society, club, group of people acting in concert, or organization. This term shall not include the federal, state, county, town, city, or local government, or any agency or institution thereof.

*Plainly audible* means any sound that can be detected by a person using his or her unaided hearing faculties, provided that a sound shall be determined to be audible even if specific words or phrases cannot be distinguished. When music is involved, the detection of rhythmic bass tones shall be sufficient to be considered plainly audible sound.

*Property boundary* means an imaginary line along the ground surface, and its vertical extension, which separates the real property owned, leased or otherwise legally controlled by one (1) person from that owned, leased or otherwise legally controlled by another person, including intra-

building real property divisions such as party walls, walls between dwelling units, or condominium unit boundaries.

Residential area means any property zoned for residential use under the Town's Land Development Regulations.

*Restaurant* means any building or structure where in the normal course of business food or drink is available for eating on the premises, in consideration for payment. For purposes of this chapter, the term restaurant includes, but is not limited to, bars, lounges, taverns, coffee shops and cafes.

Sound amplifying equipment means any machine or device used for the amplification of the human voice, music, or other sound. This term shall not include warning devices on authorized emergency vehicles, horns or other warning devices on other vehicles used only for traffic safety purposes.

# § 75.03 SPECIFIC ACTIVITIES PROHIBITED

No person shall cause or permit to be caused any of the following prohibited sounds or noises:

- (a) Large party nuisance. The creation of plainly audible sound that emanates from a gathering of ten (10) or more people, whether the gathering is completely contained within a structure or spills outdoors into balconies, yards, common areas, parking lots, or other outdoor spaces, and is plainly audible (i) across a property boundary or (ii) at a distance of one hundred (100) feet or more from its source and on property other than that which the sound originates.
- (b) Sound-producing and sound-reproducing devices. Except for commercial establishments located in areas zoned business, the use, operation or playing of any radio, phonograph, television, record, compact disc, tape, digital music, MP3 or DVD player, musical instrument, loudspeaker, sound amplifier, phone, tablet, or other machine or device capable of producing or reproducing sound, regardless of whether such sound-producing or sound-reproducing machine or device is located inside of a structure or outside of or on a structure, in such a manner or with such volume or duration that it is plainly audible between 10:00 p.m. and 7:00 a.m.:
  - (1) Inside the confines of the dwelling unit, house or apartment of another person; or
  - (2) In a residential area, at one hundred (100) or more feet from the device.
- (c) *Noisy animals*. Allowing any animal to cause any sound or noise such that it is plainly audible:
  - (1) Inside the confines of the dwelling unit, house or apartment of another person at least once a minute for ten (10) consecutive minutes; or

- (2) At one hundred (100) or more feet from the animal at least once a minute for ten (10) consecutive minutes.
- (d) Peddlers and hawkers. Yelling, shouting, whistling, screaming or crying for the purpose of attracting attention to a performance, show, sale or display of merchandise between the hours of 10:00 p.m. and 7:00 a.m. on any public street, sidewalk or parking lot or any privately owned street, sidewalk or parking lot open to the public, except to summon aid in an emergency.
- (e) Amplified sound from vehicles. Playing, using or operating, or permitting the playing, use or operation of, any radio, stereo, tape player, compact disc player, MP3 player, loudspeaker, tablet, phone, or other electronic device or mechanical equipment used for the amplification of sound, which is located on or within a motor vehicle and which is plainly audible from outside the motor vehicle at a distance of fifty (50) feet or more. The provisions of this subsection shall not apply to the playing of music or jingles by an ice cream truck or similar mobile food service vehicle, provided such vehicle may emit sounds otherwise prohibited by this subsection only between the hours of 7:00 a.m. and 10:00 p.m.
- (f) Lawn care activities. Creating any sound or noise plainly audible across a property line in a residential area between 10:00 p.m. and 7:00 a.m. in connection with lawn care, leaf removal, gardening, tree maintenance or removal or other landscaping, lawn or timbering activities. The provisions of this subsection shall not apply to (a) sound or noise generated by the maintenance of recreational facilities such as golf courses and ball or playing fields; or (b) to permitted commercial agricultural activities taking place in a residential area.
- (g) Horns, whistles, etc. Sounding or permitting the sounding of any horn, whistle or other auditory sounding device on or in any motor vehicle on any public right-of-way or public property, or on any private property accessible to the public (such as a business parking lot) except as a warning of danger, for more than ten (10) seconds, or more than three (3) times in any one minute.
- (h) Yelling, shouting, etc. Yelling, shouting, whistling or singing between the hours of 10:00 p.m. and 7:00 a.m. in such a manner as to permit sound to be plainly audible across a property boundary.
- (i) Vehicle noise other than from sound-producing and sound-reproducing devices. Operation of a motor vehicle or operation of a motorcycle within the town with a gutted muffler, muffler cutout, straight exhaust, or without an exhaust in good working order, as provided under state law.

# § 75.04 GENERAL ACTIVITIES PROHIBITED

In addition to, and not in limitation of the specific prohibitions of § 75.03, no person shall operate or permit to be operated any noise source which generates a sound level exceeding the

limits set forth in this section. The provisions of this section shall not apply to restaurants; rather, § 75.06 shall apply.

- (a) *Nighttime*. No person shall permit, operate or cause any source of sound to create a sound level in excess of fifty-five (55) dBA when measured at or outside the property boundary between the hours 10:00 p.m. and 7:00 a.m.
- (b) *Daytime*. No person shall permit, operate or cause any source of sound to create a sound level in excess of seventy (70) dBA when measured at or outside the property boundary between the hours 7:00 a.m. and 10:00 p.m.

# § 75.05 MEASUREMENT OF SOUND IN MULTIFAMILY DWELLINGS

Whenever it is necessary to determine the volume of sound for the purposes of §§ 75.03 or 75.04 inside a Two Family Dwelling or a Multi-Family Dwelling (as that term is defined in the Town's Land Development Regulations), the person making such determination may listen or take measurements to determine sound levels from common areas within or outside the structure or from other dwelling units within the structure, when requested to do so by the owner or tenant in possession and control thereof. Such measurement shall be taken at a point at least four (4) feet from the wall, ceiling or floor nearest the noise source, with doors to the receiving area closed and windows in the normal position for the season. This section applies to both measurements of the decibel level of sound, and to the determination whether a sound is plainly audible.

#### § 75.06 RESTARAUNTS

No person shall permit, operate or cause any source of sound to create a sound level emanating from a restaurant during operating hours in excess of sixty-five (65) dBA when measured at or outside the property boundary.

#### § 75.07 EXCEPTIONS

The following activities or sources of noise shall be exempt from the prohibitions set forth in this chapter:

- (a) Band performances or practices, athletic contests or practices and other school-sponsored activities on the grounds of public or private schools;
  - (b) Athletic contests and other officially sanctioned activities in Town parks;
  - (c) Bells or carillons;
  - (d) Activities for which the regulation of noise has been preempted by federal law;
  - (e) Public and private transportation, refuse collection and sanitation services;

- (f) Sound which is necessary for the protection or preservation of property or the health, safety, life or limb of any person;
- (g) Radios, sirens, horns, and bells on police, fire, or other emergency response vehicles;
  - (h) Events that the Town or the Broadway Hometown Partnership conducts;
- (i) Fire alarms and burglar alarms, including motor vehicle alarms, prior to the giving of notice and a reasonable opportunity for the owner or person in possession of the premises served by any such alarm to turn off the alarm;
  - (j) Locomotives and other railroad equipment, and aircraft;
  - (k) The striking of clocks;
- (l) The activities (including military or law enforcement activities) of any local, regional, state, or federal government; and
- (m) Any event for which a public assemblage permit has been issued under Chapter 97 of this Code, or any event for which a parade permit has been issued under Chapter 98 of this Code; provided, however, that the terms of any such permit may include a requirement that participants in the event must comply with this Chapter, and in such event, the exception contained within this clause (m) shall not apply.

# § 75.08 MEASUREMENT OF SOUND BY DECIBEL

- (a) The decibel level of any noise may be measured by the use of a sound level meter which measures sound pressure levels. Such measurements shall be accepted as prima facie evidence of the level of noise at issue in any court or legal proceeding to enforce the provisions of this Chapter.
  - (b) The accuracy of the sound level meter may be tested by a calibrator.
- (c) An individual operating a sound level meter pursuant to the provisions of this section shall issue a certificate to indicate:
  - (1) That the sound level meter used to take the decibel level reading was operated in accordance with the manufacturer's specifications;
  - (2) That the Town has on file a sworn report which states that the sound level meter has been tested within the past twelve (12) months and has been found to be accurate;
  - (3) The name of the accused;

- (4) The location of the noise;
- (5) The date and time that the reading was made; and
- (6) The decibel level reading.

Such certificate, when attested by the operator taking the decibel level reading, shall be admissible in court in any criminal or civil proceeding as evidence of the facts therein stated and of the decibel level reading. A copy of the certificate shall be delivered to the accused upon request.

#### § 75.09 PARTY PERMITS

Any person who wishes to inform the Town of a social gathering that has the potential to create noise prohibited under this Chapter, and who wishes to provide contact information to facilitate communication between the organizer of the gathering and the Town, may register such gathering with the Town. Such person shall provide the Town with the location of the gathering, the name and phone numbers of one or more persons responsible for the gathering, and any other information that the Town Manager or his designee may reasonably request. Registration does not excuse compliance with this Chapter; rather, it is intended to facilitate communication between the Town and the organizer of the gathering.

#### § 75.10 PENALTY

Any person who creates any noise in violation of the provisions of this Chapter, shall be guilty of a Class 2 misdemeanor and, upon due conviction thereof, shall be punished by a fine of not less than three hundred dollars (\$300.00) for the first offense, five hundred dollars (\$500.00) for the second offense within a twelve (12) month period, and one thousand dollars (\$1,000.00) for any subsequent offense within the same twelve (12) month period. Each day the violation is committed or permitted to continue shall constitute a separate offense and shall be punishable as such hereunder.

#### § 75.11 NUISANCE REMEDIES AVAILABLE

In addition to the remedies provided in this Chapter, noise prohibited in this Chapter shall constitute a public nuisance. The Town or a private citizen may abate such public nuisance by any remedy available to abate public nuisances under Title 9 of this Code or other applicable law.

# **CHAPTER 76: FIRE CODE**

# § 76.01 ADOPTION OF STATEWIDE FIRE CODE

The Statewide Fire Prevention Code, promulgated by the Commonwealth under state law, shall be in effect in the Town. The Rockingham County Fire Marshal shall enforce such regulations within the Town.

# § 76.02 **PENALTY**

Any person who violates the Statewide Fire Prevention Code shall be guilty of a Class 1 misdemeanor, and upon due conviction thereof, shall be fined an amount not to exceed \$2,500 or imprisoned for a term not to exceed 1 year, or both.

#### **CHAPTER 76: FIRE CODE**

# § 76.01 STATEWIDE FIRE PREVENTION CODE

The Statewide Fire Prevention Code shall be in effect in the Town. Specific provisions of this Chapter shall take precedence over conflicting provisions of such statewide code. No person shall violate any provision of the Statewide Fire Prevention Code that is in effect in the Town.

# § 76. PENALTY; ENFORCEMENT

Any person who violates any provision of the Statewide Fire Prevention Code in effect in the Town, or any other provision of this Chapter, shall be guilty of a Class 1 misdemeanor and upon due conviction thereof, shall be fined an amount not to exceed \$2,500 or imprisoned for a term not to exceed one year, or both. The County Fire Marshall shall have authority to enforce this Chapter.

# TITLE IX: PROPERTY, HEALTH, AND SAFETY

# Chapter

- 91. NUISANCES
- 92. TRASH, GRASS, WEED, AND CLUTTER ACCUMULATION
- 93. DILAPIDATED OR UNSAFE BUILDINGS AND STRUCTURES
- 94. DEFACEMENT OF BUILDINGS AND STRUCTURES
- 95. INOPERABLE MOTOR VEHICLES
- 96. SNOW REMOVAL
- 97. PUBLIC ASSEMBLAGE
- 98. TREES AND SHRUBS ON PUBLIC PROPERTY; TREE BOARD

#### **CHAPTER 91: NUISANCES**

#### § 91.01 NUISANCES GENERALLY

It shall be unlawful for any person, firm, or organization to create or maintain a nuisance on any public or private property within the Town. A nuisance shall be defined as any condition that is a nuisance under the common law of Virginia, the principles of equity, or Virginia statutory law. This definition includes, without limitation, anything that endangers life or health, gives offense to the senses, violates normal bounds of decency, or obstructs the reasonable and comfortable use of property, such as accumulations of offensive, unsanitary, or unhealthy substances in or on any place or premises; portions of any lot adjacent to a street where the difference in level between the lot and the street constitutes a danger to life and limb; unsafe, dangerous, or unsanitary public or private buildings, walls or structures; and use of any public place or property of any description in a manner not permitted to the general public without prior consent of the Town.

### § 91.02 MUNICIPAL ABATEMENT OF NUISANCES

- (a) Whenever the Town Council or Town Manager has reasonable cause to believe that any such nuisance exists on private property, the Town may notify the property owner of record of such determination by certified mail, return-receipt requested, sent to the address listed in the real estate tax records for the property. Notice shall also be sent to any tenant or other occupant of such property, if other than the record owner. The Notice shall inform the party to whom it is directed that the Town Council or Town Manager will take evidence regarding the nuisance at the date and time specified in the notice, which shall not be less than five (5) days from the date the notice is delivered.
- (b) Whenever the Town Council or Town Manager has reasonable cause to believe any such nuisance exists on public property, at the direction of the Town Council, a notice of violation as prescribed in subsection (a) shall be sent to the home or business address of the person or organization creating or maintaining such nuisance.
- (c) On the date and time fixed in the notice, the Town Council or Town Manager shall receive evidence concerning the existence of the nuisance. The Town Council or Town Manager shall provide all interested parties with an opportunity to be heard regarding the nuisance. At the conclusion of the evidence, the Town Council or Town Manager may determine whether a nuisance exists. If the Town Manager is conducting the hearing and does not wish to adjudicate the question, he may adjourn the hearing and refer the matter to the Town Council. In either event, if the Town Council or Town Manager (as the case may be) determines after all evidence has been received that a nuisance exists, the Town Manager shall notify the property owner, the property occupant, and any party creating or maintaining the nuisance.

- (d) If such nuisance is not abated, removed, or corrected within fifteen (15) days of receipt of such notice, or after the notice is returned as unclaimed mail, the Mayor or Town Manager may order the nuisance removed or corrected by the Town's own agents or employees. The cost or expense of such removal shall be chargeable to, and paid by, the owners of such private property, or the person or organization creating or maintaining the nuisance on public property, and may be collected by the Town as taxes and levies are collected and, to the full extent permitted by state law, shall constitute a lien against the property on which the nuisance occurred, on parity with tax liens, to the full extent permissible under state law, and shall also be a personal obligation of the owner of such property.
- (e) If the nuisance in question is the use of public property without prior consent of the Town, notice prior to abatement shall not be necessary.
- (f) Any person creating or maintaining a public nuisance (including, without limitation, the use of public property without the prior consent of the Town) shall be guilty of a Class 3 misdemeanor and, upon conviction, shall be fined no more than Five Hundred Dollars (\$500). Each day's continuance of the nuisance shall be a separate offense.
- (g) The notice and hearing provisions of this section shall not apply when the Town Manager or Town Council determines in writing that a public nuisance exists and constitutes an imminent or immediate threat to life or property. In such instance the Town Manager or Mayor may order the nuisance removed or corrected immediately by the Town's own agents or employees, and the Town may recover the costs thereof under the provisions of subsection (d) herein.

#### § 91.03 CIVIL ACTIONS TO ABATE NUISANCES

In addition to the remedies provided in § 91.02, the Town may maintain a civil action to compel a responsible party to abate, raze, or remove any public or private nuisance, to obtain a judicial determination of the existence of any nuisance, or to recover any damages caused by such nuisance during the period of its maintenance. Costs and reasonable attorneys' fees associated with any such action shall be assessed against the persons creating or permitting the nuisance.

#### § 91.04 REMEDIES CUMULATIVE

The remedies for nuisances contained in the Town Code shall be in addition to any other remedies available to the Town or to a private citizen under other state or local law.

# CHAPTER 92: TRASH, GRASS, WEED, AND CLUTTER ACCUMULATION

# § 92.01 REMOVAL OF WEEDS, GRASS, AND FOREIGN GROWTH

The owners of property within the Town, whether improved or unimproved, and including such property upon which buildings or other improvements are located, shall cut the grass, weeds, and other foreign growth (including running bamboo as defined under state law) on such property or any part thereof at such time or times as the Town Manager or his designee determines is reasonably necessary to maintain the safety, health, and good order of the Town. If the Town Manager or his designee deems it necessary, after ten (10) days' notice to the property owner, the Town Manager or his designee may have such grass, weeds, or other foreign growth cut by the Town's agents or employees, in which event the cost and expenses thereof shall be chargeable and paid by the owner of the property and may be collected by the Town as taxes are collected, all as described in § 92.04. Provided, however, that the Town Manager is not obligated to give more than one notice to the owner of any particular property in one calendar year. The property owner may waive any such notice and consent to the Town's cutting of such grass, weeds, and other foreign growth; in such an event, the costs thereof shall be collected in same manner as if the property owner had not waived such notice and given such consent.

# § 92.02 DISPOSAL OF TRASH, ETC.

Trash, garbage, refuse, litter, clutter, and other debris shall be disposed of in personally owned or privately owned receptacles that are provided for such use and for the use of the persons disposing of such matter or in authorized facilities provided for such purpose and in no other manner not authorized by law.

# § 92.03 REMOVAL OR TRASH, LITTER, OR CLUTTER

- (a) The owners of property shall, within ten (10) days of notice, remove from such property any and all
  - (i) trash, garbage, refuse, litter, or clutter; or
- (ii) other substances which might endanger the health or safety of other residents of the Town.
- (b) If the Town Manager deems it necessary to protect the health or safety of residents of the Town, after ten (10) days' notice, he may have such trash, garbage, refuse, litter, clutter, or other like substances which might endanger the health of other residents of the Town, removed by its own agents or employees, in which event the cost or expenses thereof shall be chargeable to and paid by the owners of such property and may be collected by the Town as taxes are collected as described in § 92.04.

(c) For purposes of this section, "clutter" shall have the broadest possible definition that is permissible under state law, but shall include, at a minimum, mechanical equipment, household furniture, containers, and similar items that may be detrimental to the well-being of a community when they are left in public view for an extended period or are allowed to accumulate.

# § 92.04 CHARGES CONSTITUTE LIEN

All costs associated with action authorized under this Chapter shall constitute a lien on any property from which weeds are cut or trash or litter removed, on parity with tax liens, and may be collected in a manner identical to the manner in which real estate tax liens are collected. Such costs shall also be a personal obligation of the owner of such property.

# § 92.05 CIVIL AND CRIMINAL PENALTIES

- (a) <u>Civil Penalties</u>. A person violating this chapter shall also be liable for a civil penalty of \$50 per day for each initial violation thereof, and of \$200 per day for each subsequent violation arising within the same twelve month period as an initial violation. Provided, however, that each day of a violation arising from the same set of operative facts in a twelve month period is penalized as an initial violation, not as a subsequent violation. In addition, a civil penalty for violation arising from the same set of operative facts may not exceed \$3,000 in any twelve month period.
- (b) <u>Criminal Penalties</u>. Any person who has had a civil penalty imposed under paragraph (a) of this section three times within any twenty-four month period, for the same or similar violations, but on three different sets of operative facts, shall be guilty of a Class 3 misdemeanor and upon due conviction thereof shall be fined an amount not to exceed \$500.

### § 92.06 NOTICE

- (a) Whenever notice is required to be given under the provisions of this Chapter or Chapter 91, such notice shall be either by:
- (1) personal and actual service of such notice on the property owner by a private process server, Town police officer or Rockingham County Sheriff Deputy;
  - (2) by certified or registered mail, return receipt requested; or
  - (3) by overnight mail or reputable overnight delivery service.
- (b) When notice is given other than by personal service, it shall be sent to the property owner at the address maintained by Rockingham County's Commissioner of the Revenue, and at any other such other address as the Town Manager believes is reasonably calculated to provide actual notice to the property owner of the Town's proposed action.

- (c) Notice shall be deemed given effective when served or when deposited with the United States Postal Service or reputable overnight delivery service, as appropriate.
- (d) Notice not given in accordance with the above provisions, but actually received by the property owner, shall be effective when received.
- (e) Service upon a person other than a natural person may be made upon any registered agent for such person, or upon any official who is an authorized recipient of service to such person under state law related to service upon persons other than natural persons.

# CHAPTER 93: DILAPIDATED OR UNSAFE BUILDINGS AND STRUCTURES

#### § 93.01 RESPONSIBILITY OF OWNERS TO PRESERVE PROPERTY

Upon notification by the Town Manager, the owners of property within the Town shall remove, repair, or secure any building, wall, or other structure that the Town Manager determines might endanger the public health or safety of other residents of the Town.

#### § 93.02 REMEDIATION OF UNSAFE STRUCTURE

If the owner of property fails to comply with a notification from the Town Manager under § 93.01, the Town may remediate the unhealthy or unsafe building, wall, or other structure through its own agents. No such remediation shall take place until thirty (30) days have passed after final notice to the landowner of the Town's proposed remedial action. Such notice shall consist of notice in writing either mailed by certified or registered mail, return receipt requested, to the property owner at the address maintained by Rockingham County's Commissioner of the Revenue, and at any other such other address as the Town Manager believes is reasonably calculated to provide actual notice to the property owner of the Town's proposed action, or by personal and actual service of such notice on the property owner by a Town police officer. Further, such notice shall be published one per week for two successive weeks in a newspaper having general circulation in the Town. Provided, however, if the Town manager believes that the structure in question poses a significant threat to public safety, and states so in the notice, then the Town may take steps to prevent public access to the unsafe structure seven (7) days after such final notice. Notice shall be considered final on the second newspaper publication, or upon the return of the receipt from the certified or registered mailing, whichever occurs last.

### § 93.03 CHARGES CONSTITUTE LIEN

All costs associated with action authorized under this Chapter shall constitute a lien on any property on which the unsafe building, wall or structure, was situate, on parity with tax liens, and may be collected in a manner identical to the manner in which real estate tax liens are collected. Such costs shall also be a personal obligation of the owner of such property.

#### § 93.04 DEMOLITION BY CONSENT

With the written consent of a property owner, the Town may demolish or remove a derelict nonresidential building or structure provided that such building or structure is neither located within or determined to be a contributing property within a state or local historic district nor individually designated in the Virginia Landmarks Register. If the property has liens, the consent shall identify the first lienholder thereon under a recorded lien (if any there be) and shall document the property owner's best reasonable efforts to obtain the consent of any such lienholder. The costs of demolition or removal shall be a lien on the property, on parity with tax liens, and to be collected in the same manner; provided, however, that in the event a first lienholder on the property does not consent to such demolition or removal, then such lien shall

be subordinate to the first lien. In any event such costs shall be a personal obligation of the owner.

# § 93.05 CIVIL PENALTY

Any person who violates this chapter shall also be liable for a civil penalty of \$1,000 per violation.

#### CHAPTER 94: DEFACEMENT OF BUILDINGS AND STRUCTURES

#### § 94.01 RESPONSIBILITY OF OWNERS TO REMOVE DEFACEMENTS

The Town Manager shall be responsible for ensuring the prompt removal of all defacements from public property. Upon notification from the Town Manager, the owners of private property shall promptly remove all defacements from such private property when such defacements are visible from a public right of way.

### § 94.02 REMEDIATION OF DEFACEMENTS

If the owner of private property fails to comply with a notification from the Town Manager under § 94.01, the Town may remove such defacements through its own agents. No such removal shall take place until fifteen (15) days have passed after final notice to the landowner of the Town's proposed removal action. Such notice shall consist either of notice in writing mailed by certified or registered mail, return receipt requested, to the property owner at the address maintained by Rockingham County's Commissioner of the Revenue, and at any other such other address as the Town Manager believes is reasonably calculated to provide actual notice to the property owner of the Town's proposed action, or of personal and actual service of such notice on the property owner by a Town police officer.

#### § 94.03 CHARGES CONSTITUTE LIEN IF PERMITTED

To the full extent permitted by state law, all costs associated with action authorized under this Chapter shall constitute a lien on any property on which the defacement was situate, on parity with tax liens, and may be collected in a manner identical to the manner in which real estate tax liens are collected. Such costs shall also be a personal obligation of the owner of such property to the full extent permitted by state law.

#### § 94.04 DEFINITION OF DEFACEMENT

For the purposes of this Chapter, "defacement" means the unauthorized application by any means of any writing, painting, drawing, etching, scratching, or marking of an inscription, word, mark, figure, or design of any type.

#### **CHAPTER 95: INOPERABLE MOTOR VEHICLES**

# § 95.01 RESPONSIBILITY OF OWNERS TO REMOVE INOPERABLE MOTOR VEHICLES

No person shall keep an inoperable motor vehicle in a residential zoning district for any period, or within a commercial zoning district for more than a two (2) week period, unless it is either within a fully enclosed building or structure, or otherwise shielded or screened from view.

#### § 95.02 TOWN REMOVAL OF INOPERABLE MOTOR VEHICLES

If an owner violates § 95.01 and the Town Manager determines that it is advisable for the Town's agents to remove the inoperable motor vehicle, he shall notify the owner that such owner must remove the vehicle, and if the owner fails to do so within the times described in this paragraph, the Town's agents will do so. Such notice shall be either by certified or registered mail, return receipt requested, to the property owner at the address maintained by Rockingham County's Commissioner of the Revenue, and may also be at any other such address as the Town Manager believes is reasonably calculated to provide actual notice to the property owner; or, in lieu of notice by mail, may be by personal and actual service of such notice on the property owner by a Town police officer or appropriate Sheriff's deputy. If the property owner is an entity, such service may be on a registered agent or other agent authorized under applicable law to receive service on behalf of an entity. Seven days after the delivery of such notice or the return of such notice as undeliverable by certified or registered mail, the Town Manager may cause such inoperable motor vehicle to be removed. Seven days after providing additional notice on the same terms, the Town Manager may dispose of such vehicle in any reasonable manner.

# § 95.03 CHARGES CONSTITUTE LIEN

All costs associated with action authorized under this Chapter, together with civil penalties duly assessed hereunder, shall constitute a lien on any property on which the inoperable vehicle was situate, on parity with tax liens, and may be collected in a manner identical to the manner in which real estate tax liens are collected. Such costs shall also be a personal obligation of the owner of such property, and of the owner of the vehicle so removed, and may be collected any way that taxes are collected. In addition, and for the recovery of the same costs, the Town shall have a lien against the proceeds (if any) from the disposal of any vehicle removed under this Chapter.

### § 95.04 DEFINITIONS

- (a) For the purposes of this Chapter 95, "inoperable motor vehicle" means
  - (i) any motor vehicle which is not in operating condition;

- (ii) any motor vehicle which for a period of 60 days or longer has been partially or totally disassembled by the removal of tires and wheels, the engine, or other essential parts required for operation of the vehicle; or
- (iii) any motor vehicle on which there are displayed neither valid license plates nor a valid inspection decal.

Notwithstanding the foregoing, this definition shall not include any motor vehicle on the property of a licensed business that, on June 26, 1970, was regularly engaged in business as an automobile dealer, salvage dealer, or scrap producer.

(b) For the purposes of this Chapter 95, "shielded or screened from view" means not visible by someone standing at ground level from outside of the property on which the subject vehicle is located.

#### § 95.04 PENALTY

- (a) <u>Civil Penalty</u>. Any person who violates § 95.01 shall be subject to a civil penalty of \$200 for the initial summons and \$500 for each additional summons. Each day during which the violation is found to have existed shall constitute a separate offense. However, specified violations arising from the same operative set of facts shall not be charged more frequently than once in any 10-day period. A series of specified violations arising from the same operative set of facts shall not result in civil penalties which exceed a total of \$5,000.
- (b) <u>Criminal Penalty</u>. Any person who violates § 95.01 after three civil penalties have previously been imposed against such person within a twenty-four (24) month period under § 95.04(a) shall be guilty of a Class 3 misdemeanor and upon due conviction thereof shall be fined an amount not to exceed \$500. To support a conviction under this paragraph, the three previous civil penalties must not have arisen from the same set of operative facts. If this paragraph applies, a criminal penalty for the conduct described shall be in lieu of a civil penalty for the same violation.
- (c) <u>Remedies Cumulative</u>. Except as otherwise provided in §§ 95.04(b) and 95.05, the remedies and penalties provided in this Chapter are cumulative to any other remedy available to the Town under the law of nuisance or otherwise.

# § 95.05 AUTHORITY OF TOWN MANAGER TO FACILIATE VOLUNTARY REMOVAL OF VEHICLES

In the event that the owner of an inoperable motor vehicle wishes to remove and abandon the vehicle but has not done so, such owner may request the Town to remove the vehicle and dispose of it. If the Town Manager believes that the removal of the vehicle would promote the public welfare, he is authorized to accept the title to such vehicle in the name of the Town, and shall promptly cause the removal of such vehicle and dispose of it. The Town shall bear the costs of

such removal and disposal, and shall be entitled to reimburse itself for the costs of such removal and disposal from any proceeds generated by the disposal of the vehicle. The Town shall return the remainder of any such proceeds to the former owner of the vehicle, and an appropriation is hereby made for the purposes described in this paragraph. The provisions of §§ 95.02 and 95.04 shall not apply to a removal under this paragraph.

# **CHAPTER 97: PUBLIC ASSEMBLAGE**

#### § 97.01 **PURPOSE**

Pursuant to the authority granted to the Town by the Code of Virginia and its general police powers, the Town has adopted this chapter in order to provide for the public health, safety and general welfare in the Town, to ensure the free and safe passage of pedestrians and vehicles on the public rights-of-way, and to ensure the safe and unimpaired use and enjoyment of public property in places open to the general public and otherwise to regulate and control the time, place and manner of activities that would otherwise threaten or impair the public health, safety, and welfare in the Town while also encouraging the exercise of the rights to free speech and assembly in the Town. The Town disavows any intention to engage in viewpoint discrimination in enacting the provisions of this chapter.

### § 97.02 **DEFINITIONS**

The following terms shall have the meanings set out herein:

*Parade* means any march, procession or motorcade consisting of people, animals, or vehicles, or a combination thereof, upon the streets, sidewalks or other public areas within the Town and that interferes with or has a tendency to interfere with the normal flow or regulation of pedestrian or vehicular traffic upon such streets, sidewalks, or other public property.

Public assembly or assembly means any meeting, demonstration, picket line, rally or gathering of more than ten people for a common purpose as a result of prior planning, that interferes with or has a tendency to interfere with the normal flow or regulation of pedestrian or vehicular traffic upon the streets, sidewalks, or other public property within the Town, or that interferes with or has a tendency to interfere with the normal use of any public property in a place open to the general public.

Spontaneous event means an unplanned or unannounced coming together of people, animals or vehicles in a parade or public assembly which was not contemplated beforehand by any participant therein and which is caused by or in response to unforeseen circumstances or events occasioned by news or affairs first coming into public knowledge within 15 days of such parade or public assembly.

# § 97.03 PERMIT REQUIRED

- (a) It shall be unlawful for any person to conduct or participate in a marathon and walking or physical endurance contest or exhibition, public assembly, demonstration or parade on the public streets, sidewalks, or other public property of the Town in a place open to the general public for which a written permit has not been issued in accordance with the provisions of this chapter.
  - (b) The provisions of this section shall not apply to:
    - (1) Spontaneous events;
    - (2) Recreational activities, including jogging or walking, that do not require closing public streets or other public rights-of-way and that do not interfere with or have a tendency to interfere with the normal use of any public property in a place open to the general public;
    - (3) Door-to-door advocacy, including canvassing, pamphleteering, religious or political proselytizing and the distribution of written materials, and similar activities that do not interfere with or have a tendency to interfere with the free passage of pedestrians and vehicles on the public rights-of-way or the normal use of any public property in a place open to the general public;
    - (4) Door-to-door sales of goods or services, and similar activities that do not interfere with or have a tendency to interfere with the free passage of pedestrians and vehicles on the public rights-of-way or the normal use of any public property in a place open to the general public; provided, however, that any persons or organizations engaging in such activities shall comply with any other applicable requirements of this Code;
    - (5) Funeral processions;
    - (6) Students going to and from school classes or participating in educational activities part of the ordinary curriculum of an accredited educational institution, provided that such conduct is under the immediate direction and supervision of the proper school authorities, and that only students and school employees participate;
    - (7) The United States Army, Navy, Air Force and Coast Guard, the military forces of the State and the police and fire divisions of the Town and Rockingham County; or

- (8) Governmental agencies acting within the scope of their functions;
- (c) Permits may be granted if they are requested by individuals or organizations who desire to have a permit, even though the permit is not required under this section.

# § 97.04 APPLICATION FOR PERMIT

- (a) Any person desiring to conduct a marathon and walking or physical endurance contest or exhibition, parade or public assembly shall make written application, in a form approved by the Town Manager, to the Town Manager, or his designee, at least 15 days prior to such parade or public assembly. Such application shall set forth the following information:
  - (1) The name, address and telephone number of the person requesting the permit;
  - (2) The name and address of any organization or group the applicant is representing;
  - (3) The name, address and telephone number of the person who will act as the parade or public assembly leader or chairperson and who will be responsible for the conduct of the parade or public assembly;
  - (4) The type of public assembly, including a description of the activities planned during the event;
  - (5) The date and time (start and ending) of the parade or public assembly;
  - (6) If an assembly, the specific location or locations of the assembly;
  - (7) If a parade, the specific assembly and dispersal locations, the specific route, and the plans, if any, for assembly and dispersal;
  - (8) The approximate number of people who, and animals and vehicles which will constitute such parade or public assembly and the type of animals and a description of the vehicles;
  - (9) A statement as to whether the parade or public assembly will occupy all or only a portion of the width of the streets or sidewalks or other public rights-of-way proposed to be traversed or used;

- (10) A description of any recording equipment, sound amplification equipment, banners, signs, or other attention-getting devices to be used in connection with the parade or public assembly;
- (11) Any conditions or undertakings regarding the event that the applicant wishes to proffer; and
- (11) Such other information as the Town Manager, or his designee, may deem reasonably necessary in order to properly provide for traffic control, street and property maintenance, administrative arrangements, police and fire protection, and for the protection of public health, safety and welfare.
- (b) The Town Manager, or his designee, shall not issue the permit if any information supplied by the applicant is false or intentionally misleading.
- (c) The Town Manager, or his designee, shall have the authority to and shall make reasonable efforts to consider an application hereunder which is filed less than 15 days before the date the parade or assembly is proposed to be conducted if, after due consideration of the date, time, place and nature of the marathon and walking or physical endurance contest or exhibition, parade or public assembly, the anticipated number of participants and the Town services required in connection with the event, and where good cause is otherwise shown, the Town Manager, or his designee, determines that the waiver of the permit application deadline will not present an undue hazard to public safety.

#### § 97.05 GRANT OR DENIAL OF PERMIT

- (a) The Town Manager, or his designee, shall issue the permit within ten days of receipt of the completed application, and in any event prior to the scheduled parade or public assembly (unless the applicant has agreed to a different deadline), if the proposed marathon and walking or physical endurance contest or exhibition, parade or public assembly will not endanger the public health, welfare or safety, after applying the following criteria and finding that:
  - (1) The time, duration, route and size of the parade or assembly will not unreasonably interrupt the safe and orderly movement of vehicular or pedestrian traffic or the normal use of public property in a place open to the general public;
  - (2) The parade or assembly is not of such a nature that it will require diversion of so great a number of police and fire personnel to properly police the line of movement in the areas contiguous thereto so as to impair the normal protection of the remainder of the Town;

- (3) The applicant has, where appropriate, designated monitors sufficient to control the orderly conduct of the parade or assembly in conformity with such permit;
- (4) The applicant has, where appropriate, agreed to be responsible for having appropriate traffic control devices installed in accordance with the Uniform Manual on Traffic Control Devices to close roadways to vehicular traffic sufficient to control the orderly conduct of the parade or assembly in conformity with such permit;
- (5) The conduct of the parade or assembly will not unduly interfere with the proper fire and police protection of, or ambulance service to, the remainder of the Town, or unreasonably disrupt other public services and protection normally provided to the Town;
- (6) The parade or assembly will not interfere with another parade or assembly for which a permit has been granted; and
- (7) The parade or assembly proposed will not violate, and will conform with all applicable State regulations and laws governing the proposed event.
- (b) For marathons and walking or physical endurance contests or exhibitions, parades or public assemblies held on a regular or recurring basis at the same location, an application for an annual permit covering all such parades or assemblies during the calendar year may be filed with the Town Manager, or his designee, at least 15 and not more than 60 days before the date and time at which the first such parade or public assembly is proposed to commence. The Town Manager, or his designee, may and shall make reasonable efforts to consider an application hereunder which is filed less than 15 days before the date and time at which the first parade or assembly is proposed to commence, after due consideration of the factors specified in Section 97.05.
- (c) If the Town Manager, or his designee, denies an application, he shall promptly attempt to call and shall promptly mail to the applicant a notice of his action, stating the reasons for his denial of the permit and notifying the applicant of his right to appeal the denial pursuant to Section 97.09.
- (d) If two or more applications are submitted requesting a permit under this article for a parade or assembly to be used at the same time and place, the application first filed shall be granted if it meets the requirements of this article.
- (e) Nothing in this article shall permit the Town Manager, or his designee, to deny a permit based upon political, social or religious grounds or reasons or based upon the content of the views expressed. Denial of a permit on such grounds is prohibited.

- (f) The Town Manager may refer any permit application to the Town Council for action, but such referral shall not affect the deadline described in § 97.05(a). The Council shall have the same powers in connection with a referred permit application as the Town Manager did prior to the referral.
- (g) If the Town Manager, his designee, or the Town Council grants an application for a permit, any conditions that an applicant proffers with respect to such permit under § 97.04(a)(11) shall become part of the permit as issued.

### § 97.06 ALTERNATIVE OR MODIFIED PERMIT

The Town Manager, or his designee, in denying a permit for a marathon and walking or physical endurance contest or exhibition, parade or public assembly shall be empowered

- (a) to authorize the conduct of the parade or assembly on a date, at a time, at a place, or over a route different from that proposed by the applicant, and
- (b) to impose reasonable time, place, or manner conditions on the applicant in order to further the purposes described in § 97.01.

An applicant desiring to accept the permit as so revised shall file a written notice of acceptance with the Town Manager, or his designee. Such permit, if accepted, shall conform to the requirements of and shall have the effect of a permit under this article.

#### § 97.07 COMPLIANCE WITH DIRECTIONS AND CONDITIONS

Every person to whom a permit is issued under this article shall substantially comply with all permit terms and conditions and with all applicable laws and ordinances. The marathon and walking or physical endurance contest or exhibition, parade or assembly chairman, or other person heading or leading the marathon and walking or physical endurance contest or exhibition, parade or assembly, shall carry the permit upon his person during the conduct of the parade or assembly and show the permit when requested to do so.

# § 97.08 REVOCATION OF PERMIT

The Town Manager, or his designee, or the Town Council, shall have the authority to revoke any permit issued pursuant to this chapter if any information supplied by the applicant is discovered to be false or intentionally misleading, if any term, condition, restriction or limitation of the permit has been substantially violated or if there is any continued violation of the terms, conditions, restrictions or limitations of the permit after the applicant or anyone acting in concert with him is notified of a violation of the permit by an appropriate law enforcement official.

# § 97.09 APPEAL

- (a) Any person aggrieved by the refusal of the Town Manager, or his designee, to grant a permit or by the revocation of a permit after one has been issued, may appeal the denial to the Town Council, by filing with the Town Manager, within five business days after the date of denial or revocation, a written notice of the appeal setting for the grounds therefor. The Town Council, or its designee, shall act upon the appeal at its next meeting following the denial.
  - (b) No appeal shall lie from a decision of The Town Council to deny a permit.

# § 97.10 PUBLIC CONDUCT DURING PERMITTED EVENTS

- (a) No person shall unreasonably hamper, obstruct, impede, or interfere with any marathon and walking or physical endurance contest or exhibition, parade, demonstration or assembly or with any person, vehicle, or animal participating or used in such an event for which a written permit has been issued in accordance with the provisions of this Chapter.
- (b) No driver of a vehicle shall drive between the vehicles, persons, or animals comprising a marathon and walking or physical endurance contest or exhibition, parade, demonstration, or assembly for which a written permit has been issued in accordance with the provisions of this Chapter, except as otherwise directed by a police officer. This shall not apply to authorized emergency vehicles such as fire apparatus, ambulances, and police vehicles.
- (c) The Town Manager or the Chief of Police shall have authority, when reasonably necessary, to prohibit or restrict the parking of vehicles along the public streets or other public rights of way constituting part of the route of a marathon and walking or physical endurance contest or exhibition, parade, demonstration or assembly. The Town Manager or the Chief of Police shall ensure that signs are posted to such effect. If such signs are duly posted, no person shall park or leave any vehicles in violation thereof.

#### § 97.11 PENALTY

- (a) Any person who
  - (1) participates in a marathon and walking or physical endurance contest or exhibition, parade, demonstration or assembly on public property or which a permit is required under this Chapter, but for which no permit has been issued;
  - (2) participates in a marathon and walking or physical endurance contest or exhibition, parade, demonstration or assembly on public property for which a permit has been issued under this Chapter, and violates any term or condition of such permit, or any other provision of this Chapter, or

# (3) violates any provision of § 97.10,

shall be guilty of a Class 4 misdemeanor for the first offence and upon due conviction thereof shall be fined an amount not more than \$250. Upon the second and subsequent violations thereof (not arising from the same set of operative facts), such person shall be likewise guilty of a Class 3 misdemeanor and shall be fined an amount not more than \$500. Provided, however, that if damage to property or injury to person results from the violation, then such person shall be guilty of a Class 1 misdemeanor and upon due conviction thereof shall be incarcerated for a period not to exceed six months and fined an amount not to exceed \$2,500, or both.

(b) Any person who intentionally makes a false statement on an application for a permit under this Chapter shall be guilty of a Class 3 misdemeanor and upon due conviction thereof shall be fined an amount not more than \$500.

# § 97.12 SUPPLEMENTAL TO OTHER LAW

The provisions of this Chapter supplement, but do not replace, the other provisions of this Code, including, without limitation, the provisions regarding nuisances, noise, and parking.

# CHAPTER 98 TREES AND SHRUBS ON PUBLIC PROPERTY; TREE BOARD

#### § 98.01 DEFINITIONS

For the purpose of this Chapter, the following words have the following meanings:

*Board* mean the Tree Board established under this Chapter.

Canopy means the crown branch area of a single tree or several combined trees.

*Drip line* means a vertical line extending from the outermost edge of the tree canopy or shrub branch to the ground.

Land-disturbing activity means defoliation or alteration of a site, or the commencement of any construction activities.

Landmark trees means trees that are designated depending on species rarity, old age, association with an historical event or person, abnormality, scenic enhancement, etc.

*Public places/property* means all streets, public rights-of-way, parks, and other real property owned by the Town of Broadway or its Industrial Development Authority (IDA), or dedicated to public use.

*Shrub* means a woody plant that usually remains low in height and produces roots or stems from the base and is usually not tree-like or single stemmed.

Street tree means a tree planted within the developed street right-of-way between the curb line of the street and the property line of the site.

*Topping* means the severe cutting back of limbs within the tree's crown to such a degree so as to remove the normal canopy and disfigure the tree.

*Tree* means any self-supporting woody plant growing upon the earth that usually provides one (1) main trunk and produces more or less a distinct and elevated head with many branches. The following are specific descriptions of three (3) sizes of trees:

Small tree: Attaining a height of less than thirty (30) feet.

Medium tree: Attaining a height of thirty (30) to forty-five (45) feet.

Large tree: Attaining a height of more than forty-five (45) feet or more.

Ordinance of February 6, 2024 added this section.

#### § 98.02 **PURPOSE**

It is the purpose of this Chapter to:

- (a) Conserve and protect public land, water, air, vegetation and other natural resources of the Town:
- (b) Promote and protect the public and private aesthetic benefits gained from trees in the Town;
- (c) Promote and protect the public health, safety, and general welfare by providing for the regulation of planting, maintenance, and removal of trees, shrubs, and other plants on public property within the Town;
- (d) Preserve and protect trees and other vegetation during all phases of any land-disturbing activity within the Town;

- (e) Encourage private landowners to enhance their property along street frontage and in public use areas;
- (f) Regulate the use of the Town's real property, including rights of way, to further the foregoing purposes; and
- (g) Ensure that the Town Council has information necessary to legislate regarding all of the foregoing with input from a board of citizens who are particularly interested in, and dedicated to, arboreal matters.

Ordinance of February 6, 2024 added this section.

## § 98.03 TREE BOARD; AUTHORITY AND DUTIES

- (a) There is hereby created and established a Tree Board for the Town, which shall consist of three (3) members that the Town Council shall appoint. The Board shall sit as an advisory board to the Town Council. Members shall serve without compensation.
- (b) The Board shall choose its own officers, develop its own operating guidelines, and keep minutes of its meetings. Board meetings will not be pre-set, but rather meetings will be scheduled on an as-needed basis at the request of a Board member or by the Mayor or Town Manager. A majority of its members shall constitute a quorum for the transaction of business. The Board shall have no executive authority, its powers being strictly advisory.
- (c) The Board, shall study, investigate and develop recommendations for the care, preservation, pruning, planting, replanting and removal of trees and shrubs in parks, along streets and in other publicly owned places. The tree board, through various public outreach and educational programs, will also encourage private landowners to enhance their property along street frontage and in private areas to which the public has access, such as parking lots.
- (d) The Town Council shall review the Board's recommendations and endorse them as is or amend them as it deems necessary.

Ordinance of February 6, 2024 added this section. The Ordinance also contained the following text: "The persons who were part of the Town's Tree Board on August 1, 2023, shall immediately become part of the Tree Board established under this Ordinance, and their terms shall continue and expire as under previous law, as through the Tree Board had continued an uninterrupted existence between August 1, 2023, and the date of the adoption of this Ordinance. The Town Council hereby validates, ratifies, and confirms any action that

the Tree Board took, or that other officers of the Town took, between August 1, 2023, and the date of the adoption of this Ordinance, *nunc pro* tunc; all such actions shall be effective to the extent that they would have been effective had this Ordinance been in effect on the date of such action." In keeping with the principle that the Code only codifies ordinances of general application, the Code does not include this text, but the attention of the reader is drawn to it by this note.

# § 98.04 TERM OF OFFICE OF TREE BOARD MEMBERS

Tree board members shall ordinarily serve a three-year term, but shall in any event hold office at the pleasure of the Town Council.

Ordinance of February 6, 2024 added this section.

#### § 98.05 APPLICABILITY

This Chapter shall apply to trees, shrubs, and other plants within street rights-of-way, parks and other public places within the Town.

Ordinance of February 6, 2024 added this section.

#### § 98.06 LICENSING AND PERMITS

It shall be unlawful for any person to engage in the business of planting, cutting, trimming, pruning, spraying, or otherwise treating trees, shrubs, or vines on public property within the Town, without first producing evidence of a business license.

Ordinance of February 6, 2024 added this section.

#### § 98.07 **INSURANCE**

All contractors in the business of planting, cutting, trimming, pruning, removing, spraying, and otherwise treating tree, shrubs or vines on public property within the Town shall show proof of liability insurance. This certificate of insurance shall be in the amount that the Town Manager determines is necessary to protect the Town. This coverage shall be for bodily injury or death and property damage, indemnifying the Town or any person injured or damaged from the pursuit of such endeavors, as herein described.

Ordinance of February 6, 2024 added this section.

#### § 98.08 NEW LANDSCAPING IN PUBLIC PLACES

In the development of commercial property and subdivisions, the Town Manager shall review landscaping plans and may require changes as recommended for street trees or trees to be planted in public places.

Ordinance of February 6, 2024 added this section.

# § 98.09 TREE PLANTING, MAINTENANCE, AND REMOVAL

(a) The Town shall have the right to plant, prune, maintain and remove trees, shrubs, and other plants within the lines of streets, alleys, avenues, lanes, and other public property as may be necessary to ensure public health or safety or to preserve or enhance the symmetry and

beauty or public property. It shall be unlawful for any person to tie any animal or bicycle to any tree or shrub or to any structure erected for the protection of the same on public property. The Town may remove or cause or order to be removed any such tree, shrub, or part thereof which constitutes a threat to public health or safety or which is injurious to public property, public buildings, sewers, electric power lines, gas lines, water lines, or other public improvements, or is affected with an injurious fungus, insect, or other pest. This section does not prohibit the planting of street trees by adjacent property owners provided that the selection and location of said trees is in accordance with this Chapter.

- (b) The tree board may develop a list of desirable tree species for planting along streets in three (3) sizes: small, medium and large. A list of undesirable tree species may be developed as well. Such lists are not all inclusive and other species or cultivars may be taken into consideration by the Town as seems reasonable. This list shall be available for the general public in assisting with landscaping of private property.
- (c) No street trees other than those listed as small trees may be planted under or within ten (10) lateral feet of an overhead utility wire, or over or within five (5) lateral feet of an underground utility line (water, sewer, gas, transmission, etc.)
- (d) No Town personnel or other person shall top a tree on public property without the express consent of the Town Manager. The Town Manager shall give such consent only upon a finding that the benefit obtained by topping such tree outweighs the damage to the purposes described in § 98.02 above, and that there is no other reasonable and feasible way to obtain such benefits. Examples of situations in which such a finding would be appropriate include: trees severely damaged by storms or other causes, certain trees under utility wires, or other obstructions where normal pruning practices are impractical.

Ordinance of February 6, 2024 added this section.

#### § 98.10 CONSTRUCTION OF SIDEWALKS TO ALLOW FOR GROWTH

Insofar as possible, when any granolithic, concrete, brick, or other permanent sidewalk is laid in any street along which trees or shrubs have been planted, care shall be taken that an opening around the same shall be left in such sidewalk large enough for the expected growth of the trees or shrubs.

Ordinance of February 6, 2024 added this section.

#### § 98.11 RESPONSIBILITIES OF ADJACENT LANDOWNERS

No person shall plant, remove, or disturb any tree on or within any public property without first obtaining written permission from the Town.

Ordinance of February 6, 2024 added this section.

#### § 98.12 TREE PROTECTION

Trees that are to be saved or kept in construction projects shall be protected as follows:

- (a) No grading shall be done within the tree's drip line.
- (b) A temporary fence shall be constructed around the tree's drip line.
- (c) No equipment or machinery shall work within the fenced area.
- (d) No tools, equipment or supplies shall be stored inside the fenced area.
- (e) Landmark trees shall be protected.

Ordinance of February 6, 2024 added this section.

#### § 98.13 ENFORCEMENT

The Town Manager shall have the power to enforce the provisions of this Chapter by appropriate action.

Ordinance of February 6, 2024 added this section.

#### § 98.14 PENALTIES, CLAIMS AND APPEALS

- (a) Any person who knowingly and willfully violates any provision of this Chapter or who fails to comply with any notice issued pursuant to provisions of this chapter shall be guilty of a Class 4 misdemeanor and upon due conviction thereof shall be subject to a fine not to exceed two hundred fifty dollars (\$250.00) for each separate offense.
- (b) Any person who violates any provision of this Chapter and causes damage to trees, shrubs, or other landscaping shall be liable for such damage to the Town or the owner thereof, in restitution or in an appropriate civil action for trespass to chattels. The value of trees and shrubs shall be determined in accordance with the latest revision of "A Guide to the Professional Evaluation of Landscape Trees, Specimen, and Evergreens," published by the International Society of Arboriculture, or by other means that the trier of fact determines are appropriate.

Ordinance of February 6, 2024 added this section.

# TITLE XI: MOTOR VEHICLES AND TRAFFIC

# Chapter

- 111. TRAFFIC REGULATION
- 112. PARKING REGULATION
- 113. OTHER PROVISIONS REGARDING MOTOR VEHICLES
- 114. PENALTIES FOR VIOLATION OF TRAFFIC OR PARKING REGULATIONS

#### **CHAPTER 111: TRAFFIC REGULATION**

#### § 111.01 INCORPORATION OF STATE LAW BY REFERENCE

Pursuant to the authority of Section 46.2-1313 of the Code of Virginia, 1950, as amended, all of the provisions and requirements of the laws of the State of Virginia contained in Title 46.2 and in Article 2, Chapter 7, Title 18.2, of the Code of Virginia, 1950, as amended and in force from time to time, except (a) those provisions and requirements the violation of which constitute a felony;(b) those provisions and requirements which by their nature can have no application to or within the municipality and (c) to the extent the Town Council specifically rejects any amendment thereto or portion thereof, are adopted and incorporated in this Ordinance by reference and made applicable within the municipality. References to "highways of the state" contained in the Code provisions and requirements adopted shall be deemed to refer to the streets, highways and other public ways within the Town of Broadway. The provisions and requirements are adopted, mutatis mutandis, and made a part of this Title as fully as though set forth at length. It shall be unlawful for any person, within the municipality, to violate or fail, neglect or refuse to comply with any provision of Title 46.2 or Article 2, Chapter 7, Title 18.2 of the Code of Virginia which is adopted by this Section. Except to the extent otherwise described in the Town Code, the penalty for each such violation shall be as prescribed in state law, including any uniform schedule thereof, but in no event shall the penalty imposed for the violation of any provision or requirement adopted exceed the penalty imposed for a similar offense under the incorporated provisions of the Code of Virginia, 1950, as amended. All future amendments to the sections of the Code of Virginia incorporated by reference under the provisions of this paragraph shall be included within the scope of this paragraph automatically upon their effective date, without any formal Town Council action. Words and phrases used and contained in this chapter shall have the meanings ascribed to them by § 46.2-100 of the Code of Virginia, except where the context clearly requires a different meaning. For the avoidance of doubt, this incorporation by reference includes, without limitation, the incorporation of all regulations of state law, contained within the above-referenced statutes, which are applicable to bicycles and mopeds.

#### § 111.02 TOWN MANAGER TO SET SPEED LIMITS

The Town Manager shall have the power to increase or decrease the speed limit on any street within the Town, provided such increase or decrease in speed shall be based upon an engineering and traffic investigation by the Town or its agent and provided such speed area or zone is clearly indicated by markers or signs. The Town Manager may also to reduce for a temporary period not to exceed sixty days, without such engineering and traffic investigation, the speed limit on any portion of any street of the Town on which work is being done or where the highway is under construction or repair. Nothing in this paragraph shall be construed to abridge the authority of the Town Council to make any such regulation directly.

#### § 111.03 SPEED LIMIT WHEN NOT MARKED

Other than as altered under § 111.02, or as marked and existing on the date of the enactment of this Code, the speed limit on any street within the Town shall be twenty-five miles per hour.

# § 111.04 STOP AND YIELD INTERSECTIONS

The Town Manager may designate intersections at which vehicles shall come to a full stop or yield the right-of-way. The Town Manager shall ensure that all such intersections are marked, and no person shall be liable for a violation of this section if at the time of the alleged violation the sign or marker placed in conformity with this section is missing or is defaced so that an ordinarily observant person under the same circumstances would not be aware of the existence of the regulation.

# § 111.05 EXISTANCE OF SIGNS EVIDENCE OF TOWN MANAGER APPROVAL; CURRENT SIGNS

The existence of signs, signals or markers at any place within the corporate limits of the town shall be prima facie evidence that such signs, signals or markers were erected or placed by and at the direction of the Town Manager in accordance with the provisions of this Chapter. The Town Council ratifies and confirms the legality and enforceability of all signs that exist on the date that this Code was adopted, but the Town Manager may alter or remove such signs if he determines to change the stop-and-yield status of an intersection.

#### § 111.06 BACKING

No vehicle operator shall back such vehicle unless such movement can be made with safety and without interfering with other traffic.

#### § 111.07 TRAFFIC SCHOOLS

To the extent that traffic school is an appropriate penalty for violations of this Ordinance under state law, any traffic school designated for Rockingham County shall be designated as a traffic school for the Town.

#### § 111.08 APPLICABILITY NOT BASED ON OWNERSHIP

The provisions of this chapter applicable to the drivers of vehicles shall apply to the drivers of all vehicles, regardless of ownership, subject to such specific exceptions as are set forth in this chapter.

# § 111.09 AUTHORITY OF FIRE DEPARTMENT OFFICIALS AND RESCUE SQUAD MEMBERS TO DIRECT TRAFFIC

Officers of a fire department and members of a rescue squad may direct or assist police in directing traffic at or in the near vicinity of a fire, vehicle accident, natural disaster or similar situations, as deemed necessary for public safety, subject in every instance to the supervision and

control of the police department. No person shall willfully fail or refuse to comply with any lawful order or direction of any police officer, or member of the fire department at the scene of a fire, who is invested by law or ordinance with authority to direct, control, or regulate traffic.

# § 111.10 DRIVING THROUGH FUNERAL OR OTHER PROCESSIONS; MANNER OF DRIVING IN FUNERAL PROCESSIONS

- (a) No operator of a vehicle shall drive between the vehicles, persons or animals comprising a funeral or other authorized procession, except when otherwise directed by a police officer. This provision shall not apply to authorized emergency vehicles.
- (b) Each driver in a funeral procession shall drive as near to the right-hand edge of the roadway as is practicable and shall follow the vehicle ahead as close as is practicable and safe.

# § 111.12 IDENTIFICATION OF VEHICLES IN FUNERAL PROCESSIONS; RIGHT-OF-WAY OF SUCH VEHICLES

- (a) All motor vehicles participating in a funeral procession, when proceeding to any place of burial, shall display illuminated head lamps thereon and such other identification as the chief of police may prescribe.
- (b) All motor vehicles so designated and traveling under police or sheriff's escort shall have the right-of-way, except for fire apparatus, ambulances and police vehicles, at any street or highway intersection within the city, and may proceed through a stop sign or signalized intersection with proper caution and safety.
- (c) No vehicle that is not properly part of a funeral procession shall join, pass through, or interfere with the passage of any funeral procession under police or sheriff's escort, as provided in this section.

#### § 111.13 CONDUCT OF PEDESTRIANS

All pedestrians shall:

- (a) obey signs and signals erected on highways therein for the direction and control of traffic, and
- (b) obey the orders of law-enforcement officers engaged in directing traffic on such highways.

# § 111.14 BARRIERS OR SIGNS PLACED BY PUBLIC UTILITIES OR TOWN DEPARTMENTS

No person, public utility or any Town department shall erect or place any barrier or sign on any highway or sidewalk unless authorized by the Town Manager, Chief of Police, Director of Public Works, or other officer designated by the Council. It shall be unlawful for any person to disobey the instructions of any barrier or sign placed in any highway or sidewalk when the barrier or sign has been first approved by the designated officer and the same has been erected pursuant to authority from such officer. No person shall ride or drive any animal or vehicle or walk over any newly made pavement or freshly painted marking in any highway when a barrier or sign is in place warning persons not to drive over such pavement or marking or when a sign is in place stating that the street or any portion thereof is closed.

### § 111.15 OFFENSES BY PERSONS OWNING OR CONTROLLING VEHICLES

Neither the owner, nor any person employing or otherwise directing the driver of any vehicle, shall require or knowingly permit the operation of such vehicle upon a street of this Town in any manner contrary to law.

#### § 111.16 [DELETED]

**Former Section 111.16** was deleted by letter from the Town Attorney to the Clerk, dated August 7, 2023, pursuant to the authority provided in Section 10.07 of the Code, because it was mistakenly included in the initial enactment, and in order to implement the manifest intention of the Town Council that Section 111.16 not be included within the Code.

#### **CHAPTER 112: PARKING REGULATIONS**

#### SUBCHAPTER A – GENERAL PARKING REGULATIONS

#### § 112.11 INCORPORATION OF STATE LAW PROVISIONS BY REFERENCE

Except as expressly modified or supplemented by this Chapter, all of the provisions and requirements of the laws of the State of Virginia contained in Title 46.2 of the Code of Virginia, 1950 concerning parking or stopping, as amended and in force from time to time, except (a) any of those provisions and requirements the violation of which constitute a felony; (b) those provisions and requirements which by their nature can have no application to or within the Town, and (c) to the extent the Town Council specifically rejects any amendment thereto or portion thereof, are adopted and incorporated in this Code by reference and made applicable within the Town. References to "highways of the state" contained in the Code provisions and requirements adopted shall be deemed to refer to the streets, highways and other public ways within the Town. The provisions and requirements are adopted, mutatis mutandis, and made a part of this Title as fully as though set forth at length. It shall be unlawful for any person, within the Town, to violate or fail, neglect or refuse to comply with any provision of Title 46.2 or Article 2, Chapter 7, title 18.2 of the Code of Virginia which is adopted by this Section. Except to the extent otherwise described in the Town Code, the penalty for each such violation shall be as prescribed in state law, including any uniform schedule thereof, but in no event shall the penalty imposed for the violation of any provision or requirement adopted exceed the penalty imposed for a similar offense under the incorporated provisions of the Code of Virginia, 1950, as amended. Words and phrases used and contained in this chapter shall have the meanings ascribed to them by § 46.2-100 of the Code of Virginia, except where the context clearly requires a different meaning. All future amendments to the sections of the Code of Virginia incorporated by reference under the provisions of this paragraph shall be included within the scope of this paragraph automatically upon their effective date, without any formal Town Council action.

#### § 112.12 AUTHORITY OF TOWN MANAGER

The Town Manager shall have the authority to make and administer parking regulations not inconsistent with this Chapter. Such authority shall include the authority to designate no parking zones and to limit the time that a vehicle may be parked in a given space. The existence of signs, signals or markers at any place within the corporate limits of the town shall be prima facie evidence that such signs, signals or markers were erected or placed by and at the direction of the Town Manager in accordance with the provisions of this Chapter. The Town Council ratifies and confirms the legality and enforceability of all signs that exist on the date that this Code was adopted, but the Town Manager may alter or remove such signs if he determines to change the parking regulations for an area. Nothing in this Chapter shall be construed to abridge the authority of the Town Council to make any such regulation directly.

#### § 112.13 PARKING IN VIOLATION OF SIGN

It shall be unlawful to park any motor vehicle, trailer, or semi-trailer on the public streets or public grounds in violation of official permanent or temporary "No Parking" signs, or in violation of a sign that limits the amount of time a vehicle may be parked in a space. Official temporary "No Parking" signs, erected to facilitate emergency repairs, special events or like activities, shall be erected 24 hours prior to the initiation of any enforcement action. Official temporary "No Parking" signs shall be erected in a uniform manner as prescribed by traffic engineering and shall be marked on the reverse side with the date and the date signs were erected. Any motor vehicle, trailer, or semi trailer parked in violation of this section is subject to ticketing and towing.

# § 112.14 OTHER PROHIBITED PARKING

- (a) It shall be unlawful to park any motor vehicle, trailer, or semi trailer, whether attended or unattended, so as to prevent the use of a curb ramp, fire hydrant, or driveway or other entrance, or mail box located on public property or on privately-owned property which is open to the public. A summons for the offense may be issued by police department employees without the necessity of a warrant being obtained by the owner of any private property.
- (b) It shall be unlawful for the owner, operator, or driver of any motor vehicle over one-ton capacity to park such vehicle or to permit it to be parked on any street, alley, or other public way in the Town for longer than thirty (30) minutes. The provisions of this section shall not apply to any vehicle while actually engaged in loading or unloading.
- (c) Except when actually loading or unloading, or as otherwise expressly permitted by law, no vehicle on a two-way street shall be parked or stopped with its left side adjacent to the curb.
- (d) Except in an emergency or to allow another vehicle or a pedestrian to cross, no vehicle shall be stopped or parked in any street, except close to and parallel with the curb. In no case shall any vehicles be parked less than four (4) feet apart. In the absence of lines marked on a street to designate the limits of a curbside parking space, no vehicle shall be stopped or parked with the curbside wheels further than six (6) inches from the curb, except where parking regulations provide for parking at an angle to the curb.
  - (e) Vehicles shall not be parked two (2) or more abreast parallel with the curb.
- (f) No vehicle shall be parked within twenty (20) feet of a street corner. The distance shall be measured from the point where the tangent line to radius of the curb is parallel with the roadway.
- (g) On all streets marked with lines on the pavement indicating spaces for the parking of vehicles, the spaces between each two (2) lines shall constitute parking room for only one (1) four-wheeled vehicle or two (2) motorcycles or moped/electric powered vehicles. No vehicle parked in such space shall be parked with wheels across the line indicating the boundaries of the space.

- (h) It shall be unlawful for any person to park a vehicle on or near a sidewalk or pedestrian crossing in such a way as to block or obstruct the free passage of pedestrians or with any portion of the vehicle on or overhanging any portion of the sidewalk.
- (i) It shall be unlawful for any person to park any vehicle alongside any curb painted yellow.
- (j) No vehicle shall be backed up to a curb, except during the time merchandise is actually being loaded or unloaded.

### § 112.15 PARKED VEHICLES; CURRENT LICENSE REQUIRED

It shall be unlawful for any person to park, keep, or permit to be parked or kept any motor vehicle, trailer, or semi trailer in or on any public highway, street, alley, public easement, or other public thoroughfare in the Town, unless:

- (a) The motor vehicle shall be currently inspected and approved in accordance with the provisions of any applicable laws of Virginia or another state;
- (b) The vehicle shall be currently registered under the applicable laws of Virginia or another state and licensed to be operated upon the highways of Virginia in accordance with the provisions of the laws of Virginia, including reciprocity laws; and
- (c) The vehicle shall meet any applicable requirement currently licensed to be operated upon the highways of the Town in accordance with § 56.18, or under a similar applicable ordinance of another Virginia locality.
- (d) Any Town vehicle license fee due in connection with such vehicle must have been paid.

# § 112.16 ROAD CONSTRUCTION WORKERS; EXEMPTIONS

The provisions of this chapter shall not apply to any vehicle owned or controlled by the state Department of Highways and Transportation or the Town, while actually engaged in the construction, reconstruction or maintenance of streets or highways.

#### § 112.17 PARKING ON INCLINE

Whenever a motor vehicle is left standing on an incline, it shall be so parked that when the brake is released the curb will act as a check to prevent its movement, except under its own power; or so arranged as to prevent movement upon release of brake.

# § 112.18 OWNERSHIP OF VEHICLE

In any prosecution for a violation of this Chapter, proof that the vehicle described in the complaint, summons, parking ticket, citation, or warrant, was parked in violation of the ordinance or regulation, together with proof that the defendant was at the time the registered owner of the vehicle, shall constitute in evidence a prima facie presumption that the registered owner of the vehicle was the person who committed the violation.

#### SUBCHAPTER B – PARKING FOR PERSONS WITH DISABILITIES

#### § 112.21 **GENERALLY**

It shall be unlawful for a vehicle not displaying disabled parking license plates, an organizational removable windshield placard, a permanent removable windshield placard, or a temporary removable windshield placard issued under Va. Code § 46.2-1241, or DV disabled parking license plates issued under Va. Code § 46.2-739(B), to be parked in a parking space reserved for persons with disabilities that limit or impair their ability to walk or for a person who is not limited or impaired in his ability to walk to park a vehicle in a parking space so designated except when transporting a person with such a disability in the vehicle.

#### § 112.22 PENALTY

Any person who violates Section § 112.21 shall be guilty of a traffic infraction and upon due conviction thereof shall be fined an amount not less than \$100 but not more than \$500.

## § 112.23 PROOF OF VIOLATION

In any prosecution under § 112.21, proof that the vehicle described in the complaint, summons, parking ticket, citation, or warrant was parked in violation of such section, together with proof that the defendant was at the time the registered owner of the vehicle, as required under state law, shall constitute prima facie evidence that the registered owner of the vehicle was the person who committed the violation.

#### § 112.24 ABSENCE OF SIGN

No violation of § 112.21 shall be dismissed for a property owner's failure to comply strictly with the requirements for disabled parking signs, provided the space is clearly distinguishable as a parking space reserved for persons with disabilities that limit or impair their ability to walk.

# § 112.25 ADOPTION BY REFERENCE OF CERTAIN PROVISIONS OF STATE LAW

The provisions of § Va. Code §§ 46.2-1247, 1248, 1249, 1250, 1251, 1252, and 1253, as amended and in force from time to time, are adopted by reference and are incorporated into this subchapter as if fully set out herein. Words and phrases used and contained in this subchapter shall have the meanings ascribed to them by § 46.2-100 of the Code of Virginia, as amended and in force from time to time, except where the context clearly requires a different meaning. All future amendments to the sections of the Code of Virginia incorporated by reference under the provisions of this paragraph shall be included within the scope of this paragraph automatically upon their effective date, without any formal Town Council action.

#### CHAPTER 113: OTHER PROVISIONS REGARDING MOTOR VEHICLES

# § 113.01 ABANDONED MOTOR VEHICLES

- (a) An "abandoned motor vehicle" is any motor vehicle, trailer, or semi trailer that:
- (i) is left unattended on public property for more than 48 hours, except where permitted by state or local law;
- (ii) has remained for more than 48 hours on private property without the consent of the property's owner, regardless of whether it was brought onto the private property with the consent of the owner or person in control of the private property, or
  - (ii) is left unattended on the shoulder of a primary highway.
- (b) No person shall cause a motor vehicle to become an abandoned motor vehicle within the Town. In any prosecution for a violation of this section, proof that the defendant was, at the time that the vehicle was found abandoned, the owner of the vehicle shall constitute in evidence a rebuttable presumption that the owner was the person who committed the violation. Such presumption, however, shall not arise if the owner of the vehicle provided notice to the Commonwealth, as provided under state law, that he had sold or otherwise transferred the ownership of the vehicle.
- (c) The Town Manager or his designee is authorized to remove any abandoned motor vehicle and to dispose of such vehicle using the procedures authorized under state law.
- (d) Any person who violates § 113.01(b) shall be subject to a civil penalty not to exceed \$500.

## § 113.02 UNATTENDED OR IMMOBILE VEHICLES

- (a) The Chief of Police is authorized to remove or to direct the removal for safekeeping any motor vehicle, trailer, semi trailer, or part thereof when:
- (i) It is left unattended on a public highway or other public property and constitutes a traffic hazard;
  - (ii) It is illegally parked;
- (iii) It is left unattended for more than 10 days either on public property or on private property without the permission of the property owner, lessee, or occupant;

- (iv) It is immobilized on a public roadway by weather conditions or other emergency situation.
  - (b) The Chief of Police shall promptly report such removal to the Town Manager.
- (c) No such motor vehicle, trailer, semi trailer, or part thereof shall be removed from private property except at the written request of the owner, lessee, or occupant of such property. The person at whose request the motor vehicle, trailer, semitrailer, or part of a motor vehicle, trailer, or semitrailer is removed from private property shall indemnify the Town against any loss or expense incurred by reason of removal, storage, or sale thereof.
- (d) Any motor vehicle, trailer, semi trailer, or part thereof described in § 113.02(a) shall be considered an abandoned vehicle, and subject to the additional provisions of § 113.01, when it:
- (i) lacks either a current license plate; or a current county, city or town license plate or sticker; or a valid state safety inspection certificate or sticker; and
  - (ii) has been in a specific location for four days without being moved.
- (e) No person shall recover possession of a motor vehicle, trailer, semi trailer, or part thereof removed under § 113.02(a) without paying to the Town all costs incidental to its removal and storage and locating the owner. If such person fails or refuses to pay the cost or if his identity or whereabouts is unknown and unascertainable after a diligent search has been made, and after notice to him at his last known address and to the holder of any lien of record with the state Department of Motor Vehicles against the motor vehicle, trailer, semitrailer, or part of a motor vehicle, trailer, or semitrailer, the vehicle shall be treated as an abandoned vehicle under the provisions of § 113.01.
- (f) The authority and powers granted under this § 113.02 are in addition to the authority and powers granted in § 113.01.

#### § 113.03 REMOVAL OF VEHICLES IN ACCIDENTS

Whenever a motor vehicle, trailer, or semitrailer involved in an accident is so located as to impede the orderly flow of traffic, Chief of Police or his specifically authorized designee may:

- (a) at no cost to the owner or operator remove the motor vehicle, trailer, or semitrailer to some point in the vicinity where it will not impede the flow of traffic; or
- (b) have the vehicle removed to a storage area for safekeeping and shall report the removal to the state Department of Motor Vehicles, the Town Manager, and to the owner of the vehicle as promptly as possible. If the vehicle is removed to a storage area under this provision, the owner shall pay to the parties entitled thereto all costs incidental to its removal and storage.

# § 113.04 REMOVAL OF VEHICLES ON PRIVATE PROPERTY

On complaint of the owner of the property on which such motor vehicle, trailer, semitrailer, or part thereof has been left for more than 72 hours, that such motor vehicle, trailer, semitrailer, or part thereof, may be removed to a storage area by or under the direction of the Chief of Police or his designee who specifically is authorized to do so by him. The owners of private property which is normally open to the public for parking shall post or cause to be posted signs warning that vehicles left on the property for more than 72 hours will be towed or removed at their owners' expense. The person at whose request the vehicle, trailer, semitrailer, or part thereof is so removed shall indemnify the Town against any loss or expense incurred by reason of removal, storage, or sale thereof. Any motor vehicle, trailer, semitrailer, or part thereof shall be disposed of as if it were a vehicle removed under § 113.02.

# CHAPTER 114: PENALTIES FOR VIOLATION OF TRAFFIC OR PARKING REGULATIONS

# § 114.01 ENFORCEMENT BY OFFICERS; OFFICERS TO BE UNIFORMED; OFFICERS TO BE PAID FIXED AND DETERMINED SALARIES

Every policeman shall enforce the provisions of this Title; provided, that such officer shall be uniformed at the time of such enforcement or shall display his badge or other sign of authority; and provided further, that all officers making arrests incident to the enforcement of this Title shall be paid fixed and determined salaries for their services and shall have no interest in, nor be permitted by law to accept the benefit of, any fine or fee resulting from the arrest or conviction of an offender against any provision of this Title.

# § 114.02 COMPLIANCE WITH CHAPTERS 111 AND 112; PENALTY FOR VIOLATION OF CHAPTERS

- (a) Any person who refuses, fails or neglects to comply with any of the provisions of this Chapters 111 (the "Traffic Ordinance") and 112 (the "Parking Ordinance") or any rule or regulation promulgated pursuant thereto, shall be guilty of a traffic infraction.
- (b) Whenever a specific penalty is provided in the Code of Virginia or a uniform fine schedule adopted thereunder for the violation of any section of the Code of Virginia adopted by reference under the Traffic or Parking Ordinance, the penalty for a violation of such section, charged on a Town warrant, shall be the same as provided by state statute.
- (c) Every person convicted of a violation of any of the provisions of the Parking Ordinance, or any rule or regulation promulgated pursuant thereto, for which no other penalty is provided, shall be fined twenty dollars (\$20.00).
- (d) Every person convicted of a violation of any of the provisions of the Traffic Ordinance, or any rule or regulation promulgated pursuant thereto, for which no other penalty is provided, for a first conviction thereof, be punished by a fine of not less than twenty dollars (\$20.00) nor more than one hundred dollars (\$100.00); for a second such conviction within one (1) year, such person shall be punished by a fine of not less than fifty dollars (\$50.00) nor more than two hundred dollars (\$200.00); for a third or subsequent conviction within one (1) year, such person shall be punished by a fine of not less than one hundred dollars (\$100.00) nor more than two hundred fifty dollars (\$500.00).

#### § 114.03 DISPOSITION OF FINES

All fines or forfeitures collected upon a finding of violations of ordinance, or upon the forfeiture of bail of any person charged with violation of any of the provisions of this Title, shall be paid into the Town's treasury and deposited in the general fund.

# § 114.04 PROCEDURES FOR SUMMONS, ARRESTS, ETC.

Procedures for the issuance of summonses, bail, arrests of offenders, and similar matters shall be as prescribed in state law for the regulation of traffic, parking, and motor vehicle offenses.

# TITLE XIII: UTILITIES

# Chapter

- 131. SEWER AND WASTEWATER ORDINANCE
- 132. WATER ORDINANCE
- 133. DROUGHT MANAGEMENT
- 134. SOLID WASTE ORDINANCE
- 135. RATES AND CHARGES

#### CHAPTER 131: SEWER AND WASTEWATER ORDINANCE

## § 131.01 PURPOSE AND POLICY

- (a) This Chapter sets forth uniform requirements for Users of the Publicly Owned Treatment Works for the Town and enables the Town to comply with all applicable State and Federal laws, including the Clean Water Act (33 U.S.C. § 1251 *et seq.*) and the Virginia Pollutant Discharge Elimination System (VPDES) Permit Regulation (9 VAC 25-31-10 *et seq.*). The objectives of this Chapter are:
- (i) To prevent the introduction of pollutants into the Publicly Owned Treatment Works that will interfere with its operation;
- (ii) To prevent the introduction of pollutants into the Publicly Owned Treatment Works that will pass through the Publicly Owned Treatment Works, inadequately treated, into receiving waters, or otherwise be incompatible with the Publicly Owned Treatment Works;
- (iii) To protect both Publicly Owned Treatment Works personnel who may be affected by wastewater and sludge in the course of their employment, and the general public;
- (iv) To improve the opportunity to reuse and recycle industrial wastewater and sludge from the Publicly Owned Treatment Works;
- (v) To provide for recovery of the cost of operation, maintenance, and improvement of the Publicly Owned Treatment Works; and
- (vi) To enable the Town to comply with its Virginia Pollutant Discharge Elimination System (VPDES) permit conditions, sludge use and disposal requirements, and any other federal or Virginia laws to which the Publicly Owned Treatment Works is subject.
- (b) This Chapter shall apply to all Users of the Publicly Owned Treatment Works. This Chapter authorizes the issuance of industrial user discharge permits; provides for monitoring, compliance, and enforcement activities; establishes administrative review procedures; and requires User reporting.
- (c) This Chapter shall be liberally construed to accomplish the objectives described in this section.

#### § 131.02 ADMINISTRATION

Except as otherwise provided herein, and subject to the supervision of the Mayor, the Town Manager shall administer, implement, and enforce the provisions of this Chapter. The Town Manager may delegate any power or authority granted to or duties imposed upon him to another Town agent, officer or employee.

#### § 131.03 ABBREVIATIONS

The following abbreviations, when used in this Chapter, shall have the designated meanings:

ASTM - American Society for Testing of Materials

BOD - Biochemical Oxygen Demand

BMP - Best Management Practice

BMR - Baseline Monitoring Report

CFR - Code of Federal Regulations

CIU - Categorical Industrial User

COD - Chemical Oxygen Demand

EPA - U.S. Environmental Protection Agency

gpd - gallons per day

IU - Industrial User

mg/l - milligrams per liter

NSCIU - Non-Significant Categorical Industrial User

POTW - Publicly Owned Treatment Works

RCRA - Resource Conservation and Recovery Act

SIU - Significant Industrial User

SNC - Significant Noncompliance

TSS - Total Suspended Solids

TRC – Technical Review Criteria

U.S.C. - United States Code

WPCF - Water Pollution Control Facilities

VDEQ – Virginia Department of Environmental Quality

VPDES - Virginia Pollutant Discharge Elimination System

#### § 131.04 DEFINITIONS

Unless a provision explicitly states otherwise, the following words and phrases, as used in this Chapter, and whether capitalized or not, shall have the meanings hereinafter designated.

- (a) Act. The Federal Water Pollution Control Act, also known as the Clean Water Act, as amended, 33 U.S.C. § 1251 *et seq*.
  - (b) Approval Authority. Virginia Department of Environmental Quality.
  - (c) Authorized or Duly Authorized Representative of the User.
    - (i) If the User is a corporation:

- (A) The president, secretary, treasurer, or a vice-president of the corporation in charge of a principal business function, or any other person who performs similar policy or decision-making functions for the corporation; or
- (B) The manager of one or more manufacturing, production, or operating facilities, provided the manager is authorized to make management decisions that govern the operation of the regulated facility including having the explicit or implicit duty of making major capital investment recommendations, and initiate and direct other comprehensive measures to assure long-term environmental compliance with environmental laws and regulations; can ensure that the necessary systems are established or actions taken to gather complete and accurate information for industrial user discharge permit requirements; and where authority to sign documents has been assigned or delegated to the manager in accordance with corporate procedures.
- (ii) If the User is a partnership or sole proprietorship: a general partner or proprietor, respectively.
- (iii) If the User is a Federal, State, or local governmental facility: a director or highest official appointed or designated to oversee the operation and performance of the activities of the government facility, or their designee.
- (iv) The individuals described in paragraphs (i) through (iii), above, may designate a different Duly Authorized Representative than the ones specified above if the authorization is in writing, the authorization specifies the individual or position responsible for the overall operation of the facility from which the discharge originates or having overall responsibility for environmental matters for the company, and the written authorization is submitted to the Town.
- (d) <u>Biochemical Oxygen Demand</u>. The quantity of oxygen utilized in the biochemical oxidation of organic matter under standard laboratory procedures for five (5) days at 20 degrees centigrade, usually expressed as a concentration (e.g., mg/I).
- (e) <u>Best Management Practices</u>. Schedules of activities, prohibitions of practices, maintenance procedures, and other management practices to implement the prohibitions listed in § 131.07 (9 VAC 25-31-770.A.1. and B.). BMPs include treatment requirements, operating procedures, and practices to control plant site runoff, spillage or leaks, sludge or waste disposal, or drainage from raw materials storage. BMPs also include alternative means (i.e., management plans) of complying with, or in place of certain established categorical Pretreatment Standards and effluent limits.
- (f) <u>Building Drain</u>. Part of the lowest horizontal piping of a drainage system which receives the discharge from soil, waste, and other drainage pipes inside the walls of the building and conveys it to the building sewer, beginning five (5) feet outside the inner face of the building wall.

- (g) <u>Building Sewer</u>. The extension from the building drain to the public sewer or other place of disposal, also called house connection.
- (h) <u>Categorical Pretreatment Standard or Categorical Standard</u>. Any regulation containing pollutant discharge limits promulgated by EPA in accordance with sections 307(b) and (c) of the Act (33 U.S.C. § 1317) that apply to a specific category of Users and that appear in 40 CFR Chapter I, Subchapter N, Parts 405-471.
- (i) <u>Categorical Industrial User</u>. An Industrial User subject to a categorical Pretreatment Standard or Categorical Standard.
- (j) <u>Chemical Oxygen Demand</u>. A measure of the oxygen required to oxidize all compounds, both organic and inorganic, in water.
  - (k) <u>Control Authority</u>. The Town of Broadway.
- (l) <u>Daily Maximum</u>. The arithmetic average of all effluent samples for a pollutant collected during a calendar day.
- (m) <u>Daily Maximum Limit</u>. The maximum allowable discharge limit of a pollutant during a calendar day. Where Daily Maximum Limits are expressed in units of mass, the daily discharge is the total mass discharged over the course of the day. Where Daily Maximum Limits are expressed in terms of a concentration, the daily discharge is the arithmetic average measurement of the pollutant concentration derived from all measurements taken that day.
  - (n) Easement. An acquired legal right for the specific use of land owned by others.
- (o) <u>Environmental Protection Agency</u>. The United States Environmental Protection Agency or, where appropriate, the Regional Water Management Division Director, the Regional Administrator, or other duly authorized official of said agency.
  - (p) <u>Existing Source</u>. Any source of discharge that is not a "New Source."
- (q) <u>Garbage</u>. The animal and vegetable waste resulting from the handling, preparation, cooking, and serving of foods.
- (r) <u>Grab Sample</u>. A sample that is taken from a waste stream without regard to the flow in the waste stream and over a period of time not to exceed fifteen (15) minutes.
- (s) <u>Indirect Discharge or Discharge</u>. The introduction of pollutants into the POTW from any nondomestic source.

- (t) <u>Instantaneous Limit</u>. The maximum concentration of a pollutant allowed to be discharged at any time, determined from the analysis of any discrete or composited sample collected independent of the industrial flow rate and the duration of the sampling event.
- (u) <u>Interference</u>. A discharge that, alone or in conjunction with a discharge or discharges from other sources, inhibits or disrupts the POTW, its treatment processes or operations or its sludge processes, use or disposal; and therefore, is a cause of a violation of the Town's VPDES permit or of the prevention of sewage sludge use or disposal in compliance with any of the following statutory or regulatory provisions or permits issued hereunder, or any more stringent Virginia or Town regulations: Section 405 of the Act; the Solid Waste Disposal Act, including Title II commonly referred to as the Resource Conservation and Recovery Act (RCRA); any Virginia regulations contained in any Virginia sludge management plan prepared pursuant to Subtitle D of the Solid Waste Disposal Act; the Clean Air Act; the Toxic Substances Control Act; and the Marine Protection, Research, and Sanctuaries Act.
- (v) <u>Local Limit</u>. Specific discharge limits developed and enforced by the Town upon industrial or commercial facilities to implement the general and specific discharge prohibitions listed in 9 VAC 25-31-770.A.1. and B.
- (w) <u>Medical Waste</u>. Isolation wastes, infectious agents, human blood and blood products, pathological wastes, sharps, body parts, contaminated bedding, surgical wastes, potentially contaminated laboratory wastes, and dialysis wastes.
- (x) <u>Monthly Average</u>. The sum of all "daily discharges" measured during a calendar month divided by the number of "daily discharges" measured during that month.
- (y) <u>Monthly Average Limit</u>. The highest allowable average of "daily discharges" over a calendar month, calculated as the sum of all "daily discharges" measured during a calendar month divided by the number of daily discharges" measured during that month.
- (z) <u>Natural Outlet</u>. Any outlet, including storm sewers, into a watercourse, pond, ditch, lake, or other body of surface or groundwater.

#### (aa) New Source.

- (i) Any building, structure, facility, or installation from which there is (or may be) a discharge of pollutants, the construction of which commenced after the publication of proposed Pretreatment Standards under section 307(c) of the Act that will be applicable to such source if such Standards are thereafter promulgated in accordance with that section, provided that:
- (A) The building, structure, facility, or installation is constructed at a site at which no other source is located; or

- (B) The building, structure, facility, or installation totally replaces the process or production equipment that causes the discharge of pollutants at an Existing Source; or
- (C) The production or wastewater generating processes of the building, structure, facility, or installation are substantially independent of an Existing Source at the same site. In determining whether these are substantially independent, factors such as the extent to which the new facility is integrated with the existing plant, and the extent to which the new facility is engaged in the same general type of activity as the Existing Source, should be considered.
- (ii) Construction on a site at which an Existing Source is located results in a modification rather than a New Source if the construction does not create a new building, structure, facility, or installation meeting the criteria of Section (i)(B) or (i)(C) above but otherwise alters, replaces, or adds to existing process or production equipment.
- (iii) Construction of a New Source as defined under this paragraph has commenced if the owner or operator has:
- (A) Begun, or caused to begin, as part of a continuous onsite construction program,
- (1) any placement, assembly, or installation of facilities or equipment; or
- (2) significant site preparation work including clearing, excavation, or removal of existing buildings, structures, or facilities which is necessary for the placement, assembly, or installation of new source facilities or equipment; or
- (B) Entered into a binding contractual obligation for the purchase of facilities or equipment which are intended to be used in its operation within a reasonable time. Options to purchase or contracts which can be terminated or modified without substantial loss, and contracts for feasibility, engineering, and design studies do not constitute a contractual obligation under this paragraph.
- (bb) <u>Noncontact Cooling Water</u>. Water used for cooling that does not come into direct contact with any raw material, intermediate product, waste product, or finished product.
- (cc) <u>Pass Through</u>. A discharge which exits the POTW into waters of the United States in quantities or concentrations which, alone or in conjunction with a discharge or discharges from other sources, is a cause of a violation of any requirement of the Town VPDES permit, including an increase in the magnitude or duration of a violation.
- (dd) <u>Person</u>. Any individual, partnership, co-partnership, firm, company, limited liability company, corporation, association, joint stock company, trust, estate, governmental entity,

or any other legal entity; or their legal representatives, agents, or assigns. This definition includes all Federal, State, and local governmental entities.

- (ee) pH. A measure of the acidity or alkalinity of a solution, expressed in standard units.
- (ff) <u>Pollutant</u>. Dredged spoil, solid waste, incinerator residue, filter backwash, sewage, garbage, sewage sludge, munitions, Medical Wastes, chemical wastes, biological materials, radioactive materials, heat, wrecked or discarded equipment, rock, sand, cellar dirt, municipal, agricultural and industrial wastes, and certain characteristics of wastewater (e.g., pH, temperature, TSS, turbidity, color, BOD, COD, toxicity, or odor).
- (gg) <u>Pretreatment</u>. The reduction of the amount of pollutants, the elimination of pollutants, or the alteration of the nature of pollutant properties in wastewater prior to, or in lieu of, introducing such pollutants into the POTW. This reduction or alteration can be obtained by physical, chemical, or biological processes; by process changes; or by other means, except by diluting the concentration of the pollutants unless allowed by an applicable Pretreatment Standard.
- (hh) <u>Pretreatment Requirements</u>. Any substantive or procedural requirement related to pretreatment imposed on a User, other than a Pretreatment Standard.
- (ii) <u>Pretreatment Standards or Standards</u>. Pretreatment Standards shall mean prohibited discharge standards, Categorical Pretreatment Standards, and Local Limits.
- (jj) <u>Prohibited Discharge Standards or Prohibited Discharges</u>. Absolute prohibitions against the discharge of certain substances as described in this Chapter.
- (kk) <u>Publicly Owned Treatment Works</u>. A treatment works, as defined by section 212 of the Act (33 U.S.C. § 1292), which is owned by the Town. This definition includes any devices or systems used in the collection, storage, treatment, recycling, and reclamation of sewage or industrial wastes of a liquid nature and any conveyances, which convey wastewater to a treatment plant.
- (ll) <u>Sanitary Sewer</u>. Any sewer that carries liquid and water-carried wastes from residences, commercial buildings, industrial plants, and institutions together with minor quantities of ground, storm, and surface waters that are not admitted intentionally.
- (mm) <u>Septic Tank Waste</u>. Any sewage from holding tanks such as vessels, chemical toilets, campers, trailers, and septic tanks.
- (nn) <u>Sewage</u>. Human excrement and gray water (household showers, dishwashing operations, etc.).
- (00) <u>Significant Industrial User</u>. Except as provided in paragraphs (iii) and (iv) of this paragraph, a Significant Industrial User is:

- (i) An Industrial User subject to categorical Pretreatment Standards;
- (ii) An Industrial User that:
- (A) Discharges an average of twenty-five thousand (25,000) gpd or more of process wastewater to the POTW (excluding sanitary, noncontact cooling and boiler blowdown wastewater);
- (B) Contributes a process wastestream which makes up five (5) percent or more of the average dry weather hydraulic or organic capacity of the POTW treatment plant; or
- (C) Is designated as such by the Town on the basis that it has a reasonable potential for adversely affecting the POTW's operation or for violating any Pretreatment Standard or Requirement.
- (iii) The Town may determine that an Industrial User subject to categorical Pretreatment Standards is a Non-Significant Categorical Industrial User rather than a Significant Industrial User on a finding that the Industrial User never discharges more than 100 gallons per day (gpd) of total categorical wastewater (excluding sanitary, non-contact cooling and boiler blowdown wastewater, unless specifically included in the Pretreatment Standard) and the following conditions are met:
- (A) The Industrial User, prior to the Town's finding, has consistently complied with all applicable categorical Pretreatment Standards and Requirements;
- (B) The Industrial User annually submits the certification statement required in § 91.212 L (see 9 VAC 25-31-840.Q.), together with any additional information necessary to support the certification statement; and
- (C) The Industrial User never discharges any untreated concentrated wastewater.
- (iv) Upon a finding that a User meeting the criteria in Subsection (ii) of this paragraph has no reasonable potential for adversely affecting the POTW's operation or for violating any Pretreatment Standard or Requirement, the Town may at any time, on its own initiative or in response to a petition received from an Industrial User, and in accordance with procedures in 9 VAC 25-31-800.F.6., determine that such User should not be considered a Significant Industrial User.
- (pp) <u>Slug Load or Slug Discharge</u>. Any Discharge at a flow rate or concentration, which could cause a violation of the prohibited discharge standards described in § 131.07 of this Chapter. A Slug Discharge includes any Discharge of a non-routine, episodic nature, including but not limited to an accidental spill or a non-customary batch discharge, which has a reasonable potential

to cause Interference or Pass Through, or in any other way violate the POTW's regulations, Local Limits or Permit conditions.

- (qq) <u>Storm Water</u>. Any flow occurring during or following any form of natural precipitation, and resulting from such precipitation, including snowmelt.
- (rr) <u>Total Suspended Solids or Suspended Solids</u>. The total suspended matter that floats on the surface of, or is suspended in, water, wastewater, or other liquid, and that is removable by laboratory filtering.
  - (ss) <u>Town</u>. The Town of Broadway, Virginia, which serves as the Control Authority.
- (tt) <u>Town Manager</u>. The Town Manager for the Town, or his or her duly authorized agent designed to administer, implement, and enforce the provisions of this Chapter.
  - (uu) User or Industrial User. A source of indirect discharge.
- (vv) <u>Wastewater</u>. Liquid and water-carried industrial wastes and sewage from residential dwellings, commercial buildings, industrial and manufacturing facilities, and institutions, whether treated or untreated, which are contributed to the POTW.
- (ww) <u>Wastewater Treatment Plant or Treatment Plant</u>. That portion of the POTW which is designed to provide treatment of municipal sewage and industrial waste. This definition includes, without limitation, the Town's wastewater treatment plant located at 5524 New Market Road, Timberville, Virginia.
- (xx) A reference in this Chapter to any repealed or recodified statute, ordinance, or regulation shall be taken as a reference to the corresponding successor statute, ordinance, or regulation.

# § 131.05 USE OF PUBLIC SEWERS REQUIRED; LIMITED EXCEPTIONS

- (a) It shall be unlawful for any person to place, deposit, or permit to be deposited in an unsanitary manner on public or private property within the Town or in any area under its jurisdiction any human or animal excrement, garbage, or objectionable waste.
- (b) It shall be unlawful for any person to discharge to any natural outlet within the Town, or in any area under its jurisdiction, any wastewater or other polluted waters, except where suitable treatment has been provided in accordance with subsequent provisions of this Chapter.
- (c) Except as hereinafter provided, it shall be unlawful for any person to construct or maintain any privy, privy vault, septic tank, cesspool, or other facility intended or used for the disposal of wastewater.

- (d) The owner of any house, structure, buildings or camp used for human occupancy, employment, recreation, or other purposes, situated within the Town and abutting on any street, alley, or right-of-way, is hereby required to install, at the owner's expense, suitable toilet facilities therein, and to connect such facilities directly with the public sewer in accordance with the provisions of this Code, within (90) days after the date of official notice to do so, unless the Town Manager has determined that it is not feasible to make public sewer service available. This paragraph shall not apply to any property serviced by a private wastewater disposal system on the date that the Town determined to require connection to the public sewer system, and which has continually maintained such private wastewater disposal system since that date.
- (e) No connections to the public sewer outside the corporate limits of the Town shall be made without the express approval of the Town Council.
- (f) When a public sewer is available to a property served by a private wastewater disposal system and such private system fails or malfunctions, the Town Manager upon determining that such failure or malfunction exists shall direct the owner to connect to the public sewer within ninety (90) days, and any septic tanks, cesspools, and similar private wastewater disposal facilities shall be cleared of sludge and filled with suitable material.
- (g) It shall be unlawful for any person to repair a private wastewater disposal system or any part thereof when public sewer is available, unless the Town Manager has determined that it is not feasible to make public sewer service available to such property.
- (h) No statement contained in the Code shall be construed to interfere with any additional requirements that may be imposed by the County health officer.
- (i) The Town Manager may discontinue sewer service to a consumer for repairs to the sewer system and shall give as much notice of such discontinuance to the consumer as is practicable under the circumstances.

#### § 131.06 BUILDING SEWERS AND CONNECTIONS

- (a) No unauthorized person shall uncover, make any connections with or opening into, use, alter, or disturb any public sewer or appurtenance thereof without first obtaining a written permit from the Town Manager.
- (b) All costs and expenses incidental to the installation and connection of a building sewer shall be borne by the owner. Each owner, jointly and severally, shall indemnify the Town from any loss or damage that may be directly or indirectly occasioned by the installation of the building sewer.
- (c) A separate and independent building sewer shall be provided for every building, except where one building stands at the rear of another on an interior lot and no private sewer is

available or can be constructed to the rear building through an adjoining alley, court, yard, or driveway. In such a circumstance the building sewer from the front building may be extended to the rear building and the whole considered one building sewer. The Town does not and will not assume any obligation or responsibility for damage caused by or resulting from any such connection aforementioned.

- (d) Old building sewers may be used in connection with new buildings only when they are found, on examination and test by the Town Manager to meet all requirements of this Chapter.
- (e) The size, slopes, alignment, materials of construction of a building sewer, and the methods to be used in excavating, placing of the pipe, jointing, testing, and backfilling of the trench, shall all conform to the requirements of the building and plumbing code or other applicable rules and regulations of the Town, including without limitation the Town's specifications manual, as the Town may amend such manual from time to time. In the absence of such provisions or in amplification thereof, the materials and procedures set forth in appropriate specifications of the ASTM and WPCF Manual of Practice No. 9, shall apply.
- (f) Whenever possible, the building sewer shall be brought to the building at an elevation below the basement floor. In all buildings in which any building drain is too low to permit gravity flow to the public sewer, sanitary sewage carried by such building drain shall be lifted by an approved means and discharged to the building sewer.
- (g) No person shall make connection of roof downspouts, foundation drains, areaway drains, or other sources of surface runoff or groundwater to a building sewer or building drain, which in turn is connected directly or indirectly to the POTW.
- (h) The connection of the building sewer into the public sewer shall conform to the requirements of the building and plumbing code or other applicable rules and regulations of the Town, including without limitation the Town's specifications manual (as the Town may amend such manual from time to time) or, to the extent applicable and not in conflict with the Town's specifications manual, and in amplification of all such standards, the procedures set forth in appropriate specifications of the ASTM and the WPCF Manual of Practice No. 9. All such connections shall be made gastight and watertight, and verified by proper testing to prevent the inclusion of infiltration/inflow. Any deviation from the prescribed procedures and materials, or the specifications manual, must be expressly approved by the Town in writing prior to installation and must specifically reference a waiver of the requirements of this section. No note on any approved subdivision plat or other submission from a developer to the Town shall constitute approval of any such deviation; rather, such approval may only be given in a separate writing specifically granting such approval, and executed by an authorized official of the Town.
- (i) The applicant for the building sewer permit shall notify the Town Manager when the building sewer is ready for inspection and connection to the public sewer. The connection and inspection shall be made under the supervision of the Town Manager or authorized representative thereof.

- (j) Prior to any excavation for the building of sewer installations, notice shall be given to all authorities having underground utility installations in the vicinity. All excavations for building sewer installation shall be adequately guarded with barricades and lights so as to protect the public from hazard. Streets, sidewalks, parkways, and other public property disturbed in the course of the work, shall be restored in a manner satisfactory to the Town and the relevant state authorities.
- (k) All main lines and service lines lying in public property or in a utility or public right of way easement shall follow the route prescribed by the Town Manager.
- (l) All connections shall be made by the Town or by a licensed plumber approved by the Town, at the expense of the applicant, who shall pay to the Town, or to the licensed plumber approved by the Town, the cost of making such connections, including all labor and materials necessary for such connections.
  - (m) The Town shall have the right to inspect all work done before service commences.
- (n) Developers of new properties to be served by the Town sewer system shall construct the main lines to the property being developed and shall construct them on a route prescribed by the Town. The sizes of lines, materials, and configuration and construction methods shall be in accordance with the Town's specifications manual (as the Town may amend such manual from time to time), shall be approved by the Town in advance of construction, and shall be subject to final acceptance contingent upon inspection and approval by the Town. The Town may connect to any necessary extensions from existing main lines to the property line of the property being developed.
- (o) The Town may require that main or service lines be dedicated to the Town and that easements or rights-of-way be granted for the benefit of the Town. If such dedication is accepted by the Town, the owner or installer of such lines so dedicated shall be responsible for defects in such lines for a period of five years, commencing on the date of dedication. The Town may require a bond covering such defects.
  - (p) Cost of Constructing Lines.
- (i) In connection with existing lots not created as part of a Major Subdivision, or lots created in the future as part of a Minor Subdivision, the Town shall run all sewer service lines at its cost under a public right of way or public property for a distance that is the lesser of (1) the distance between an existing sewer main and the applicant's property line; or (2) a distance of fifty (50) feet from the existing sewer main. If the distance exceeds fifty (50) feet, the applicant shall pay all costs of the extension beyond fifty (50) feet. The property owner shall pay all costs associated with the construction of the sewer line on its own property.

- (ii) If the need for sewer lines arose or arises because of a Major Subdivision, the applicant shall bear the cost of all sewer lines (mains, laterals, and service lines) associated with any necessary sewer extension.
- (iii) The terms "Major Subdivision" and "Minor Subdivision", as used in this section, shall have the same meaning as in the Land Development Regulations.

#### § 131.07 PROHIBITED DISCHARGE STANDARDS

- (a) <u>General and Specific Prohibitions</u>. No User shall introduce or cause to be introduced into the POTW any pollutant or wastewater, which causes Pass Through or Interference. These general prohibitions apply to all Users of the POTW whether or not they are subject to categorical Pretreatment Standards or any other National, State, or local Pretreatment Standards or Requirements. No person shall process or store Pollutants, substances, or wastewater prohibited by this Section 131.07 in such a manner that they could be discharged to the POTW. Furthermore, no user may contribute any of the following substances to the POTW:
- (i) Pollutants which create a fire or explosive hazard in the POTW, including, but not limited to, wastestreams with a closed-cup flashpoint of less than 140°F (60° C) using the test methods specified in 40 C.F.R. 261.21. Prohibited materials include but are not limited to gasoline, kerosene, naphtha, benzene, toluene, xylene, ethers, alcohols, ketones, aldehydes, peroxides, chlorates, perchlorates, bromates, carbides, hydrides, and sulfides;
- (ii) Wastewater having a pH less than 5.5 or more than 9, or otherwise causing corrosive structural damage to the POTW or equipment;
- (iii) Solid or viscous substances in amounts which will cause obstruction of the flow in the POTW resulting in Interference. Solids shall not exceed one-half inch (1/2") or one and twenty-seven hundredths centimeters (1.27 cm) in any dimension;
- (iv) Pollutants, including oxygen-demanding pollutants (BOD, etc.), released in a discharge at a flow rate and/or pollutant concentration which, either singly or by interaction with other pollutants, will cause Interference with the POTW;
- (v) Wastewater having a temperature greater than 150° F (66°C), or which will inhibit biological activity in the treatment plant resulting in Interference, but in no case wastewater which causes the temperature at the introduction into the treatment plant to exceed 104° F (40° C);
- (vi) Petroleum oil, non-biodegradable cutting oil, or products of mineral oil origin, in amounts that will cause Interference or Pass Through;
- (vii) Pollutants which result in the presence of toxic gases, vapors, or fumes within the POTW in a quantity that may cause acute worker health and safety problems;

- (viii) Trucked or hauled pollutants including septic tank waste;
- (ix) Noxious or malodorous liquids, gases, solids, or other wastewater which, either singly or by interaction with other wastes, are sufficient to create a public nuisance or a hazard to life, or to prevent entry into the sewers for maintenance or repair;
- (x) Wastewater which imparts color, which cannot be removed by the treatment process, such as, but not limited to, dye wastes and vegetable tanning solutions, which consequently imparts color to the treatment plant's effluent, thereby violating the Town 's VPDES permit;
- (xi) Wastewater containing any radioactive wastes or isotopes except specifically approved by the Town Manager in compliance with applicable State or Federal regulations;
- (xii) Storm Water, surface water, ground water, artesian well water, roof runoff, subsurface drainage, swimming pool drainage, condensate, deionized water, Noncontact Cooling Water, and unpolluted wastewater, unless specifically authorized by the Town Manager;
- (xiii) Sludges, screenings, or other residues from the pretreatment of industrial waste;
  - (xiv) Medical wastes, except as the Town Manager may specifically authorize;
- (xv) Wastewater causing, alone or in conjunction with other sources, the treatment plant's effluent to fail any applicable toxicity test;
- (xvi) Detergents, surface-active agents, or other substances which that might cause excessive foaming in the POTW;
- (xvii) Fats, oils, or greases of animal or vegetable origin in concentrations that might pass through at the treatment plant, thereby violating the Town's VPDES permit;
- (xviii) Wastewater causing two readings on an explosion hazard meter at the point of discharge into the POTW, or at any point in the POTW, of more than five percent (5%) or any single reading over ten percent (10%) of the Lower Explosive Limit of the meter.
- (xix) Any material identified as hazardous waste according to 40 C.F.R. 261 except as may be specifically authorized by the Town Manager.
- (xx) There shall be no discharge of bulk, expired, outdated, or concentrated prescriptions or non-prescription drugs, hazardous waste pharmaceuticals, or DEQ controlled

substances to the POTW by a healthcare facility or reverse distributor pursuant to 40 C.F.R. 266.505.

- (b) <u>National Categorical Pretreatment Standards</u>. Users must comply with the Categorical Pretreatment Standards found at 40 CFR Chapter I, Subchapter N, Parts 405--471. When wastewater subject to a categorical Pretreatment Standard is mixed with wastewater not regulated by the same Standard, the Town Manager shall impose an alternate limit in accordance with 9 VAC 25-31-780.E.
- (c) <u>State Pretreatment Standards</u>. Users must comply with Virginia Regulations codified at Part VII-General Pretreatment Regulations for Existing and New Sources of Pollution, 9VAC 25-31.

# (d) Local Limits.

- (i) In addition to local limits prescribed elsewhere in this Section 131.07, the Town Manager is authorized to establish Local Limits pursuant to 9 VAC 25-31-770.C.
- (ii) The following pollutant limits are established to protect against Pass Through and Interference. No person shall discharge into the POTW wastewater containing in excess of the following monthly average limits: 0.39 mg/1 chromium; 0.039 mg/1 copper; 0.79 mg/l lead; 0.39 mg/l nickel; 1849 mg/l total suspended solids; 0.31 mg/l zinc. The above limits apply at the point where the wastewater is discharged to the POTW. All concentrations for metallic substances are for total metal. The Town Manager may impose mass limitations in addition to the concentration-based limitations above.
- (iii) The Town may develop Best Management Practices (BMPs), by Chapter or in industrial user discharge permit, to implement any additional Local Limits and the requirements stated herein.
- (iv) The Town may set additional local limits in Industrial User Discharge Pennits, where necessary.
- (v) The Town reserves the right to establish, by amendment to this Code or in industrial user discharge permits, more stringent Standards or Requirements on discharges to the POTW consistent with the purpose of this Chapter. No User shall have a vested right in any particular Standard, Requirement, or discharge limit; rather, the Town may alter or amend such Standards, Requirements, or limits at any time.
- (e) <u>Dilution</u>. No User shall ever increase the use of process water, or in any way attempt to dilute a discharge, as a partial or complete substitute for adequate treatment to achieve compliance with a discharge limitation unless expressly authorized by an applicable Pretreatment Standard or Requirement. The Town Manager may impose mass limitations on Users who are

using dilution to meet applicable Pretreatment Standards or Requirements, or in other cases when the imposition of mass limitations is appropriate.

(f) <u>Planning for Plant Failure</u>. Users shall control production of all discharges to the extent necessary to maintain compliance with categorical Pretreatment Standards upon reduction, loss, or failure of its treatment facility until the facility is restored or an alternative method of treatment is provided. This requirement applies in the situation where, among other things, the primary source of power of the treatment facility is reduced, lost, or fails.

## § 131.08 VARIANCES AND AGREEMENTS

The Town shall have the right to enter into special agreements with Categorical Industrial Users setting forth terms under which the user may discharge into the POTW. In no case will a special agreement waive compliance with a pretreatment standard. However, the industrial user may request a net/gross adjustment to a categorical standard in accordance with 9 VAC 25-31-870 in the event that such industrial user would like to obtain credit for pollutants present in the industrial user's influent water. Industrial users may also request a fundamentally different factor variance from the Categorical Pretreatment Standards of the EPA. Such a request must be either approved or not objected to by the EPA and State. An industrial user requesting a fundamentally different factor variance must comply with procedural and substantive provisions set forth in 9 VAC 25-31-850.

#### § 131.09 VANDALISM; UNLAWFUL ENTRY

- (a) No person shall maliciously, willfully, or negligently break, damage, destroy, uncover, deface, interfere with the proper functioning of, or tamper with any structure, appurtenance or equipment which is a part of the Town's POTW or any other element of the Town's sewer or wastewater system.
- (b) No person shall, without lawful authority, climb over, get through or break into the enclosure of the Town's POTW any other facility associated with the Town's water system.

#### § 131.10 PRETREATMENT OF WASTEWATER

(a) Pretreatment Facilities. Users shall provide wastewater treatment as necessary to comply with this Chapter and shall achieve compliance with all Categorical Pretreatment Standards, Local Limits, and the prohibitions set out in § 131.07 of this Chapter within the time limitations specified by EPA, the State, or the Town Manager, whichever is most stringent. Any facilities necessary for compliance shall be provided, operated, and maintained at the User's expense. Detailed plans describing such facilities and operating procedures shall be submitted to the Town Manager for review, and shall be acceptable to the Town Manager before such facilities are constructed. The review of such plans and operating procedures shall in no way relieve the

User from the responsibility of modifying such facilities as necessary to produce a discharge acceptable to the Town under the provisions of this Chapter.

## (b) Additional Pretreatment Measures.

- (i) Whenever he deems necessary, the Town Manager may require Users to restrict their discharge during peak flow periods, designate that certain wastewater be discharged only into specific sewers, relocate and/or consolidate points of discharge, separate sewage waste streams from industrial waste streams, and impose such other conditions as may be necessary to protect the POTW and determine the User's compliance with the requirements of this Chapter.
- (ii) The Town Manager may require any person discharging into the POTW to install and maintain, on their property and at their expense, a suitable storage and flow control facility to ensure equalization offlow. An industrial user discharge permit may be required and issued solely for flow equalization.
- (iii) Grease, oil, and sand interceptors shall be provided when, in the opinion of the Town Manager, they are necessary for the proper handling of wastewater containing excessive amounts of grease and oil, or sand; except that such interceptors shall not be required for residential users. All interception units shall be of a type and capacity approved by the Town Manager, and shall be so located to be easily accessible for cleaning and inspection. Such interceptors shall be inspected, cleaned, and repaired by the User at such User's expense. The User shall maintain records of the dates and means of disposal used, which shall be made available to the Town Manager upon request. Any removal and hauling of the collected materials not performed by the User through its own employees must be performed by a currently licensed waste disposal film.
- (iv) The Town Manager may require Users with the potential to discharge flammable substances to install and maintain an approved combustible gas detection meter.
- (c) <u>Accidental Discharge/Slug Discharge Control Plans</u>. The Town Manager shall evaluate whether each SIU needs an accidental discharge/slug discharge control plan or other action to control Slug Discharges. The Town Manager may require any User to develop, submit for approval, and implement such a plan or take such other action that may be necessary to control Slug Discharges. These actions may include construction of equipment or implementation of procedures that will significantly reduce the potential for occurrences of slug discharges. In the alternative, the Town Manager may develop such a plan for any User. An accidental discharge/slug discharge control plan shall address, at a minimum, the following circumstances:
  - (i) Description of discharge practices, including non-routine batch discharges;
  - (ii) Description of stored chemicals;
- (iii) Procedures for immediately notifying the Town Manager of any accidental or Slug Discharge, as required by § 131.27; and

(iv) Procedures to prevent adverse impact from any accidental or Slug Discharge. Such procedures include, but are not limited to, inspection and maintenance of storage areas, handling and transfer of materials, loading and unloading operations, control of plant site runoff, worker training, building of containment structures or equipment, measures for containing toxic organic pollutants, including solvents, and/or measures and equipment for emergency response.

#### § 131.11 PERMITS GENERALLY

No User shall uncover, make any connections with, use, alter, or disturb any sanitary sewer or storm sewer without first obtaining a written permit from the Town Manager. There are two classes of permits for connections to the POTW: (1) residential/commercial, and (2) industrial. In all cases, the owner shall make an application for a permit to connect to the POTW on a form that the Town Manager furnishes. The Owner shall submit the application with a connection fee established by the Town Council and appearing on the fee schedule attached to the Town Code. The Town may reject any applications for connection if they are not consistent with the long range growth plans of the Town or would adversely affect service to existing customers. The Town is not obligated to permit any User outside the limits of the Town to connect to the POTW absent explicit Town Council approval of such obligation.

#### § 131.12 INDUSTRIAL PERMITS

(a) <u>Industrial Waste Surveys</u>. When requested by the Town Manager, a User holding or seeking an industrial user discharge permit must submit information on the nature and characteristics of its wastewater by completing an industrial waste survey. The User has 60 days to complete and return the survey.

#### (b) <u>Industrial User Discharge Permit Requirements.</u>

- (i) No Significant Industrial User shall discharge wastewater into the POTW without first obtaining an industrial user discharge permit from the Town Manager, except that a Significant Industrial User that has filed a timely application pursuant to § 131.12(c) may continue to discharge for the time period specified therein.
- (ii) The Town Manager may require other Users to obtain industrial user discharge permits as necessary to carry out the purposes of this Chapter.
- (iii) Any violation of the terms and conditions of an industrial user discharge permit shall constitute a violation of this Chapter and subjects the wastewater discharge permittee to the sanctions set out in § 131.34 through § 131.37 of this Chapter. Obtaining an industrial discharge permit does not relieve a permittee of its obligation to comply with all Federal and State Pretreatment Standards or Requirements or with any other requirements of Federal, State, and local law.

- (c) <u>Industrial User Discharge Permitting: Existing Connections.</u> Any User required to obtain an industrial user discharge permit who was discharging wastewater into the POTW prior to the effective date of former Town Code Chapter 91 and who wishes to continue such discharges in the future, shall, within 90 days after such date, apply to the Town Manager for an industrial user discharge permit in accordance with § 131.12(d), and shall not cause or allow discharges to the POTW to continue after one hundred and eighty (180) days such date except in accordance with said discharge permit issued by the Town Manager.
- (d) <u>Industrial User Discharge Permitting: New Connections</u>. Any User required to obtain an industrial user discharge permit who proposes to begin or recommence discharging into the POTW must obtain such permit prior to the beginning or recommencing of such discharge. An application for this industrial user discharge permit, in accordance with § 131.12(e) must be filed at least 90 days prior to the date upon which any discharge will begin.
- (e) <u>Industrial User Discharge Permit Application Contents</u>. All Users required to obtain an industrial user discharge permit must submit a permit application. Incomplete or inaccurate applications will not be processed and will be returned to the User for revision. The Town Manager may require Users to submit all or some of the following information as part of a permit application:
- (i) Identifying information to include name and address of the facility, including the name of the operator and owner.
- (ii) Contact information, and description of activities, facilities, and plant production processes on the premises;
- (iii) Environmental permits to include a list of any environmental control permits held by or for the facility;
  - (iv) A description of operations to include:
- (A) A brief description of the nature, average rate of production (including each product produced by type, amount, processes, and rate of production), and standard industrial classifications of the operation(s) carried out by such User. This description should include a schematic process diagram, which indicates points of discharge to the POTW from the regulated processes;
- (B) Types of wastes generated, and a list of all raw materials and chemicals used or stored at the facility which are, or could accidentally or intentionally be, discharged to the POTW;

- (C) Number and type of employees, hours of operation, and proposed or actual hours of operation; type and amount of raw materials processed (average and maximum per day);
- (D) Site plans, floor plans, mechanical and plumbing plans, and details to show all sewers, floor drains, and appurtenances by size, location, and elevation, and all points of discharge; and
- (E) flow measurements to include information showing the measured daily and maximum daily flow in gallons per day to the POTW from regulated process streams and other streams, as necessary.

#### (v) Measurement of Pollutants, to include:

- (A) The categorical Pretreatment Standards applicable to each regulated process and any new categorically regulated processes for Existing Sources.
- (B) The results of sampling and analysis identifying the nature and concentration, anc1lor mass, where required of regulated pollutants in the discharge from each regulated process.
- (C) Instantaneous, Daily Maximum, and long-term average concentrations, or mass, where required, shall be reported.
- (D) The sample shall be representative of daily operations and shall be analyzed in accordance with procedures set out in § 131.30 of this Chapter. Where the Standard requires compliance with a BMP or pollution prevention alternative, the User shall submit documentation as required by the Town Manager or the applicable Standards to determine compliance with the Standard.
- (E) Sampling must be performed in accordance with procedures set out in § 131.30 of this Chapter.
- (vi) Any requests for a monitoring waiver (or a renewal of an approved monitoring waiver) for a pollutant neither present nor expected to be present in the discharge based on § 131.25(a) (9 VAC 25-31-840.E.2.).
- (vii) Any other information that the Town Manager may deem necessary to evaluate the permit application.
  - (f) <u>Application Signatories and Certifications.</u>

- (i) All wastewater discharge permit applications, User reports, and certification statements must be signed by an Authorized Representative of the User and contain the certification statement in § 131.30(i).
- (ii) If the designation of an Authorized Representative is no longer accurate because a different individual or position has responsibility for the overall operation of the facility or overall responsibility for environmental matters for the company, a new written authorization satisfying the requirements of this Section must be submitted to the Town Manager prior to or together with any reports to be signed by an Authorized Representative.
- (iii) A facility that the Town Manager determines to be a Non-Significant Categorical Industrial User must annually submit the signed certification statement in § 131.30(j)
- (g) <u>Industrial User Discharge Permit Decisions</u>. The Town Manager will evaluate the data that the User furnishes and may require additional information. Within thirty (30) days of receipt of a complete permit application, or such longer time as the Town Manager reasonably requires to adjudicate such application, the Town Manager will determine whether to issue an industrial user discharge permit and shall communicate such decision to the User in writing. The Town Manager may deny any application for an industrial user discharge permit.

#### § 131.13 CONTENTS OF INDUSTRIAL USER DISCHARGE PERMITS

- (a) The Town Manager may include in an industrial user discharge permit any conditions that the Town Manager deems reasonably necessary to prevent Pass Through or Interference, protect the quality of the water body receiving the treatment plant's effluent, protect worker health and safety, facilitate sludge management and disposal, and protect against damage to the POTW.
  - (b) Industrial user discharge permits must contain:
- (i) A statement that indicates the wastewater discharge permit issuance date, expiration date and effective date;
- (ii) A statement that the wastewater discharge permit is nontransferable without prior notification to the Town in accordance with § 131.15, and provisions for furnishing the new owner or operator with a copy of the existing wastewater discharge permit;
- (iii) Effluent limits, including Best Management Practices, based on applicable Pretreatment Standards;
- (iv) Self-monitoring, sampling, reporting, notification, and record-keeping requirements. These requirements shall include an identification of pollutants (or best management practice) to be monitored, sampling location, sampling frequency, and sample type based on Federal, State, and local law;

- (v) The process for seeking a waiver from monitoring for a pollutant neither present nor expected to be present in the discharge in accordance with § 131.25(a);
- (vi) A statement of applicable civil and criminal penalties for violation of Pretreatment Standards and Requirements, and any applicable compliance schedule. Such schedule may not extend the time for compliance beyond that required by applicable Federal, State, or local law:
- (vii) Requirements to control Slug Discharge, if the Town Manager determines such requirements are necessary
  - (viii) Any grant of the monitoring waiver by the Town Manager.
  - (c) Industrial user discharge permits may contain the following conditions:
- (i) Limits on the average and/or maximum rate of discharge, time of discharge, and/or requirements for flow regulation and equalization;
- (ii) Requirements for the installation of pretreatment technology, pollution control, or construction of appropriate containment devices, designed to reduce, eliminate, or prevent the introduction of pollutants into the POTW;
- (iii) Requirements for the development and implementation of spill control plans or other special conditions including management practices necessary to adequately prevent accidental, unanticipated, or non-routine discharges;
- (iv) Development and implementation of waste minimization plans to reduce the amount of pollutants discharged to the POTW;
- (v) The unit charge or schedule of User charges and fees for the management of the wastewater discharged to the POTW;
- (vi) Requirements for installation and maintenance of inspection and sampling facilities and equipment, including flow measurement devices;
- (vii) A statement that compliance with the industrial user discharge permit does not relieve the permittee of responsibility for compliance with all applicable Federal and State Pretreatment Standards, including those which become effective during the term of the industrial user discharge permit; and
- (viii) Any other condition that the Town Manager deems to ensure compliance with this Chapter, and State and Federal laws, rules, and regulations.

#### § 131.14 DURATION OF INDUSTRIAL USER DISCHARGE PERMITS

An industrial user discharge permit shall be issued for a specified time period, not to exceed five (5) years from the effective date of the permit. An industrial user discharge permit may be issued for a period less than five (5) years, at the discretion of the Town Manager. Each industrial user discharge permit will indicate a specific date upon which it will expire.

# § 131.15 MODIFICATIONS OF INDUSTRIAL USER DISCHARGE PERMIT

Upon request of a User of whenever he otherwise deems it appropriate, the Town Manager may modify an industrial user discharge permit for good cause, including, but not limited to, the following reasons:

- (a) To incorporate any new or revised Federal, State, or local Pretreatment Standards or Requirements;
- (b) To address significant alterations or additions to the User's operation, processes, or wastewater volume or character since the time of the industrial user discharge permit issuance;
- (c) A change in the POTW that requires either a temporally or permanent reduction or elimination of the authorized discharge;
- (d) Information indicating that the permitted discharge poses a threat to the Town, the POTW, Town personnel, or the receiving waters;
  - (e) Violation of any terms or conditions of the permit;
- (f) Misrepresentations or failure to fully disclose all relevant facts in the wastewater discharge permit application or in any required reporting;
- (g) Revision of or a grant of variance from categorical Pretreatment Standards pursuant to 9 VAC 25-31-850;
  - (h) To correct typographical or other errors in the industrial user discharge permit; or
- (i) To reflect a transfer of the facility ownership or operation to a new owner or operator where requested in accordance with § 131.16.

#### § 131.16 INDUSTRIAL USER DISCHARGE PERMIT TRANSFER

Industrial user discharge permits may be transferred to a new owner or operator only if the permittee gives at least thirty (30) days' advance notice to the Town Manager and the Town Manager approves the industrial user discharge permit transfer. Failure to provide advance notice of a transfer, and to receive the approval of the Town Manager, renders the industrial user

discharge permit void as of the date of the facility's transfer. The notice to the Town Manager must include a written certification by the new owner or operator which:

- (a) States that the new owner and/or operator has no immediate intent to change the facility's operations and processes;
  - (b) Identifies the specific date on which the transfer is to occur; and
- (c) Acknowledges full responsibility for complying with the existing industrial user discharge permit.

# § 131.17 INDUSTRIAL USER DISCHARGE PERMIT REVOCATION

The Town Manager may revoke an industrial user discharge permit for good cause, including, but not limited to, the following reasons:

- (a) Failure to notify the Town Manager of significant changes to the wastewater prior to the changed discharge;
- (b) Failure to provide prior notification to the Town Manager of changed conditions pursuant to § 131.26;
- (c) Misrepresentation or failure to fully disclose all relevant facts in the wastewater discharge permit application;
  - (d) Falsifying self-monitoring reports and certification statements;
  - (e) Tampering with monitoring equipment;
- (f) Refusing to allow the Town Manager timely access to the facility premises and records:
  - (g) Failure to meet effluent limitations;
  - (h) Failure to pay fines;
  - (i) Failure to pay sewer charges;
  - (j) Failure to meet compliance schedules;
- (k) Failure to complete a wastewater survey or the wastewater discharge permit application; or

(l) Violation of any Pretreatment Standard or Requirement, or any terms of the wastewater discharge permit or this Chapter.

# § 131.18 AUTOMATIC REVOCATION OF INDUSTRIAL USER DISCHARGE PERMIT

- (a) An industrial user discharge permits shall be revoked automatically upon cessation of operations.
- (b) An industrial user discharge permits shall be revoked automatically upon transfer to a new owner or operator without compliance with the provisions of § 131.16.
- (c) All industrial user discharge permits issued to a User are void upon the issuance of a new industrial user discharge permit to that User.

#### § 131.19 INDUSTRIAL USER DISCHARGE PERMIT REISSUANCE

A User with an expiring industrial user discharge permit shall apply for an industrial user discharge permit reissuance by submitting a complete permit application, in accordance with § 131.12(e) *et seq.*, as if applying for a new permit, a minimum of one hundred and eighty (180) days prior to the expiration of the User's existing industrial user discharge permit.

# § 131.20 PUBLIC NOTICES; ADMINISTRATIVE APPEALS

- (a) <u>Pre-Issuance</u>. The Town Manager will publish in a newspaper of general circulation that provides meaningful public notice with the jurisdictions served by the POTW, or on a Web page, a notice of a proposal to
- (i) issue an industrial user discharge permit as an initial matter under § 131.12(e), or
- (ii) modify an existing industrial user discharge permit under § 131.15, in a manner that would not constitute a minor modification under 9VAC25-31-400; or
  - (iii) reissue or renew an industrial user discharge permit under § 131.19.

The Town Manager shall issue such notice at least thirty (30) days prior to the issuance of the proposed permit. The notice will indicate a location where the draft permit may be reviewed and an address where written comments may be submitted.

(b) <u>Post-Issuance</u>. After taking an action described in § 131.20(a), the Town Manager shall provide public notice of having taken such action in the same manner as described in §131.20(a).

(c) <u>Inadvertent Failure to Give Notice</u>. The inadvertent failure to give a notice required under §§ 131.20(a) or (b) shall not affect the validity of the permit.

# (d) <u>Administrative Appeals</u>.

- (i) Any person, including the User, who is aggrieved by a final action that the Town Manager has taken under §§ 131.12 through 131.19 may petition the Town Manager to reconsider such action within thirty (30) days of such action by filing a written statement of the reasons therefor with the Town Manager. Failure to submit a timely petition for review shall be deemed to be a waiver of both the administrative appeal and the right to any subsequent judicial review.
- (ii) In its petition, the appealing party must indicate the action objected to, the reasons for this objection, and the alternative action or the alternative condition, if any, it seeks to place in industrial user discharge permit. If the petition asks the Town Manager to reconsider his determination to deny a permit *in toto*, the appealing party shall state the conditions and other information that the petitioner believes should have been included in the issued permit.
- (iii) If the appeal seeks reconsideration of a determination of the Town Manager to issue an industrial development discharge permit, the effectiveness of the industrial user discharge permit as the Town Manager issued it shall not be stayed pending the appeal.
- (iv) If the Town Manager fails to act within thirty (30) days after receiving the petition, he shall be deemed to have denied the petition for reconsideration.

## (e) Judicial Review.

- (i) A person aggrieved by a final action that the Town Manager has taken under §§ 131.12 through 131.19, and who has exhausted the administrative remedy provided in § 131.20(d), may seek judicial review of such action in the Rockingham County Circuit Court. Such person must do so by filing a complaint within the time frames prescribed by law.
- (ii) For the purposes of judicial review, the Town Manager's decision to deny an appeal under § 131.20(d) constitutes a final administrative action to implement the decision that has been appealed.
- (iii) Except to the extent that state law requires a different standard of review, the Circuit Court may modify or reverse the Town Manager's action only if it is arbitrary and capricious, or not otherwise permitted by law.

# § 131.21 REGULATION OF WASTE RECEIVED FROM OUTSIDE THE TOWN

- (a) If another municipality, or User located outside of the Town's jurisdiction, contributes wastewater to the POTW, the Town Manager shall enter into a written agreement with the contributing municipality or User.
- (b) Prior to entering into an agreement required by paragraph (a), above, the Town Manager shall request the following information from the contributing municipality or User:
- (i) A description of the quality and volume of wastewater to be discharged to the POTW;
- (ii) If a municipality, an inventory of all Users located within the contributing municipality that are discharging to the POTW; and
  - (iii) Such other information as the Town Manager may deem necessary.
  - (c) Any agreement described above shall contain the following conditions:
- (i) A requirement that a contributing municipality to adopt a sewer use ordinance which is at least as stringent as this Chapter and Local Limits, which are at least as stringent as those set out in § 131.07 and reporting requirements at least as stringent as the ones contained in this Chapter. The requirement shall specify that such Chapter and limits must be revised as necessary to reflect changes made to the Town's ordinances or Local Limits;
- (ii) A requirement that a contributing municipality submit a revised User inventory on at least an annual basis;
- (iii) A provision specifying which pretreatment implementation activities, including industrial user discharge permit issuance, inspection and sampling, and enforcement, will be conducted by the contributing municipality or User; which of these activities will be conducted by the Town Manager; and which of these activities will be conducted jointly by the contributing municipality or User and the Town Manager;
- (iv) A requirement for the contributing municipality or User to provide the Town Manager with access to all information that the contributing municipality or User obtains as part of its pretreatment activities;
- (v) Limits on the nature, quality, and volume of the contributed wastewater at the point where it discharges to the POTW;
- (vi) Requirements for monitoring the contributing municipality or User's discharge;

- (vii) A provision ensuring the Town Manager access to the facilities of Users located within a contributing municipality's jurisdictional boundaries for the purpose of inspection, sampling, and any other duties or activities that the Town Manager deems necessary; and
- (viii) A provision specifying remedies available for breach of the terms of the agreement.

#### § 131.22 BASELINE MONITORING REPORTS

- (a) Within either one hundred eighty (180) days after the effective date of a categorical Pretreatment Standard, or the final administrative decision on a category determination under 9 VAC 25-31-780.AA., whichever is later, existing Categorical Industrial Users currently discharging to or scheduled to discharge to the POTW shall submit to the Town Manager a report which contains the information required below. At least ninety (90) days prior to commencement of their discharge, New Sources, and sources that become Categorical Industrial Users subsequent to the promulgation of an applicable categorical Standard, shall submit to the Town Manager a report which contains the information listed below. A New Source shall report the method of pretreatment it intends to use to meet applicable categorical Standards. A New Source also shall give estimates of its anticipated flow and quantity of pollutants to be discharged.
- (b) Users described above shall submit all information required in § 131.12(e), paragraphs (i), (ii), (iii), (iv), (v), and (viii).
- (c) The User shall take a minimum of one representative sample to compile that data necessary to comply with the requirements of this § 131.22. Samples should be taken immediately downstream from pretreatment facilities if such exist or immediately downstream from the regulated process if no pretreatment exists. If other wastewaters are mixed with the regulated wastewater prior to pretreatment the User should measure the flows and concentrations necessary to allow use of the combined wastestream formula in 9 VAC 25-31-780.E. to evaluate compliance with the Pretreatment Standards. Where an alternate concentration or mass limit has been calculated in accordance with 9 VAC 25-31-780.E this adjusted limit along with supporting data shall be submitted to the Town;
  - (d) Sampling and analysis must be performed in accordance with § 131.30;
- (e) The Town Manager may allow the submission of a baseline report which utilizes only historical data so long as the data provides information sufficient to determine the need for industrial pretreatment measures.
- (f) The baseline report shall indicate the time, date and place of sampling and methods of analysis, and shall certify that such sampling and analysis is representative of normal work cycles and expected pollutant Discharges to the POTW.

- (g) The baseline report shall include a statement, reviewed by the User's Authorized Representative as defined in § 131.04 and certified by a qualified professional, indicating whether Pretreatment Standards are being met on a consistent basis, and, if not, whether additional operation and maintenance (O&M) and/or additional pretreatment is required to meet the Pretreatment Standards and Requirements.
- (h) If additional pretreatment and/or O&M will be required to meet the Pretreatment Standards, the baseline report must include shortest schedule by which the User will provide such additional pretreatment and/or O&M must be provided. The completion date in this schedule shall not be later than the compliance date established for the applicable Pretreatment Standard. The schedule shall contain progress increments in the form of dates for the commencement and completion of major events leading to the construction and operation of additional pretreatment required for the User to meet the applicable Pretreatment Standards (such events include, but are not limited to, hiring an engineer, completing preliminary and final plans, executing contracts for major components, commencing and completing construction, and begimling and conducting routine operation). No such progress increment shall exceed nine (9) months. The User shall submit a progress report to the Town Manager no later than fourteen (14) days following each date in the schedule and the final date of compliance including, as a minimum, whether or not it complied with the increment of progress, the reason for any delay, and, if appropriate, the steps being taken by the User to return to the established schedule. In no event shall more than nine (9) months elapse between such progress reports to the Town Manager.
- (i) All baseline monitoring reports must be certified in accordance with § 131.30(i) signed by an Authorized Representative of the User.

# § 131.23 REPORTS ON COMPLIANCE WITH CATEGORICAL PRETREATMENT STANDARD DEADLINE

Within ninety (90) days following the date for final compliance with applicable categorical Pretreatment Standards, or in the case of a New Source following commencement of the introduction of wastewater into the POTW, any User subject to such Pretreatment Standards and Requirements shall submit to the Town Manager a report containing the information described in § 131.12(e)(iv)E and e(v) 91.210, and § 131.22(b) of this Chapter. This report shall include the User's actual production during the appropriate sampling period. All compliance reports must be signed and certified in accordance with § 131.30(i) of this Chapter. All sampling must be done in conformance with § 131.30.

## § 131.24 PERIODIC COMPLIANCE REPORTS

(a) All Significant Industrial Users must submit, at a frequency determined by the Town Manager that is no less frequently than twice per year (June and December), reports indicating the nature, concentration of pollutants in the discharge which are limited by Pretreatment Standards and the measured or estimated average and maximum daily flows for the reporting period. In cases where the Pretreatment Standard requires compliance with a Best

Management Practice (BMP) or pollution prevention alternative, the User must submit documentation required by the Town Manager or the Pretreatment Standard necessary to determine the compliance status of the User.

- (b) All periodic compliance reports must be signed and certified in accordance with § 131.30(i).
- (c) All wastewater samples used to submit a periodic compliance report, or otherwise conducted or submitted under this Title, must be representative of the User's discharge. Wastewater monitoring and flow measurement facilities shall be properly operated, kept clean, and maintained in good working order at all times. The failure of a User to keep its monitoring facility in good working order shall not be grounds for the User to claim that sample results are unrepresentative of its discharge.
- (d) If a User subject to the reporting requirement in this section monitors any regulated pollutant at the appropriate sampling location more frequently than required by the Town Manager using the procedures prescribed in § 131.30(f), the results of this monitoring shall be included in the report.

## § 131.25 WAIVERS OF SAMPLING

- (a) The Town may authorize an Industrial User subject to a categorical Pretreatment Standard to forego sampling of a pollutant regulated by a categorical Pretreatment Standard if the Industrial User has demonstrated through sampling and other technical factors that the pollutant is neither present nor expected to be present in the Discharge, or is present only at background levels from intake water and without any increase in the pollutant due to activities of the Industrial User (see 9 VAC 25-31-840.E.2).
  - (b) Such authorization is subject to the following conditions:
- (i) The Town will authorize the waver only where a pollutant is determined to be present solely due to sanitary wastewater discharged from the facility provided that the sanitary wastewater is not regulated by an applicable categorical Standard and otherwise includes no process wastewater.
- (ii) The monitoring waiver is valid only for the duration of the effective period of the industrial user discharge permit, but in no case longer than 5 years. The User must submit a new request for the waiver before the waiver can be granted for each subsequent industrial user discharge permit.
- (iii) In making a demonstration that a pollutant is not present, the Industrial User must provide data from at least one sampling of the facility's process wastewater prior to any treatment present at the facility and that is representative of all wastewater from all processes.

- (iv) The request for a monitoring waiver must be signed by an Authorized Representative in accordance with § 131.04(c), and include the certification statement in § 131.30(i) (9 VAC 25-31780.A.2.b.).
- (v) Non-detectable sample results may be used only as a demonstration that a pollutant is not present if the EPA approved method from 40 CFR Part 136 with the lowest minimum detection level for that pollutant was used in the analysis.
- (c) Any grant of the monitoring waiver by the Town Manager must be included as a condition in the User's permit. The reasons supporting the waiver and any information submitted by the User in its request for the waiver must be maintained by the Town Manager for 3 years after expiration of the waiver.
- (d) Upon approval of the monitoring waiver and revision of the User's permit by the Town Manager, the Industrial User must certify on each report, subject to the statement in § 131.30(k) below, that there has been no increase in the pollutant in its waste stream due to activities of the Industrial User.
- (e) In the event that a waived pollutant is found to be present or is expected to be present because of changes that occur in the User's operations, the User must immediately:
- (i) Comply with the monitoring and reporting requirements that would apply in the absence of the waiver, or other more frequent monitoring requirements imposed by the Town Manager, in connection with such pollutant; and
  - (ii) notify the Town Manager.
- (f) This section does not supersede certification processes and requirements established in categorical Pretreatment Standards, except as otherwise specified in the categorical Pretreatment Standard.

#### § 131.26 REPORTS OF CHANGED CONDITIONS

- (a) Each User must notify the Town Manager of any significant changes to the User's operations or system which might alter the nature, quality, or volume of its wastewater at as soon as possible before the change.
- (b) The Town Manager may require the User to submit such information as may be deemed necessary to evaluate the changed condition, including the submission of a wastewater discharge permit application to conform with § 131.12(e).
- (c) The Town Manager may issue an industrial user discharge permit or modify an existing wastewater discharge permit in response to changed conditions or anticipated changed conditions, in each instance after complying with the notice requirements of § 131.20.

(d) No Industrial User shall implement the planned changed condition(s) until the Town Manager has responded to the Industrial User's notice.

### § 131.27 REPORTS OF POTENTIAL PROBLEMS

- (a) In the case of any discharge, including, but not limited to, accidental discharges, discharges of a non-routine, episodic nature, a non-customary batch discharge, a Slug Discharge or Slug Load, that might cause potential problems for the POTW, the User shall immediately telephone and notify the Town Manager of the incident. This notification shall include the location of the discharge, type of waste, concentration and volume, if known, and corrective actions taken by the User.
- (b) Unless the Town Manager excuses compliance with this paragraph, within five (5) days following such discharge, the User shall submit a detailed written report describing the cause(s) of the discharge and the measures the User will undertake to prevent similar future occurrences. Such notification shall not relieve the User of any expense, loss, damage, or other liability the User might incur as a result of damage to the POTW, natural resources, or any other damage to person or property; nor shall such notification relieve the User of any fines, penalties, or other liability which may be imposed pursuant to this Chapter or other applicable law.
- (c) The User shall cause a notice to be permanently posted on the User's bulletin board or other prominent place advising employees whom to call in the event of a discharge described in § 131.27 (a). Users shall ensure that all employees who could cause such a discharge to occur are advised of the emergency notification procedure.
- (d) Significant Industrial Users are required to notify the Town Manager immediately of any changes at its facility affecting the potential for a Slug Discharge.

## § 131.28 REPORTS FROM UNPERMITTED USERS

All Users not required to obtain an Industrial User Discharge Permit shall provide appropriate reports to the Town Manager as the Town Manager may reasonably require.

## § 131.29 REPORT OF VIOLATION/REPEAT SAMPLING AND REPORTING

If sampling performed by a User indicates a violation, the User must notify the Town Manager within twenty-four (24) hours of becoming aware of the violation. The User shall also repeat the sampling and analysis and submit the results of the repeat analysis to the Town Manager within thirty (30) days after becoming aware of the violation. Resampling by the Industrial User is not required if the Town performs sampling at the User's facility at least once a month, or if the Town performs sampling at the User between the time when the initial sampling was conducted and the time when the User or the Town receives the results of this sampling, or if the Town has performed the sampling and analysis in lieu of the Industrial User.

## § 131.30 ANALYTICAL AND PROCEDURAL REQUIREMENTS RELATED TO SAMPLING AND REPORTS

- (a) The provisions of this section apply to all sampling undertaken at a User's facility, and to all reports submitted to the Town Manager in connection with a User's facility.
  - (b) All wastewater samples must be representative of the User's discharge.
- (c) Wastewater monitoring and flow measurement facilities shall be properly operated, kept clean, and maintained in good working order at all times. The failure of a User to keep its monitoring facility in good working order shall not be grounds for the User to claim that sample results are unrepresentative of its discharge.
- (d) If a User subject to the reporting requirement in this section monitors any regulated pollutant at the appropriate sampling location more frequently than required by the Town Manager using the procedures prescribed in § 131.30(f), the results of this monitoring shall be included in the reports made under this Chapter.
- (e) All pollutant analyses, including sampling techniques, to be submitted as part of a wastewater discharge permit application or report shall be performed in accordance with the techniques prescribed in 40 CFR Part 136 and amendments thereto, unless otherwise specified in an applicable categorical Pretreatment Standard. If 40 CFR Part 136 does not contain sampling or analytical techniques for the pollutant in question, or where the EPA determines that the Part 136 sampling and analytical techniques are inappropriate for the pollutant in question, sampling and analyses shall be performed by using validated analytical methods or any other applicable sampling and analytical procedures, including procedures suggested by the Town Manager, the Commonwealth, or other parties approved by EPA.
- (f) Samples collected to satisfy reporting requirements must be based on data obtained through appropriate sampling and analysis performed during the period covered by the report, based on data that is representative of conditions occurring during the reporting period. The following requirements shall also apply:
- (i) Except as indicated in paragraphs (ii) and (iii) below, the User must collect wastewater samples using 24-hour flow-proportional composite sampling techniques, unless time-proportional composite sampling or grab sampling is authorized by the Town. Where time-proportional composite sampling or grab sampling is authorized by the Town the samples must be representative of the discharge. Using protocols (including appropriate preservation) specified in 40 CFR Part 136 and appropriate EPA guidance, multiple grab samples collected during a 24-hour period may be composited prior to the analysis as follows: for cyanide, total phenols, and sulfides the samples may be composited in the laboratory or in the field; for volatile organics and oil and grease, the samples may be composited in the laboratory. Composite samples for other parameters unaffected by the compositing procedures as documented in approved EPA methodologies may

be authorized by the Town as appropriate. In addition, grab samples may be required to show compliance with Instantaneous Limits.

- (ii) Samples for oil and grease, temperature, pH, cyanide, total phenols, sulfides, and volatile organic compounds must be obtained using grab collection techniques.
- (iii) For sampling required in support of baseline monitoring and 90-day compliance reports required in § 131.22 and § 131.23 (9 VAC 25-31-840.B. and D.), a minimum of four (4) grab samples must be used for pH, cyanide, total phenols, oil and grease, sulfide and volatile organic compounds for facilities for which historical sampling data do not exist; for facilities for which historical sampling data are available, the Town may authorize a lower minimum. For the reports required in § 131.24 (9 VAC 25-31-840.E. and 840.H.), the Industrial User is required to collect the number of grab samples necessary to assess and assure compliance with applicable Pretreatment Standards and Requirements.
- (g) Written reports shall be hand-delivered to the Town Office, sent to the Town Office via reputable overnight delivery service or overnight mail, or mailed to the Town Office via first class mail. Reports shall be deemed submitted on the date of actual receipt (in the case of hand delivery or delivery by first class mail) or one day after deposit with a reputable overnight delivery service or with United States Postal Service Express Mail.
- (h) Users subject to any reporting requirement under this Chapter shall retain, and make available to the Town Manager or his designee for inspection and copying, all records of information obtained pursuant to any monitoring activities required by this Code, any additional records of information obtained pursuant to monitoring activities undertaken by the User independent of such requirements, and documentation associated with Best Management Practices established under § 131.07(d)(iii). Records shall include the date, exact place, method, and time of sampling, and the name of the person(s) taking the samples; the dates analyses were performed; who perfol111ed the analyses; the analytical techniques or methods used; and the results of such analyses. These records shall remain available for a period of at least three (3) years. This period shall be automatically extended for the duration of any litigation concerning the User or the Town, or where the Town Manager has specifically assigned the User a longer retention period.
- (i) The following certification statement is required to be signed and submitted by Users submitting permit applications in accordance with § 131.12; Users submitting baseline monitoring reports under § 131.22; Users submitting reports on compliance with the categorical Pretreatment Standard deadlines under § 131.23; Users submitting periodic compliance reports required by § 131.24; and Users submitting an initial request to forego sampling of a pollutant on the basis of § 131.25. The following certification statement must be signed by an Authorized Representative as defined in § 131.04(c):

I certify under penalty of law that this document and all attachments were prepared under my direction or supervision in accordance with a system designed to assure that qualified personnel properly gather and evaluate the information submitted. Based on my inquiry of the person or persons who manage the system, or those persons directly responsible for gathering the information, the information submitted is, to the best of my knowledge and belief, true, accurate, and complete. I am aware that there are significant penalties for submitting false information, including the possibility of fine and imprisonment for knowing violations.

(j) A facility that the Town Manager determines to be a Non-Significant Categorical Industrial User pursuant to Section §131.04(oo)(iii) and § 131.12(f)(iii) must annually submit the following certification statement signed by an Authorized Representative in Section § 131.04(c). This certification must also accompany an alterative report required by the Town Manager:

Based on my inquiry of the person or persons directly responsible for managing compliance with the categorical Pretreatment Standards under 40 CFR\_\_, I certify that, to the best ofmy knowledge and belief that during the period from \_\_ to \_\_ [months, days, year]:

- (a) The facility described as \_\_\_\_\_ [facility name] met the definition of a Non-Significant Categorical Industrial User as described in § 131.04(00)(iii); and
- (b) The facility complied with all applicable Pretreatment Standards and requirements during this reporting period; and
- (c) the facility never discharged more than 100 gallons of total categorical wastewater on any given day during this reporting period.

This compliance certification is based on the following information.

[compliance information follows]

(k) Users that have an approved monitoring waiver based on § 131.25 must certify on each report with the following statement that there has been no increase in the pollutant in its waste stream due to activities of the User:

Based on my inquiry of the person or persons directly responsible for managing compliance with the Pretreatment Standard for 40 CFR [specify applicable National Pretreatment Standard part(s)], I certify that, to the best of my knowledge and belief, there has been no increase in the level of [list pollutant(s)] in the wastewaters due to the activities at the facility since filing of the last periodic report under § 131.25.

## § 131.31 COMPLIANCE MONITORING

- (a) Right of Entry: Inspection and Sampling.
- (i) The Town Manager, bearing credentials and identification, shall have the right to enter the facility and property of any User to determine whether the User is complying with all requirements of this Chapter and their industrial user discharge permit (if applicable) or order issued hereunder. Users shall allow the Town Manager ready access to all parts of the premises for the purposes of inspection, sampling, records examination and copying, and the performance of any additional duties. By accepting access to the public sewer system of the Town, or by accepting any industrial user discharge permit, each User is deemed to have consented to the inspection and sampling provisions of this Section.
- (ii) Where a User has security measures in force, which require proper identification and clearance before entry into its premises, the User shall make necessary arrangements with its security guards so that, upon presentation of suitable identification, the Town Manager shall be permitted to enter without delay for the purposes of performing specific responsibilities.
- (iii) The Town Manager shall have the right to set up on the User's property, or require installation of, such devices as are necessary to conduct sampling and/or metering of the User's operations.
- (iv) The Town Manager may require the User to install monitoring equipment as necessary. The facility's sampling and monitoring equipment shall be maintained at all times in a safe and proper operating condition by the User at its own expense. All devices used to measure wastewater flow and quality shall be calibrated periodically as required to ensure their accuracy.
- (v) Any temporary or permanent obstruction to safe and easy access to the facility to be inspected and/or sampled shall be promptly removed by the User at the written or verbal request of the Town Manager and shall not be replaced. The User shall bear the costs of clearing such access.
- (vi) While performing the necessary work, the Town Manager shall observe all safety rules applicable to the premises of the user. The User shall not responsible for injury or death to the Town employees or agents, except as such may be caused by negligence or failure of the User to maintain safe conditions.
- (vii) The Town Manager and other duly authorized employees of the Town bearing proper credentials and identification shall be permitted to enter all private property through which the Town holds a duly negotiated easement for the purposes of, but not limited to, inspection, observation, measurement, sampling, repair, and maintenance of any portion of the

wastewater facilities lying within said easement. All work performed on any easement shall be done in full accordance with the terms of the duly negotiated easement pertaining to the property involved.

- (viii) No person shall cause an unreasonable delay in allowing the Town Manager access to the User's premises.
- (b) Search Warrants. If the Town Manager has been refused access to a building, structure, or property, or any part thereof, and is able to demonstrate probable cause to believe that there may be a violation of this Chapter, or that there is a need to inspect and/or sample as part of a routine inspection and sampling program of the Town of Broadway designed to verify compliance with this Chapter or any permit or order issued hereunder, or to protect the overall public health, safety and welfare of the community, the Town Manager may seek a search warrant from an appropriate officer to order such inspection.

#### § 131.32 CONFIDENTIAL INFORMATION

Information and data on a User obtained from reports, surveys, wastewater discharge permit applications, industrial user discharge permits, and monitoring programs, and from the Town Manager's inspection and sampling activities, shall be available to the public without restriction, unless the User specifically requests, and is able to demonstrate to the satisfaction of the Town Manager, that the release of such information would divulge information, processes, or methods of production entitled to protection as trade secrets or otherwise under applicable State law. Any such request must be asserted at the time of submission of the information or data, and must state specifically the State law basis for such exemption by citing the appropriate statute. When requested and demonstrated by the User furnishing a report that such information should be held confidential, and if the Town determines that State law permits the Town to withhold the information, the portions of a report which might disclose trade secrets or secret processes shall not be made available for inspection by the public. Notwithstanding the foregoing, any information in the Town's possession shall be made available immediately upon request to government agencies for uses related to the VPDES program or pretreatment program, and in enforcement proceedings involving the person furnishing the report. Wastewater constituents and characteristics and other effluent data, as defined at 40 CFR 2.302, shall not be recognized as confidential information and shall be available to the public without restriction.

#### § 131.33 PUBLICATION OF USERS IN SIGNIFICANT NONCOMPLIANCE (SNC)

The Town Manager shall publish annually, in a newspaper of general circulation that provides meaningful public notice within the jurisdictions served by the POTW, a list of the Users which, at any time during the previous twelve (12) months, were in Significant Noncompliance with applicable Pretreatment Standards and Requirements. The term Significant Noncompliance shall be applicable to all Significant Industrial Users (or any other Industrial User that violates paragraphs (c), (d) or (h) of this Section) and shall mean:

- (a) Chronic violations of wastewater discharge limits, defined here as those in which sixty-six percent (66%) or more of all the measurements taken for the same pollutant parameter taken during a six (6) month period exceed (by any magnitude) a numeric Pretreatment Standard or Requirement, including Instantaneous Limits as defined in § 131.07;
- (b) Technical Review Criteria (TRC) violations, defined here as those in which thirty-three percent (33%) or more of wastewater measurements taken for each pollutant parameter during a six (6) month period equals or exceeds the product of the numeric Pretreatment Standard or Requirement including Instantaneous Limits, as defined by § 131.07 multiplied by the applicable criteria (I.4 for BOD, TSS, fats, oils and grease, and 1.2 for all other pollutants except pH);
- (c) Any other violation of a Pretreatment Standard or Requirement as defined by § 131.07 (Daily Maximum, long-term average, Instantaneous Limit, or narrative standard) that the Town Manager determines has caused, alone or in combination with other discharges, Interference or Pass Through, including endangering the health of POTW personnel or the general public;
- (d) Any discharge of a pollutant that has caused imminent endangerment to the public or to the environment, or has resulted in the Town Manager's exercise of his emergency authority to halt or prevent such a discharge;
- (e) Failure to meet, within ninety (90) days of the scheduled date, a compliance schedule milestone contained in an industrial user discharge permit or enforcement order for starting construction, completing construction, or attaining final compliance;
- (f) Failure to provide within forty-five (45) days after the due date, any required reports, including baseline monitoring reports, reports on compliance with categorical Pretreatment Standard deadlines, periodic self-monitoring reports, and reports on compliance with compliance schedules;
  - (g) Failure to accurately report noncompliance; or
- (h) Any other violation, which may include a violation of Best Management Practices, that the Town Manager determines will adversely affect the operation or implementation of the local pretreatment program.

#### § 131.34 ADMINISTRATIVE ENFORCEMENT REMEDIES

(a) Notification of Violation. When the Town Manager finds that a User has violated, or continues to violate, any provision of this Chapter, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement, the Town Manager may serve upon that User a written Notice of Violation. Within five (5) days of the receipt of such notice, the User shall submit to the Town Manager an explanation of the violation and a plan for the satisfactory correction and prevention thereof, to include specific required actions. Submission

of such a plan in no way relieves the User of liability for any violations occurring before or after receipt of the Notice of Violation. Nothing in this paragraph shall limit the authority of the Town Manager to take any action, including emergency actions or any other enforcement action, without first issuing a Notice of Violation.

- (b) Consent Orders. The Town Manager may enter into Consent Orders, assurances of compliance, or other similar documents establishing an agreement with any User responsible for noncompliance. Such documents shall include specific action to be taken by the User to correct the noncompliance within a time period specified by the document. Such documents shall have the same force and effect as the administrative orders issued and shall be judicially enforceable.
- (c) Show Cause Hearing. The Town Manager may order a User which has violated, or continues to violate, any provision of this Chapter, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement, to appear before the Town Manager and show cause why the proposed enforcement action should not be taken. Notice shall be served on the User specifying the time and place for the meeting, the proposed enforcement action, the reasons for such action, and a request that the User show cause why the proposed enforcement action should not be taken. The notice of the meeting shall be served personally or by registered or certified mail (return receipt requested) at least ten (10) days prior to the hearing. Such notice may be served on any Authorized Representative of the User as defined in § 131.04 or as otherwise permitted by law for service of process. A show cause hearing shall not be a bar against, or prerequisite for, taking any other action against the User.
- (d) Compliance Orders. When the Town Manager finds that a User has violated, or continues to violate, any provision of this Chapter, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement, the Town Manager may issue an order to the User responsible for the discharge directing that the User come into compliance within a specified time. If the User does not come into compliance within the time provided, the Town may discontinue sewer service unless adequate treatment facilities, devices, or other related appurtenances are installed and properly operated. Compliance orders also may contain other requirements to address the noncompliance, including additional self-monitoring and management practices designed to minimize the amount of pollutants discharged to the sewer. A compliance order may not extend the deadline for compliance established for a Pretreatment Standard or Requirement, nor does a compliance order relieve the User of liability for any violation, including any continuing violation. Issuance of a compliance order shall not be a bar against, or a prerequisite for, taking any other action against the User.
- (e) Cease and Desist Orders. When the Town Manager finds that a User has violated, or continues to violate, any provision of this Chapter, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement, or that the User's past violations are likely to recur, the Town Manager may issue an order to the User directing it to cease and desist all such violations and directing the User to: (1) Immediately comply with all requirements; and (2) Take such appropriate remedial or preventive action as may be needed to properly address a continuing or threatened violation, including halting operations and/or

terminating the discharge. Issuance of a cease and desist order shall not be a bar against, or a prerequisite for, taking any other action against the User.

- Administrative Fines and Cost Recovery Fees. When the Town Manager finds that a User has violated, or continues to violate, any provision of this Chapter, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement, the Town Manager may fine such User in an amount not to exceed \$5,000 per day, and may assess against such User a cost recovery fee to compensate the Town for its administrative and legal costs associated with the enforcement activity that the Town undertakes to address noncompliance. Such fines shall be assessed on a per-violation, per-day basis. In the case of monthly or other long-term average discharge limits, fines shall be assessed for each day during the period of violation. Unpaid charges, fees, fines, and penalties shall, after thirty (30) calendar days, be assessed an additional penalty of ten percent (10%) of the unpaid balance, and interest shall accrue thereafter at a rate of two percent (2%) per month. To secure unpaid fines, fees, charges, and penalties, the Town shall a lien against the User's property, collectable in the same manner as unpaid real estate taxes. Users desiring to dispute such fines or fees must file a written request for the Town Manager to reconsider the fine along with full payment of the amount within ten (10) days of being notified of the fine. In any appeal the Town Manager shall provide the User with an opportunity to be heard with respect to the fine or fee, and shall issue a written determination regarding the appeal stating the reasons for the granting or denial of such appeal. In the event the User's appeal is successful, the payment, together with any interest accruing thereto at the rate described above, shall be returned to the User. If the appeal is unsuccessful, the Town Manager may add the additional costs of adjudicating such appeal administrative enforcement actions, such as notices and orders, to the fine. Issuance of an administrative fine shall not be a bar against, or a prerequisite for, taking any other action against the User.
- Emergency Suspensions. The Town Manager may immediately suspend a User's discharge, after notice to the User appropriate under the circumstances, whenever such suspension is necessary to stop an actual or threatened discharge, which reasonably appears to present, or cause an clear and present danger to the health or welfare of persons. The Town Manager may also immediately suspend a User's discharge, after notice and opportunity to respond, that threatens to interfere with the operation of the POTW, or which presents, or may present, an endangerment to the environment. Any User notified of a suspension of its discharge shall immediately stop or eliminate its contribution. In the event of a User's failure to immediately comply voluntarily with the suspension order, the Town Manager may take such steps as deemed necessary, including immediate severance of the sewer connection, to prevent or minimize damage to the POTW, its receiving stream, or endangerment to any individuals. The Town Manager may allow the User to recommence its discharge when the User has demonstrated to the satisfaction of the Town Manager that the period of endangerment has passed, unless the Town commences termination proceedings under paragraph (h) below. A User that is responsible, in whole or in part, for any discharge presenting a clear and present danger shall submit a detailed written statement, describing the causes of the harmful contribution and the measures taken to prevent any future occurrence, to the Town Manager prior to the date of any show cause or termination hearing.

Nothing in this Section shall be interpreted as requiring a hearing prior to any Emergency Suspension under this Section.

- (h) Termination of Access to Sewer System. Any User who violates the following conditions is subject to termination of its right to discharge into the Town's sewer system:
  - (i) Violation of industrial user discharge permit conditions;
  - (ii) Failure to accurately report the wastewater constituents and characteristics;
- (iii) Failure to report significant changes in operations or wastewater volume, constituents, and characteristics prior to discharge;
- (iv) Refusal of reasonable access to the User's premises for the purpose of inspection, monitoring, or sampling;
  - (v) Violation of the Pretreatment Standards in § 131.07 of this Code; or
- (vi) Making a materially false statement on any application, report, or other documentation submitted to the Town under this Chapter.

Such User will be notified of the proposed termination of its discharge rights and shall receive an opportunity to show cause under paragraph (c) of this Section why the termination should not occur. Proceedings under this paragraph shall not be a bar to, or a prerequisite for, taking any other action against the User.

## § 131.35 JUDICIAL ENFORCEMENT REMEDIES

- (a) Injunctive Relief. When the Town Manager finds that a User has violated, or continues to violate, any provision of this Chapter, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement, the Town may commence a civil action in the nature of a suit in equity for the issuance of a temporary or permanent injunction, as appropriate, which restrains or compels the specific performance of the industrial user discharge permit, order, or other requirement that this Chapter imposes on activities of the User. The Town may also seek such other action as is appropriate for legal or equitable relief, including a requirement for the User to conduct environmental remediation. An action for injunctive relief shall not be a bar against, or a prerequisite for, taking any other action against a User.
- (b) Civil Penalties. A User who has violated, or continues to violate, any provision of this Chapter, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement shall be liable to the Town for a maximum civil penalty of \$1,000 per violation, per day. In the case of a monthly or other long-term average discharge limit, penalties shall accrue for each day during the period of the violation. The Town may recover

reasonable attorneys' fees, court costs, and other expenses associated with enforcement activities, including sampling and monitoring expenses, and the cost of any actual damages incurred by the Town. In determining the amount of civil liability, the Court shall take into account all relevant circumstances, including, but not limited to, the extent of harm caused by the violation, the magnitude and duration of the violation, any economic benefit gained through the User's violation, corrective actions by the User, the compliance history of the User, and any other factor as justice requires. The imposition of civil penalties, or an action to recover civil penalties, shall not be a bar against, or a prerequisite for, taking any other action against a User.

## (c) Criminal Prosecution.

- (i) A User who willfully or negligently violates any provision of this Chapter, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement shall be guilty of a Class One Misdemeanor, and upon due conviction thereof, shall be punished by a fine of not more than \$2,500 or imprisonment for not more than one (1) year, or both. Each day and violation constitute a separate violation of this paragraph.
- (ii) A User who willfully or negligently introduces any substance into the POTW, which causes personal injury or property damage, shall be guilty of a Class One Misdemeanor, and upon due conviction thereof, shall be punished by a fine of not more than \$2,500 or imprisonment for not more than one (1) year, or both. Each day and violation constitute a separate violation of this paragraph. This penalty shall be in addition to any other cause of action for personal injury or property damage available under State law.
- (iii) Any person who knowingly makes any false statements, representations, or certifications in any application, record, report, plan, or other documentation filed, or required to be maintained, pursuant to this Chapter, an industrial user discharge permit, or order issued hereunder, or who falsifies, tampers with, or knowingly renders inaccurate any monitoring device or method required under this Chapter shall be guilty of a Class One Misdemeanor, and upon due conviction thereof, shall be punished by a fine of not more than \$2,500 or imprisonment for not more than one (1) year, or both. Each day and violation constitute a separate violation of this paragraph.

#### § 131.36 SUPPLEMENTAL ENFORCEMENT ACTION

- (a) Penalties for Late Reports. A penalty of \$500 shall be assessed to any User for each day that a report required by this Chapter, a permit or order issued hereunder is late, beginning five days after the date the report is due. Action to collect late reporting penalties shall not limit the Town Manager's authority to initiate other enforcement actions that may include penalties for late reporting violations.
- (b) Performance Bonds. The Town Manager may decline to issue or reissue an industrial user discharge permit to any User who has failed to comply with any provision of this Chapter, a previous industrial user discharge permit, or order issued hereunder, or any other

Pretreatment Standard or Requirement, unless such User first files a bond or other collateral, in form and substance satisfactory to the Town Attorney, payable to the Town, in a sum not to exceed a value determined by the Town Manager to be necessary to achieve consistent compliance, and conditioned upon the User's compliance.

- (c) Liability Insurance; Repair of Prior Damage. The Town Manager may decline to issue or reissue an industrial user discharge permit to any User who has failed to comply with any provision of this Chapter, a previous industrial user discharge permit or order issued hereunder, or any other Pretreatment Standard or Requirement, unless the User first submits proof that it has financial resources to complete the process of repair and restoration of any damage that a prior discharge caused to the POTW, and/or obtained has insurance sufficient to restore or repair damage to the POTW caused that a future discharge might cause.
- (d) Payment of Outstanding Fees, Fines, and Penalties. The Town Manager may decline to issue or reissue an industrial user discharge permit to any User who has failed to pay any outstanding fees, fines or penalties incurred as a result of any provision of this Chapter, a previous industrial user discharge permit or order issued hereunder.
- (e) Water Supply Severance. Whenever a User has violated or continues to violate any provision of this Chapter, an industrial user discharge permit or order issued hereunder, or any other Pretreatment Standard or Requirement, the Town may sever water service to the User. Service will recommence, at the User's expense, only after the User has satisfactorily demonstrated its ability to comply.
- (f) Public Nuisances. A violation of any provision of this Code, an industrial user discharge permit, or order issued hereunder, or any other Pretreatment Standard or Requirement is hereby declared a public nuisance and may be corrected or abated though any provision of State or Town law relating to public nuisances.
- (\$1,000) for information leading to the discovery of noncompliance by a User. In the event that the information provided results in a civil penalty levied against the User, the Town Manager may disperse up to fifty percent (50%) of the collected fine or penalty to the informant. However, a single reward payment may not exceed one thousand dollars (\$1,000).
- (h) Contractor Debarment. Users that have violated or continue to violate any provision of this Chapter, an industrial user discharge permit or order issued hereunder, or any other Pretreatment Standard or Requirement, are debarred from the award of Town contracts. The Town may terminate any such contract that already exists on the date the Town Manager finds any such violation or noncompliance.

#### § 131.37 REMEDIES NONEXCLUSIVE

The remedies contained in this Chapter are not exclusive. The Town Manager may take any, all, or any combination of these actions against a noncompliant User. Moreover, the Town Manager is empowered to take more than one enforcement action against any noncompliant User.

#### § 131.38 AFFIRMATIVE DEFENSES TO DISCHARGE VIOLATIONS

## (a) Upset

- (i) For the purposes of this paragraph (a), an "upset" means an exceptional incident in which there is unintentional and temporary noncompliance with Categorical Pretreatment Standards because of factors beyond the reasonable control of the User. An upset does not include noncompliance to the extent caused by operational error, improperly designed treatment facilities, inadequate treatment facilities, lack of preventive maintenance, or careless or improper operation.
- (ii) An upset shall constitute an affirmative defense to an action brought for noncompliance with categorical Pretreatment Standards if the requirements of (iii) below, are met.
- (iii) Users who wish to establish the affirmative defense of upset shall show, through a preponderance of the evidence, that
  - (A) An upset occurred and the User can identify the cause of the upset;
- (B) The facility was at the time being operated in a prudent and workman-like manner and in compliance with applicable operation and maintenance procedures, including the requirement to plan for the loss or failure of its treatment facility and the source of power thereto; and
- (24) hours of becoming aware of the upset, and with a follow-up written submission provided within five (5) days, a description of the indirect discharge and cause of the noncompliance, the period of noncompliance, including exact dates and times or, if not corrected, the anticipated time the noncompliance is expected to continue; and a list of the steps the User is taking or planning to reduce, eliminate, and prevent recurrence of the noncompliance.
- (b) Prohibited Discharge Standards. A User shall have an affirmative defense to an enforcement action brought against it for noncompliance with the general or the specific prohibitions in § 91.206 of this Code if it can prove that it did not know, or have reason to know, that its discharge, alone or in conjunction with discharges from other sources, would cause Pass Through or Interference and that either:
- (i) A Local Limit exists for each pollutant discharged and the User was in compliance with each limit directly prior to, and during, the Pass Through or Interference; or

- (ii) No Local Limit exists, but the discharge did not change substantially in nature or constituents from the User's prior discharge when the Town was regularly in compliance with its VPDES permit, and in the case of Interference, was in compliance with applicable sludge use or disposal requirements.
- (c) Burden of Proof. In any enforcement proceeding, the User seeking to establish an affirmative defense shall have the burden of proof.

#### § 131.39 BYPASS

- (a) For the purposes of this Section:
- (i) Bypass means the intentional diversion of waste streams from any portion of a User's treatment facility.
- (ii) Severe property damage means substantial physical damage to property, damage to the treatment facilities which causes them to become inoperable, or substantial and permanent loss of natural resources which can reasonably be expected to occur in the absence of a bypass. Severe property damage does not mean economic loss caused by delays in production.
- (b) A User may allow any bypass to occur which does not cause Pretreatment Standards or Requirements to be violated, but only if it also is for essential maintenance to assure efficient operation. These bypasses are not subject to the provision of paragraphs (c) and (d) of this Section.

## (c) Bypass Notifications

- (i) If a User knows in advance of the need for a bypass that exceeds applicable Pretreatment Standards, it shall submit prior notice to the Town, at least ten (10) days before the date of the bypass, if possible.
- (ii) A User shall submit oral notice to the Town of an unanticipated bypass that exceeds applicable Pretreatment Standards within twenty-four (24) hours from the time it becomes aware of the bypass. The User shall also provide a written submission within five (5) days of the time the User becomes aware of the bypass. The written submission shall contain a description of the bypass and its cause; the duration of the bypass, including exact dates and times, and, if the bypass has not been corrected, the anticipated time it is expected to continue; and steps taken or planned to reduce, eliminate, and prevent reoccurrence of the bypass. The Town Manager may waive the written report on a case-by-case basis if he has received the oral report within twenty-four (24) hours of the bypass.
- (d) No bypass shall occur that causes violation of Pretreatment Standards or Requirements unless

- (i) Bypass was unavoidable to prevent loss of life, personal injury, or severe property damage;
- (ii) There were no feasible alternatives to the bypass, such as the use of auxiliary treatment facilities, retention of untreated wastes, or maintenance during normal periods of equipment downtime. This condition is not satisfied if adequate back-up equipment should have been installed in the exercise of reasonable engineering judgment to prevent a bypass which occurred during normal periods of equipment downtime or preventive maintenance; and
  - (iii) The User submits notices as required under paragraph (c) of this section.
- (e) For any planned bypass that will cause a violation of Pretreatment Standards or Requirements, the User shall seek approval from the Town Manager. The Town Manager may approve such bypass, after considering its adverse effects, if the Town Manager determines that it will meet the three conditions listed in paragraph (d) of this Section.

## § 131.40 PROVISIONS FOR FEES AND DEPOSITS

- (a) Before making any sewer connection, the User shall pay the connection fee prescribed on the fee schedule to the Town Code.
- (b) Prior to receiving service, any prospective consumer not holding title to the property served by public sewer shall pay a deposit of an amount that appears on the fee schedule to the Town Code. Such deposits are refundable upon termination of service and the payment of all outstanding sewer and or water bills including any penalty and interest. If any person shall apply for sewer connection, paying all such deposits and fees, and such charges are increased prior to the actual introduction of sewer services into the applicant's premises, he shall also pay the difference between the current and prior rates.
- (c) The Town Council shall set sewer rates, which shall appear on the rate schedule to the Town Code. Sewer rates shall be based upon water use, with no discount for filling of pools, watering lawns, or other water uses that do not implicate sewer use.

#### § 131.41 SANITARY SEWER BACKUPS

(a) It is the policy of the Town that the Town should quickly resolve and pay certain small claims, filed in good faith by citizens of this Town and others, even though, as a technical matter, the law may not force the Town to pay such claims. To this end, the Town Manager shall have the authority, but not the duty, to compromise, settle, or pay in full, claims filed in good faith against the Town, its officers, officials, employees, volunteers, and agents, based on sanitary sewer backups, where the blockage or problem is determined by the Town to be located on Town property, and where the proposed payment to the claimant is \$1,000 or less.

- (b) Persons proposing to make a claim against the Town, its officers, officials, agents, or employees, for damage to property alleged to have been sustained by reason of a sewer backup within the Town, shall file within 90 days after the damage is alleged to have occurred, with the Town Manager, a written statement of the nature of the claim and of the time and place at which the backup damage is alleged to have occurred, and a description of any insurance coverage available in connection with such claim.
- (c) The Town Manager shall cause a factual investigation to be made of the claims and, if necessary, contact the Town Attorney for counsel and for research of the law governing the claim, and may deny or offer to settle the claim or make a recommendation with respect thereto, accordance with this Section.
- (d) In reviewing any such claim, the Town Manager shall consider any available insurance coverage with respect to such claim, and shall not pay a claim or which insurance proceeds have made the claimant whole, unless otherwise required by law.
- (e) All claims not disposed of under this policy shall be handled, decided, and disposed of, according to the law of the Commonwealth, the Town, and the United States of America. No settlement shall be made under this policy which would be prohibited by law.

#### **CHAPTER 132: WATER ORDINANCE**

## § 132.01 CONNECTION MANDATORY; FEES

- (a) Unless the Town Manager has determined that a connection is not feasible, and except for Accessory Buildings (as that term is defined in the Town's Land Development Regulations), every owner of property with improvements within the Town shall cause the improvements thereon to be connected to the Town's water system.
- (b) The Town Council shall set a fee for water obtained through the Town's water system, in an amount shown on the rate schedule to the Town Code. Any person who does not own title to the property to be serviced shall pay a deposit before obtaining service, in the amount shown on the rate schedule to the Town Code.
- (c) The use of water from the Town's system operates as consent for the agents of the Town to enter upon the user's property at periodic intervals to read the water meter.

#### § 132.02 PRIVATE SYSTEMS PROHIBITED

For the promotion of the health and general welfare of the general public, it shall be unlawful for any person other than the Town of Broadway to furnish any water to any other person within the corporate limits of the Town, where Town water is available. It shall further be unlawful for any person to furnish water for personal use within the corporate limits of the Town where Town water is available, without the approval of the Town Council.

## § 132.03 SUPERVISION AND CONTROL

- (a) The water system of the Town shall be under the immediate supervision and control of the Town Manager. The Town Manager shall have supervision over the water system and all pipes, plugs and other properties in any way connected therewith. He shall keep all parts of the machinery and equipment in connection with the water system in good order and repair. He shall keep a map showing the location of the main pipes and length and size of each of them. When there is an extension of any main pipe, he shall mark on the map the place of extension, size of the pipe used and other related information and shall keep an account of the cost of material and labor used in making such extension. He shall keep a record of all branch pipes, hydrants and other connections with the water system.
- (b) The Town Manager may delegate any of his duties or powers under this Section or this Chapter generally to any other Town employee or official.
- (c) The Town Manager may discontinue water service to a consumer for repairs to the water system or because of the inadequacy or pollution of the water supply and shall give as much notice of such discontinuance to the consumer as is practicable under the circumstances.

#### § 132.04 STATE LAW

The Mayor and the Town Manager shall see that all provisions of this Code and other provisions of the Town and State law that safeguard the public health against injurious effects arising from impurities, contamination or other unsanitary conditions, are strictly enforced.

#### § 132.05 CONNECTIONS GENERALLY

All connections to the Town water system shall be made as provided in this Chapter and only upon order of the Town Manager. The Town Manager shall have the power to promulgate reasonable regulations and rules, not arbitrary and capricious, to carry out his responsibilities as described in this Chapter and to amplify the provisions herein.

#### § 132.06 APPLICATION

Any property owner along the line of any water main who desires the introduction of water into such property shall make written application therefore to the Town Manager. Such application shall set forth the name of the applicant, the location of the property into which the water is to be introduced, the purposes for which the water is to be used and the name of the plumber who is to do the work incident to the introduction of water into the premises. Such application shall also contain a provision which, upon acceptance of the application by the Town Manager and subsequent connection of the service pipe with the main, will obligate the owner of the premises to pay for all water furnished pursuant to such application. The Town Manager shall not cause water to be furnished at any water connection until the owner has agreed to pay for all water furnished. Without limiting the generality of the foregoing, the act of connection to the Town's water system indicates a connector's consent to pay for water drawn through such connection.

## § 132.07 SEPARATE CONNECTIONS OR METERS FOR INDIVIDUAL RESIDENCES OR PROPERTIES

Each individual residence or property shall be required to have a separate connection and meter unless the Town Manager otherwise authorizes. In addition, a building containing two or more units of occupancy served by one water meter shall be charged according to the number of units in the building, at a rate equal to the individual metering of each unit.

## § 132.08 CONNECTIONS OUTSIDE OF TOWN

The Town shall not supply water to any person outside the corporate limits of the Town where such connection would require the extension of any water main, without the express permission of the Town Council.

## § 132.09 CONNECTION DEPOSITS AND CHARGES

- (a) Before a connection to the Town water system shall be made, the applicant therefore shall pay to the Town Treasurer the material deposits, service fee deposits, and connection fees described on the rate schedule to the Town Code. A service fee deposit is refundable upon termination of service and the payment of all outstanding sewer and/or water bills including any penalty and interest.
- (b) If any person shall apply for water and sewer connection, paying all required deposits and fees and such charges are increased prior to the actual introduction of water services into the applicant's premises, he shall pay the difference between the current and prior rates.

# § 132.10 EXCAVATION FOR WATER INSTALLATION; NOTICE, PROTECTION, AND RESTORATION

Prior to any excavation for the building of water installations, notice shall be given to all authorities having underground utility installations in the vicinity. All such excavations shall be adequately guarded with barricades and lights so as to protect the public from hazard. Streets, sidewalks, parkways, and other public property disturbed in the course of the work shall be stored in a manner satisfactory to the Town and the relevant state authorities.

## § 132.11 COST OF WATER INSTALLATIONS; CONSTRUCTION SPECIFICATIONS: DEDICATION

The Town will provide the tap of the water main, at its cost, together with a water meter for each connection, as part of the water connection fee. All other costs associated with the connection, including the costs of the construction of water mains and other water lines, shall be the responsibility of the applicant. The construction of all such mains and lines shall be in accordance with the specifications prescribed by the Town Manager, and in every instance in accordance with the specifications contained within the Town's specifications manual, as the Town may amend the same from time to time. Upon proper installation as determined by the Town, such water mains in the subdivision shall become the property of and shall be maintained by the Town. Prior to the commencement of water service, the owner or developer shall execute and deliver appropriate instruments providing for the dedication of the same.

## § 132.12 RIGHT OF ENTRY OF TOWN MANAGER

Every person occupying a lot or tenement into which water is conveyed, from the Town water system, shall permit the Town Manager or his authorized agent to enter such lot or tenement, at reasonable hours, to inspect the work therein or to see if the provisions of this Code and other ordinances of the Town relative to the water system are being or have been violated. By connection to the Town's water system or drawing water from the same, the owner or occupant of such lot or dwelling consents to allowing the Town Manager or his authorized agent the access contained in this Section.

## § 132.13 UNAUTHORIZED USE, ETC. OF WATER, FIRE PLUG OR HYDRANT; INTERFERANCE WITH WATER SYSTEM

- (a) No person, without obtaining the prior consent and approval of the Town Manager, shall use water from the Town's water system, open any pipe, cock or other part of the system so as to waste water, or remove or injure any part of the system.
- (b) No person shall use or manipulate any of the fireplugs or hydrants of the Town unless by the order, or with the consent, of the Town Manager.
- (c) No person shall interfere with the proper operation of, damage, destroy, remove, or replace any water meter, water main or line, or any other element of the Town's water system without the permission of the Town Manager.
- (d) No person shall place any obstruction on public property that interferes with a service line, meter, or other Town water system element, nor shall any person do so on private property with the specific intention of interfering with the Town's inspection rights under § 132.12.
  - (e) In cases of a fire emergency, this Section shall not apply to the fire personnel.

## § 132.14 BATHING, POLLUTION OF RESERVOIR, ETC.

No person shall bathe in the reservoirs or springs of the Town water system. No person shall deposit any offensive matter or any sticks, mud or rubbish in the reservoir or springs of the Town's water system. No person shall knowingly or willfully cast any dead animals or any other noxious substance or matter, or what, is commonly known as bait, into any of the streams, springs, reservoirs, or other sources of the town water supply, or drown and leave or cause to be drowned and left, any animal therein, or do any other similar act by reason of which the health of any person is affected, or the water supply of the Town is rendered impure or offensive.

## § 132.15 UNAUTHORIZED ENTRY

No person shall, without lawful authority, climb over, get through or break into the enclosure of the reservoir or springs of the Town water system, the Town's water plant, or any other facility associated with the Town's water system.

# § 132.16 PLACING BUILDING MATERIAL AT STOPCOCK, ETC.; OBSTRUCTING ACCESS TO FIXTURES.

No person shall place any building material, rubbish or other matter at a stopcock, street main or service pipe or obstruct access to any fixture connected with the waterworks.

#### § 132.17 BACKFLOW PREVENTION

- (a) An approved backflow prevention device shall be installed on each service line to a consumer's water system where, in the judgment of the Town Manager, a health, pollution or system hazard to the waterworks exists, and as a minimum, but not limited to, where the following conditions exist:
  - (1) Premises having an auxiliary water system;
- (2) Premises on which any substance is handled in such a manner as to create an actual or potential hazard to a waterworks;
- (3) Premises having internal cross-connections that, in the judgment of the Town Manager, may not be easily correctable, or intricate plumbing arrangements which make it impracticable to determine whether or not cross-connections exist;
- (4) Premises where, because of security requirements or other prohibitions or restrictions, it is impossible or impractical to make a complete cross-connection survey; or
- (5) Premises having a repeated history of cross-connections being established or re-established.
- (b) The Town will have available, upon request, a list of approved backflow prevention devices. The Town will consider devices not appearing on the list provided they have been tested by a recognized testing laboratory or evaluation agency, and the Town Manager determines they are of satisfactory materials.

#### § 132.18 TURNING ON WATER

Water shall not be turned on in any connection except by the order of the Town Manager, provided, however, that this provision shall not be construed to prohibit duly licensed and bonded plumbers from turning water into any such pipes to test same or for purposes of testing the plumbing connected thereto. The Town Manager shall not authorize water to be turned on while any outstanding connection, deposit, fees, interest, or other charges associated with a connection remain unpaid.

#### § 132.19 NO RESALE OR RECONVEYANCE

- (a) No owner or occupant of any premises upon which water has been conveyed shall permit such water to be used, taken or received by any person not lawfully entitled to its use.
- (b) No owner or the occupant of any premises upon which water has been conveyed shall or to resell water that the Town has furnished.

#### § 132.19 PENALTY

Any person who violates the provisions of §§ 132.13 through 132.18, inclusive, or who draws water from the Town's water system without appropriate authorization under this Chapter, shall be guilty of a Class 1 Misdemeanor and, upon due conviction thereof, shall be fined an amount not to exceed Twenty-Five Hundred Dollars (\$2,500) or imprisoned for a term not to exceed one (1) year, or both.

#### **CHAPTER 133: DROUGHT MANAGEMENT**

#### § 133.01 PURPOSE

The purpose of this Chapter is to implement the water shortage and supply policies of the Town, as required under state law, and to provide for the enforcement of the Plan, and to voluntarily and mandatorily restrict the use of the Town's public water supply system during declared water shortages or water emergencies.

## § 133.02 INCORPORATION OF LOCAL AND REGIONAL WATER SUPPLY PLAN

Water supply regulations enacted by the State Water Control Board pursuant to 9 VAC § 25-780 and § 62.1-44.38:1 of the Code of Virginia are hereby incorporated into this Chapter by reference as if fully set out.

#### § 133.03 DEFINITIONS

Unless the context specifically indicates otherwise, the meaning of terms in this Chapter shall be as follows:

*Plan* shall mean the Upper Shenandoah River Basin Drought Preparedness and Response Plan as adopted by the Town on September 6, 2011.

CSPDC shall mean the Central Shenandoah Planning District Commission.

Council shall mean the Broadway Town Council.

*Town* shall mean the Town of Broadway, Virginia.

Town Manager shall mean the Town Manager of the Town.

Capitalized terms appearing in this Chapter and not otherwise defined shall have the same meaning as in the Plan.

#### § 133.04 DROUGHT INDICATORS

The indicators used to designate drought severity shall be as described in Appendix A of the Plan. Upon determination that indicators exceed the threshold of a drought stage, as set forth in Appendix A of the Plan, the Town Manager may declare a specific drought stage. The Town Manager shall also declare a drought stage upon a corresponding declaration by the Commonwealth of Virginia.

#### § 133.05 DROUGHT STAGE RESPONSES

Upon declaration of a Drought Watch or Drought Warning, voluntary conservation measures may be requested of residents and businesses served by the Town's public water supply as set forth in Appendix C of the Plan. Upon declaration of a Drought Emergency, mandatory restrictions shall apply to residents and businesses served by the Town's public water supply as set forth in Appendix C of the Plan.

## § 133.06 WAIVER OF RESTRICTIONS

Upon prior written request by an individual, business, or other water user, the Town Manager may permit less than full compliance with any drought restrictions upon good cause shown, including evidence that the applicant is affected in a substantial manner not common to other businesses or persons generally. The Town Manager shall not grant any waiver unless he determines that the public health, safety, and welfare will not be adversely affected by the waiver. The Town Manager shall report all waivers to the Council at its next regular meeting.

#### § 133.07 ENFORCEMENT

- (a) Any person who violates the provisions of this Plan, who fails to carry out duties and responsibilities imposed by this Plan, or who impedes or interferes with any action undertaken or ordered pursuant to this Plan shall be guilty of a Class 3 misdemeanor and upon due conviction thereof shall be fined an amount not to exceed five hundred dollars (\$500.00). Each act or each day's continuation of the violation shall be considered a separate offense
- (b) In addition, if the Town Manager learns of any violation of any water use restriction imposed pursuant to this Chapter, he shall cause a written notice of the violation to be affixed to the property where the violation occurred and mailed to the customer of record. Such notice shall describe the violation and order that it be corrected, cured, or abated immediately or within such specified time as the Town Manager determines is reasonable under the circumstances. If the violation continues after such time, the Town Manager may restrict or terminate water service to the customer by giving notice by mail or personal delivery that, due to the violation, water services will be restricted or discontinued within a specified time. A fee of \$50 shall be paid for with the reconnection of any water service restricted or terminated pursuant to this paragraph. In the event of subsequent violations, the reconnection fee shall be \$200 for the second violation and \$400 for each additional violation.

By letter from the Town Attorney to the Clerk, dated September 5, 2023, pursuant to the authority provided in Section 10.07 of the Code, the word "Policy" was changed to "Plan" in three locations in section (a), in order to correct a scriviner's error in the re-drafting of the Ordinance, which uses the word "Plan" rather than "Policy".

#### **CHAPTER 134: SOLID WASTE ORDINANCE**

#### § 134.01 DEFINITIONS

For the purposes of this Chapter, the following words and phrases shall have the following meanings:

Combustible refuse means refuse accumulations of paper, excelsior, rags or wooden or paper boxes of containers, sweepings, and all other accumulations of a nature other than garbage, which are usual to housekeeping and to the operation of stores, offices and other business places.

Garbage means every refuse accumulation of animal, fruit or vegetable matter that attends the preparation, use, cooking and dealing in, or storage of meats, fruit or vegetables, and any other matter of any nature whatsoever, which is subject to decay, putrefaction and the generation of noxious or offensive gases or odors, or as breeding or feeding material for flies or other germ-carrying insects. "Garbage" also includes any container which can retain water contents and may serve as a breeding place for flies, mosquitoes or other water-breeding insects.

Garden cuttings means tree limbs that cannot be placed in containers and accumulations of grass or shrubbery, cuttings, and other refuse attending the care of lawns, shrubbery, vines and trees.

Non-combustible refuse means refuse materials that are unburnable at ordinary flame temperatures such as metals, mineral matter, large quantities of glass or crockery, metal furniture and auto bodies or parts.

Trash means garbage, combustible and non-combustible refuse, and garden cuttings.

Refuse means combustible and non-combustible refuse.

# § 134.02 AUTHORITY OF TOWN AND TOWN MANAGER AS TO COLLECTION, ETC., OF TRASH; PUBLIC POLICY REGARDING TRASH

(a) The Town Manager shall have full authority to supervise the collection and disposal of all trash in the Town; to make rules and regulations not in conflict with the provisions of this Chapter or other law, and not arbitrary and capricious, regarding the collection, removal and disposition of trash, the items eligible to be collected as part of the public waste disposal program, and the times of collection. The Town Manager may issue such regulations informally by posting them on the Town's website as trash collection procedures.

- (b) No person shall fail, neglect or refuse to comply with the rules and regulations that the Town Manager has promulgated under this section.
- (c) The Town has enacted this Chapter to provide for the public health and safety by requiring the orderly storage, collection, and removal of trash. The Town Council finds and declares that the storage, collection, and removal of trash in an improper way, or its accumulation, or the presence of trash containers on public streets for more than an appropriate amount of time, can all attract unwelcome animals, harbor germs, interfere with traffic, and obstruct parking. Accordingly, in enacting the provisions of this Chapter to prevent the same, and the civil and criminal penalties contained in § 134.13, the Town asserts its authority under Virginia Code §§ 15.2-901, 15.2-928, and other law to prescribe times for the removal of trash, and to enforce the same.

## § 134.03 PURPOSE AND PROHIBITED USE OF PUBLIC TRASH CONTAINERS

Trash containers placed upon the streets or sidewalks of the Town by the Town or by any private organization, under the direction and approval of the Town, shall be for the incidental use of the general public. No person shall load any such container with household or business trash.

## § 134.04 LITTERING

- (a) No person shall throw into the streets or alleys or upon the sidewalks or other public places or private premises within the Town, or upon any private property within the Town without the consent of the owner thereof, any trash, rubbish, trash, paper, dirt, filth, vegetables, metals, fruit, broken glass or glass of any nature, or any other solid or substance or thing that renders the streets, sidewalks, alleys or private premises unclean or unsightly or unsafe to any person, animal or vehicle.
- (b) No person shall place in the streets or alleys or upon the sidewalks, or other public places or private premises anything of a nature that could injuriously affect the health or welfare of the general community.
- (c) No person shall cast, place, sweep, or deposit within the Town any trash in such amount that it may be carried or deposited by the elements upon any street, sideway, alley, or other public place, or into any occupied premises within the Town, except as the Town Manager may permit at certain times during clean up weeks or as part of a program to collect garden cuttings.
- (d) Any person who violates this section shall be guilty of a Class 1 misdemeanor and upon due conviction thereof shall be subject to a fine not to exceed \$2,500 or to incarceration for a period not to exceed one year, or both.

# § 134.05 STORAGE OF TRASH, ETC., IN STREET; REMOVAL OF TRASH, ETC., SO STORED

Except as otherwise provided, no alley or street may be used to store any kind of trash, or any other material which causes any unsightly appearance in the Town. Anyone violating this provision shall have the trash, refuse or material in question removed within seventy-two hours after the Town Manager issues a removal notice. Procedures for such notice shall be the same as in § 92.06.

## § 134.06 DEPOSITING TRASH, ETC., NOT ORIGINATING WITHIN THE TOWN WITHOUT A PERMIT

No person shall deposit or cause to be deposited on private or public property in the Town, any trash not originating within the Town without first obtaining a permit from the Town Manager. Any person who violates this section shall be guilty of a Class 1 misdemeanor and upon due conviction thereof shall be subject to a fine not to exceed \$2,500 or to incarceration for a period not to exceed one year, or both.

#### § 134.07 CONTAINERS FOR GARBAGE SET OUT FOR COLLECTION

- (a) Except as described herein, all trash set out for collection in the Town shall be placed in sturdy containers not less than ten gallons nor more than thirty- five gallons in capacity, or in heavy-duty trash bags. Residents, occupants or places of business in the Town shall provide containers equipped with handles and covers, except as such containers may be provided by a contractor collecting such trash as described in § 134.10. Residents, occupants or places of business using heavy-duty trash bags shall securely fasten all trash bags. Filled containers or trash bags shall not contain any liquid or exceed fifty (50) pounds by weight.
- (b) Garden cuttings or bulk pick-up items set out for collection in the Town shall be collected under regulations that the Town Manager may establish under § 134.02(a).
  - (c) Town employees will not collect hot ashes or any material burning in any container.

# § 134.08 SETTING OUT GARBAGE, ETC., CONTAINERS, ETC., FOR COLLECTION

- (a) Except as provided herein, all containers for garbage to be collected shall be set out by the alley adjoining the property or set out next to the street where convenient to those making the collection. Residents and occupants shall place the containers by 7:00 a.m. on the day of collection but not earlier than 4:00 p.m. on the day before collection.
- (b) For periodic "bulk" collection outside the purview of garbage normally collected on a weekly basis, trash (or trash containers) shall not be placed by the alley adjoining the property or set out next to the street where convenient to those making the collection except for the 60 hours before the start of collection. It may be kept in place until collected.

- (c) All containers for trash shall be kept covered at all times. After emptying containers, collectors shall replace the covers or tops and replace the containers where they were prior to being emptied.
- (d) After being emptied, each container for garbage, trash, refuse and other similar matter shall be removed from the sidewalk, curb, street or alley by the owner before noon of the day following the day of collection.

# § 134.09 REMOVAL OF WASTE MATERIALS RESULTING FROM CONSTRUCTION, ETC., OF BUILDINGS

All waste materials resulting from construction, repairs or demolition of buildings, such as lumber, stone, brick, cement, sand, glass, gravel, roofing, plaster and foundation stone or gravel, shall be removed by the owners of the buildings or their agents.

## § 134.10 PRIVATE CONTRACTORS

- (a) The Town may engage a private contractor to collect garbage from some or all properties within the Town. The contractor will collect all such garbage, as provided in its contract with the Town.
- (b) From time to time, the Town Manager may schedule the collection of items qualitatively or quantitatively beyond what the contractor will collect, within limits to be set by the Manager.
- (c) Additionally, the Town Manager may authorize accepting such recyclable materials as he deems practicable.

#### § 134.11 MAINTENANCE OF TRASH CONTAINERS

The owners and occupants of real property shall be responsible for the proper maintenance of dumpsters, recycle containers, grease dumpsters, and similar storage bins, regardless whether a private contractor, the Town, or another person provides the container, and regardless whether the owner or occupant participates in the Town's trash collection program. No owner or occupant shall allow garbage, grease, or other materials to accumulate in the immediate area of any such bin.

#### § 134.12 PARTICIPATION IN PROGRAM; FEES

(a) All residential properties within the Town, and which do not receive trash collection services through a property owner's association, shall participate in the Town's trash collection program.

- (b) From time to time, the Town will establish a fee for the services to be provided under this Chapter. All properties required to participate in the Town's trash collection program shall pay the fees that the Town Council establishes for trash collection, and which appear on the fee schedule to the Town Code.
- (c) In the case of a Two Family Dwelling or a Multi-Family Dwelling (as those terms are defined in the Town's Land Development Regulations) that is required to participate in the Town's trash collection program, the fee shall be applied to each dwelling unit therein.
  - (d) Fees due under this Chapter shall also be subject to the provisions of Chapter 135.

#### **§ 134.13 PENALTIES**

- (a) <u>Civil Penalties</u>. A person violating any provision of this Chapter for which a penalty is not otherwise provided shall be liable for a civil penalty of \$50 per day for each initial violation thereof, and of \$200 per day for each subsequent violation arising within the same twelve month period as an initial violation. Provided, however, that each day of a violation arising from the same set of operative facts in a twelve month period is penalized as an initial violation, not as a subsequent violation. In addition, a civil penalty for violation arising from the same set of operative facts may not exceed \$3,000 in any twelve month period. No person shall be assessed a civil penalty under this section unless the Town Manager has previously provided seventy-two hours' notice of his intent to assess a civil penalty, and determined that the condition giving rise to the penalty has not been remedied before he assesses such penalty. Notice shall be provided by personal delivery to the owner or occupant of the property in question, or by posting of such notice on the front door of the principal structure of the property on which the violation is occurring. In the event that notice by personal delivery cannot practicably be made, the Town Manager may mail notice to the address of the property own as shown on the Rockingham County
- (b) <u>Criminal Penalties</u>. Any person who has had a civil penalty imposed under paragraph (a) of this section three times within any twenty-four month period, for the same or similar violations, but on three different sets of operative facts, shall be guilty of a Class 3 misdemeanor and upon due conviction thereof shall be fined an amount not to exceed \$500. This paragraph shall not apply to violations of § 134.08, for which only a civil penalty may be imposed.
- (c) <u>Collection</u>. All charges authorized by this section, including the civil penalties described in paragraph (a), shall constitute a lien against the property on which the violation took place, ranking on a parity with liens for unpaid local real estate taxes and enforceable in the same manner. Without limiting the generality of the foregoing, a civil or criminal penalty assessed for violations of §§ 134.07, 134.08, and 134.11 shall constitute a lien on the property on which the dumpster or other container is situate.

#### **CHAPTER 135: RATES AND CHARGES**

## § 135.01 APPLICIBILITY

Except as otherwise provided in separate contracts with the Town, the provisions of this title apply to any water, sewer, or solid waste fee charged under this Title 13.

#### § 135.02 BILLING

Billing for water, sewer, and solid waste fees shall be on a bi-monthly basis beginning on January 1 of each calendar year. Bills shall be due on the 25<sup>th</sup> day of the calendar month following the end of each bi-monthly period.

### § 135.03 LATE FEES AND PENALTIES

Any bill for water, sewer, or solid waste fees not paid when due shall incur a late fee of 5% of the amount unpaid, or \$3.00, whichever is greater.

#### § 135.04 DISCONTINUATION OF SERVICE

If any bill (including the associated late fee) is not paid on or before the 4<sup>th</sup> day of the second calendar month following the end of a bimonthly billing cycle, the Town shall discontinue service to such property. Service shall be reconnected only after all outstanding bills are paid. The Town may charge a reconnection fee in the amount shown on the rate schedule to the Town Code before reconnecting any service.

#### § 135.05 TAMPERING WITH METER; DAMAGED METERS

If at any time a meter shall be found to have been tampered with, shall fail to register for any reason, or shall be found to be defective in registering since the last previous reading, the water and sewer consumption for such period may be based on the Town Manager's best estimate of such consumption, taking into account the average of past readings from that connection.

#### § 135.06 LEAKAGE

In the event of a confirmed leak, customers shall, upon proof of repair within thirty (30) days of notification of such leak, be given credit against their bill for one-half (1/2) of the usage over and above their normal average consumption. Such credit shall apply both to water and sewer charges.

#### § 135.07 METER ACCURACY

Any water or sewer user having a complaint in connection with the assessment of his water or sewer bill shall report the same to the Town Manager. Any water user who believes in good faith

that he is over-assessed may have his meter tested by the Town. If the meter is found to over-register, the most recent invoice shall be adjusted.

## § 135.08 CHARGES CONSTITUE A LIEN

To the fullest extent permitted by law, every charge that this Title 13 authorizes with which the owner of any such property has been assessed and which remains unpaid shall constitute a lien against such property ranking on a parity with liens for unpaid local taxes and enforceable in the same manner. All such liens shall remain a personal obligation of the owner of the property at the time the liens were imposed.

## § 135.09 PERMITTING FEE FOR ACCESS TO PUBLIC RIGHT OF WAY

- (a) Except to the extent that state law prohibits the imposition of a permitting fee, any public service company that wishes to access the public right of way within the Town shall obtain a permit to so do from the Town, and shall pay a Public Right of Way Access Permit review fee to the Town in the amount shown on the rate schedule to the Town Code. In requesting such permit, the public service company shall submit to the Town any information that the Town Manager may reasonably request concerning the nature and extent of such proposed access.
- (b) The Town Manager shall review such information and shall issue the permit if the Town Manager determines that access to the public right of way will not unreasonably interfere with the public's use of such right of way or the property or utility lines of the Town or another public service company within such right of way.
- (c) Any public service company that accesses a public right of way within the Town shall restore such right of way to the condition it was in immediately before such access, shall not disturb the property or utility lines of the Town or another public service company except as necessary to perform the proposed work, and shall restore such property or utility lines after such work to the same condition as immediately before the work. Every public service company that accesses a public right of way within the Town shall provide a bond, with surety, to secure its full performance of its obligations under this paragraph.

## § 135.10 INCREASE OF WATER, SEWER, AND TRASH FEES BASED ON RATE OF INFLATION

When preparing the budget for any fiscal year, and without any further or other action of the Town Council, the Town Manager shall (a) include in such budget an increase in the Town's regular recurring water, sewer, and trash rates based on the percentage positive change (if any) in the Consumer Price Index for All Urban Consumers (CPI-U), U. S. city average, as the U.S. Department of Labor reports it, between (i) January 1 of the calendar year immediately preceding the calendar year in which the new fiscal year begins, and (ii) January 1 of the calendar year in which the new fiscal year begins; and (b) shall advertise a public hearing reflecting such proposed

increase for the May Council meeting preceding the beginning of such fiscal year. The Clerk shall include such public hearing, an discussion of an ordinance to implement the proposed increases, on the agenda for such May Council meeting, and the Town Attorney shall prepare an appropriate ordinance for consideration at such May Council meeting.

**Ordinance of May 7, 2024** added this section. The Ordinance also contained an uncodified transitional provision that covers only the fiscal year 2025-26.

## TITLE XV: BUSINESS REGULATION

# Chapter

- 151. REGULATION OF SHORT TERM RENTALS
- 152. ITINERANT MERCHANTS, PEDDLERS, AND SOLICITORS
- 153. DEALERS IN PRECIOUS METALS
- 154. MASSAGE PARLORS

#### **CHAPTER 151: REGULATION OF SHORT TERM RENTALS**

#### § 151.01 DEFINITIONS

For the purpose of this Chapter, the following definitions apply:

"Operator" means the proprietor of any dwelling, lodging, or sleeping accommodations offered as a short-term rental, whether in the capacity of owner, lessee, sublessee, mortgagee in possession, licensee, or any other possessory capacity.

"Short term rental" means the provision of a room or space that is suitable or intended for occupancy for dwelling, sleeping, or lodging purposes on a more or less transient or temporary basis, for a average period of fewer than 30 consecutive days, in exchange for a charge for the occupancy. The term includes "Hotel", and "Bed and Breakfast or Short Term Rental" establishments as those terms are defined in the Town's Land Development Regulations, and potentially includes a "Boarding House" as that term is defined in the Town's Land Development Regulations, if the Boarding House's average length of stay is fewer than 30 consecutive days.

## § 151.02 REGISTRATION REQUIREMENT

- (a) A short term rental operator shall register with the Town Treasurer within thirty (30) days of first holding itself out as offering lodging within the Town.
- (b) After initial registration under the provisions of this Chapter, a short term rental operator shall re-register annually before July 1 of each year.
- (c) The registration form shall require the operator of the short term rental to provide the name of the owner of each property so rented, the address of the property, and the Rockingham County tax map number of the property. The operator shall also provide the mailing and physical address of the operator (if different from the property address), the telephone number, and the email address of the operator. If the operator of the property is a not a natural person, the operator shall also provide the name of a natural person who is responsible for the operator (who may be an owner, member, officer, manager, or agent of the operator), and shall provide contact information for that person.
- (d) If the owner, operator, or responsible natural person changes in the middle of a registration year, but short term rentals will continue, the operator shall file a new registration form with the updated information within thirty (30) days of the change.
- (e) No person shall be required to register if applicable state law exempts that person from the registration requirement.

## § 151.03 FEE

Upon each registration or re-registration of an operator of a short term rental under this Chapter, the Treasurer shall assess a fee as indicated on the Town's Schedule of Taxes, Fees, and Rates. If a single registration registers multiple properties, the Treasurer shall assess the fee only once.

### § 151.04 ENFORCEMENT

- (a) No person shall offer the short term rental of a location within the Town without timely complying with § 151.02 above.
  - (b) A person who violates § 151.04(b):
- (i) shall be assessed a civil penalty of \$500 for each instance of non-compliance;
- (ii) shall not offer to rent, or actually rent, a short term rental during the period of such non-compliance, or during the period when the civil penalty related to such non-compliance remains unpaid.
- (c) Upon repeated violations of this Chapter as it relates to a specific property, the Town Manager may prohibit from registering and offering that property for short-term rental.
- (d) The Town Manager may prohibit an operator required to register from offering properties for short-term rental in the Town upon multiple violations on more than three occasions of applicable state and local laws, ordinances, and regulations, as they relate to a short-term rental. Violations to be considered are violations both under this Chapter and under other applicable Virginia, Rockingham County, and Town laws, ordinances, and regulations.
- (e) The Town Manager may promulgate appropriate regulations, not arbitrary and capricious, including forms, to implement this Chapter.

### § 151.05 INTERACTION WITH OTHER LAW

Compliance with this Chapter does not excuse compliance with any other applicable Virginia, Rockingham County, or Town law, ordinance, or regulation; rather, the requirements of this Chapter are in addition to requirements under any other such law, ordinance, or regulation. Specifically, but without limitation, compliance with this Chapter does not excuse compliance with the Town's land development regulations, transient occupancy tax ordinance, or business and professional license ordinance. Registration under this Chapter does not create any land use or other vested rights, and does not operate to allow a short term rental in a zoning district not otherwise permitted in the Land Development Regulations.

## **CHAPTER 152: ITINERANT MERCHANTS, PEDDLERS AND SOLICITORS**

## § 152.01 DECLARATION OF LEGISLATIVE INTENT AND PURPOSE

The Town Council recognizes that it is expressly empowered under Virginia law to regulate peddlers, and, in addition, to provide such other regulation as it deems appropriate to prevent fraud and deceit in the sale of property. The Town Council hereby finds and declares that the risks to the public of fraud and deceit in the sale of property are higher when a merchant has no fixed and determinable place of business, and, therefore is not easily accessible to the public in the period after the consummation of the transactions. Accordingly, the Town Council finds and declares that the public good necessitates the regulations in this Chapter for the purpose of preventing fraud or deceit in the sale of property.

## § 152.02 DEFINITIONS

For the purpose of this Chapter the following definitions shall apply unless the context clearly indicates or requires a different meaning.

<u>Itinerant Merchant</u>. "Itinerant merchant" means any person, whether as owner, agent, or consignee, who transports an inventory of new merchandise to a building, vacant lot, or other location within the Town and who, at that location, displays, sells or offers to sell the new merchandise to the public on a temporary basis. The term "itinerant merchant" shall not include a merchant with an established store, regularly open to the public; or a licensed merchant with a regularly serviced supply route or location.

New Merchandise means goods or products which are not used but are in a similar condition as the goods or products wholesaled by manufacturers or suppliers to established retail stores for first-time purchase by consumers. New merchandise shall not include (i) crafts or goods made by the seller or his own household; (ii) food stuffs; (iii) the seller's own household personal property; (iv) merchandise sold by nonprofit charitable, educational or religious organizations or at events sponsored by such organizations; (v) merchandise sold during parades, festivals, sporting or entertainment events, civic or fundraising activities sponsored by nonprofit charitable, educational or religious organizations; or (vi) any goods sold at the Broadway Farmers Market during a day on which the Farmers Market is open in the regular course of its business.

<u>Peddler</u>. Any person, not an itinerant merchant who: (i) travels from place to place but at times within the Town, by any means, carrying new merchandise for sale, or making sales of new merchandise, or making deliveries of new merchandise; or (ii) without traveling from place to place, sells or offers new merchandise for sale from any public place within the Town.

<u>Solicitor</u>. Any person who travels by any means from place to place but at times within the Town, taking or attempting to take orders for sale of new merchandise to be delivered in the future or for services to be performed in the future. A person who is a solicitor is not a peddler.

## § 152.03 LICENSE REQUIRED

- (a) No itinerant merchant, peddler, or solicitor shall operate within the Town without a duly issued license under this Chapter.
- (b) The fee for the license required by this Chapter shall be as set on the fee schedule appended to the Town Code
  - (c) No license issued under this Chapter shall be transferable.
  - (d) All licenses issued under this Chapter shall expire 90 days after the date of issuance.

## § 152.04 APPLICATION PROCEDURE

All applicants for licenses required by this Chapter shall file an application with the Clerk. This application shall be signed by the applicant if an individual, or by all partners if a partnership, or by the president if a corporation, or by a manager or other authorized member of a limited liability company. The applicant shall provide information concerning the following items:

- (a) The name and address of the applicant;
- (b) The name of the individual, a natural person, who has management authority or supervision of the applicant's business during the time that it is proposed to be carried on in the Town;
  - (c) The local address of such individual;
  - (d) The permanent address of such individual;
  - (e) The capacity in which such individual will act;
- (f) The name and address of the person, if any, for whose purpose the business will be carried on, and, if an entity, the state of organization of the entity;
- (g) The time period or periods during which it is proposed to carry on applicant's business;
- (h) The nature, character, and quality of the goods or services to be offered for sale or delivered;
- (i) If goods, their invoice value and whether they are to be sold by sample as well as from stock;

- (j) If goods, where and by whom such goods are manufactured or grown, and where such goods are at the time of application;
  - (k) The nature of the advertising proposed to be done for the business;
- (l) Whether or not the applicant, or the individual identified in (b) above has been convicted of any crime or misdemeanor (not including a traffic offense) and, if so, the nature of each offense and the penalty assessed for each offense.

In addition, applicants for peddler or solicitor licenses must provide further information concerning the following items, in addition to that requested above:

- (m) A description of the applicant;
- (n) A description of any vehicle proposed to be used in the business, including its registration number, if any.

All applicants for licenses required by this Chapter shall attach to their application, if requested by the Clerk, credentials from the person, if any, for which the applicant proposes to do business, authorizing the applicant to act as such representative.

Applicants who propose to handle foodstuffs shall also attach to their application, in addition to any attachments required above, a statement from a licensed physician, dated not more than ten days prior to the date of application, certifying the applicant to be free of contagious or communicable disease.

#### § 152.05 ISSUANCE OF LICENSE

- (a) Upon receipt of an application, the Town Manager shall investigate an investigation the applicant's business reputation and moral character.
- (b) The Town Manager shall approve the application, and issue a license, unless such investigation discloses tangible evidence that the conduct of the applicant's business would pose a substantial threat to the public welfare. In particular, the Town Manager shall deny the application upon tangible evidence that the applicant has been convicted of a crime of moral turpitude; has made willful misstatements in the application; has committed prior violations of applicable law pertaining to itinerant merchants, peddlers, solicitors, and the like; has committed prior fraudulent acts; or has a record of continual breaches of solicited contracts.
- (c) If the Town Manager denies a license application, the Town Manager shall notify the applicant via personal delivery to the supervisor described above or to the applicant or other agent of the applicant, or by certified mail, return receipt requested, so to the local address given on the application described above.

## § 152.06 REVOCATION OF LICENSE

The Town Manager may revoke a license issued under this Chapter for a reason stated in § 152.07 below, after delivering a notice of his intent to do so via personal delivery to the supervisor described above or to the applicant or other agent of the applicant, or by certified mail, return receipt requested, so to the local address given on the application described above, and after giving the applicant five days after receiving such notice to respond to the Town Manager regarding the notice. Such notice shall state, in brief, the Town Manager's reasons for his intent to revoke the license. Once the Town Manager has revoked the license, he shall notify the applicant using the one of the same procedures permissible for delivering the notice of intent to revoke.

### § 152.07 STANDARDS FOR REVOCATION

The Town Manager may revoke a license granted under this Chapter for any of the following reasons:

- (a) Any fraud or misrepresentation contained in the license application; or
- (b) Any fraud, misrepresentation, or false statement made in connection with the business being conducted under the license; or
  - (c) Any violation of this Chapter or other applicable law; or
- (d) Conviction of the licensee of any felony, or conviction of the licensee of any misdemeanor involving moral turpitude; or
- (e) Conducting the business licensed in an unlawful manner or in such a way as to constitute a menace to the health, safety, morals, or general welfare of the public.

## § 152.08 APPEAL; BARRING OF REPETITIVE APPLICATIONS

- (a) An applicant aggrieved by the Town Manager's determination under §§ 152.05 or 152.06 may file a petition in Rockingham County Circuit Court for judicial review of the Town Manager's determination. The applicant shall file any such petition within ten (10) days of the applicant's receipt of the notice of the Town Manager's determination, or such petition shall be barred. The Circuit Court shall review the Town Manager's determination, but shall uphold such determination unless the applicant shows that such determination was arbitrary and capricious.
- (b) The Town Manager may summarily deny any application for a license under this Chapter if, during the one-year period preceding the application, the applicant (i) was denied a license by the Town; (ii) has a license revoked by the Town, in either instance with or without judicial review. A summary denial under this paragraph is not reviewable.

### § 152.09 EXHIBITION OF LICENSE

- (a) Any license issued to an itinerant merchant under this Chapter shall be posted conspicuously in or at the place named therein. In the event more than one place within the Town shall be used to conduct the business licensed, separate licenses shall be issued for each place.
- (b) The Clerk shall issue a license to each peddler or solicitor licensed under this Chapter. The license shall contain the words "Licensed Peddler" or "Licensed Solicitor," the expiration date of the license, and the number of the license. The license shall be kept with the licensee during such time as he is engaged in the business licensed.

### § 152.10 POLICY ON PEDDLING AND SOLICITING

The Town Council declares the policy of the Town that the occupants of the residences in the Town shall make the determination of whether peddlers or solicitors shall be, or shall not be, invited to their respective residences.

#### § 152.11 NOTICE REGULATING PEDDLING OR SOLICITING

- (a) Notice of the refusal of invitation to solicitors or peddlers to any residence shall be given on a weatherproof card, approximately three inches by four inches in size, exhibited upon or near the main entrance door to the residence, indicating the determination by the occupant, containing the applicable words, as follows: "NO SOLICITORS INVITED". The letters shall be at least 1/3-inch in height. For the purpose of uniformity, the cards shall be provided by the Chief of Police to persons requesting, at the cost thereof.
- (b) The card so exhibited shall constitute sufficient constructive notice to any peddler or solicitor of the determination by the occupant of the residence not to invite peddlers or solicitors to such residence.

### § 152.12 DUTY OF SOLICITORS AND PEDDLERS TO ASCERTAIN NOTICE

- (a) Every solicitor and peddler has a duty, upon going onto any premises in the Town upon which a residence is located, first to examine the notice provided for in § 152.11 if any is attached, and to be governed by the statement contained on the notice. If the notice states "NO SOLICITORS INVITED" as provided above, then the solicitor, whether registered or not, shall immediately and peacefully depart from the premises.
- (b) Any solicitor who has gained entrance to any residence, whether invited or not, shall immediately and peacefully depart from the premises when requested to do so by the occupant.

## § 152.13 PROHIBITED SOLICITATION

No person shall go upon any premises and ring the doorbell upon or near any door, or create any sound in any manner calculated to attract the attention of the occupant of such residence, for the purpose of securing an audience with the occupant thereof and engage in soliciting or

peddling in defiance of the notice exhibited at the residence in accordance with the provisions of § 152.11 above.

## § 152.14 COMPLAINCE WITH STATE LAW

No itinerant merchant, peddler, or solicitor operating within the Town shall violate any applicable provision of state law.

## § 152.15 IMPLIED CONSENTS

- (a) By submitting an application for a license under this Chapter, an applicant appoints the Secretary of the Commonwealth as its statutory agent for service of process in any civil action brought to enforce the provisions of this Chapter, or brought by a private citizen out of any transaction or contract entered into as a result of the business conducted in the Town under a license issued under this Chapter.
- (b) By submitting an application for a license under this Chapter, an applicant consents to the personal jurisdiction of courts of the Commonwealth of Virginia in connection with any civil action brought to enforce the provisions of this Chapter, or brought by a private citizen out of any transaction or contract entered into as a result of the business conducted in the Town under a license issued under this Chapter.
- (c) By submitting an application for a license under this Chapter, an applicant waives (i) any objection to venue in the state or federal courts sitting in Harrisonburg, Virginia, and (ii) agrees to a jury trial to the extent provided for under Virginia law, in each instance in connection with any civil action brought to enforce the provisions of this Chapter, or brought by a private citizen out of any transaction or contract entered into as a result of the business conducted in the Town under a license issued under this Chapter. The provisions of this paragraph apply regardless of any contrary provisions in any contract between the applicant or another party, and a purchaser of new merchandise sold under the license contemplated in this Chapter.

### § 152.16 PENALTIES AND ENFORCEMENT

- (a) Any person who violates this Chapter (whether acting on his own behalf, or as an agent of an entity) shall be guilty of a Class 4 misdemeanor and upon due conviction thereof shall be filed \$250. Each solicitation by a solicitor or peddler, or each day's business by an itinerant merchant, shall constitute a separate violation.
- (b) The provisions of paragraph (a) shall be cumulative to any civil remedy in equity. The Town Attorney is authorized, at the direction of the Town Manager, to bring a civil action in the nature of a suit in equity to restrain any violation of the provisions of this Chapter.

## § 152.17 RELATION TO THIRD PARTIES

- (a) Nothing in this Chapter shall affect bona fide purchaser status as to title to new merchandise acquired from an itinerant merchant or peddler, or through the solicitation of a solicitor.
  - (b) This Chapter does not create any private right of action to enforce its provisions.

## § 152.18 RELATION TO OTHER LAW

The provisions of this Chapter are in addition to the requirements of other applicable state, federal, and local law including, without limitation, the Town's ordinance regarding business licenses and the Town's ordinances regarding trespassing.

#### **CHAPTER 153: DEALERS IN PRECIOUS METALS**

#### § 153.01 APPLICABILITY AND DEFINITIONS

This Chapter applies to all dealers in precious metals or gems, referred to in this Chapter as "dealers".

"Coin" means any piece of gold, silver or other metal fashioned into a prescribed shape, weight and degree of fineness, stamped by authority of a government with certain marks and devices, and having a certain fixed value as money.

"Dealer" means any person, firm, partnership, or entity engaged in the business of (i) purchasing secondhand precious metals or gems; (ii) removing in any manner precious metals or gems from manufactured articles not then owned by the person, firm, partnership, or corporation; or (iii) buying, acquiring, or selling precious metals or gems removed from manufactured articles. "Dealer" includes all employers and principals on whose behalf a purchase is made, and any employee or agent who makes any purchase for or on behalf of his employer or principal.

The definition of "dealer" shall not include persons engaged in the following:

- 1. Purchases of precious metals or gems directly from other dealers, manufacturers, or wholesalers for retail or wholesale inventories, provided that the selling dealer has complied with the provisions of this chapter.
- 2. Purchases of precious metals or gems from a qualified fiduciary who is disposing of the assets of an estate being administered by the fiduciary.
- 3. Acceptance by a retail merchant of trade-in merchandise previously sold by the retail merchant to the person presenting that merchandise for trade-in.
- 4. Repairing, restoring or designing jewelry by a retail merchant, if such activities are within his normal course of business.
- 5. Purchases of precious metals or gems by industrial refiners and manufacturers, insofar as such purchases are made directly from retail merchants, wholesalers, dealers, or by mail originating outside the Commonwealth.
- 6. Persons regularly engaged in the business of purchasing and processing nonprecious scrap metals which incidentally may contain traces of precious metals recoverable as a by-product.

"Gems" means any item containing precious or semiprecious stones customarily used in jewelry.

"Precious metals" means any item except coins composed in whole or in part of gold, silver, platinum, or platinum alloys.

## § 153.02 DEALERS TO FOLLOW STATE LAW

Every dealer operating within the Town shall follow all applicable provisions of state law relative to dealers, including the provisions of state law requiring each dealer to have a license duly issued by the Town before engaging in business within the Town.

### § 153.03 AUTHORITY OF POLICE CHIEF

The Police Chief of the Town is hereby designated to have all authority in connection with dealers that state law provides to the chief law enforcement officer of the Town. This authority includes, without limitation, the authority to issue or revoke licenses to dealers as described in state law.

### § 153.04 STATEMENT OF OWNERSHIP

When providing the statement of ownership required by state law regarding a precious metal or a gem to be sold to a dealer, the seller thereof shall include, at a minimum, a statement from the seller about the provenance of the item including, if the seller knows, the name and contact information of the owner of the item immediately preceding the seller.

#### § 153.05 BOND

Bonds required by state law in connection with the issuance of a license to a dealer shall be made out for the benefit of the Town.

## § 153.06 PENALTY

Any person who violates any provision of this Chapter shall be guilty of a Class 2 misdemeanor and upon due conviction thereof, shall be fined an amount not to exceed \$1,000. Any person who violates any provision of this Chapter and has been convicted previously on a separate offence under this Chapter, a state law regulating dealers, or the laws of another locality regulating dealers (within or without the Commonwealth) shall be guilty of a Class 1 misdemeanor and upon due conviction thereof, shall be filed an amount not to exceed \$2,500 or incarcerated for a period not to exceed one (1) year, or both. Such criminal penalties shall be in addition to any revocation of such dealer's license as authorized or required under state law.

#### **CHAPTER 154: MASSAGE ESTABLISHMENTS**

### § 154.01 DEFINITIONS.

For the purpose of this chapter, the following definitions shall apply unless the context clearly indicates or requires a different meaning.

<u>Erogenous Area</u>. The pubic area, penis, scrotum, vulva, perineum or anus.

<u>Health Club</u>. Any establishment which offers service in the form of massage, baths, exercises or similar services, in combination, to club members, or to the public for a charge. The terms "health club" and "massage establishment" do not include:

- (a) Hospitals, nursing homes, medical clinics, or other offices or quarters of a physician, a surgeon, a chiropractor or an osteopath or a physical therapist duly licensed by the state;
- (b) Exercise clubs exclusively for the purpose of maintaining physical fitness and health where the services are performed, without massage in any form for pay and without any physical touching with any part of one person's body with an erogenous area of any other person;
- (c) Barbershops and beauty parlors in which massages are administered only to the scalp, the face, the neck the shoulder or below the knee;

### (d) Any establishment:

- (i) Offering or providing facilities for, or instruction in, controlled exercise, weight lifting, calisthenics, or general physical fitness;
- (ii) Which is actually occupying the premises of not less than 5,000 square feet, of which not more than 5% is used for massages; and
- (ii) Whose gross income from massages is less than 15% of the total gross business income derived from physical fitness sales contracts at each business location.

<u>Massage Establishment</u>. An establishment having a fixed place of business where massages are administered for pay including, but not limited to, massage parlors or salons, health clubs, sauna baths and steam baths.

<u>Massage Therapist</u>. A person licensed as a massage therapist under the general laws of the Commonwealth.

Massage Therapy shall have the same definition as in state law.

## § 154.02 COMPLIANCE WITH CHAPTER

- (a) No person shall operate a massage establishment or health club, or practice as a massage therapist within the Town, without a massage permit.
- (b) No person shall practice massage therapy within the Town other than in Massage Establishment or Health Club, or in an establishment described in subsection (d) of the definition of "Health Club" in § 154.01.

## § 154.03 LAW-ENFORCEMENT OFFICERS EXEMPT

No provision of this chapter shall apply to police officers while in performance of official duties relating to any violation of this chapter or of any law of the United States, the Commonwealth of Virginia, or the Town.

### § 154.04 ELIGIBILITY FOR PERMITS

- (a) No massage permit shall be issued for any health club or massage establishment except to an owner or manager who is
  - (1) A physician or chiropractor;
  - (2) A registered nurse with at least two years' nursing experience;
  - (3) A licensed practical nurse with at least four years' nursing experience; or
  - (4) A registered physical therapist
  - (5) A certified massage therapist
  - (6) Other person authorized under state law to practice massage therapy.
- (b) Upon receipt of the application, the Town may refer the application to the appropriate local building official, the fire department, or other appropriate Town official, each of which shall, with a period of 21 days from the date of application, review records and make an inspection of the premises proposed to be used as a massage establishment or health club and make a written recommendation to the Town concerning compliance with the law.
- (c) A change of location of a licensed massage establishment may be approved by the Town if the applicant complies with all applicable provisions of this Code, including this chapter.

### § 154.05 PERMIT TO BE POSTED OR AVAILABLE FOR INSPECTION

The permit must be conspicuously posted in the massage parlor or health club, so that the permit may be readily seen by persons entering the premises in the public room thereof or, if the massage is given in a permitted area other than a health club or massage establishment, the permit must be available on request.

## § 154.06 MASSAGE PERMIT TERM

Each permit shall be valid for one year. All provisions of this chapter shall apply to renewals in the same manner as they apply to application for a granting of initial permits.

### § 154.07 PERMIT APPLICATION

- (a) Each application for a massage permit shall be upon a form provided by the Town. Each form shall contain the following information:
  - (1) The name and address of the massage establishment or health club;
- (2) A list of all the beneficial owners of such massage establishment or health club;
- (3) The name of the primary owner or manager of the massage establishment or health club, and the names and addresses of all previous massage establishments where such owner or manager has been employed as a massage technician;
- (4) The criminal record, if any, other than misdemeanor traffic violations, of such primary owner or manager;
- (5) Whether any permit to provide massage services has previously been denied to such owner or manager or revoked and, if so, the circumstances of such denial or revocation;
- (6) A complete set of the primary owner or manager's fingerprints, which shall be taken by the chief of police or his agent, and a photograph of the primary owner or manager;
- (7) Written proof that the primary owner or manager is 18 years of age or older; and
  - (8) Proof of licensure of the primary owner or manager.

- (9) A list of all employees or others who will be practicing massage therapy or otherwise administering massages at such massage establishment or health club, together with the information described in sections (3) through (8) of such employees or others.
- (10) Plans and specifications of the quarters proposed to be occupied. Such plans shall show details of entrances, partitions, windows, openings, ventilation, plumbing or fixtures, water supply and waste and vent connections.
- (b) Each holder of a massage permit shall report to the Town any change or supplementation (including, without limitation, new employees or independent contractors) in any of the information required in division (a) of this section, such report to be made within 14 days of learning of the change. Failure to report such a change or supplemental information after learning of it shall be grounds for revocation of the permit.

## § 154.08 INVESTIGATION OF APPLICANTS; GROUNDS FOR DENIAL

- (a) In addition to all other requirements of this Chapter, every application or supplement required under this Chapter shall include a sworn statement as to whether or not the primary owner or manager, and any employees or other persons practicing massage therapy, have been convicted, pleaded *nolo contendere*, or suffered a forfeiture of any felony charge or on a charge of violating any provision included in Va. Code § 18.2-344, *et seq.* and Va. Code § 18.2-372, *et seq.*, which laws relate to sexual offenses, or on a charge of violating a similar law of any other jurisdiction, or on a charge of violating any provision of this chapter or similar ordinance in any other jurisdiction. The application shall state thereon, "It is unlawful for any person to make a false statement on this application, and discovery of a false statement shall constitute grounds for denial of an application or renovation of a permit."
- (b) Upon receipt of the application fee as provided by the preceding sections, the Town shall make or cause to be made a thorough investigation relative to the application. The Town Manager shall deny any application for a permit after notice and hearing if he finds that the primary owner of manager, or an employee or other persons practicing massage therapy, has been convicted, pleaded *nolo contendere*, or suffered a forfeiture on any felony charge or on a charge of violating any provision included in Va. Code § 18.2-344, *et seq.*, which laws relate to sexual offenses, or on a charge of violating any similar law of any other jurisdiction. Provided, however, that in lieu of denial the Town Manager may permit the application to amend the permit application to remove any employee or other person who initially appeared in the application and has violated the statutes cited, but who will not be employed by the applicant.
- (c) The Town Manager may also deny an application, or revoke a permit previously issued, if he finds that an application contains a materially false statement.
- (d) The Town Manager shall revoke the permit of any person after notice and hearing if the Town Manager finds that such person has been convicted, pleaded *nolo contendere*, or suffered a forfeiture of any felony charge or on a charge of violating any provision included in Va. Code § 18.2-344, *et seq.*, and Va. Code § 18.2-372, *et seq.*, which laws relate to sexual offenses,

or on a charge of violating any similar law of any other jurisdiction. Provided, however, that in lieu of revocation the Town Manager may permit the applicant to amend the permit to remove any employee or other person who initially appeared in the application and has violated the statutes cited, but who will no longer be employed by the applicant.

- (e) Before any permit is denied or revoked, the applicant or holder is entitled to notice and a hearing. Notice of the hearing before the Town Manager shall be given in writing, setting forth the grounds of the proposed denial or revocation of the permit and the time and place of the hearing. Such notice shall be mailed by certified mail to the address on the application at least 5 days prior to the date set for the hearing, or by personal delivery to the primary owner or manager identified in the permit by a Town police officer. The Town Manager shall transmit his decision in writing to the applicant or holder within 5 days of the hearing by the same method.
- (f) An applicant or permit holder aggrieved by the Town Manager's determination under (e) above may file a petition in Rockingham County Circuit Court for judicial review of the Town Manager's determination. The applicant or holder shall file any such petition within ten (10) days of such person's receipt of the notice of the Town Manager's determination, or such petition shall be barred. The Circuit Court shall review the Town Manager's determination, but shall uphold such determination unless the applicant shows that such determination was arbitrary and capricious.

## § 154.09 EQUIPMENT STANDARDS

- (a) All tables, tubs, shower stalls, and floors, except reception and administrative areas, shall be made of non-porous materials which may be readily disinfected.
  - (b) Closed containers shall be provided for wet towels and waste materials.

## § 154.10 BUILDING INSPECTION AND REQUIREMENTS

- (a) No health club or massage establishment shall begin operations until the building occupied or to be occupied shall have been approved by the Town Manager as to the requirements of the building, plumbing, electric and fire prevention codes, and any other requirements of this chapter or state law.
- (b) No exterior or interior doors in any health club or massage establishment shall be equipped with any remotely controlled locking device.

### § 154.11 SANITATION AND HYGIENE

(a) All equipment, shower stalls, toilets, lavatories and any other such accounterments of the establishment shall be regularly treated with disinfectants, and shall be maintained in a clean and sanitary condition at all times.

- (b) Health, plumbing, electric and other inspectors shall be given access to any part of the quarters of a health club or massage establishment for purposes of inspections at all reasonable times.
- (c) No person shall practice any of the services of a massage therapist without a certificate of good health issued by a duly licensed physician, commensurate with the nature of the services rendered. Such certificate shall be renewed every 6 months.
- (d) No health club or massage establishment shall knowingly serve any patron infected with any fungus or other skin infections; nor shall service be performed on any patron exhibiting skin inflammation or eruptions; provided that a duly licensed physician may certify that a person may be safely served, prescribing the conditions thereunder.
- (e) All personnel shall wash their hands in hot running water, using a proper soap or disinfectant, before giving any service or treatment to each separate patron.
- (f) All towels and tissues, all sheets or other coverings shall be used singularly for each patron, and discarded for laundry or disposal immediately after use.
  - (g) Non-disposable tools of the trade shall be disinfected after use upon one patron.

## § 154.12 SLEEPING QUARTERS

No part of any quarters of any health club or massage establishment shall be used for, or connected with, any bedroom or sleeping quarters; nor shall any person sleep in such club or establishment except for limited periods incidental to and directly related to a massage or bath. This provision shall not preclude the location of a health club or massage establishment in separate quarters of a building housing a hotel or other separate businesses or clubs.

# § 154.13 PROHIBITED CONDUCT

- (A) No owner or manager of a health club or massage establishment shall tolerate in his or her club or establishment any activity or behavior prohibited by the laws of this Commonwealth, particularly, but not exclusive of, Va. Code § 18.2-346, *et seq.*, proscribing prostitution and bawdy places, etc.; and Va. Code § 18.2-372, *et seq.*, being the state statute on obscenity, dealing with shameful or morbid interest in nudity or sex, which appeals to prudent interest, and which covers obscene items, exhibitions, performances, indecent exposure, illicit photographs or photographic activity, and illicit modeling.
- (B) No owner or manager of a health club or massage establishment shall tolerate in his or her club or establishment any activity or behavior which violates any other applicable local or federal laws or regulations similar in nature or subject to those listed in division (A) of this section.

## § 154.14 PENALTY

- (a) Any person who violates any of the provision of this chapter other than §§ 154.02 or 154.13 shall be guilty of a Class 3 misdemeanor, and upon due conviction thereof shall be fined not more than \$500.
- (b) Any person who violates §§ 154.02 or 154.13 shall be guilty of a Class 2 misdemeanor for the first offence, and upon due conviction thereof shall be fined not more than \$1,000, or confined in jail for a period not to exceed six months, or both. Upon conviction of any subsequent offense, such person shall be guilty of a Class 1 misdemeanor, and upon due conviction thereof shall be fined not more than \$2,500, or confined in jail for a period not to exceed six months, or both.
- (c) Upon the first conviction by any court for violation of any provision of this chapter, the Town Manager may revoke such person's license to engage in business under this chapter for a period of one full year from the date of conviction. Such revocation shall be mandatory upon a second conviction.